# Signalling Pathways Mediated by the Bombesin/GRP Receptor

Amanda Charlesworth

A Thesis submitted for the degree of Doctor of Philosophy at the University of London

1996

Department of Oncology University College, London **Growth Regulation Laboratory** 



ProQuest Number: 10106528

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10106528

Published by ProQuest LLC(2016). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code. Microform Edition © ProQuest LLC.

> ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346

### ABSTRACT

Neuropeptides, and in particular peptides of the bombesin family, including gastrin releasing peptide, stimulate DNA synthesis and cell proliferation in cultured cells and are implicated as cellular growth factors in a variety of fundamental biological processes including development, tissue regeneration and tumourigenesis.

Bombesin elicits multiple signalling pathways in various cell types, including activation of phospholipase C, which leads to activation of protein kinase C and elevation of intracellular calcium; tyrosine phosphorylation of intracellular substrates and reorganisation of the actin cytoskeleton, and cell proliferation. It is not clear however whether these responses are mediated by a single receptor subtype or by different subtypes that couple preferentially to specific pathways. To resolve this the mouse bombesin/GRP receptor was transfected into Rat-1 fibroblasts and the pathways activated by bombesin were investigated. Here it is shown that phospholipase C activation, cell growth and tyrosine phosphorylation emanate from a single class of bombesin receptor.

Mitogen Activated Protein Kinase (MAPK) activity is a common point of convergence for mitogenic signals. While the mechanism by which receptor tyrosine kinases activate MAPK is well established, the mechanism by which G protein-coupled receptors activate MAPK is less well understood. The involvement of Ras and Raf in activation of MAPK has not been described in bombesin-induced signalling. This thesis shows that in Rat-1 cells transfected with the bombesin/GRP receptor, bombesin can activate MAPK and Raf-1 in a protein kinase C-independent fashion. Also it is demonstrated for the first time the bombesin stimulates GTP loading of p21<sup>ras</sup>.

Calcium mobilisation from intracellular calcium stores has been observed to be coincident with cell proliferation induced by a variety of growth factors including neuropeptides and especially bombesin. However, a direct role of calcium in this event has not been described. Here it is shown that partial and persistent depletion of intracellular calcium stores, using specific inhibitors of the endoplasmic reticulum calcium pump, thapsigargin and 2,5-di-tert-butylhydroquinone, stimulated reinitiation of DNA synthesis in synergy with either phorbol-12,13-dibutyrate or bombesin in Swiss 3T3 cells. This stimulation of DNA synthesis was dependent on entry of extracellular calcium and hence this suggests a role for calcium in growth control.

#### ACKNOWLEDGEMENTS

I would like to thank everyone who made this work possible. First and foremost I would like to thank Henry for his insight, supervision and guidance, and for the many valuable lessons he has taught me about staying alive in science.

In addition I would like to thank Mark Bouzyk, Simon Broad, Theresa Higgins and Jim Sinnett-Smith for getting me started in the lab and also Thomas Seuffelein and Dominic Withers for keeping me going.

Special thanks go to Sara Rankin for showing me what can be achieved and together with Theresa and Alison McDowell for my "personal development". Many thanks to Perugino's for keeping me fed during my long evenings by the computer and to the Canadian Muffin Co. who kept me going during the long Sundays.

I would also like to acknowledge everyone who had to put up with my bad moods and constant moaning during the home straight. In particular I am most grateful to the Lab-down-the-end who took me away when I needed it most, and when I didn't.

And I must not forget the guys in photography who had to do just-one-moreemergency on several occaisions. Finally I must thank Sharon Matthews for giving this thesis a rather excellent once over.

It just leaves me now to say "here Mum and Dad, this is for you."

## **CONTENTS**

ABSTRA	СТ		2
ACKNOWLEDGEMENTS			3
CONTENTS			4
LIST OF FIGURES			9
LIST OF	TABL	ES	12
ABBREVI	IATIO	NS	13
Chapte	er 1	INTRODUCTION	16
1.1. G	ROWI	TH REGULATION	17
1.1.1.	Gr	owth Factors	18
1.1.2.	Sy	nergistic Signalling Pathways	21
1.1.3.	Re	gulatory Signals and Obligatory Events	22
1.1.4.	Re	ceptor Tyrosine Kinases	22
1.1.5.	G	Protein-Coupled Receptors	24
1.	1.5.1.	Growth Regulation by G Protein-Coupled Receptors	25
1.	1.5.2.	Structure	26
1.1.6.	He	eterotrimeric G-Proteins	28
1.	1.6.1.	Structure	28
1.	1.6.2.	The G Protein Cycle	29
1.	1.6.3.	Classes of G Protein	31
1.	1.6.4.	βγ subunits	31
1.	1.6.5.	Modifying Toxins	33
<b>1.2.</b> B	OMBE	SIN	34
1.2.1.	Bo	mbesin/GRP Receptor	36
<b>1.3.</b> B	OMBE	SIN-MEDIATED SIGNAL TRANSDUCTION	37
1.3.1.	Ph	ospholipase C	37
1.3.2.	In	ositol 1,4,5-trisphosphate and Ca <sup>2+</sup> Mobilisation	
1.	3.2.1.	Calcium Homeostasis	39
1.	3.2.2.	Calcium Release from Intracellular Stores	
1.	3.2.3.	Calcium Influx from the Extracellular Medium	
1.	3.2.4.	Calcium Gradients	46
1.	3.2.5.	Calcium in Growth Control	46
1.	3.2.6.	Calcium Effectors	49
1.3.3.	Pr	otein Kinase C	50
1.	3.3.1.	Structure and Activation	51
1.	3.3.2.	Substrates	52
1.	3.3.3.	80K/MARCKS	53

	1.3.3.4	4. Protein Kinase D	53
1.3	<b>3.4.</b> 1	Monovalent Ion Fluxes	54
1.3	<b>3.5.</b> 1	Phospholipase A <sub>2</sub> and Arachidonic Acid Release	55
1.3	<b>3.6.</b>	Adenylyl Cyclase, cAMP and PKA	56
1.3	<b>3.</b> 7. <sup>7</sup>	Tyrosine Phosphorylation	57
	1.3.7.1	1. Tyrosine Phosphorylation of Multiple Substrates	57
	1.3.7.2	2. Focal Adhesions	58
	1.3.7.3	3. Stress Fibres	59
	1.3.7.4	4. Src	60
	1.3.7.5	5. Focal Adhesion Kinase	61
	1.3.7.6	6. p130 <sup>cas</sup>	61
	1.3.7.2	7. Paxillin	61
	1.3.7.8	8. Bombesin-Induced FAK and Paxillin Tyrosine Phosphorylat	ion62
	1.3.7.9	9. Role of FAK and Paxillin Tyrosine Phosphorylation	63
1.3	3.8. <sup>-</sup>	The MAPK Pathway	64
	1.3.8.1	1. Overview	64
	1.3.8.2	2. MAPK	65
	1.3.8.3	3. MEK	66
	1.3.8.4	4. Raf	66
	1.3.8.5	5. Ras	68
	1.3.8.6	6. Effects of MAPK	70
	1.3.8.7	7. Activation of the MAPK Pathway by G Protein-Coupled Rece	ptors.70
	1.3.8.8	8. Bombesin-Induced MAPK Activation	73
	1.3.8.9	9. Cellular Context	74
	1.3.8.1	10. Function of $G\beta\gamma$	75
	1.3.8.1	11. Stress-Activated Protein Kinases	
1.3	<b>3.9</b> . 1	Induction of Early Genes	77
	1.3.9.1	1. AP-1	77
	1.3.9.2	2. Induction of AP-1 Components	78
1.4.	COUF	PLING TO MULTIPLE PATHWAYS	80
1.5.	AIMS	S OF THESIS	83
Chaj	pter 2	2 – MATERIALS AND METHODS	85
2.1.	MATI	ERIALS	85
2.2.	METH	HODS	86
2.2	2.1.	Transfection of Rat-1 cells	86
2.2	2.2.	Cell Culture	
	2.2.2.1	1. Rat-1 Cells	86

2.2.2	.2.	Swiss 3T3 Cells	86
2.2.3.	[ <sup>12</sup>	<sup>5</sup> I]GRP binding assay	87
2.2.4.	Me	asurement of Total Inositol Phosphates	87
2.2.5.	Me	easurement of Ca <sup>2+</sup>	88
2.2.5	.1.	Using the fluorescent Ca <sup>2+</sup> indicator, Fura-2	88
2.2.5	.2.	Measurement of Total <sup>45</sup> Ca <sup>2+</sup> Content	89
2.2.6.	Me	asuring PKC Activation by 80K/MARCKS Phosphorylation	89
2.2.7.	M/	APK Activation	90
2.2.7	.1.	Immune Complex Assay for p42 <sup>mapk</sup> Activation	90
2.2.7	.2.	p42 <sup>mapk</sup> Bandshift	90
2.2.8.	Ra	f-1 Kinase Assay	91
2.2.8	.1.	Preparation of Substrates	91
2.2.8	.2.	Raf-1 Kinase Assay	91
2.2.9.	GT	P Loading of Ras	92
2.2.10.	An	alysis of Tyrosine Phosphorylated Proteins	93
2.2.1	0.1	Tyrosine Phosphorylation of Multiple Proteins	93
2.2.1	0.2	Tyrosine Phosphorylation of FAK	93
2.2.1	0.3	Tyrosine Phosphorylation of Paxillin	93
2.2.11.	Me	asurements of cellular proliferation	94
2.2.1	1.1	. [ <sup>3</sup> H]Thymidine Incorporation Assay	94
2.2.1	1.2	. Autoradiography of Labelled Nuclei	95
2.2.1	1.3	. Measurement of cell proliferation	95
2.2.12.	Pro	etreatments	95

# Chapter 3 – Signalling Pathways Coupled to The Bombesin/GRP Receptor 97

3.1. Bindi	ng Characteristics of the Expressed Bombesin/GRP Receptor97
3.1.1.	Expression of the bombesin/GRP receptor in Rat-1 cells
3.1.2.	Timecourse of [125I]GRP Binding to Transfected Rat-1 Cells98
3.1.3.	Ligand Recognition profile of the Transfected Bombesin/GRP
	Receptor98
3.2. The 7	Transfected Bombesin/GRP Receptor Functionally Couples to
PLC-A	Activated Signalling Pathways103
3.2.1.	Bombesin Stimulates Intracellular Ca <sup>2+</sup> Mobilisation
3.2.2.	Bombesin stimulates 80K/MARCKS Phosphorylation 103
3.3. Expre	ession of the bombesin/GRP receptor confers a growth response
to bo	mbesin in Rat-1 cells
3.3.1.	Bombesin Stimulates DNA Synthesis

3.3.	2. Bombesin Induces Cell Proliferation
3.4. Be	mbesin Stimulates Tyrosine Phosphorylation in Rat-1 cells
T	ansfected with the Bombesin/GRP Receptor111
3.4.	1. Bombesin induces tyrosine phosphorylation of multiple proteins
	in BOR 5 and BOR 15 cells 111
3.4.	2. Bombesin-Induced Tyrosine Phosphorylation is inhibited by the
	specific bombesin receptor antagonist116
3.4.	3. Bombesin-Induced Tyrosine Phosphorylation is Dependent on the
	Integrity of the Actin Cytoskeleton116
3.4.	4. Tyrosine Phosphorylation induced by Bombesin is independent
	of PKC120
3.4.	5. Bombesin stimulates tyrosine phosphorylation of FAK and
	paxillin in BOR 5 and BOR 15 cells 120
3.5. Sı	mmary and Discussion
3.5.	Multiple Coupling from one Receptor         123
3.5.	2. Multiple Coupling via βγ Subunits
3.5.	3. Multiple coupling via a distinct Ga subunit
Chanter	A = The MADK Dethway Stimulated By
Chapter	$\mathbf{T} = \mathbf{I} \mathbf{I} \mathbf{C} \mathbf{M} \mathbf{A} \mathbf{I} \mathbf{A} $
<b>1</b>	Rombesin in Rat-1 Cells 130
	Bombesin in Rat-1 Cells 130
4.1. Bo	Bombesin in Rat-1 Cells 130 ombesin Stimulation of MAPK activity in Rat-1 Cells Transfected
4.1. Bo w	Bombesin in Rat-1 Cells 130 ombesin Stimulation of MAPK activity in Rat-1 Cells Transfected ith the Bombesin/GRP Receptor
4.1. Bo w 4.1.	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptor1.Bombesin Stimulates MAPK activity130
4.1. Bo w 4.1. 4.1.	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptor1301. Bombesin Stimulates MAPK activity1302. Dose Response of Bombesin-Induced MAPK activity in BOR 15
<b>4.1.</b> Bo w 4.1. 4.1.	Bombesin in Rat-1 Cells       130         Imbesin Stimulation of MAPK activity in Rat-1 Cells Transfected       130         In Bombesin/GRP Receptor       130         I. Bombesin Stimulates MAPK activity       131
<b>4.1. B0</b> <b>w</b> 4.1. 4.1. 4.1.	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptor1301.Bombesin Stimulates MAPK activity1302.Dose Response of Bombesin-Induced MAPK activity in BOR 15Cells1313.Bombesin-stimulated MAPK activity is independent of PKC in
<b>4.1. Bo</b> <b>w</b> 4.1. 4.1. 4.1.	Bombesin in Rat-1 Cells       130         Imbesin Stimulation of MAPK activity in Rat-1 Cells Transfected       130         In Bombesin/GRP Receptor       130         In Bombesin Stimulates MAPK activity       130         In Bose Response of Bombesin-Induced MAPK activity in BOR 15       131         In Bombesin-stimulated MAPK activity is independent of PKC in BOR 15       131
<ul> <li>4.1. Bo</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Bo</li> </ul>	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptor1301.Bombesin Stimulates MAPK activity1302.Dose Response of Bombesin-Induced MAPK activity in BOR 15Cells1313.Bombesin-stimulated MAPK activity is independent of PKC in BOR 15BOR 15131ombesin Stimulation of Raf-1 Activity in Transfected Rat-1 cells135
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> </ul>	Bombesin in Rat-1 Cells       130         Imbesin Stimulation of MAPK activity in Rat-1 Cells Transfected       130         In Bombesin/GRP Receptor       130         In Bombesin Stimulates MAPK activity       131         In Bombesin Stimulated MAPK activity is independent of PKC in BOR 15       131         In Bombesin Stimulation of Raf-1 Activity in Transfected Rat-1 cells       135         In Bombesin Stimulates Raf-1 activation in BOR 15 Cells       135
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> </ul>	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptorith the Bombesin/GRP Receptor1301.Bombesin Stimulates MAPK activity1302.Dose Response of Bombesin-Induced MAPK activity in BOR 15CellsCells1313.Bombesin-stimulated MAPK activity is independent of PKC in BOR 15BOR 15131ombesin Stimulation of Raf-1 Activity in Transfected Rat-1 cells1351.Bombesin Stimulates Raf-1 activation in BOR 15 Cells1352.Bombesin-Induced Raf-1 Activation is not Dependent on PKC
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> <li>4.2.</li> </ul>	Bombesin in Rat-1 Cells       130         Imbesin Stimulation of MAPK activity in Rat-1 Cells Transfected       130         Ith the Bombesin/GRP Receptor       130         I. Bombesin Stimulates MAPK activity       130         2. Dose Response of Bombesin-Induced MAPK activity in BOR 15       131         3. Bombesin-stimulated MAPK activity is independent of PKC in BOR 15       131         3. Bombesin Stimulates Raf-1 Activity in Transfected Rat-1 cells       135         1. Bombesin Stimulates Raf-1 activation in BOR 15 Cells       135         2. Bombesin-Induced Raf-1 Activity is not Dependent on PKC       135         3. Pertussis Toxin does not Inhibit Bombesin-Induced Raf-1       135
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> <li>4.2.</li> </ul>	Bombesin in Rat-1 Cells130mbesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptor1301.Bombesin Stimulates MAPK activity1302.Dose Response of Bombesin-Induced MAPK activity in BOR 15CellsCells1313.Bombesin-stimulated MAPK activity is independent of PKC in BOR 15BOR 15131Bombesin Stimulates Raf-1 Activity in Transfected Rat-1 cells1352.Bombesin Stimulates Raf-1 activation in BOR 15 Cells2.Bombesin-Induced Raf-1 Activity is not Dependent on PKC3.Pertussis Toxin does not Inhibit Bombesin-Induced Raf-1 ActivityActivityActivity
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.3. Be</li> </ul>	Bombesin in Rat-1 Cells       130         Imbesin Stimulation of MAPK activity in Rat-1 Cells Transfected       130         In Bombesin/GRP Receptor       130         In Bombesin Stimulates MAPK activity       130         In Bombesin Stimulates MAPK activity in BOR 15       131         In Bombesin-stimulated MAPK activity is independent of PKC in BOR 15       131         In Bombesin Stimulates Raf-1 Activity in Transfected Rat-1 cells       135         In Bombesin Stimulates Raf-1 activation in BOR 15 Cells       135         In Bombesin Stimulates Raf-1 Activation is not Dependent on PKC       135         In Bombesin Stimulates Raf-1 Activation is not Dependent on PKC       135         In Bombesin Stimulates Raf-1 Activation is not Dependent on PKC       136         In Hibbit Bombesin-Induced Raf-1       138         In Hibbit Bombesin Stimulation of Ras.GTP Loading in Transfected Rat-1 Cells.       138
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.3. Be 4.4. Su</li> </ul>	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfectedith the Bombesin/GRP Receptor1301.Bombesin Stimulates MAPK activity1302.Dose Response of Bombesin-Induced MAPK activity in BOR 15CellsCells1313.Bombesin-stimulated MAPK activity is independent of PKC in BOR 15BOR 15131ombesin Stimulation of Raf-1 Activity in Transfected Rat-1 cells1352.Bombesin-Induced Raf-1 activation in BOR 15 Cells1353.Pertussis Toxin does not Inhibit Bombesin-Induced Raf-1 Activity138ombesin Stimulation of Ras.GTP Loading in Transfected Rat-1 Cells.138mmary and Discussion
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> <li>4.4.</li> </ul>	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfected130ith the Bombesin/GRP Receptor
<ul> <li>4.1. Be w</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.1.</li> <li>4.2. Be 4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.2.</li> <li>4.3. Be 4.4.</li> <li>4.4.</li> </ul>	Bombesin in Rat-1 Cells130ombesin Stimulation of MAPK activity in Rat-1 Cells Transfected130ith the Bombesin/GRP Receptor130i.Bombesin Stimulates MAPK activity1302.Dose Response of Bombesin-Induced MAPK activity in BOR 151313.Bombesin-stimulated MAPK activity is independent of PKC in BOR 151313.Bombesin Stimulates Raf-1 Activity in Transfected Rat-1 cells1351.Bombesin Stimulates Raf-1 activation in BOR 15 Cells1352.Bombesin-Induced Raf-1 Activity in Transfected Rat-1 cells1353.Pertussis Toxin does not Inhibit Bombesin-Induced Raf-1 Activity138ombesin Stimulation of Ras.GTP Loading in Transfected Rat-1 Cells.138mmary and Discussion1422.Bombesin activation of Ras. Mediation by Gβγ?143

# Chapter 5 - The Calcium Signal

5.1. Effec	t of Thapsigargin and DBHQ on PDB-Stimulated DNA Synthe	sis 146
5.1.1.	Thapsigargin and DBHQ Stimulate [ <sup>3</sup> H]Thymidine Incorporat	tion
	in synergy with PDB	146
5.1.2.	Kinetics of Thapsigargin and DBHQ Stimulated [ <sup>3</sup> H]thymidin	e
	Incorporation is Consistent with Entry to S-Phase	150
5.1.3.	Thapsigargin and DBHQ increase the Proportion of Cells tha	t
	Reinitiate DNA Synthesis	150
5.2. Effec	t of DBHQ with other mitogens	153
5.2.1.	Sub-maximal Concentrations of Bombesin Synergise with	
	Thapsigargin and DBHQ	154
5.3. Exan	nination of the Synergism Between PKC and Thapsigargi	n or
DBH	Q	154
5.3.1.	The Synergistic Stimulation of DNA Synthesis Requires	
	Functional PKC	154
5.3.2.	Thapsigargin and DBHQ do not potentiate PDB-mediated sig	nal
	transduction	158
5.4. Effec	t of Thapsigargin and DBHQ on intracellular calcium stores	161
5.4.1.	Thapsigargin and DBHQ inhibit Bombesin-Induced Ca <sup>2+</sup>	
	Mobilisation	161
5.4.2.	Kinetics of Thapsigargin and DBHQ inhibition of Bombesin-	
	Induced Ca <sup>2+</sup> Mobilisation	161
5.4.3.	Thapsigargin and DBHQ do not Prevent Bombesin-Induced	
	Inositol Phosphate Accumulation	165
5.4.4.	Thapsigargin and DBHQ cause Prolonged Depletion of Total	
	Cellular Ca <sup>2+</sup>	165
5.5. Effec	t of extracellular calcium on DNA synthesis	167
5.6. Sumr	nary and Discussion	172
5.6.1.	Capacitative Ca <sup>2+</sup> Entry	172
5.6.2.	Synergy with PKC	173
5.6.3.	Cytotoxic effects	174
5.6.4.	Possible Pathways Activated	175
Chapter 6	- SUMMARY AND PERSPECTIVES	178
REFERENC	`FS	182
IVELLIVEINC		103

PUBLICATIONS	225

# LIST OF FIGURES

Figure 1.1.	Diagram of PDGF receptor with sites of interaction with signal
	transduction molecules
Figure 1.2.	Schematic diagram of a seven transmembrane G-protein-coupled
	receptor
Figure 1.3.	Probable arrangement of helices in seven transmembrane domain
	receptors
Figure 1.4.	G protein cycle
Figure 1.5.	Amino acid sequence of mammalian and amphibian bombesin-like
	peptides
Figure 1.6.	Bombesin-stimulated signal transduction pathways
Figure 1.7.	Calcium Homeostasis and Release
Figure 1.8.	Structure of PKC isozymes
Figure 1.9.	Linear diagram of paxillin structure
Figure 1.10.	Receptor tyrosine kinase activation of the MAPK pathway71
Figure 1.11.	Mechanism of MAPK activation by G protein coupled receptors
Figure 1.12.	Multiple mammalian MAPK modules77
Figure 1.13.	Activation of AP-1 transcription factor
Figure 1.14.	Possible mechanisms of multiple coupling for bombesin receptors 81
Figure 3.1.	Specific [125I]GRP binding to different clones
Figure 3.2.	Binding of [ <sup>125</sup> I]-GRP as a function of GRP concentration100
Figure 3.3.	Time-course of [125I]GRP binding to BOR 5 cells101
Figure 3.4.	Competition Binding of [125]GRP to BOR 5 cells
Figure 3.5.	Bombesin-induced increase in $[Ca^{2+}]_{cyt}$ in BOR 5 cells 104
Figure 3.6.	Antagonist inhibition of bombes in-induced increase in $[{\rm Ca}^{2+}]_{cyt}105$
Figure 3.7.	Bombesin-induced phosphorylation of 80K/MARCKS 106
Figure 3.8.	Kinetics of [ <sup>3</sup> H]thymidine incorporation induced by bombesin108
Figure 3.9.	Effect of antagonist on bombesin-induced [ <sup>3</sup> H]thymidine
	incorporation109
Figure 3.10.	Cell proliferation induced by bombesin in BOR 5 cells110
Figure 3.11.	Bombesin-induced tyrosine phosphorylation of multiple proteins
	(I)
Figure 3.12.	Bombesin-induced tyrosine phosphorylation of multiple proteins
	(II)
Figure 3.13.	Kinetics of bombesin induction of tyrosine phosphorylation of
	multiple proteins (I)

Figure 3.14.	Kinetics of bombesin induction of tyrosine phosphorylation of
	multiple proteins (II)
Figure 3.15.	Effect of antagonist on bombesin-induced tyrosine
	phosphorylation of multiple proteins117
Figure 3.16.	Effect of Cytochalasin D on bombesin-induced tyrosine
	phosphorylation of multiple proteins
Figure 3.17.	Effect of GF 109203X on bombesin-induced tyrosine
	phosphorylation of multiple proteins
Figure 3.18.	Bombesin-induced tyrosine phosphorylation of FAK and Paxillin
	(I)
Figure 3.19.	Bombesin-induced tyrosine phosphorylation of FAK and Paxillin
	(II)
Figure 3.20.	Different areas in the 3rd cytoplasmic loop of $\alpha_{2A}$ -AR-c10 are
	required for coupling to $G\alpha_{14}$ vs $G\alpha_{q/11}$
Figure 4.1.	p42 <sup>mapk</sup> activation induced by various factors
Figure 4.2.	Bombesin-induced increase in p42 <sup>mapk</sup> activity133
Figure 4.3.	Effect of PKC inhibition and down regulation on bombesin-
	stimulated MAPK activity134
Figure 4.4.	Time course of bombesin-induced Raf-1 activity
Figure 4.5.	Effect of PKC inhibition on Raf-1 activity induced by various
	mitogens
Figure 4.6.	Effect of pertussis toxin on Raf-1 activity induced by various
	mitogens
Figure 4.7.	Bombesin stimulation of loading (I)
Figure 4.8.	Bombesin stimulation of Ras.GTP loading (II)141
Figure 5.1.	Dose response curve for the stimulation of DNA synthesis by
	DBHQ or thapsigargin147
Figure 5.2.	Effect of the length of preincubation with DBHQ and thapsigargin
	on the subsequent stimulation of DNA synthesis by these
	compounds in synergy with PDB148
Figure 5.3.	Dose response curve for the stimulation of DNA synthesis by PDB 149
Figure 5.4.	Time course for incorporation of [ <sup>3</sup> H]thymidine in the presence of
	thapsigargin and DBHQ151
Figure 5.5.	Effect of thapsigargin and DBHQ on DNA synthesis assessed by
	autoradiography and liquid scintillation counting of
	[ <sup>3</sup> H]thymidine incorporation152
Figure 5.6.	Effect of DBHQ on DNA synthesis induced by bombesin assessed
	both by autoradiography and liquid scintillation counting of
	[ <sup>3</sup> H]thymidine incorporation as comparison

Figure 5.7.	Effect of down regulating PKC on [ <sup>3</sup> H]thymidine incorporation	
	induced by DBHQ and either PDB or bombesin	156
Figure 5.8.	Effect of inhibiting PKC on [ <sup>3</sup> H]thymidine incorporation induced	
	by DBHQ and either PDB or bombesin	157
Figure 5.9.	Effect of thapsigargin and DBHQ on PDB-mediated signal	
	transduction (I)	159
Figure 5.10.	Effect of thapsigargin and DBHQ on PDB-mediated signal	
	transduction (II)	160
Figure 5.11.	Reduction of bombesin induced Ca <sup>2+</sup> mobilisation by thapsigargin	
	and DBHQ	162
Figure 5.12.	The mean increase in [Ca <sup>2+</sup> ] <sub>cyt</sub> induced by bombesin after	
	pretreatment with thapsigargin or DBHQ	163
Figure 5.13.	Time course of depletion of bombesin induced calcium response	
	by thapsigargin and DBHQ	164
Figure 5.14.	Effect of Thapsigargin and DBHQ on total intracellular calcium	
	measured by <sup>45</sup> Ca <sup>2+</sup>	166
Figure 5.15.	Effect of extracellular calcium on [ <sup>3</sup> H]thymidine incorporation	
	induced by different mitogens	169
Figure 5.16.	Effect of econozole on [ <sup>3</sup> H]thymidine incorporation induced by	
	different mitogens	170
Figure 5.17.	Effect of extracellular calcium on [ <sup>3</sup> H]thymidine incorporation	
	induced by bombesin	171
Figure 6.1.	The bombesin/GRP receptor couples to multiple signal	
	transduction pathways	182

# LIST OF TABLES

Table 1.I.	Stimulation of Swiss 3T3 fibroblast DNA synthesis by various factors 20
Table 1.II.	Examples of G protein coupled receptors and their heterogeneity
Table 1.III	Examples of heterotrimeric G protein subunits
Table 1.IV.	Function of βγ subunits
Table 1.V.	PKC isoforms in mammalian tissues
Table 3.I.	Seven transmembrane receptors that couple to multiple G proteins 126
Table 3.II.	Comparison of receptor-ligand $EC_{50}$ values between different $G\alpha$
	mediated signals127
Table 3.III.	Comparison of $EC_{50}$ values of cellular effects induced by bombesin 128
Table 5.I.	Effect of DBHQ on DNA synthesis induced by various mitogens
Table 5.II.	Effect of thapsigargin on total inositol phosphates generated by
	various mitogens

## **ABBREVIATIONS**

5-HT	5-hydroxytryptamine (serotonin)
80K/MARCKS	80K/myristoylated alanine rich C-kinase substrate
AA	arachidonic acid
Ach	acetylcholine
ADP	adenosine diphosphate
aPKC	atypical protein kinase C
AR	adrenergic receptor e.g. $\alpha_1$ -AR : $\alpha_1$ adrenergic receptor
ATP	adenosine triphosphate
βARK	β-adrenergic receptor kinase
BSA	bovine serum albumin
[Ca <sup>2+</sup> ] <sub>cyt</sub>	concentration of cytoplasmic Ca <sup>2+</sup>
[Ca <sup>2+</sup> ] <sub>out</sub>	concentration of extracellular Ca <sup>2+</sup>
CaM	calmodulin
СаМК	calmodulin kinase
cAMP	cyclic adensoine 5' monophosphate
CIF	calcium influx factor
сРКС	classical/conventional protein kinase C
CRE	cAMP response element
CREB	cAMP response element binding protein
CTX	cholera toxin
DBHQ	di- <i>tert</i> -butylhydroquinone
D-F <sub>5</sub> -Phe <sup>6</sup> , D-Ala <sup>11</sup> -	D-F5-Phe-Gln-Trp-Ala-Val-D-Ala-His-Leu-OMe
Bombesin(6-13)OMe	
DMEM	Dulbecco's modified Eagle's medium
DMSO	dimethylsulphoxide
DTT	dithiothreitol
EC <sub>50</sub>	concentration that gives half maximum effect
ED <sub>50</sub>	dose that gives half maximium effect
EDTA	ethylenediamine tetraacetic acid
EGF	epidermal growth factor
EGTA	ethylene glycol-bis[β-aminoethyl ether]-N,N,N',N',-
	tetraacetic acid
ER	endoplasmic reticulum
ERK	extracellular signal regulated protein kinase
FAK	focal adhesion kinase
FBS	foetal bovine serum

-	FGF	fibroblast growth factor
	FDGF	fibroblast-derived growth factor
	Fura-2/AME	fura-2 acetoxymethylester
	G protein	guanine nucleotide binding protein
	GAP	GTPase activating protein
	GDP	guanosine diphosphate
	GEF	guanine nucleotide exchange factor
	GPCR	G protein-coupled receptor
	GRP	gastrin releasing peptide
	GST	glutathione-S-transferase
	GTP	guanosine triphosphate
	HEPES	N-[2-hydroxyethyl]piperazine-N'-[2ethanesulfonic acid]
	IBMX	isobutylmethylxanthine
	IC <sub>50</sub>	concentration that gives half maximum inhibition
	Ins	inositol
	Ins(1,3,4,5)P <sub>4</sub>	inositol 1,3,4,5-tetrakisphosphate
	$Ins(1,4)P_2$	inositol 1,4-bisphosphate
	Ins(1,4,5)P <sub>3</sub>	inositol 1,4,5-trisphosphate
	IPTG	isopropylthio-β-D-galactoside
	M-Ach receptor	muscarinic acetylcholine receptor e.g. $M_1$ -Ach receptor : $M_1$
		muscarinc acetylcholine receptor
	mAb	monoclonal antibody
	MAP-2	microtubule associated protein 2
	MAPK	microtubule associated protein 2 kinase/mitogen activated
		protein kinase
	MBP	myelin basic protein
	MEK	mitogen activated protein kinase/extracellular signal
		regulated protein kinase kinase
	NGF	nerve growth factor
	NmB	neuromedin B
	NmC	neuromedin C
	nPKC	novel/new protein kinase C
	PBS	phosphate buffered saline
	PDB	phorbol 12,13-dibutyrate
	PDGF	platelet-derivedgrowth factor
	PH	pleckstrin homology
	РІЗК	phosphatidylinositol 3' kinase
	PIP <sub>2</sub>	phosphatidylinositol 4,5-bisphosphate
	PKA	protein kinase A

protein kinase C
protein kinase D
phospholipase A <sub>2</sub>
phospholipase C
phenylmethylsulfonyl fluoride
phosphatidyl serine
pertussis toxin
small cell lung cancer
sodium dodecyl sulphate
SDS-polyacrylamide gel electrophoresis
standard error
sarcoplasmic or endoplasmic reticulum Ca <sup>2+</sup> ATPase
sis inducible element
son of sevenless
serum response element
src homology 2
src homology 3
thapsigargin
tetradecanoyl phorbol acetate
thyroid stimulating hormone
phosphotyrosyl residue

# Chapter 1 INTRODUCTION

## **1.1. GROWTH REGULATION**

Multicellular organisms have developed highly efficient regulatory networks to control cell proliferation. These involve cellular interactions with positive and negative diffusible modulators as well as with the extracellular matrix proteins. In fully mature organisms, the cells of many tissues and organs are maintained in a non-proliferating state (the  $G_0/G_1$  phase of the cell cycle), but can be stimulated to resume DNA synthesis and cell division in response to external stimuli such as hormones, antigens or growth factors. In this manner the growth of individual cells is regulated according to the needs of the whole organism. The regulation of normal cell proliferation is therefore central to many physiological processes including embryogenesis, growth and development, selective cell survival, haemopoeisis, tissue repair and immune responses.

It has become evident that cultured cancer cells which are characterised by unrestrained proliferation acquire complete or partial independence of mitogenic signals in the extracellular environment through different mechanisms (Cross and Dexter, 1991; Westermark and Heldin, 1991). These include production of growth factors that act on the same cells that produced them (autocrine loop) or on adjacent cells (paracrine communication), alterations in the number or structure of cellular receptors and changes in the activity of post-receptor signalling pathways that either stimulate or suppress cell growth (Sager, 1989; Bishop, 1991). For these reasons the identification of the extracellular factors which modulate cell proliferation and the elucidation of the molecular mechanisms involved have emerged as fundamental problems in cancer biology.

The observation that animal cells duplicate their DNA during a discrete interval between cell divisions allowed the cell cycle to be divided into four classical phases:  $G_1$ (growth phase), S (DNA synthetic period),  $G_2$  and M (mitosis).  $G_1$  is the gap between mitosis and the initiation of DNA synthesis, and G2 is the period between S and M (reviewed by Baserga 1976 and Pardee *et al.*, 1978). It was found that, in most cases, cells respond to sub-optimal growth conditions by reversibly arresting in the  $G_1$  or  $G_0/G_1$ ) phase of the cell cycle (Pardee, 1974; Baserga, 1976). This arrest is observed in normal fibroblasts in culture (Pardee, 1974). Since the crucial control events for the regulation of growth seem to reside in  $G_1$  (Pardee, *et al.*, 1978; Pardee, 1989), elucidation of the mechanisms leading to restimulation of DNA synthesis is therefore a necessary step in the understanding of the control of cell proliferation *in vivo*.

It became evident from the first attempts to propagate animal cells *in vitro* that serum was an essential requirement for the proliferation of most cells. Cells ceased to

proliferate and became quiescent in  $G_1/G_0$  when they depleted serum of its growth promoting activity. Furthermore, addition of fresh serum caused the cells to reinitiate a program of RNA and protein synthesis, DNA synthesis and cell division (reviewed in Pardee, *et al.*, 1978). Many virally and chemically transformed cells were found to have either partially or wholly lost the ability to arrest in  $G_1$  and exhibited a decreased dependence on serum for proliferation. These findings led to the hypothesis that animal sera contained "growth factors" that controlled cellular proliferation (Wolstenholme and Knight, 1971; Temin, *et al.*, 1972; Pardee and Rozengurt, 1975).

#### **1.1.1. GROWTH FACTORS**

Within the last decade rapid progress has been made toward the elucidation of the mechanisms of action of the diverse extracellular factors that control the growth of cells. The availability of cell culture in nutrient media together with the purification of various polypeptide growth factors (reviewed in Rozengurt, 1980; Rozengurt and Collins, 1983; James and Bradshaw, 1984) allowed the investigation of mitogen action under chemically defined conditions (Bottenstein, *et al.*, 1979; Shipley and Ham, 1981). From this work it has become clear that a variety of mitogenic factors can regulate the proliferation of normal cells. In this respect the Swiss 3T3 fibroblast has proved a particularly useful system for identification of growth promoting factors (Rozengurt, 1980; Rozengurt, 1986).

Swiss 3T3 cells are a murine fibroblast cell line that was established by repeated sub-culture of disaggregated embryonic cells (Todaro and Green, 1963). Cultures of this 3T3 cell line become "quiescent" in the  $G_1$  to  $G_0$  phase of the cell cycle when the cultures reach confluence at a saturation density determined by the serum concentration (Holley, 1975). This arrest appears therefore to be due to depletion of growth promoting activity in the serum. Readdition of fresh serum to quiescent cultures of these cells stimulates cellular metabolism and after a variable lag period (10 - 15 hours) initiation of DNA synthesis.

The use of Swiss 3T3 cells in culture has a number of advantages for the study of growth control. Identical monolayer cultures of a clonal population of cells can be easily and consistently produced, allowing for reproducibility of results. In addition a chemically-defined nutrient media has been devised for these cells that allows large and reproducible increases in DNA synthesis to be monitored in the presence of growth promoting agents. Also, since most cells are arrested in the same phase of the cell cycle ( $G_1/G_0$ ), on restimulation they may be regarded as a synchronised cell population. It should be noted, however, that established cell lines often display abnormal karyotypes and therefore cannot be directly correlated with any cell type *in vivo*. In this respect, the 3T3 cell can be considered as a general model for growth control, but results obtained form this cell line necessarily require further investigation to determine their relevance *in vivo*. However, many mechanisms of growth control identified initially in 3T3 cells have been demonstrated to be of general importance. For example, monovalent ion transport across the plasma membrane (Rozengurt and Heppel, 1975) has since been accepted as a universally important part of growth factor action (Moolenaar, *et al.*, 1983).

The 3T3 cell system has proved particularly useful in two major areas of growth control research. The first is in the identification of agents that modulate the growth state of the cell, and their purification from various biological sources. The second is the elucidation of the mechanisms by which serum, purified growth factors and other mitogens initiate DNA synthesis. Attention has been focused on the early signalling events in mitogenesis as the initial steps in the proliferative response.

It is now recognised that the proliferation of 3T3 cells may be regulated by several classes of mitogen (table 1.I). These include members of a large family of polypeptide growth factors such as PDGF, EGF and insulin-like growth factors and members of a family of small regulatory peptides that classically behave as local hormones or fast acting neurotransmitters. In addition various pharmacological agents including phorbol esters and synthetic diacylglycerols can act as mitogens for these cells (table 1.I).

Both the polypeptide growth factors and the neuropeptides stimulate their growth promoting effects by binding to specific receptors located on the cell surface and activating multiple signalling pathways. At least two major signal transduction pathways initiate the cascades of molecular events stimulated by these two classes of mitogens. The polypeptide growth factors e.g. PDGF and EGF bind to receptors with intrinsic tyrosine kinase activity while the neuropeptides act through receptors coupled to heterotrimeric guanine nucleotide binding proteins (G proteins) to effector activation.

Some experiments presented in this thesis were performed on a 3T3-like cell line, Rat-1 fibroblasts. These cells also retain the ability to quiesce and to reinitiate DNA synthesis when stimulated with growth promoting agents (Higgins, *et al.*, 1992). Because Rat-1 fibroblasts do not respond to as many growth factors as Swiss 3T3 cells, this makes them an ideal candidate for the investigation of exogenous expression of defined receptors.

Class of Growth Factor	Name of Factor	Reference
Polypeptide growth factors	Epidermal Growth Factor Insulin-like Growth Factors Platelet Derived Growth Factor	Carpenter and Cohen, 1979 Froesch, <i>et al.</i> , 1979 Rozengurt, <i>et al.</i> , 1983c; Heldin and Westermark, 1984; Lopez- Rivas, <i>et al.</i> , 1984
Neuropeptides	Adrenomedullin Bombesin	Withers, <i>et al.</i> , 1996 Rozengurt and Sinnett-Smith, 1983
	Bradykinin Endothelin	Woll and Rozengurt, 1988b Brown and Littlewood, 1989; Takuwa. <i>et al.</i> , 1989
	Vasopressin	Rozengurt, <i>et al.</i> , 1979; Dicker and Rozengurt, 1980; Rozengurt and Mendoza, 1980;
	Vasoactive Intestinal	Fabregat and Rozengurt, 1980
	Contractor Vasoactive Intestinal Peptide	Zurier, <i>et al.</i> , 1988
Activators of Protein kinase C	Phorbol esters, teleocidin, 1-Oleoyl-2-acetylglycerol	Dicker and Rozengurt, 1978; Dicker and Rozengurt, 1980; Dicker and Rozengurt, 1981; Collins and Rozengurt, 1982a; Collins and Rozengurt, 1982b; Rozengurt, <i>et al.</i> , 1984
Cyclic nucleotide elevating agents	Cholera toxin, Adenosine agonists, cAMP derivatives, Forskolin, IBMX, Prostaglandin E <sub>1</sub>	Rozengurt, 1981; Rozengurt, <i>et al.</i> , 1981b; Rozengurt, 1982b; Rozengurt, 1982a; Rozengurt, <i>et al.</i> , 1983a; Rozengurt, <i>et al.</i> , 1983a
Microtubule disrupting agents	Colchicine, Colemid, Nocodazole, Podolotoxin, Vinblastine	Friedkin and Rozengurt, 1981; Wang and Rozengurt, 1983
Vitamin A derivatives	Retinoic acid	Dicker and Rozengurt, 1979; Dicker and Rozengurt, 1980)
Permeability modulators	Mellitin, Mastoparan	Gelehrter and Rozengurt, 1980; Gil, <i>et al.</i> , 1991
Bacterial Toxins	Pasteurella multocida toxin	Rozengurt. <i>et al.</i> , 1990

#### Table 1.I. Stimulation of Swiss 3T3 fibroblast DNA synthesis by various factors

#### **1.1.2.** Synergistic Signalling Pathways

The use of Swiss 3T3 cells as a model for the investigation of growth factors has revealed the fundamental concept of synergistic signal transduction pathways. Defined mitogens such as those summarised in table 1.I all exhibit potent synergistic interactions when added to cell cultures maintained in medium devoid of serum (reviewed in Rozengurt, 1986). In contrast to many other mitogens PDGF and bombesin stimulate DNA synthesis and cell division in the absence of any other synergistic factor. Studies with PDGF identified that this growth factor activates PKC (Rozengurt, et al., 1983b) and causes elevation of cAMP (Rozengurt, et al., 1983c). It was predicted that if an agent elicits the same set of early events as PDGF it should act as a growth factor for Swiss 3T3 cells in medium devoid of serum or other mitogens. Agents that elicit part of the early responses stimulated by PDGF should become mitogenic when added in combinations that reconstitute the other early events (Rozengurt, 1986). The availability of a panel of defined mitogens that are biologically active (see table 1.I) provides a tool for elucidating the nature of the regulatory signals and molecular events implicated in these synergistic effects (reviewed in Rozengurt, 1986). For example, a group of agents such as phorbol esters, diacylglycerol and vasopressin elicit a common set of early events; namely they activate PKC but do not alter the basal level of cAMP. Addition of any of these agents either individually or in combination to Swiss 3T3 cells fails to induce a mitogenic response. Agents that increase intracellular cAMP such as prostaglandin  $E_1$ , cholera toxin and cAMP derivatives do not activate PKC and do not stimulate DNA synthesis when added singly or in combination. Therefore agents that share a common signalling system cannot act synergistically to stimulate initiation of DNA synthesis.

Crucially the agents mentioned above become potent mitogens when added to quiescent Swiss 3T3 cells in combinations that elicit the generation of both types of signals and thereby reconstitute the complex pattern of signalling events elicited by PDGF. Insulin which can synergise with both groups of extracellular factors at supramaximal concentrations (that is in lieu of insulin-like growth factor) does not act identically to either group of agents. In fact this hormone does not activate PKC, increase intracellular Ca<sup>2+</sup> or increase the level of cAMP in intact 3T3 cells (Rozengurt, 1986). Similar conclusions can be drawn when EGF is added instead of insulin.

It is important to note at this point that elevation of cAMP and activation of PKC are not the only early signals induced by PDGF and other growth factors and this thesis will describe more recent signalling pathways elicited by G protein-coupled receptors. However the main principle of synergistic pathways leading to mitogenesis, which is illustrated above by PKC and cAMP, remains relevant.

#### **1.1.3. REGULATORY SIGNALS AND OBLIGATORY EVENTS**

In line with the hypothesis of growth control discussed above the key events elicited by growth factors in quiescent cells can be broadly divided into two major categories: regulatory signals and obligatory events (Rozengurt, 1986). The former class represent intracellular processes that mediate the action of specific growthpromoting agents; although they are crucial in eliciting biological responses by a given factor they can be bypassed by another group of factors. In contrast, obligatory events are envisaged as molecular steps that must take place for the stimulation of DNA synthesis and cell division to occur regardless of the regulatory signals utilised to activate the cells. This distinction is particularly relevant to section 1.3.2.5. In order to distinguish between regulatory signals and obligatory events one approach is to ascertain the effect of a wide panel of defined mitogenic agents. While regulatory signals are elicited by certain mitogens but not others, obligatory events should be stimulated by all mitogenic combinations. This principle is utilised in section 5.5.

The next section illustrates the mechanism by which receptors with intrinsic tyrosine kinase activity can elicit multiple signals; this is due to their multiple docking sites for signal transduction molecules. The mechanism by which G protein-coupled receptors can elicit multiple signal transduction pathways is more elusive. This aspect of mitogenic signalling will be one of the major subjects of the present thesis. In particular, because bombesin, like PDGF, is a sole mitogen for Swiss 3T3 cells much attention has been given to elucidating its cellular mechanisms of activation. This thesis will describe signalling pathways mediated by the bombesin/GRP receptor which is a seven transmembrane, G protein coupled receptor. However, at points throughout this thesis G protein-coupled receptor signalling shall be compared and contrasted to pathways elicited by receptor tyrosine kinases. The salient points of receptors with intrinsic tyrosine kinase activity shall be briefly described first and then G protein-coupled receptors shall be described in more detail.

#### **1.1.4. Receptor Tyrosine Kinases**

Receptor tyrosine kinases bind large polypeptide growth factors such as PDGF and EGF. PDGF is a 30 kDa homo- or heterodimer of A or B chains (reviewed in Ross, *et al.*, 1986). It is not a widely circulating growth factor but it is released by platelets in response to injury and targets mesenchymal cells i.e. smooth muscle and fibroblasts to stimulate their proliferation and migration in to the wound. The role of

PDGF as a growth factor is highlighted by the v-*sis* oncogene from simian sarcoma virus which is homologous to the PDGF-B chain. The inappropriate production of v-*sis* by infected cells may result in autocrine or paracrine stimulation which could contribute to the tumourigenesis observed with this oncogene (Westermark and Heldin, 1991). Unlike PDGF, EGF is ubiquitous being found in nearly all body fluids and the EGF receptor is found in many cell types (Carpenter, 1985). The v-*erbB* oncogene from avian erythroblastosis virus is a truncated EGF receptor which lacks part of the EGF regulated extracellular domain (Downward, *et al.*, 1984).

Receptors for EGF and PDGF contain a large extracellular ligand binding domain, a single transmembrane region and a cytoplasmic portion with a conserved protein tyrosine kinase domain (reviewed in Fantl, *et al.*, 1993; Malarkey, *et al.*, 1995). Upon ligand binding the receptor chains dimerise and this activates their intrinsic protein tyrosine kinase activity. The active receptor autophosphorylates multiple sites by an intermolecular mechanism. The tyrosine residues that become phosphorylated serve as docking sites for multiple signal transduction molecules. This has been best characterised for the PDGF $\beta$  receptor. A series of mutations mapped the sites at which the different molecules interact with the intracellular portion of this receptor, see figure 1.1.

The PDGF receptor binds Src, p85, SHPTP-2 Grb2, GAP and PLC $\gamma$ . Some of these proteins have enzymatic activity such as PLC $\gamma$  and Src, whilst others appear to have no enzymatic activity but instead serve as linkers or adapters between receptor tyrosine kinases and enzymes, e.g. p85 and Grb2. It is possible that these associations simply serve to localise the signalling molecules to sites near their substrates e.g. phospholipids in membranes are hydrolysed by PLC $\gamma$  or phosphorylated by PI3K. In addition phosphorylation of substrates may modify the conformation and activity of these molecules (Malarkey, *et al.*, 1995). It has been shown that tyrosine phosphorylation of PLC $\gamma$  regulates its enzymatic activity (Lee and Rhee, 1995) and see section 1.3.1.

Receptor tyrosine kinases and their associated substrates use protein-protein interactions, mediated by distinct conserved domains, to build up signalling complexes. These protein modules (reviewed in Cohen, *et al.*, 1995) have homology with regions in the non-receptor tyrosine kinase, Src, i.e. Src homology 2 (SH2) and Src homology 3 (SH3) domains. SH2 and SH3 domains are true protein domains: they form compact units that maintain their structure in isolation and each has its N- and C-termini in close apposition so that they can be plugged into the surface of proteins. SH2 domains bind to phosphorylated tyrosine residues. There is some degree of specificity in this interaction which is determined by the three residues immediately carboxyl to the phosphotyrosine (Cohen, *et al.*, 1995). SH3 domains appear to recognise a stretch of about 10 amino acids that are proline rich. It is thought that

like SH2 domains there is specificity between proline rich sequences and SH3 domains determined by variable amino acids that surround the conserved motif (Cohen, *et al.*, 1995). The adaptor molecules Grb2 is composed of several of these modules, SH3—SH2—SH3 (Lowenstein, *et al.*, 1992).



# Figure 1.1. Diagram of PDGF receptor with sites of interaction with signal transduction molecules.

Based on Malarkey, *et al.*, (1995). For simplicity only one of the chains is shown. The phosphorylation of the receptor occurs upon ligand induced dimerisation (see text for details).

#### **1.1.5. G PROTEIN-COUPLED RECEPTORS**

This class of receptor responds to mitogenic neuropeptides, biogenic amines, glycoproteins and sensory stimuli such as light and odorants. These receptors constitute the classic hormone signalling pathway of receptor - G protein - effector - second messenger. A common theme among this class of receptor is the diversity of receptor subtypes for each ligand which has reached eleven for the biogenic amine, serotonin. Examples of G-protein diversity are shown in table 1.II.

Receptor/ligand	Subtype	predominant G protein coupling
muscarinic acetylcholine	M <sub>1</sub> , M <sub>3</sub> , M <sub>5</sub> M <sub>2</sub> , M <sub>4</sub>	G <sub>q/11</sub> G <sub>i/o</sub>
adenosine	A <sub>1</sub> , A <sub>3</sub> A <sub>2A</sub> , A <sub>2B</sub>	$G_{q/11}$ $G_{s}$
adrenaline/noradrenaline	$\alpha_{1A}, \alpha_{1B}, \alpha_{1D}$ $\alpha_{2A}, \alpha_{2B}, \alpha_{2C}$ $\beta_1, \beta_2, \beta_3$	G <sub>q/11</sub> G <sub>q/11</sub> G <sub>s</sub>
angiotensin	AT <sub>1</sub> AT <sub>2</sub>	G <sub>q/11</sub> ↓cGMP
bombesin	GRP (BB <sub>2</sub> ), NmB (BB <sub>1</sub> )	, BRS3 G <sub>q/11</sub>
bradykinin	B <sub>1</sub> , B <sub>2</sub>	G <sub>q/11</sub>
cholecystokinin/gastrin	CCK <sub>A</sub> , CCK <sub>B</sub>	G <sub>q/11</sub>
dopamine	D <sub>1</sub> , D <sub>5</sub> D <sub>2</sub> , D <sub>3</sub> , D <sub>4</sub>	G <sub>s</sub> G <sub>q/11</sub>
endothelin	$ET_A$ , $ET_B$	G <sub>q/11</sub>
glutamate	mGlu₁, mGlu₅ mGlu₂, mGlu₃, mGlu₄	, mGlu <sub>6</sub> , mGlu <sub>7</sub> $G_{q/11}$
histamine	H <sub>1</sub> H <sub>2</sub>	$G_{q/11}$ $G_{s}$
serotonin	5-HT <sub>1A</sub> , 5-HT <sub>1B</sub> , 5-HT 5-HT <sub>2A</sub> , 5-HT <sub>2B</sub> , 5-HT 5-HT <sub>4</sub> , 5-HT <sub>6</sub> , 5-HT <sub>7</sub>	$G_{q/11}$ $G_{q/11}$ $G_{q/11}$ $G_{q/11}$ $G_{q/11}$ $G_{q}$
prostanoid	DP, IP, EP <sub>2</sub> , EP <sub>4</sub> FP, TP, EP <sub>1</sub> , EP <sub>3</sub> EP <sub>3</sub>	G <sub>s</sub> G <sub>q/11</sub> (alternatively spliced) G <sub>i/o</sub>
somatostatin	sst <sub>1</sub> , sst <sub>2</sub> , sst <sub>3</sub> , sst <sub>4</sub> , s	st <sub>5</sub> G <sub>i/o</sub>

#### Table 1.II. Examples of G protein coupled receptors and their heterogeneity

NK1 - Substance P receptor, NK2 - Substance K receptor, NK3 - neurokinin B receptor. Table based on TIPS receptor and ion channel nomenclature supplement 1995.

#### 1.1.5.1. GROWTH REGULATION BY G PROTEIN-COUPLED RECEPTORS

Neuropeptides, which are ligands for this class of receptor, have been implicated in normal growth and development and are increasingly recognised as playing a role in sustaining the proliferation of cancer cells (Sethi, *et al.*, 1992). Studies in SCLC cells revealed that these cells have functional receptors for a variety of Ca<sup>2+</sup> mobilising neuropeptides such as bombesin, GRP, vasopressin, bradykinin, cholecystokinin, gastrin, galanin and neurotensin (Woll and Rozengurt, 1989; Sethi and Rozengurt, 1991a; Sethi and Rozengurt, 1992; Sethi , *et al.*, 1993). Furthermore, these neuropeptides stimulate clonal growth of SCLC cells in semisolid media (Sethi and

Rozengurt, 1991b; Sethi and Rozengurt, 1991a; Sethi and Rozengurt, 1992; Sethi , *et al.*, 1993). It is known that GRP, vasopressin, cholecystokinin and neurotensin are secreted by some SCLC tumours (for refs. see Sethi *et. al.* 1992). Other peptides may be released by a variety of normal cells in the lung, or like bradykinin, produced extracellularly as a result of the proteolytic cleavage of plasma precursors in the damaged tissue surrounding tumours (Steranka, *et al.*, 1989). Collectively these findings support the hypothesis that SCLC growth is sustained by an extensive network of autocrine and paracrine interactions involving multiple neuropeptides which signal through G protein-coupled receptors (Woll and Rozengurt, 1988a; Sethi, *et al.*, 1992).

In addition, receptors that are coupled to G proteins have been shown to transform fibroblasts *in vitro*. The serotonin receptor 5-HT<sub>1C</sub> was shown to be a conditional oncogene by having agonist-dependent transforming potential (Julius, *et al.*, 1989) and this was also shown for certain muscarinic receptors (Gutkind, *et al.*, 1991). In other studies mutated  $\alpha_{1B}$ -AR receptors which were constitutively activated and ligand independent were transforming (Allen, *et al.*, 1991) and activated mutants of the TSH receptor (Parma, *et al.*, 1993) were linked to thyroid hyperplasia. Thus the study of cellular events elicited by G protein-coupled receptors is important for understanding neoplastic growth.

#### 1.1.5.2. STRUCTURE

The main characteristic of G protein-coupled receptors is seven hydrophobic regions of approximately 23 - 28 amino acids and the periodic distribution of the hydrophobicity is consistent with an  $\alpha$ -helical conformation. These helices, which contain the areas of highest homology, are postulated to span the plasma membrane and are connected by hydrophilic extracellular and intracellular loops that are less well conserved (Strader, *et al.*, 1995), see figure 1.2.

The arrangement of the helices within the membrane is modelled upon the projection map of rhodopsin (Schertler, *et al.*, 1993) and the probable order deduced from the chemical and physical properties of conserved features of all G-protein linked receptors sequenced at that time (Baldwin, 1993).

The constraints established from detailed sequence analysis were that each helix must be positioned next to its neighbours in the sequence and helices I, IV and V must be the most exposed to lipid and III the least exposed. Helices IV - VII are proposed to be perpendicular to the plasma membrane although conserved prolines may kink them and helices I - III lay slanted in the membrane (Baldwin, 1993), see figure 1.3.



Figure 1.2. Schematic diagram of a seven transmembrane G-proteincoupled receptor



# Figure 1.3. Probable arrangement of helices in seven transmembrane domain receptors.

Viewed from the extracellular surface and modelled on rhodopsin projection map. The shaded area between helices I, II and III represents these helices lying at a slant in the membrane.

Biogenic amines bind within the IV - VII pocket. For peptide and protein ligands helix II and III, and the extracellular domains are used as well. For example the C-terminal of substance P binds in the IV - VII pocket and the N-terminal lays across the slanted helix II (Strader, *et al.*, 1995).

Mutational studies have implicated residues at the cytoplasmic ends of all helices and portions of the loops closest to the membrane as critical for overall interaction with the G protein and for specificity determination (Strader, *et al.*, 1994) and especially sequences in the 3rd and 4th intracellular domains. In fact short

synthetic peptides derived from the C-terminal of the 3rd cytoplasmic domain are capable of efficiently stimulating G-proteins *in vitro* (Lefkowitz, *et al.*, 1993). Thus the receptor must shield these peptides from the G protein when in an inactive, ligand free conformation.

The other main feature of G protein-coupled receptors is that they have no enzymatic activity themselves, unlike the receptor tyrosine kinases, but they are nucleotide exchange factors for heterotrimeric G proteins.

#### **1.1.6.** HETEROTRIMERIC G-PROTEINS

Seven transmembrane receptors are coupled to heterotrimeric G-proteins that are comprised of an  $\alpha$ , a  $\beta$  and a  $\gamma$  subunit. The  $\beta$  and  $\gamma$  subunits can only be separated upon denaturation and so they are generally considered a functional monomer. Molecular cloning has identified 20  $\alpha$  subunits, 5  $\beta$  subunits and 12  $\gamma$  subunits to date. See table 1.III for examples of diversity and tissue distribution.

#### 1.1.6.1. STRUCTURE

The structure of heterotrimeric G proteins has been reviewed by Neer (1995). G $\alpha$  are proteins of 39 - 50 kDa and consist of a GTPase domain and a largely  $\alpha$ -helical domain. Areas where the most conformational change occurs upon nucleotide exchange are called switch regions and this is where the  $\beta$  subunit binds to G $\alpha$ . The receptor binds to the C-terminal end of G $\alpha$ .

Gβ are approximately 35 kDa and are members of the family of WD-repeat proteins (Neer, *et al.*, 1994) characterised by 4 - 8 recurring highly conserved units ending with Trp-Asp (WD). In β subunits the length of the repeat is about 40 amino acids and so is sometimes known as the WD-40 repeat. Very recently the crystal structure of β has been described (Wall, *et al.*, 1995; Lambright, *et al.*, 1996; Sondek, *et al.*, 1996). The β subunit resembles a 7-bladed propeller with each blade made of 4 antiparallel strands in a β-sheet. The WD-40 repeat comprises the outer strand of one β-sheet together with the 3 inner strands of the next β-sheet. The most variable sequences in the repeats are found in the outer strand and this is consistent with the proposal that WD proteins build multiprotein complexes (Neer, *et al.*, 1994). At the Nterminal is an α-helix which forms a coiled coil with the N-terminal α-helix of the γ subunit.

 $G\gamma$  is small, about 6 - 9 kDa and shows the most heterogeneity of the G protein subunits and thus has been postulated to determine the specificity of  $\beta\gamma$  combinations (Neer, 1995). Not all  $\beta\gamma$  combinations can occur. For example  $\beta_1$  forms dimers with  $\gamma_1$  and  $\gamma_2$  but  $\beta_2$  does not form a dimer with  $\gamma_1$ . Also  $\beta_3$  does not dimerise with either  $\gamma_1$ 

or  $\gamma_2$  (Iniguez-Lluhi, *et al.*, 1992; Pronin and Gautam, 1992; Schmidt, *et al.*, 1992), and see section 1.1.6.4.

Subunit		Toxin	Distribution	Typical receptor	Effector		
G <sub>s</sub>	$lpha_{s(s)} \ lpha_{s(l)} \ lpha_{olf}$	CTX "	ubiquitous " olfactory neuroepithelium	β-AR, TSH " odorant	stimulate adenylyl cyclase " "		
Gi	$\begin{array}{c} \alpha_{i1} \\ \alpha_{i2} \\ \alpha_{i3} \\ \alpha_{oA} \\ \alpha_{oB} \\ \alpha_{t1} \\ \alpha_{t2} \\ \alpha_{g} \\ \alpha_{z} \end{array}$	PTX " " CTX, PTX CTX, PTX CTX(?), PTX	nearly ubiquitous ubiquitous nearly ubiquitous brain, others brain, others retinal rods retinal cones taste buds brain, adrenal, platelets	M <sub>2</sub> -Ach, α <sub>2</sub> -AR " α <sub>2</sub> -AR " rhodopsin cone opsin taste (?) M <sub>2</sub> -Ach	inhibit adenylyl cyclase " " " stimulate cGMP phosphodiesterase " ? inhibit adenylyl cyclase		
Gq	$\begin{array}{c} \alpha_{q} \\ \alpha_{11} \\ \alpha_{14} \\ \alpha_{16} \\ \alpha_{15} \end{array}$		nearly ubiquitous nearly ubiquitous lung, liver, kidney T cells, myeloid cells B cells, myeloid cells	M <sub>1</sub> -Ach, α <sub>1</sub> -AR " ? ? ?	stimulate PLCβ " " ?		
G <sub>12</sub>	α <sub>12</sub> α <sub>13</sub>		ubiquitous ubiquitous	? ?	? ?		
β	β <sub>1</sub> β <sub>2</sub> β <sub>3</sub> β <sub>4</sub>		ubiquitous nearly ubiquitous "				
γ	Υ1 Υ2 Υ3 Υ4 Υ5 Υ6 Υ7		retina, other (?) brain, adrenal, other (?) brain, testis kidney, retina (?) widely expressed brain, other (?) widely expressed				

#### Table 1.III Examples of heterotrimeric G protein subunits

Based on Hepler and Gilman (1992), additional information from Cali, et al. (1992)

#### 1.1.6.2. THE G PROTEIN CYCLE

This has been reviewed by Neer (1995) and Hamm and Gilchrist (1996), and the main points shall be described. In the inactive state the  $\alpha$  subunit of the heterotrimeric G protein has GDP bound and is associated with the receptor. Upon ligand binding the receptor activates G $\alpha$ . This is thought to occur by changes in the receptor conformation which may expose a peptide at the C-terminus of the 3rd intracellular loop (Lefkowitz, *et al.*, 1993). Active G $\alpha$  has lower affinity for GDP which then dissociates. Because the cellular concentration of GTP is higher than GDP, GTP binds to the guanine nucleotide binding site of G $\alpha$ . This causes a conformational

#### Chapter 1

#### **INTRODUCTION**

change in the switch region of  $G\alpha$  at the  $\beta\gamma$  binding site and causes dissociation of the complex into  $G\alpha$ ,  $G\beta\gamma$  and receptor. The crystal structure of  $\beta\gamma$  does not change when  $G\alpha$  is dissociated (Lambright, *et al.*, 1996; Sondek, *et al.*, 1996). It has been proposed that the activity of  $\beta\gamma$  is due to the active binding site being exposed (Clapham, 1996). G $\alpha$  has intrinsic GTPase activity and is therefore deactivated after a certain time. Phospholipase C can enhance the intrinsic GTPase activity of  $G\alpha_q$  (Berstein, *et al.*, 1992). When GTP is hydrolysed to GDP,  $G\alpha$  reassociates with  $\beta\gamma$  and the receptor. Thus the activity of both  $\alpha$  and  $\beta\gamma$  is limited by the hydrolysis of GTP (see figure 1.4) (Neer, 1995; Hamm and Gilchrist, 1996).



#### Figure 1.4. G protein cycle.

When ligand binds to receptor  $G\alpha$  exchanges GDP for GTP. In this activated state it dissociates from receptor and  $\beta\gamma$  subunit. Both  $G\alpha$  and  $G\beta\gamma$  stimulate effectors.

30

The assembled receptor-G protein complex results in a high affinity state of the receptor for its ligand. However, when the G protein in its GTP bound state has dissociated, the receptor agonist complex is less stable and the receptor changes to a lower affinity state. Thus, in membrane preparations in the presence of GTP there is a decrease in the affinity of the receptor for its ligand (Spiegel, 1988). Use has been made use of this observation to identify receptor coupling to G proteins (e.g. see Sinnett-Smith, *et al.*, 1990 and section 1.2.1).

#### 1.1.6.3. CLASSES OF G PROTEIN

Although  $\beta\gamma$  transduce signals also (Clapham and Neer, 1993) (see below) it is the  $\alpha$  subunit which determines the functional class of G protein. These fall into four groups (see table 1.III): G<sub>s</sub> because they stimulate adenylyl cyclase, G<sub>i</sub> because they inhibit adenylyl cyclase, G<sub>q</sub> which activates PLC $\beta$  and G<sub>12</sub> for which the direct effector is unknown although recent work has implicated the G<sub>12</sub> family in Na<sup>+</sup>/H<sup>+</sup> exchange and stress fibre assembly (Dhanasekaran, *et al.*, 1994; Voyno-Yasenetskaya, *et al.*, 1994a; Buhl, *et al.*, 1995). G $\alpha$  regulation of these effectors will be described in later sections.

It was originally thought that receptors only coupled to one species of  $G\alpha$  but this now appears not to be the case. This will be discussed further in Chapter 3.

#### **1.1.6.4.** βγ SUBUNITS

The role of  $\beta\gamma$  was originally thought to be as a chaperone or negative regulator to guide the  $\alpha$  subunit back to the receptor. However in recent years it has been shown that  $\beta\gamma$  subunits can activate effectors themselves (Clapham and Neer, 1993). The effectors for which direct association has been shown are adenylyl cyclase and PLC $\beta$ . Another important effector of  $\beta\gamma$  are G protein receptor kinases (GRK) that desensitise seven transmembrane receptors (Inglese, et al., 1993). They were originally identified for  $\beta$ -adrenergic receptors ( $\beta$ ARK). The involvement of  $\beta\gamma$  subunits in cellular events has been demonstrated experimentally by  $\beta$ ARK and G $\alpha_t$  which act as inhibitors by sequestering the  $\beta\gamma$  subunits (Inglese, *et al.*, 1995).  $\beta\gamma$  subunits bind to the PH domain of  $\beta$ ARK (Gibson, *et al.*, 1994; Luttrell, *et al.*, 1995). PH domains were first defined as two repeats of approximately 100 residues in pleckstrin, the major PKC substrate in platelets. Now PH domains are identified in a wide variety of signalling and cytoskeletal proteins many of which are found associated with plasma or organella membranes (reviewed in Gibson, et al., 1994; Cohen, et al., 1995). Although the primary sequence homology is low between individual PH domains the tertiary structure of this domain is virtually the same. It is not clear what peptide sequences PH domains bind to although  $G\beta\gamma$  subunits will bind to carboxyl portions of some PH domains (Gibson, et al., 1994; Luttrell, et al., 1995). Also it is suggested that

the amino terminal portion of the PH domain binds specifically to  $PIP_2$  (Harlan, *et al.*, 1994) in keeping with the majority of PH domain-containing proteins being found localised to cellular membranes.

It was mentioned above that not all  $\beta\gamma$  combinations can occur. This could indicate that specific  $\beta\gamma$  combinations are required for coupling to distinct receptors or for activating distinct effectors. During this thesis aspects of  $\beta\gamma$  signalling shall be discussed, therefore some evidence regarding  $\beta\gamma$  specificity shall be presented here.

A series of experiments in which selected endogenous subunits were targeted by microinjection of antisense oligonucleotides revealed some degree of specificity. It was shown that in neuronal cells the M<sub>4</sub> muscarinic receptor couples to  $G\alpha_{o1}\beta_{3}\gamma_{4}$ whereas a somatostatin receptor couples to  $G\alpha_{o2}\beta_{1}\gamma_{3}$  (Kleuss, *et al.*, 1991; Kleuss, *et al.*, 1992; Kleuss, *et al.*, 1993). More recently Yan, *et al.* (1996) have used the yeast two hybrid system to show differential ability of  $\beta$  and  $\gamma$  subunits to form the G protein  $\beta\gamma$ complex. All isoforms were tested ( $\beta_{1}$ - $\beta_{5}$  and  $\gamma_{1}$ - $\gamma_{5}$  and  $\gamma_{7}$ ) and it was demonstrated that  $\beta_{1}$  and  $\beta_{2}$  can form tight complexes with most  $\gamma$  except ( $\gamma_{1}$  and  $\gamma_{2}$ ) and that there is wide heterogeneity between the other combinations. This shows that there could be specific  $\beta\gamma$  combinations but does not reveal anything about the selection of specific signalling pathways.

The effect of  $\beta\gamma$  subunits on various enzymes and signalling pathways has been the subject of many studies. Table 1.IV shows some events which specific  $\beta\gamma$  subunits regulate. It should be noted that not all  $\beta\gamma$  combinations have been assessed. In general the combinations which have been tested so far show similar activities. Some  $\beta\gamma$  combinations appear to show specificity:  $\beta_1\gamma_1$ , the retinal  $\beta\gamma$  for  $G\alpha_t$  (Iniguez-Lluhi, *et al.*, 1992; Ueda, *et al.*, 1994) is recognised to be different (see Clapham and Neer, 1993) and  $\beta_2\gamma_2$  which does all the signalling described for other  $\beta\gamma$  subunits except it does not activate the tyrosine kinases, Tsk and Btk. Some aspects of specificity have been shown in other ways. Luttrell, *et al.* (1995) looked at the effect of several PH domains on the signalling mediated by  $\beta\gamma$  subunits from  $G_q$  and found that it was only affected by the PH domain from PLC $\gamma$ . This suggested some specificity in PH domain recognition of  $\beta\gamma$ . It should be noted that most of the above studies are performed with ectopic expression of  $\beta\gamma$  subunits which may compromise specificity. To conclude, there is some evidence for  $\beta\gamma$  specificity but this is not well documented at the functional level.

		exist	PTX	AC	PLC	MAPK	shc	Tsk	Btk	actin
$\beta_1$	γ1	+	-	-	-/+	+		-		
	γ2	+	+	+	+	+	+	+	+	-
	γ3	+	+	+	+	+		+	+	-
	γ4	+								
	γ5	+	+	+	+					
	γ7	+	+	+	+					
β2	γ1	-	-		-	-		-		
	γ2	+	+	+	+	+	+	-	-	-
	γ3	+	+	+	+	+		+	+	-
	γ5	+	+	+	+					
	γ7	+	+	+	+					
ßa	2/-	-			_	_				
P3	/1 Vo	_			_	_				
	12				_	_				
	13				-	-				
β₄	Ŷι				-	-				
F 7	γ2				-	-				
	γ <u>2</u> γ <sub>3</sub>				-	-				
	1.5									
refs.		а	b	С	d	e	f	g	h	i

Table 1.IV. Function of  $\beta\gamma$  subunits

The table shows a summary of the documented effects of βγ subunits, (+) denotes an effect, (-) denotes no effect. **exist**: not all βγ combinations can exist. (+) shows that they form a physical dimer and (-) shows they do not; **PTX**: enhancement of pertussis toxin ADP ribosylation of G<sub>i</sub> or G<sub>o</sub>; **AC**: regulation of adenylyl cyclase. This includes both inhibition and stimulation of the relevant subtypes as described in the introduction, both effects have been grouped; **PLC**: activation of PLCβ isoforms; **MAPK**: activation of MAP kinase; **shc**: tyrosine phosphorylation of shc; **Tsk**: activation of Tsk; **Btk**: activation of Btk; **actin**: formation of stress fibres. refs.: **a** (Iniguez-Lluhi, *et al.*, 1992; Kleuss, *et al.*, 1994); **c** (Iniguez-Lluhi, *et al.*, 1992; Ueda, *et al.*, 1994); **d** (Ueda, *et al.*, 1994; Hawes, *et al.*, 1995); **e** (Faure, *et al.*, 1994; Hawes, *et al.*, 1995); **f** (Touhara, *et al.*, 1995).

#### **1.1.6.5.** MODIFYING TOXINS

Bacterial toxins have been used to help dissect G protein-mediated signal transduction pathways such as cholera toxin from *Vibrio cholerae* which activates  $G\alpha_s$  through inhibition of its GTPase activity (Cassel and Selinger, 1977). This is achieved by ADP-ribosylating a specific arginine residue (van Dop, *et al.*, 1984b). Another useful

tool is pertussis toxin from *Bordetella pertussis*. This toxin ADP-ribosylates a specific cysteine residue near the C-terminal of  $G_i$  and  $G_o$  (West, *et al.*, 1985).  $G_s$  and  $G_q$  do not have the necessary cysteine and are insensitive to pertussis toxin. The preferred substrate of pertussis toxin is the trimeric complex rather than just the  $\alpha$  subunit which means that  $\beta\gamma$  is no longer liberated from  $G_i$ . Thus cholera toxin locks  $G_s$  in the GTP state whereas pertussis toxin locks  $G_i$  in the GDP state and is used as an inhibitor of  $G_i$  (van Dop, *et al.*, 1984a; Spiegel, 1988).

### **1.2. BOMBESIN**

Bombesin is a bioactive tetradecapeptide that was originally discovered in extracts from the skin of the European toad *Bombina bombina* (Anastasi, *et al.*, 1971) which had pleiotrophic pharmacological effects in a range of different mammals (for general review see Tache, *et al.*, 1988). Bombesin produced a rise in systemic blood pressure and had stimulant action on isolated smooth muscle preparations of the intestine, colon, uterus and bladder. When administered centrally it affected thermoregulation, gastric function and behaviour. Bombesin also caused pancreatic secretions of digestive enzymes (e.g. amylase), insulin and glucagon (Tache, *et al.*, 1988), and the release of several gut hormones (Ghatei, *et al.*, 1982) Bombesin-like immunoreactivity is found in the brain, spinal cord, in nerve fibres along the gastrointestinal tract, where it is thought to act as a neurotransmitter, and in neuroendocrine cells of the lung and thyroid (for refs. see Solcia, *et al.*, 1988; Sunday, 1988).

The search for mammalian counterparts to bombesin led to the isolation of a gastrin releasing peptide from porcine gastric tissue (McDonald, *et al.*, 1979) and two other peptides from porcine spinal cord, neuromedin B (NmB) (Minamino, *et al.*, 1983) and neuromedin C (NmC) (Minamino, *et al.*, 1984), which had contractile activity on uterine tissue. Bombesin-like peptides fall into 3 groups on the basis of their primary structure, pharmacological activity and receptor affinity. The bombesin subfamily is characterised by the C-terminal tetrapeptide Gly-His-<u>Leu-Met-NH<sub>2</sub></u>, the litorin/ranatensin subfamily by Gly-His-<u>Phe-Met-NH<sub>2</sub></u> and the phyllitorin subfamily by Gly-Ser-Phe(Leu)-Met-NH<sub>2</sub> (examples relevant for this thesis are shown in figure 1.5).

Bombes	in subfamily				
	GRP	Ala- Pro- Val- Ser- Val- Gly-Gly-Gly-Thr- Val- Leu-Ala- Lys-			
		Met-Tyr- Pro-Arg-Gly-Asn- His-Trp-Ala- Val-Gly-His-Leu-Met-NH <sub>2</sub>			
	GRP-10 or Neuromedin C	Gly-Asn- His-Trp-Ala- Val-Gly-His-Leu-Met-NH <sub>2</sub>			
	Bombesin	pGlu-Gln-Arg- Leu-Gly-Asn-Gln-Trp-Ala- Val-Gly-His-Leu-Met-NH <sub>2</sub>			
Litorin subfamily					
	Litorin	pGlu- Gln- Trp-Ala- <i>Val-</i> Gly-His-Phe-Met-NH <sub>2</sub>			
	Neuromedin B	Gly-Asn-Leu-Trp-Ala- <i>Thr-</i> Gly-His-Phe-Met-NH <sub>2</sub>			

# Figure 1.5. Amino acid sequence of mammalian and amphibian bombesin-like peptides

Amino acids in bold show the C-terminal tetrapeptide that identify the subfamily. Amino acids in *italics* show where NmB and litorin differ from bombesin and GRP.

Bombesin, litorin, ranatensin and phyllitorin are all amphibian peptides. GRP, NmB and NmC are mammalian peptides. GRP and NmC have 9 of the last 10 amino acids identical to bombesin and NmC is also known as GRP-10 or  $\text{GRP}_{18-27}$ . NmB belongs to the litorin subfamily and although the first found had 9 amino acids a large NmB of 32 amino acids has also been described (Minamino, *et al.*, 1988). By submitting 27 bombesin-like peptides to parallel bioassay it was shown that the minimum amino acid sequence required for bombesin-like activity was the C-terminal heptapeptide, the tryptophan residue being of crucial importance. Full activity was seen with the C-terminal nonapeptide (Erspamer, *et al.*, 1988).

Bombesin immunoreactivity was found in the endocrine cells of the developing human lung (Wharton, *et al.*, 1978) and in small-cell lung carcinoma (SCLC) (Moody, *et al.*, 1981; Wood, *et al.*, 1981; Erisman, *et al.*, 1982) a cancer of pulmonary neuroendocrine cells characterised by a rapid growth rate, high metastatic potential and ectopic hormone production (Greco and Oldham, 1979). This raised the possibility that bombesin could participate in the control of cell proliferation and indeed it was shown that bombesin was a sole mitogen for Swiss 3T3 cells in culture (Rozengurt and Sinnett-Smith, 1983). This was followed by the observation that bombesin/GRP was a growth factor for normal bronchial epithelial cells (Willey, *et al.*, 1984). SCLC cells release bombesin-like peptides (for refs. see Moody and Korman, 1988) and it was shown by use of antibodies towards bombesin (Cuttitta, *et al.*, 1987) that
bombesin acted as an autocrine growth factor for SCLC. In addition [Leu<sup>13</sup>- $\psi$ (CH<sub>2</sub>NH)Leu<sup>14</sup>]bombesin, a specific bombesin/GRP receptor antagonist (Woll, *et al.*, 1988), was shown to inhibit the growth of SCLC cell lines *in vitro* and inhibit xenographt formation in nude mice (Mahmoud, *et al.*, 1991). Since then bombesin/GRP has been implicated in other tumours such as pancreatic (Avis, *et al.*, 1993) and gastric tumours (Qin, *et al.*, 1994) and has also been shown to be a growth factor for gastric carcinoma cell lines (Bold, *et al.*, 1994).

# **1.2.1.** BOMBESIN/GRP RECEPTOR

Bombesin/GRP mediates its effects through specific receptors first characterised on pancreatic acinar cells (Jensen, et al., 1978). The bombesin/GRP receptor in Swiss 3T3 cells is a 75 - 80 kDa cell surface glycoprotein that was identified by affinity crosslinking (Zachary and Rozengurt, 1987) with a protein core of 43 kDa (Sinnett-Smith, et al., 1988) and functionally coupled to a GTP binding protein (Fischer and Schonbrunn, 1988; Sinnett-Smith, et al., 1990) demonstrated by the ability of GTP or GTP analogues to reduce the affinity of bombesin for its receptor (see section 1.1.6.2). Two types of bombesin receptor were pharmacologically distinguished in rat oesophagus, a bombesin/GRP preferring receptor ( $K_i$  bombesin = 4 nM, K<sub>i</sub> NmB = 156 nM) and a NmB preferring receptor (K<sub>i</sub> NmB = 0.3 nM, K<sub>i</sub> bombesin = 2 nM) (von Schrenck, et al., 1989). Bombesin and GRP differ from NmB by 2 amino acids in the C-terminal heptapeptide. One of these differences, the Phe of NmB compared to the Leu of bombesin and GRP at the penultimate residue, is also found in litorin (see figure 1.5), but litorin has equal potency to bombesin in smooth muscle preparations and in Swiss 3T3 cells (Erspamer, et al., 1988; Millar and Rozengurt, 1990b; Sinnett-Smith, et al., 1990). The residue that differs between NmB and bombesin, GRP or litorin is the Thr of NmB five residues from the C-terminal end, see figure 1.5, and this could be the key residue that confers receptor specific recognition.

Molecular cloning has identified both the bombesin/GRP receptor (Spindel, *et al.*, 1990; Battey, *et al.*, 1991; Corjay, *et al.*, 1991) and the NmB preferring receptor (Corjay, *et al.*, 1991; Wada, *et al.*, 1991). Sequence analysis revealed that these receptors did belong to the superfamily of G-protein linked receptors. More recently another receptor has been cloned that shows sequence similarity to the bombesin/GRP receptor but the physiological ligand has not yet been defined (Gorbulev, *et al.*, 1992; Fathi, *et al.*, 1993).

# **1.3.** BOMBESIN-MEDIATED SIGNAL TRANSDUCTION

Binding of bombesin/GRP to its receptor initiates a cascade of intracellular signals (summarised in figure 1.6) culminating in DNA synthesis 10 - 15 hours later. These signals shall be described in this next section.





# **1.3.1. PHOSPHOLIPASE C**

One of the earliest events to occur after the binding of bombesin to its receptor is the rapid stimulation of phospholipase C (PLC) catalysed hydrolysis of phosphatidylinositol 4,5-bisphosphate (PIP<sub>2</sub>) in the plasma membrane (Mendoza, *et al.*, 1986; Lopez-Rivas, *et al.*, 1987; Nanberg and Rozengurt, 1988). This reaction produces inositol 1,4,5-trisphosphate and 1,2-diacylglycerol (Berridge, 1984), the functions of which are described below.

Phosphatidylinositol specific PLC has multiple isoforms which fall into 3 families,  $\beta$ ,  $\gamma$  and  $\delta$  (reviewed in Exton, 1994; Lee and Rhee, 1995). PLC isozymes all contain the highly homologous X and Y regions which are believed to be the catalytic domain. In addition they all contain an N-terminal PH domain and PLC $\gamma$  also contains

additional PH, SH2 and SH3 domains between the X and Y regions (Lee and Rhee, 1995). There are 4 isoforms of  $PLC\beta_{(1-4)}$  and these are activated by members of the  $G\alpha_q$  family. In addition  $PLC\beta_{(1-3)}$  are also activated by  $G\beta\gamma$ . This could explain why receptors that are coupled to  $G_i$  proteins which are classically linked to inhibition of adenylyl cyclase, can cause  $PIP_2$  hydrolysis in a pertussis toxin-sensitive fashion (Abdel-Baset, *et al.*, 1992). The different isozymes of  $PLC\beta$  are activated to different extents by  $G\alpha_q$  or  $G\beta\gamma$ .  $\alpha_q$  activates  $PLC\beta_1$  the most and  $PLC\beta_2$  the least, whereas  $\beta\gamma$  activate  $PLC\beta_3$  the most and  $PLC\beta_1$  the least (Exton, 1994).  $\alpha$  and  $\beta\gamma$  subunits of heterotrimeric G proteins bind to different areas on  $PLC\beta$ ,  $\alpha_q$  binds to the C-terminal region (Lee, *et al.*, 1993b) and  $\beta\gamma$  appears to bind to the Y region (Kuang, *et al.*, 1996). The hydrolysis of  $PIP_2$  in response to various agonists that signal through G protein-coupled receptors depends upon the cellular expression of the different isozymes.

PLC<sub>γ</sub> binds to activated phosphorylated growth factor receptors via its SH2 domain, and is then phosphorylated on tyrosine residues which lie between the X and Y regions (Kim, *et al.*, 1991). Tyrosine phosphorylated PLC<sub>γ</sub> is the active form (Lee and Rhee, 1995). The mechanism of activation of PLC $\delta$  is not yet known (Exton, 1994).

Bombesin stimulates phosphatidylinositol specific PLC in a pertussis toxininsensitive manner in Swiss 3T3 cells (Zachary, *et al.*, 1987) suggesting that bombesin does not couple to members of the  $G_{i/o}$  family. Indeed it has been shown that addition of bombesin to Swiss 3T3 cells induces [ $\alpha$ -<sup>32</sup>P]GTP azidoanilide photolabelling of  $G_q$ and  $G_{11}$  (Offermanns, *et al.*, 1994a).

As well as being a substrate for PLC,  $PIP_2$  is recognised to play a role in actin cytoskeletal regulation (Janmey, 1994) and serve as a membrane attachment site for proteins with PH domains (Harlan, *et al.*, 1994). Thus  $PIP_2$  depletion by PLC-mediated hydrolysis could also have important consequences for the cell (Lee and Rhee, 1995) including initiation of cytoskeletal rearrangement (Divecha and Irvine, 1995).

# 1.3.2. INOSITOL 1,4,5-TRISPHOSPHATE AND Ca<sup>2+</sup> MOBILISATION

Hydrolysis of PIP<sub>2</sub> produces  $Ins(1,4,5)P_3$  which, as a second messenger, binds to an intracellular receptor and induces the release of Ca<sup>2+</sup> from internal stores (Berridge, 1993) and below. The  $Ins(1,4,5)P_3$  signal produced by receptor-mediated PLC activation is switched off by two mechanisms which are described in detail by Majerus (1992) and Shears (1992). The first is due to dephosphorylation by inositol polyphosphate 5-phosphatase which converts  $Ins(1,4,5)P_3$  to  $Ins(1,4)P_2$ , the latter of which has no Ca<sup>2+</sup> mobilising ability.  $Ins(1,4)P_2$  is further dephosphorylated to Ins by phosphatases that are inhibited by Li<sup>+</sup>. The second mechanism is phosphorylation to Ins $(1,3,4,5)P_4$  by Ins $(1,4,5)P_3$  3-kinase. There are some reports which suggest that Ins $(1,3,4,5)P_4$  is also involved in Ca<sup>2+</sup> fluxes, this is further described in section 1.3.2.3. Ins $(1,3,4,5)P_4$  can also be dephosphorylated by 3-phosphatases and 5-phosphatases and eventually back to Ins. It is thought that some of these metabolites have biological functions although it is not clear what these are (Berridge and Irvine, 1989; Menniti, *et al.*, 1993).

Bombesin causes a rapid increase in cytoplasmic  $Ins(1,4,5)P_3$  which co-incides with a transient increase in the intracellular Ca<sup>2+</sup> concentration ([Ca<sup>2+</sup>]<sub>cyt</sub>) and with Ca<sup>2+</sup> efflux from the cells (Mendoza, *et al.*, 1986; Lopez-Rivas, *et al.*, 1987; Nanberg and Rozengurt, 1988). This efflux of Ca<sup>2+</sup> lasts for approximately 2 min, after which there is a 50% depletion of total cellular Ca<sup>2+</sup> which is persistent, lasting for at least 1 h (Mendoza, *et al.*, 1986). This pattern of Ca<sup>2+</sup> signalling is also seen with other mitogens, for example, addition of serum or vasopressin to fibroblasts loaded with <sup>45</sup>Ca<sup>2+</sup> causes the efflux of <sup>45</sup>Ca<sup>2+</sup> from an intracellular pool followed by depletion in total cellular <sup>45</sup>Ca<sup>2+</sup> that persists for more than 1 h (Lopez-Rivas and Rozengurt, 1983; Lopez-Rivas and Rozengurt, 1984).

The early transient increase in  $[Ca^{2+}]_{cyt}$  upon bombesin stimulation is due to release from intracellular stores. There is also a second phase to the Ca<sup>2+</sup> signal which is the sustained elevation of  $[Ca^{2+}]_{cyt}$  which is dependent upon extracellular Ca<sup>2+</sup>. These two phases shall be described in detail below.

# **1.3.2.1. CALCIUM HOMEOSTASIS**

Calcium concentration in blood plasma is about 3 mM of which about half is ionised Ca<sup>2+</sup> (Carafoli, 1987). In contrast, the amount of ionised Ca<sup>2+</sup> in the cytoplasm is 100 - 200 nM even though the amount of calcium in intracellular organelles such as the endoplasmic reticulum (ER) can reach millimolar concentrations. This results in a large electrochemical gradient across the plasma membrane. Ionised Ca<sup>2+</sup> can be kept at these low concentrations by two mechanisms: (i) binding to proteins that can act as Ca<sup>2+</sup> buffers, and (ii) by active transport across cellular membranes. The cytoplasm contains many proteins which bind Ca<sup>2+</sup> with high affinity but some of these are Ca<sup>2+</sup> regulated proteins rather than Ca<sup>2+</sup> buffers.  $[Ca^{2+}]_{cyt}$  is kept low mainly by removal from the cell across the plasma membrane or by uptake into intracellular organelles such as the ER or mitochondria, the former of which is believed to be the hormonally regulated Ca<sup>2+</sup> store (Carafoli, 1987).

#### Hormonally Regulated Intracellular Ca<sup>2+</sup> Store

The main characteristic of a hormone-responsive  $Ca^{2+}$  store is the colocalisation of low affinity, high capacity  $Ca^{2+}$  binding proteins and the  $Ins(1,4,5)P_3$ receptor. The lumen of the ER contains several major proteins including Grp78 (BiP or immunoglobulin heavy chain binding protein), protein disulphide isomerase (PDI), endoplasmin and reticulin. Endoplasmin and reticulin have been estimated to bind, in the mM range, about 10 and 20 molecules of calcium per molecule, respectively (Koch, 1990). All four molecules have the KDEL retention signal of the ER preceded by clusters of acidic residues, characteristic of Ca<sup>2+</sup> binding proteins. It is proposed that Ca<sup>2+</sup> binding to these proteins contributes both to Ca<sup>2+</sup> storage and to facilitate the binding of the KDEL sequence to its receptor and thus control the retention of protein in the lumen of the ER (Milner, *et al.*, 1992).

Ins $(1,4,5)P_3$  only releases about 40% of cellular Ca<sup>2+</sup> compared to the amount released by a Ca<sup>2+</sup> ionophore suggesting that there are multiple intracellular Ca<sup>2+</sup> pools some of which could be mitochondria or nuclei (Gill, *et al.*, 1992). Subcellular fractionation shows that the Ins $(1,4,5)P_3$ -releasable Ca<sup>2+</sup> store is in microsomes that are not mitochondria or nuclei. Within the microsome population there is heterogeneity in their ability to sequester and release Ca<sup>2+</sup> (Meldolesi, *et al.*, 1990). It has been proposed that the Ca<sup>2+</sup> store is an organelle called a "calciosome". This may or may not be distinct from the ER, however the current feeling is that the ER is not homogenous and discrete areas could function specifically as the calciosome (Koch, 1990; Meldolesi, *et al.*, 1990; Lytton and Nigam, 1992). Indeed it has been documented that endoplasmin is not distributed uniformly throughout the ER supporting the idea of separate functional areas (Koch, 1990).

#### **Calcium Pumps**

Calcium is actively pumped into the stores by the <u>sarcoplasmic/endoplasmic</u> <u>reticulum Ca<sup>2+</sup>A</u>TPase (SERCA), reviewed in Thomas and Hanley (1994). Five SERCA isoforms have been identified that are derived from 3 genes. SERCA 1 and 2 are alternatively spliced with isoform 2b being considered the main ER-type Ca<sup>2+</sup> pump of non-muscle cells. The role these pumps play in Ca<sup>2+</sup> homeostasis has been investigated using specific inhibitors., which are are thapsigargin, di-*tert*-butylhydroquinone (DBHQ) and, to a lesser extent, cyclopiazonic acid. The first two inhibitors have been used in this thesis and they shall be described.

**Thapsigargin** — Thapsigargin was originally isolated from the Western Mediterranean plant *Thapsia garganica* and identified as the major active component when the plant was used as a treatment for muscle and joint inflammation (Rasmussen, *et al.*, 1978). Its skin irritation properties were linked to histamine secretion from mast cells (Rasmussen, *et al.*, 1978). Thapsigargin was then shown to activate other cells in the inflammatory response (Ali, *et al.*, 1985). Because thapsigargin had structural similarities with TPA this led to its identification as a tumour promoter, but it was not a phorbol ester type tumour promoter (Hakii, *et al.*, 1986). The mechanism of action of thapsigargin was linked to  $Ca^{2+}$  fluxes (Ali, *et al.*, 1985) and this was demonstrated to be via increasing cytoplasmic  $Ca^{2+}$  through the discharge of intracellular stores (Thastrup, *et al.*, 1989) without the generation of inositol phosphates (Jackson, *et al.*, 1988). This was achieved by specifically inhibiting SERCA pumps (Thastrup, *et al.*, 1990; Sagara and Inesi, 1991). Thapsigargin is very potent, it is effective in subnanomolar concentrations, and specific, it has no action on the plasma membrane  $Ca^{2+}ATPase$  or any other ion-motive ATPase and appears to form an irreversible dead-end complex with SERCA pumps (Thomas and Hanley, 1994).

DBHQ — Investigation into quinone toxicity in hepatocytes revealed that DBHQ impaired microsomal Ca<sup>2+</sup> sequestration. It was shown to inhibit ATP-dependent Ca<sup>2+</sup> sequestration and Ca<sup>2+</sup>-stimulated ATPase activity of microsomal fractions but not plasma membrane or mitochondrial fractions (Moore, *et al.*, 1987). Like thapsigargin, DBHQ could rapidly elevate cytosolic Ca<sup>2+</sup> without producing inositol phosphates or altering the passive efflux of Ca<sup>2+</sup> from Ca<sup>2+</sup> loaded microsomes (Kass, *et al.*, 1989). DBHQ is less potent than thapsigargin, IC<sub>50</sub> for inhibition of Ca<sup>2+</sup>ATPase is about 1  $\mu$ M (Moore, *et al.*, 1987; Kass, *et al.*, 1989) and like thapsigargin, DBHQ stabilises the non-Ca<sup>2+</sup> bound state of the enzyme (Thomas and Hanley, 1994).

Addition of SERCA inhibitors to cells results in the elevation of  $[Ca^{2+}]_{cyt}$  within 15 s - 2 min, followed by a sustained elevation for many minutes. This reveals that Ca<sup>2+</sup> homeostasis is in a constant dynamic state that balances Ca<sup>2+</sup> uptake with a constant passive leak from internal stores, see figure 1.7. Thus the Ca<sup>2+</sup> pump must be working continuously against a native high Ca<sup>2+</sup> permeability of the ER membrane (Thomas and Hanley, 1994). The leak "channel" has not been identified.

#### Calcium Channel

The Ca<sup>2+</sup> release channel is the Ins(1,4,5)P<sub>3</sub> receptor (Mikoshiba, 1993). This is a large protein (known as P<sub>400</sub>) and upon cloning it was shown to have a large Nterminal cytoplasmic domain and a C-terminal transmembrane domain. The cDNA encoded a protein that bound Ins(1,4,5)P<sub>3</sub> and had Ca<sup>2+</sup> channel activity, hence the Ins(1,4,5)P<sub>3</sub> receptor is an Ins(1,4,5)P<sub>3</sub>-gated Ca<sup>2+</sup> channel. The transmembrane domain contains 6 or 8 membrane spanning regions as predicted from the hydropathy profile of the amino acid sequence. The functional Ins(1,4,5)P<sub>3</sub> receptor is a homotetramer which is regulated by Ca<sup>2+</sup>. With increasing amounts of Ca<sup>2+</sup> the Ins(1,4,5)P<sub>3</sub> receptor Ca<sup>2+</sup> channel is activated in a positive feedback manner but when [Ca<sup>2+</sup>]<sub>cyt</sub> gets high this inhibits the Ins(1,4,5)P<sub>3</sub> receptor in a negative feedback manner (Taylor and Marshall, 1992). Ins $(1,4,5)P_3$  receptor distribution is mainly on the ER supporting the role of the ER as the Ca<sup>2+</sup> store (Mikoshiba, 1993).

#### **1.3.2.2. CALCIUM RELEASE FROM INTRACELLULAR STORES**

At rest  $[Ca^{2+}]_{cyt}$  is low due to removal by the plasma membrane ATPase and uptake into the ER by SERCA. Upon agonist stimulation PLC<sub>Y</sub> or PLC<sub>β</sub> hydrolyses PIP<sub>2</sub> to yield Ins(1,4,5)P<sub>3</sub> which binds to the Ins(1,4,5)P<sub>3</sub> receptor on the ER. The intrinsic  $Ca^{2+}$  channel opens and  $Ca^{2+}$  flows down the concentration gradient resulting in a large increase in intracellular  $Ca^{2+}$  from about 0.1 to 0.5 - 1 µM (reviewed by Berridge, 1993; Petersen, *et al.*, 1994). This increased  $[Ca^{2+}]_{cyt}$  is transient and  $Ca^{2+}$  is removed by reuptake into the ER and extrusion through the plasma membrane. This is the  $Ca^{2+}$ efflux observed (Lopez-Rivas and Rozengurt, 1983; Lopez-Rivas and Rozengurt, 1984). Agonists activate the plasma membrane pump both by  $Ca^{2+}$ -dependent (calmodulin) and independent (phosphorylation by PKA or PKC) mechanisms (Carafoli, 1992). The ER  $Ca^{2+}ATPase$  is also stimulated by an increase in  $[Ca^{2+}]_{cyt}$  and inhibited when the ER stores are full (IC<sub>50</sub> = 300 µM) (Petersen, *et al.*, 1994).



#### Figure 1.7. Calcium Homeostasis and Release

Calcium homeostasis is a balance between the Ca<sup>2+</sup>ATPase and the passive Ca<sup>2+</sup> leak from the ER. Thapsigargin (TG) and DBHQ can interrupt this balance by inhibiting the Ca<sup>2+</sup>ATPase. Upon agonist stimulation of e.g. the bombesin/GRP receptor, PLC $\beta$  produces Ins(1,4,5)P<sub>3</sub> which binds to its receptor on the ER causing the release of Ca<sup>2+</sup>. The ER store is replenished by the Ca<sup>2+</sup>ATPase.

#### Chapter 1

# **Calcium in Single Cells**

The use of patch clamp electrophysiological techniques and video or confocal fluorescent microscopy has shown that  $Ca^{2+}$  signals within a single cell are more complex than just a single transient elevation. Cytosolic  $Ca^{2+}$  can oscillate and spread throughout the cell in waves. This is believed to be due to the biphasic nature of  $Ca^{2+}$  on the regulation of the  $Ins(1,4,5)P_3$  receptor (Taylor and Marshall, 1992), and see below.

Ca<sup>2+</sup> mobilising agents can evoke Ca<sup>2+</sup> oscillations in several exocrine glands, fibroblasts, hepatocytes, endothelial cells, macrophages and oocytes (Tepikin and Petersen, 1992). The spatiotemporal behaviour of Ca<sup>2+</sup> observed in a variety of cells varies from whole-cell oscillations or repetitive spiking to intracellular target patterns and spiral waves (Clapham and Sneyd, 1995). There are basically two types of oscillations that can be produced by various agonists, low frequency baseline spikes and higher frequency sinusoidal oscillations superimposed on an elevated baseline, but is not known how different agonists can produce these different oscillatory profiles (Tepikin and Petersen, 1992; Putney and Bird, 1993). A variety of models have been proposed to explain the basic oscillatory phenomena (see Clapham and Sneyd, 1995) including oscillations of  $Ins(1,4,5)P_3$  (Ciapa, *et al.*, 1994), although currently favoured models are explained by the biphasic effect of Ca<sup>2+</sup> on the  $Ins(1,4,5)P_3$ receptor (Clapham and Sneyd, 1995; Petersen, *et al.*, 1994). The principles are as follows.

Agonists cause the production of  $Ins(1,4,5)P_3$  which activates receptors on a subset of the ER resulting in a low level of Ca<sup>2+</sup> release. When the Ca<sup>2+</sup>ATPase has filled the ER store the luminal [Ca<sup>2+</sup>] inactivates the pump. The low level of Ca<sup>2+</sup> release increases the local [Ca<sup>2+</sup>]<sub>cyt</sub> until the threshold is reached where Ca<sup>2+</sup> synergises with  $Ins(1,4,5)P_3$  to open all the  $Ins(1,4,5)P_3$  sensitive channels. This results in an explosive release of Ca<sup>2+</sup>. However as Ca<sup>2+</sup> increases, Ca<sup>2+</sup> will slowly bind to an inactivating domain on the receptor inhibiting the Ca<sup>2+</sup> flux. This results in a net uptake of Ca<sup>2+</sup> back into the ER by the Ca<sup>2+</sup>ATPase and Ca<sup>2+</sup> is extruded by the plasma membrane ATPase. There would be a refractory period as  $[Ca<sup>2+</sup>]_{cyt}$  is lowered and the  $Ins(1,4,5)P_3$  receptor becomes activated again. This could explain oscillations if constant  $Ins(1,4,5)P_3$  was present. Also the positive effect of Ca<sup>2+</sup> on its own release is the reason for the propagation of waves in a 2-dimensional spatial model.

To compensate for the extrusion of  $Ca^{2+}$  from the cytoplasm  $Ca^{2+}$  entry is required and this will be discussed in the next section. However this compensation is not complete and in prolonged stimulation by an agonist there is a balance between  $Ca^{2+}$  release, extrusion and influx. This results in the stores being maintained in a depleted state in Swiss 3T3 cells after prolonged stimulation by serum or vasopressin (Lopez-Rivas and Rozengurt, 1983; Lopez-Rivas and Rozengurt, 1984).

Note that the above events can only be measured in single cells. Populations of cells in suspension will not oscillate synchronously.

# **1.3.2.3.** CALCIUM INFLUX FROM THE EXTRACELLULAR MEDIUM

It has been known for a long time that the Ca<sup>2+</sup> signal is biphasic (Putney, et al., 1981; Putney and Bird, 1993). There is a transient increase in  $[Ca^{2+}]_{cyt}$  from intracellular stores, followed by a prolonged influx from the extracellular medium. The Ca<sup>2+</sup> channels activated in this process operate with extracellular Ca<sup>2+</sup> in the mM range (Mauger, et al., 1984; Poggioli, et al., 1985). The mechanism of activation of these channels is still controversial. Originally it was thought that it was the increase in [Ca<sup>2+</sup>]<sub>cvt</sub> which activated the influx (von Tscharner, et al., 1986), or that the inositol trisphosphate metabolite,  $Ins(1,3,4,5)P_4$ , was the intracellular messenger for this process (Irvine and Moor, 1986; Irvine and Moor, 1987). This mechanism was revived as a potential candidate more recently when patch-clamp techniques were used (Luckhoff and Clapham, 1992; Hashii, et al., 1993). Also a combination of both Ca<sup>2+</sup> and inositol phosphates has been proposed (Pittet, et al., 1989). It was thought that the mechanism of action could involve a conformational coupling proposed by (Putney, 1986) by which the  $Ins(1,4,5)P_3$  receptor could join the ER and plasma membranes, a process regulated by inositol phosphates. Once anchored by the  $Ins(1,4,5)P_3$  receptor the plasma membrane calcium channel could join to the  $Ca^{2}$ +ATPase of the ER forming a structure analogous to that seen in a gap junction (Irvine and Moor, 1987). This would mean that  $Ca^{2+}$  entry would bypass a journey through the cytoplasm. There is now evidence against this model and it is now accepted that Ca<sup>2+</sup> crosses the cytoplasm first (Putney and Bird, 1993).

The currently favoured mechanism for  $Ca^{2+}$  influx is that the empty state of the  $Ca^{2+}$  store regulates the influx and this has been described as Capacitative Calcium Influx (Putney, 1986). This was supported later in a study by Takemura and Putney (1989). These investigators used agonists to stimulate the cells in the absence of extracellular  $Ca^{2+}$  followed by antagonists to stop this stimulation thus bringing inositol phosphate levels back to the resting state.  $Ca^{2+}$  influx was monitored on replacing extracellular  $Ca^{2+}$  in the absence of raised inositol phosphates. Capacitative calcium influx was substantiated when thapsigargin and DBHQ were used to deplete the ER store because these agents do not cause the release of inositol phosphates (Takemura, *et al.*, 1989; Robinson, *et al.*, 1992). The  $Ca^{2+}$  influx activated by agonists, or by depletion of agonist sensitive stores with thapsigargin or DBHQ, was indistinguishable supporting the role of the filling state of the store as the regulator of

Ca<sup>2+</sup> influx (Demaurex, *et al.*, 1992; Schilling, *et al.*, 1992; Breittmayer, *et al.*, 1993; Montero, *et al.*, 1993a).

How does the empty store activate influx? It was proposed that a small signalling molecule is released from the ER (rather than a metabolite of  $Ins(1,4,5)P_3$ ) that activates calcium influx because thapsigargin and DBHQ do not cause release of inositol phosphates but they do activate Ca<sup>2+</sup> influx (Takemura, et al., 1989). This putative signalling molecule has become the focus of intense investigation. Cytochrome P-450 was suggested to have a role in production of this messenger (Alvarez, et al., 1991) which was estimated to have a half life of 7 s (Montero, et al., 1992). However, it is now thought that *P*-450 inhibitors block the influx channel itself rather than blocking the production of the messenger (Vostal and Fratantoni, 1993). Parekh, et al, (1993) then described a diffusible messenger which could be washed away in patch clamp studies with a half life of 10 s and was deactivated by an okadaic acid-sensitive phosphatase. At the same time Randriamampita and Tsien (1993) prepared a crude Jurkat cell extract which had Ca<sup>2+</sup> influx properties when applied extracellularly to several cell types and called the active agent(s) CIF (calcium influx factor). It had a phosphate group which was consistent with the observation of Parekh, et al. and Mr of under 500 (Randriamampita and Tsien, 1993). This extract was further investigated by different groups who showed it to be a heterogeneous mix of intra- and extra-cellular acting factors (Gilon, et al., 1995; Thomas and Hanley, 1995). Thomas and Hanley (1995) applied the extracts intracellularly by microinjection of oocytes then purified the extract further to obtain 3 calcium mobilising molecules one of which fulfilled criteria for a CIF (Kim, et al., 1995).

However other observations suggest that CIF is not the only regulator of Ca<sup>2+</sup> influx. PKC activation was shown to greatly enhance the influx induced by thapsigargin in RINm5F insulin secreting cells (Bode and Goke, 1994) but PKC activation in neutrophils, megakaryoblastic cells and Jurkats inhibited thapsigargin-induced Ca<sup>2+</sup> influx (Montero, *et al.*, 1993b; Conroy, *et al.*, 1994; Xu and Ware, 1995). Tyrosine phosphorylation has also been proposed as a mechanism for Ca<sup>2+</sup> influx. In human foreskin fibroblasts and lymphocytes tyrosine kinase inhibitors blocked Ca<sup>2+</sup> influx but not release (Lee, *et al.*, 1993a; Tepel, *et al.*, 1994). A role for GTP hydrolysis was described in the initiation of Ca<sup>2+</sup> influx in lachrymal acinar cells and a leukaemic mast cell line (Bird and Putney, 1993; Fasolato, *et al.*, 1993). Also metabolic poisons which decrease ATP levels, and hence GTP levels, can inhibit influx (Gamberucci, *et al.*, 1994). Pertussis toxin was shown to inhibit Ca<sup>2+</sup> influx (Fernando and Barritt, 1994). This was substantiated by the observation that antibodies to Ga<sub>11-2</sub> and peptides against Ga<sub>12</sub> but not Ga<sub>13</sub> blocked Ca<sup>2+</sup> influx (Berven, *et al.*, 1995).

hypothesis has not been presented yet. This could mean that there are multiple mechanisms to  $Ca^{2+}$  influx (Neher, 1992) which may differ subtly in their biological function and the experimental procedure used may bias for the measurement of one pathway over the other. Alternatively these differences could reflect variation between cell types (Thomas and Hanley, 1994) or even steps in a sequential process for the production or action of CIF.

#### **1.3.2.4.** CALCIUM GRADIENTS

It is important to point out that free  $Ca^{2+}$  is a very localised signal. In the cytoplasm there are many buffering proteins which will limit the range of the  $Ca^{2+}$  ion. It has been estimated that whereas the diffusion constant is 24 µm for  $Ins(1,4,5)P_3$ , that for unbuffered  $Ca^{2+}$  is 0.1 µm and buffered  $Ca^{2+}$  is 5 µm. Therefore, in cells smaller than 20 µm,  $Ins(1,4,5)P_3$  is a global messenger whereas  $Ca^{2+}$  acts in restricted domains (Allbritton, *et al.*, 1992). Thus within the cell there are  $Ca^{2+}$  gradients (Petersen, *et al.*, 1994). Micro-gradients can occur within a spatial dimension of 1 - 10 µm and over a concentration range of 0.1 - 1 µM and can be visualised with current  $Ca^{2+}$  imaging techniques. Also Petersen, *et al.* (1994) suggest the possibility of nano-gradients which could appear around  $Ca^{2+}$  channels and may reach concentrations of 100 µM. This may occur if  $Ca^{2+}$  mobilisation through the channel is faster than diffusion through the cytoplasm and may only last milli seconds depending on the state of the  $Ca^{2+}$  channel. Nano-gradients cannot be visualised with current  $Ca^{2+}$  imaging techniques.

# **1.3.2.5. CALCIUM IN GROWTH CONTROL**

It has been known for more than 20 years that transformed cells require less extracellular calcium for growth than non-transformed cells (Balk, 1971; Balk, *et al.*, 1973; Boynton, *et al.*, 1977; Paul and Ristow, 1979; Nicholson, *et al.*, 1984). This indicated that the effect of calcium is regulatory rather than permissive and that a  $Ca^{2+}$ -dependent step could be bypassed in transformed cells. The ability of cells to take up  $Ca^{2+}$  from the extracellular medium is crucial for normal cell proliferation. This has been studied using  $Ca^{2+}$ -deficient culture medium or blood plasma in parathyroidectomised rats, or by using  $Ca^{2+}$  channel antagonists (reviewed by Takuwa, *et al.* (1995). The importance of  $Ca^{2+}$  influx through non-voltage sensitive channels is further illustrated by the clinical investigation of a blocker of these channels in patients with cancer (Kohn, *et al.*, 1996). However, these studies show that  $Ca^{2+}$  is essential but it is more difficult to demonstrate that  $Ca^{2+}$  is a regulatory signal rather than an obligatory event, see section 1.1.3. The intracellular calcium chelator dmBAPTA has been used to quench increases in  $[Ca^{2+}]_{cyt}$  induced by serum to show a causal relationship between increases in  $[Ca^{2+}]_{cyt}$  and re-entry to the cell cycle (Wahl and Gruenstein, 1993). However, it is difficult to distinguish between permissive and regulatory effects of  $[Ca^{2+}]_{cyt}$ , also dmBAPTA chelates other ions and this introduces non-specific effects. Takuwa, *et al.* (1991a) showed that bombesin required extracellular Ca<sup>2+</sup> to reinitiate DNA synthesis in Swiss 3T3 cells. However this study did not distinguish between permissive and regulatory requirements for Ca<sup>2+</sup> either.

Because many early signalling events induced by bombesin were independent of the  $[Ca^{2+}]_{out}$  (Takuwa, *et al.*, 1991a) it was concluded that  $Ca^{2+}$  has an obligatory role at  $G_1/S$  and/or  $G_2/M$  (Takuwa, *et al.*, 1995).  $Ca^{2+}$  is now recognised to be essential at various points throughout the cell cycle including  $G_1/S$  and  $G_2/M$  but it has been difficult to ascribe anything more than a supporting or housekeeping role to this ion (Means, 1994; Berridge, 1995; Takuwa, *et al.*, 1995).

The role of  $Ca^{2+}$  in initiating the transition between  $G_0/G_1$  is not clear, with the exception of lymphocytes. Changes in  $Ca^{2+}$  fluxes in lymphocytes have been identified as stimulating proliferation but this is because an increase in  $[Ca^{2+}]_{cyt}$  induces the IL-2 gene which then acts as an autocrine growth factor (Crabtree, 1989). The role of  $Ca^{2+}$  in initiating fibroblast proliferation has not been defined, however, several observations indicate that  $Ca^{2+}$  could be important.

1. Neuropeptides such as bombesin, vasopressin, endothelin and bradykinin, which are potent mitogens for Swiss 3T3 cells (Rozengurt, 1986), induce a rapid and transient increase in  $[Ca^{2+}]_{cyt}$  followed by a persistent depletion of  $Ca^{2+}$  from stores in the ER (Lopez-Rivas and Rozengurt, 1984; Mendoza, *et al.*, 1986; Lopez-Rivas, *et al.*, 1987; Nanberg and Rozengurt, 1988).

2. Receptors that are coupled to  $G_q$  have oncogenic potential. These include the conditional oncogenes 5-HT<sub>1C</sub> receptor (Julius, *et al.*, 1989) and M<sub>1</sub>-, M<sub>3</sub>- and M<sub>5</sub>-Ach receptors (Gutkind, *et al.*, 1991) which all require ligand activation to induce transformation. Also mutants of the  $\alpha_{1B}$ -AR that constitutively activate  $G_q$  enhance mitogenesis and transform fibroblasts (Allen, *et al.*, 1991).

3. M<sub>2</sub>- and M<sub>4</sub>-Ach receptors which couple to G<sub>i</sub> are not transforming (Gutkind, *et al.*, 1991). Interestingly transfected 5-HT<sub>1A</sub> receptors which couple to pertussis toxin-sensitive G<sub>i</sub> proteins were shown to be conditionally transforming, although activation of these receptors did induce PIP<sub>2</sub> hydrolysis (Abdel-Baset, *et al.*, 1992), probably mediated by  $\beta\gamma$  subunits as described earlier.

4. Constitutively activated  $G_q$  mutants transform fibroblasts (Kalinec, *et al.*, 1992).

5. Microinjection of antibodies raised against  $G_q$  blocks the mitogenic activity of bradykinin (LaMorte, *et al.*, 1993a).

6. A bacterial toxin from *Pasteurella multocida* which causes a striking activation of inositol phosphate production and Ca<sup>2+</sup> mobilisation, also promotes cell proliferation including anchorage-independent growth (Rozengurt, *et al.*, 1990; Higgins, *et al.*, 1992; Murphy and Rozengurt, 1992).

7. Microinjection of PLC into quiescent NIH 3T3 induces entry into S phase (Smith, *et al.*, 1989).

Thus Ca<sup>2+</sup> mobilisation is coincident with mitogenic stimulation. But this does not prove a role for Ca<sup>2+</sup> fluxes in the mitogenic activation of quiescent fibroblasts because hydrolysis of PIP<sub>2</sub> also generates DAG which activates PKC. The importance of PKC in cellular proliferation has been previously described (Rozengurt, *et al.*, 1984; Nishizuka, 1988 and see section 1.3.3). Furthermore, insulin and EGF can stimulate DNA synthesis in Swiss 3T3 cells without inducing Ca<sup>2+</sup> mobilisation (Rozengurt, *et al.*, 1983b; Vara and Rozengurt, 1985). Similarly, mutant receptors for platelet-derived growth factor and fibroblast growth factor defective in PLC<sub>Y</sub> activation have been shown to retain mitogenic activity upon ligand activation (Mohammadi, *et al.*, 1992; Peters, *et al.*, 1992; Valius, *et al.*, 1993; Valius and Kazlauskas, 1993). Also activated Ga mutants that do not stimulate PIP<sub>2</sub> hydrolysis are transforming such as  $\alpha_{12}$  and  $\alpha_{13}$ (Xu, *et al.*, 1993; Voyno-Yasenetskaya, *et al.*, 1994b) and  $\alpha_i$  (*gip*2 oncogene), although the latter is only weakly transforming and only in certain fibroblasts (Pace, *et al.*, 1991; Gupta, *et al.*, 1992). Thus, the contribution of Ca<sup>2+</sup> to mitogenic signalling remains unclear.

Another approach for assessing the role of  $Ca^{2+}$  in the transition from  $G_0$  into DNA synthesis, is the use of specific inhibitors of the  $Ca^{2+}ATPase$  of the ER to create an isolated  $Ca^{2+}$  signal. Addition of these compounds to intact cells induces mobilisation of  $Ca^{2+}$  from internal stores bypassing PLC-mediated formation of  $Ins(1,4,5)P_3$ .

This has advantages over the use of the Ca<sup>2+</sup> ionophores A23187 or ionomycin which increase  $[Ca^{2+}]_{cyt}$  by permeablising the plasma, ER and mitochondrial membranes to Ca<sup>2+</sup> and uncouple oxidative phosphorylation (Reed and Lardy, 1972; Liu and Herman, 1978). Ca<sup>2+</sup> homeostasis is therefore disrupted in non-specific manner so it is not suprising that induction of DNA synthesis in fibroblasts has not been described for the Ca<sup>2+</sup> ionophores. Thapsigargin and DBHQ will selectively deplete the hormonally-responsive store causing a situation which resembles an agonist response more closely.

Previous studies have shown that the addition of either thapsigargin or DBHQ, at micromolar concentrations, profoundly inhibited cell proliferation or induced cytotoxic effects (Ghosh, *et al.*, 1991; Short, *et al.*, 1993). These findings are difficult

to reconcile with the fact that mitogenic neuropeptides and other growth promoting factors also cause persistent depletion of Ca<sup>2+</sup> from the ER and that thapsigargin has been shown to be a tumour promoter.

The experiments presented in Chapter 5 were designed to assess the effect of gradual depletion of  $Ca^{2+}$  from the ER on the ability of quiescent cells to exit from  $G_0$  and enter DNA synthesis.

# **1.3.2.6.** CALCIUM EFFECTORS

The main  $Ca^{2+}$  binding protein in the cytoplasm is the 17 kDa ubiquitous multifunctional calmodulin (CaM) that binds  $Ca^{2+}$  with a K<sub>d</sub> in the µM range consistent with activation at high  $[Ca^{2+}]_{cyt}$  (reviewed in Gnegy (1993). CaM acts as a  $Ca^{2+}$  dependent regulator of cyclic nucleotide metabolism,  $Ca^{2+}$  transport, protein phosphorylation and dephosphorylation cascades, ion transport, cytoskeletal function and cell proliferation. Takuwa, *et al.* (1995) describes the evidence that CaM is the major mediator of  $Ca^{2+}$ -dependent proliferation. In particular CaM levels start to rise in late  $G_1$  and S phase and reach maximum in late  $G_2$ . CaM may mediate its effects in the nucleus during S phase. Also a variety of transformed cells and tumour tissues show elevated levels of CaM and overexpression of CaM shortens  $G_1$  and increases cell density at growth saturation (Takuwa, *et al.*, 1995).

At the protein level CaM also binds to a number of other, predominately cytoskeletal, proteins including MAP-2, MARCKS, adducin and tubulin (reviewed in Gnegy (1993). Interestingly many of the cytoskeletal CaM binding proteins can be phosphorylated by PKC. CaM also activates isozymes of adenylyl cyclase, cyclic nucleotide phosphodiesterase and several protein kinases including myosin light chain kinase, which is involved in muscle contraction, and multifunctional CaM kinase (CaMK).

Multifunctional CaMK, like CaM, is ubiquitous and as the name suggests it phosphorylates many substrates. These are involved in many cellular processes such as neurotransmitter release, muscle contraction, glycogen-, fatty acid- and carbohydrate metabolism, microtubule assembly and signal transduction (reviewed in Braun and Schulman (1995), including the cAMP response element binding protein, CREB, see section 1.3.9.2. An interesting aspect to CaMK activity has been proposed recently in which it acts as  $Ca^{2+}$  spike frequency detector (Hanson, *et al.*, 1994). CaMK exists as a multimeric enzyme of 8 - 10 monomers that are arranged like spokes on a wheel with the kinase domain at the periphery (Braun and Schulman, 1995). The subunits are active when  $Ca^{2+}/CaM$  is bound and in this state they can phosphorylate the neighbouring subunit (which also needs  $Ca^{2+}/CaM$  bound to be an effective substrate). This autophosphorylation within the holoenzyme converts CaMK to a  $Ca^{2+}/CaM$ -independent state. Computer simulations predict that if  $[Ca^{2+}]_{cyt}$  oscillates at a higher than threshold frequency, the rate at which  $Ca^{2+}/CaM$  associates with CaMK is greater than the rate of dissociation of  $Ca^{2+}/CaM$  and subsequent dephosphorylation of CaMK subunits. Higher frequency  $Ca^{2+}$  spikes will increase the likelihood of having  $Ca^{2+}/CaM$  on two neighbouring subunits: the "kinase" and the "substrate". In these conditions effective intersubunit phosphorylation can occur and hence  $Ca^{2+}/CaM$ -independent CaMK activity. This model suggests that CaMK could act as a frequency decoder that becomes activated in response to agonists that induce  $Ca^{2+}$  oscillations above a certain frequency (Hanson, *et al.*, 1994).

# **1.3.3. PROTEIN KINASE C**

PLC-mediated hydrolysis of PIP<sub>2</sub> also generates 1,2-diacylglycerol (DAG). DAG, which can also be generated from other sources such as phosphatidylcholine hydrolysis, acts as a second messenger in the activation of protein kinase C (PKC) by multiple extracellular stimuli (Nishizuka, 1988; Nishizuka, 1995) including bombesin (Erusalimsky, *et al.*, 1988). In accord with this, bombesin strikingly increases the phosphorylation of an acidic, myristoylated protein that migrates with an apparent molecular mass of 80 kDa which has been identified as a prominent substrate for PKC in cultured cells and several tissues, (Rozengurt, *et al.*, 1983b; Erusalimsky, *et al.*, 1991) see section 1.3.3.3. PKC activation can vary depending on the mitogen. For example PKC activation by bradykinin shows a transient phosphorylation of 80K which is almost back to basal levels after 2 min (Issandou and Rozengurt, 1990), whereas bombesin-stimulated 80K phosphorylation remains elevated for at least 10 min (Zachary, *et al.*, 1986).

Calcium-dependent protein kinase (PKC) was originally identified as a phospholipid-, diacylglycerol- and Ca<sup>2+</sup>-dependent protein kinase which phosphorylates serine residues. It has long been implicated in growth regulation because it is the site of action of the potent tumour-promoting phorbol esters (Nishizuka, 1984). Phorbol esters stimulate DNA synthesis and cell division in synergy with insulin and other growth factors (Dicker and Rozengurt, 1978; Dicker and Rozengurt, 1980). Also the addition of the synthetic DAG, 1-oleoyl-2-acetylglycerol (OAG) mimics the action of phorbol esters in stimulating reinitiation of DNA synthesis and cell division (Rozengurt, *et al.*, 1984).

An approach to testing the role of PKC in biological responses is to exploit the selective removal of PKC caused by prolonged pretreatment of the cells with phorbol ester. Chronic exposure to phorbol esters leads to a marked decrease in the number of specific phorbol ester-binding sites (Collins and Rozengurt, 1984) and to the disappearance of measurable PKC activity in cell-free preparations (Rodriguez-Pena

INTRODUCTION

and Rozengurt, 1984). Accordingly the cells become desensitised to the mitogenic effects elicited by phorbol esters (Collins and Rozengurt, 1982b; Collins and Rozengurt, 1984) or OAG (Rozengurt, *et al.*, 1984).. When cells are pretreated with PDB to downregulate PKC, DNA synthesis stimulated by bombesin alone is abolished but in the presence of insulin DNA synthesis is unaffected (Rozengurt and Sinnett-Smith, 1987).

# **1.3.3.1.** STRUCTURE AND ACTIVATION

To date there are 10 PKC isoforms which fall into 3 categories, classical or conventional PKC (cPKC), new or novel PKC (nPKC) and atypical PKC (aPKC) see table 1.V.

	Subspecies	Activators	Tissue expression
cPKC	α	Ca <sup>2+</sup> , DAG, PS	universal
	βI	11	some tissues
	βII	"	many tissues
	γ	n	brain only
nPKC	δ	DAG, PS	universal
	3	DAG, PS, PIP <sub>3</sub>	brain and others
	η(L)	DAG, PS, PIP <sub>3</sub>	skin, lung, heart
	θ	?	muscle, hemopoetic cells
aPKC	ζ	PS, PIP <sub>3</sub>	universal
	λ(ι)	?	many tissues

#### Table 1.V. PKC isoforms in mammalian tissues

From Nishizuka (1995) PS - phosphatidylserine; PIP<sub>3</sub> - phosphatidylinositol 3,4,5-trisphosphate.

All PKCs require phosphatidylserine (PS) for activation. This is an acidic phospholipid which is located exclusively on the cytoplasmic face of membranes. There is variation between the other co-activators and this is reflected in differences in the domain structure (C1 - C4) of the different classes of PKCs (reviewed in Nishizuka, 1995) see figure 1.8.

The C1 domain comprises an auto-inhibitory pseudosubstrate sequence followed by two tandem repeats of a cysteine rich, zinc finger motif which is responsible for DAG and phorbol ester binding. The aPKC group lack one cysteine rich domain and do not bind DAG and phorbol esters. The C2 domain, which is conserved among other proteins and is known as the CalB domain, is only present in cPKCs and is required for Ca<sup>2+</sup>-dependent phospholipid binding. Hence the classical PKCs are Ca<sup>2+</sup> dependent. The C3 region contains the ATP binding consensus sequence and the C4 domain binds the protein substrate.

The C-terminal catalytic domain is separated from the N-terminal regulatory domain via a flexible hinge region. Substrate binding displaces the pseudosubstrate region from the substrate binding site of the catalytic domain. This was deduced from the increased susceptibility to proteolysis of the pseudosubstrate region upon substrate binding. Cleavage of the pseudosubstrate sequence or antibodies directed against it result in constitutively activated PKC enzymes (Newton, 1995).



Figure 1.8. Structure of PKC isozymes

Swiss 3T3 cells contain PKCs  $\alpha$ ,  $\delta$ ,  $\varepsilon$  and  $\zeta$ . Prolonged treatment of Swiss 3T3 cells with phorbol esters therefore down regulates PKCs  $\alpha$ ,  $\delta$  and  $\varepsilon$  (Olivier and Parker 1992).

# 1.3.3.2. SUBSTRATES

It is believed that the function of different isozymes of PKC is to phosphorylate different substrates although the affinity of PKCs for known substrates and rate of phosphorylation do not differ much between isozymes. Attention has now turned to the location within the cell to explain specificity and it appears that PKCs are targeted to distinct subcellular locations (Jaken, 1996). It has been widely observed that PKC translocates from the cytoplasm to the membrane upon activation (Dekker and Parker, 1994), and DAG increases the affinity of PKC for PS (Nishizuka, 1995). These observations are consistent with the suggested importance of localisation of the enzymes. Substrates for PKC are 80K/MARCKS, vinculin, talin, adducins and annexins

which are involved in attaching the cytoskeleton with the cellular membranes; G protein-coupled receptors such as muscarinic receptors and receptor tyrosine kinases such as the EGF receptor; and enzymes involved in metabolism and signal transduction (Liu, 1996).

# 1.3.3.3. 80K/MARCKS

One of the earliest events occurring within seconds after phorbol ester and growth factor stimulation of cells is the phosphorylation of an acidic 80 kDa protein termed "80K" (Rozengurt, et al., 1983b). This was shown to be a specific PKC substrate and hence a useful tool for assaying PKC activity (Rodriguez-Pena and Rozengurt, 1986). Subsequently, 80K was purified and cloned (Brooks et al., 1990; Erusalimsky, et al. (1991). Further studies revealed that 80K is the murine homologue of the bovine myristoylated alanine rich C-kinase substrate, MARCKS (Herget, et al., 1992). The function of 80K/MARCKS is not entirely clear. Upon agonist-stimulated phosphorylation it translocates from the membrane to the cytosolic fraction (Herget and Rozengurt, 1994) and then it is down regulated by a post-transcriptional mechanism (Brooks, et al., 1991; Brooks, et al., 1992). Also the expression of 80K/MARCKS is very low when cells are dividing and its re-expression coincides with growth arrest in G<sub>0</sub> (Herget, et al., 1993). MARCKS has actin and calmodulin binding sites and is implicated in regulation of the cytoskeleton (Aderem, 1992). Recently it was shown that 80K/MARCKS deficiency in mice lead to abnormal brain development and perinatal death (Stumpo, et al., 1995) suggesting an important role for 80K/MARCKS in the regulation of growth and development.

## 1.3.3.4. PROTEIN KINASE D

PKC enzymes are not the only targets of the second messenger DAG. Recently a novel serine/threonine protein kinase has been cloned (Valverde, *et al.*, 1994) that binds and is activated by PDB (Valverde, *et al.*, 1994: van Lint, *et al.*, 1995) and has been named protein kinase D (PKD). This kinase has a catalytic domain that is most similar to myosin light chain kinase and the Ca<sup>2+</sup>/CaM regulated kinases. In particular PKD exhibits only a low degree of similarity to the highly conserved regions of PKCs and thus PKD represents a distinct class of protein kinases. PKD has two cysteine rich domains that are responsible for binding phorbol esters (Valverde, *et al.*, 1994), furthermore phorbol esters and synthetic diacylglycerols activate PKD in the presence of PS (van Lint, *et al.*, 1995). The identification of another target for phorbol esters raises the possibility that some of the cellular actions of phorbol esters could be mediated by PKD.

# **1.3.4.** MONOVALENT ION FLUXES

Growth factors stimulate monovalent cation fluxes (Rozengurt and Ober, 1988). This was first observed by Rozengurt and Heppel (1975) who found that addition of serum rapidly stimulated the influx rate of <sup>86</sup>Rb<sup>+</sup> (a K<sup>+</sup> tracer) into quiescent Swiss 3T3 cells. Most of the increased <sup>86</sup>Rb<sup>+</sup> uptake was inhibited by ouabain indicating that it was mediated by the plasma membrane Na<sup>+</sup>-K<sup>+</sup> pump. Subsequently, a variety of agents which are mitogenic in Swiss 3T3 cells were also shown to stimulate Na<sup>+</sup>-K<sup>+</sup> pump activity. These include PDGF, FDGF, vasopressin, phorbol esters and bombesin. This was also observed for a wide variety of mitogens in a range of cell types (reviewed in Rozengurt and Ober, 1988).

The Na<sup>+</sup>-K<sup>+</sup> pump activity is limited by the availability of intracellular Na<sup>+</sup> and changes in the rate of Na<sup>+</sup> entry therefore play a primary role in controlling the rate of Na<sup>+</sup>-K<sup>+</sup> pump activity. Studies with Li<sup>+</sup> uptake identified a second Na<sup>+</sup>-specific transport system which was activated by serum and was amiloride-sensitive (Smith and Rozengurt, 1978) suggesting that this Na<sup>+</sup> influx was mediated by the Na<sup>+</sup>/H<sup>+</sup> antiport. Since Na<sup>+</sup>/H<sup>+</sup> exchange can operate in either direction, the stimulation of Na<sup>+</sup> entry via the Na<sup>+</sup>/H<sup>+</sup> antiport by mitogenic agents could result from at least two alternative mechanisms. Growth factors could directly activate the antiport system leading to an increase in Na<sup>+</sup>/H<sup>+</sup> exchange and subsequent increase in pH<sub>i</sub>. Alternatively the enhancement of Na<sup>+</sup>/H<sup>+</sup> antiport activity by mitogens could represent a compensatory mechanism to reduce cytoplasmic acidification caused by excess protons generated by growth factor-enhanced cellular metabolism. Schuldiner and Rozengurt (1982) demonstrated that incubation of Swiss 3T3 cells in the presence of EGF, vasopressin and insulin increases the pH<sub>i</sub> and this was found to be a general response to mitogenic stimulation.

The fluxes of Na<sup>+</sup>, H<sup>+</sup> and K<sup>+</sup> across the membrane of Swiss 3T3 cells and other cell types can be envisaged as a "Na<sup>+</sup> cycle". Stimulation of the Na<sup>+</sup>/H<sup>+</sup> antiport by mitogens increases Na<sup>+</sup> influx and H<sup>+</sup> efflux, raising intracellular Na<sup>+</sup> and pH. The resulting increase in intracellular Na<sup>+</sup> stimulates the Na<sup>+</sup>-K<sup>+</sup> pump, elevating intracellular K<sup>+</sup> and restoring the electrochemical gradient for Na<sup>+</sup> (Rozengurt and Ober, 1988).

Several Na<sup>+</sup>/H<sup>+</sup> exchangers have been cloned recently (NHE1, NHE2, NHE3 and NHE4) but only NHE1 is ubiquitously expressed, the others appear to have a specialised distribution in epithelia (Bianchini and Pouyssegur, 1994). It is thought that NHE1 is regulated by phosphorylation or by a regulatory protein that may also be phosphorylated (reviewed in Bianchini and Pouyssegur, 1994). The involvement of PKC is implicated because phorbol esters and OAG enhance both ouabain-sensitive

<sup>86</sup>Rb<sup>+</sup> uptake and amiloride-sensitive <sup>22</sup>Na<sup>+</sup> entry (Vara, *et al.*, 1985). However, bombesin-stimulated monovalent ion fluxes is only partially inhibited after down regulation of PKC by prolonged treatment with phorbol ester, suggesting the presence of PKC-independent pathways (Mendoza, *et al.*, 1986). In support of this CaM has been proposed to have a role in activating NHE1 (Bianchini and Pouyssegur, 1994), and more recently, the newly identified Ga subunits  $\alpha_{12}$  and  $\alpha_{13}$  have been implicated in control of monovalent ion fluxes. Overexpression of activated mutants of  $\alpha_q$ ,  $\alpha_{12}$  and  $\alpha_{13}$  cause intracellular alkalisation (Dhanasekaran, *et al.*, 1994; Voyno-Yasenetskaya, *et al.*, 1994a).  $\alpha_q$  and  $\alpha_{13}$  were dependent upon PKC for this effect but  $\alpha_{12}$ -induced intracellular alkalisation was PKC-independent. Thus, it appears that Na<sup>+</sup>/H<sup>+</sup> can be regulated by multiple mechanisms.

Monovalent ion fluxes have been proposed to play a role in the initiation of cell proliferation. Inhibition of ion fluxes prevents the initiation of DNA synthesis and increased ion fluxes are associated with initiation of DNA synthesis (reviewed in (Rozengurt and Ober, 1988).

# **1.3.5.** PHOSPHOLIPASE A<sub>2</sub> AND ARACHIDONIC ACID RELEASE

While bombesin stimulates DNA synthesis in the absence of other growth factors, vasopressin is mitogenic for Swiss 3T3 cells only in synergistic combination with other factors (Rozengurt, *et al.*, 1979; Rozengurt and Sinnett-Smith, 1983). Binding of vasopressin to its distinct receptor on quiescent cultures of Swiss 3T3 cells causes a rapid production of  $Ins(1,4,5)P_3$ , mobilisation of  $Ca^{2+}$  from intracellular stores and sustained activation of PKC via a G-protein linked transduction pathway (reviewed in Rozengurt, 1991b). Since the initiation of DNA synthesis is triggered by independent signal-transduction pathways that act synergistically in mitogenic stimulation, the ability of bombesin to act as a sole mitogen could be due to activation of a signalling pathway not stimulated by vasopressin.

Recently, bombesin, but not vasopressin, has been shown to induce a marked, biphasic release of arachidonic acid into the extracellular medium (Millar and Rozengurt, 1990a; Domin and Rozengurt, 1993). A first phase involves rapid activation of phospholipase A<sub>2</sub> (PLA<sub>2</sub>). These results showed a clear difference in the pattern of early signals induced by the neuropeptides bombesin and vasopressin in Swiss 3T3 cells. The stimulation of arachidonic acid release by bombesin is likely to contribute to bombesin-induced mitogenesis because externally applied arachidonic acid potentiates mitogenesis induced by agents that stimulate PIP<sub>2</sub> hydrolysis but not arachidonic acid release, for example vasopressin (Millar and Rozengurt, 1990a). Arachidonic acid release in Swiss 3T3 cells stimulated by bombesin is biphasic with a rapid phase which lasts for 10 min followed by a sustained release which lasts for at least 60 min (Takuwa, *et al.*, 1991b). Bombesin-induced activation of PLA<sub>2</sub> involves multiple mechanisms, including intracellular Ca<sup>2+</sup> release for the first phase, and PKC activation and Ca<sup>2+</sup> influx for the second phase. However there also appears to be other unidentified mechanisms involved (Takuwa, *et al.*, 1991b).

Phospholipase  $A_2$  (PLA<sub>2</sub>) enzymes can generally be divided into two major groups, secretory PLA<sub>2</sub> (sPLA<sub>2</sub>) which are non-specific in their hydrolysis of phospholipid substrate and cytoplasmic PLA<sub>2</sub> (cPLA<sub>2</sub>) which specifically hydrolyse phospholipids that contain arachidonic acid (Glaser, *et al.*, 1993). cPLA<sub>2</sub> contains a CalB domain which mediates its Ca<sup>2+</sup>-dependent translocation to membrane vesicles (Clark, *et al.*, 1991). Possible signals involved in the activation of PLA<sub>2</sub> include Ca<sup>2+</sup> fluxes, PKC activation and activation of G<sub>i/o</sub> proteins (Piomelli, 1993). It is thought that PKC mediates its effect by activating MAPK which has been shown to phosphorylate and activate cPLA<sub>2</sub> *in vitro* (Lin, *et al.*, 1993). After hydrolysis free arachidonic acid can either diffuse out of the cell, be recycled back into phospholipids or be metabolised to prostaglandins, prostacyclins or thromboxanes. These metabolites may act as intracellular second messengers or as paracrine/autocrine factors which bind to specific seven transmembrane prostanoid receptors (Piomelli, 1993) and see table 1.II).

Arachidonic acid released by bombesin is believed to be converted to E-type prostaglandins which acting in an autocrine and paracrine manner enhance cAMP accumulation in the cell (Millar and Rozengurt, 1988; Millar and Rozengurt, 1990a). Since elevated cAMP levels constitute a mitogenic signal for Swiss 3T3 cells, see table 1.I, at least one consequence of arachidonic acid release may be the modulation of intracellular cAMP levels. However, other arachidonic acid metabolites may also play a role in mitogenic signal transduction by bombesin.

# **1.3.6.** ADENYLYL CYCLASE, CAMP AND PKA

Bombesin causes enhancement of forskolin stimulated adenylyl cyclase activity (Millar and Rozengurt, 1988) that is believed to be due to arachidonic acid release (see previous section), which is partially dependent on PKC. The intracellular levels of cAMP are controlled by membrane bound adenylyl cyclase which converts intracellular ATP to cAMP (reviewed in Taussig and Gilman, 1995). At present there are 7 full length cloned adenylyl cyclases all approximately 120 kDa and all activated by  $G\alpha_s$  and the diterpene forskolin. However they differ in their localisation and finer regulatory mechanisms. For instance  $G\beta_{\gamma}$  can stimulate type II and IV adenylyl cyclase but inhibit type I. Ca<sup>2+</sup>/CaM can activate types I, VIII and III whereas the others are insensitive to

 $Ca^{2+}/CaM$ , and types V and VI are inhibited by  $Ca^{2+}$  ions. In a similar fashion to the PLC $\beta$  isozymes in section 1.3.1 the cAMP response to various agonists depends upon which adenylyl cyclases are expressed (Taussig and Gilman, 1995).

The downstream effector of cAMP is the cAMP-dependent protein kinase (PKA). This consists of two inhibitory regulatory subunits and two catalytic subunits in the inactive state. When cAMP binds to the regulatory subunit, the catalytic subunits dissociate and their inhibition is released. PKA has many substrates (see Walsh and van Patten (1994) a number of which are involved in the breakdown of glycogen for metabolic energy. Genes which are responsive to increases in cellular cAMP have a cAMP response element (CRE) in their promoter to which the cAMP response element binding protein (CREB) binds upon phosphorylation by PKA (Karin and Hunter, 1995), for example see section 1.3.9.2.

# **1.3.7. TYROSINE PHOSPHORYLATION**

# **1.3.7.1. TYROSINE PHOSPHORYLATION OF MULTIPLE SUBSTRATES**

Recently it was shown that agonists which signal through G protein-coupled receptors increase protein tyrosine phosphorylation. These include the peptides bombesin, angiotensin II, endothelin, vasopressin and bradykinin (Huckle, *et al.*, 1990; Force, *et al.*, 1991; Leeb-Lundberg and Song, 1991; Tsuda, *et al.*, 1991; Zachary, *et al.*, 1991a), the protease thrombin (Force, *et al.*, 1991) and the biological amines acetylcholine, serotonin and norepinephrine (Tsuda, *et al.*, 1991; Gutkind and Robbins, 1992). They all rapidly stimulated the tyrosine phosphorylation of bands that ran diffusely on SDS-PAGE around 60 - 80 kDa and 110 - 130 kDa. In addition, increases in tyrosine kinase activity were also shown (Zachary, *et al.*, 1991b; Huckle, *et al.*, 1992). These substrates appeared to be a subset of those stimulated by EGF (Huckle, *et al.*, 1990; Force, *et al.*, 1991) but distinct from those of PDGF such as PLC<sub>γ</sub>, ras-GAP or PI3K (Zachary, *et al.*, 1991a; Gutkind and Robbins, 1992).

It was shown that there was no tyrosine kinase activity in Swiss 3T3 membrane preparations stimulated with bombesin even though they exhibited specific bombesin binding that could be modulated by guanine nucleotides. In contrast addition of PDGF to Swiss 3T3 membranes did stimulate tyrosine kinase activity (Zachary, *et al.*, 1991a). The cell surface M<sub>r</sub> 75 - 85 kDa receptor for bombesin which had been affinity labelled with <sup>125</sup>I-GRP was not immunoprecipitated by anti-phosphotyrosine residues. These results indicate that stimulation of tyrosine phosphorylation is not a consequence of receptor associated kinase activity (Zachary, *et al.*, 1991a) consistent with bombesin binding to a seven transmembrane receptor.

It was unclear at first whether tyrosine phosphorylation was downstream of PLC because all these agonists were acting on receptors which were coupled to PI In rat liver epithelial cells angiotensin II stimulated tyrosine hydrolysis. phosphorylation in a Ca<sup>2+</sup> dependent manner and Ca<sup>2+</sup> mobilising agents such as ionophores and thapsigargin were able to mimic the effects of agonist in smooth muscle cells and glomerula mesangial cells, although to a lesser extent (Huckle, et al., 1990; Force, et al., 1991; Tsuda, et al., 1991). This effect of Ca<sup>2+</sup> was not seen in all cell types (Zachary, et al., 1991a). Also activation of PKC by phorbol ester produced a similar pattern of tyrosine phosphorylated proteins, but again to a lower level than receptor agonist stimulation (Force, et al., 1991; Tsuda, et al., 1991; Zachary, et al., 1991a) or not at all (Huckle, et al., 1990; Leeb-Lundberg and Song, 1991). The role of PKC was examined in agonist stimulation of events. The PKC inhibitors H7 and staurosporine or down regulation of phorbol ester-sensitive isoforms did not affect agonist-induced protein tyrosine phosphorylation although they did abolish phorbol ester-induced tyrosine phosphorylation and tyrosine kinase activity (Force, *et al.*, 1991; Leeb-Lundberg and Song, 1991; Zachary, et al., 1991b). PKC and Ca<sup>2+</sup> therefore did not account for the full effects of tyrosine phosphorylation. Thus tyrosine phosphorylation of 60 - 80 kDa and 110 - 130 kDa represented a new pathway stimulated by G protein-coupled receptors which was not downstream of PLCmediated events.

The identification of these bands was aided by studies in v-src transformed cells in which a number of proteins were phosphorylated on tyrosine. To identify these putative v-src substrates, monoclonal antibodies were raised against the proteins that were tyrosine phosphorylated (Glenney and Zokas, 1989; Kanner, *et al.*, 1990) and the proteins they recognised were affinity purified. This procedure resulted in the isolation of two new proteins both localised to focal adhesions, paxillin (Turner, *et al.*, 1990) and a novel tyrosine kinase, p125 Focal Adhesion Kinase (FAK) (Schaller, *et al.*, 1992). Subsequently some of the heterogeneity regarding Ca<sup>2+</sup> stimulation of the tyrosine phosphorylation of a 120 kDa band, e.g. by angiotensin, has been explained by the demonstration that it is in fact not FAK but an unidentified protein (Earp, *et al.*, 1995). Before describing FAK and paxillin in more detail focal adhesions shall be explained.

# **1.3.7.2.** FOCAL ADHESIONS

Cells in culture adhere strongly to the substratum at places called focal contacts or focal adhesions (reviewed by Jockusch, *et al.*, 1995). These structures are where actin bundles, also known as stress fibres, of the cytoskeleton are anchored to the extracellular matrix and they are essential *in vivo* for cellular attachment in tissue

formation during embryogenesis and wound healing. This link between intra- and extracellular regions is mediated by integrins.

Integrins are heterodimers consisting of an  $\alpha$  and a  $\beta$  subunit. There are many isoforms of each subunit which could in theory associate to give more than 100 integrin heterodimers but the actual diversity appears much more restricted. Specific  $\alpha\beta$  combinations bind to specific extracellular matrix proteins such as collagen or fibronectin (reviewed in Hynes, 1992). Integrins are transmembrane proteins and have a cytoplasmic tail which interacts with proteins found in focal adhesions. The most abundant protein involved in the construction of the cytoplasmic face of the focal adhesion is actin from stress fibres which terminate there. These are actin filaments in an antiparallel arrangement which are thick enough to be clearly visible using fluorescence microscopy. Other high abundance structural proteins found in focal adhesions are  $\alpha$ -actinin, filamin, talin and vinculin which bind and cross link actin.  $\alpha$ actinin also binds to  $\beta_1$ -integrin, vinculin and zyxin. Vinculin, as well as binding actin and  $\alpha$ -actinin, will also bind talin and paxillin (Jockusch, *et al.*, 1995).

Proteins in focal adhesions each interact with many other proteins to build these large structures, hence a regulatory element is built in to regulate their rapid assembly and disassembly. This is described further in later sections. Zyxin and paxillin are substrates for phosphorylation and can bind  $\alpha$ -actinin and vinculin respectively. Kinases associated with focal adhesions include tyrosine kinases, src and FAK and serine kinases, PKC $\alpha$  and PKC $\delta$  (Jockusch, *et al.*, 1995).

The dynamics of these focal adhesions have been the focus of much attention due to transformed cells having reduced adhesion to the extracellular matrix, showing fewer focal adhesions and stress fibres, and being morphologically altered to a rounder and less flattened state (Ben-Ze'ev, 1985; Vasiliev, 1985). Similarly cells with low metastatic ability have well developed focal contacts and actin bundles, while high metastatic cells have fewer and smaller contacts and less actin bundles (Ben-Ze'ev, 1985; Vasiliev, 1985). The link between cell adhesion and cell proliferation has been recognised for many years. Untransformed cells will not grow in suspension but require a solid substrate for growth - this was described as anchorage-dependent growth. Transformation of these fibroblasts with polyoma virus negated the requirements for a solid support and this was described as anchorage-independent growth (Stoker, *et al.*, 1968). Thus the regulation of adhesion and cellular morphology could play an important role in cancer.

# 1.3.7.3. STRESS FIBRES

Focal adhesions are where actin stress fibres are anchored to the extracellular matrix. Quiescent cells and the majority of cells *in vivo* do not contain stress fibres

but they appear upon mitogenic stimulation in vitro and in instances in vivo such as wound healing in blood vessels (Ben-Ze'ev, 1985; Vasiliev, 1985; Jockusch, et al., 1995). To co-ordinate stress fibre assembly the diffuse actin filament network must be broken down by  $Ca^{2+}$  and phospholipid dependent gelsolin and profilin and repolymerised in thick bundles. The first clue to a stress fibre regulator was the observation that the bacterial exoenzyme C3 from *Clostridium botulinum*, when introduced into cells, caused them to round up due to disappearance of actin microfilaments. This was attributed to the ADP-ribosylation of the small GTP-binding protein, p21<sup>rho</sup> (Rubin, et al., 1988; Chardin, et al., 1989), a member of the ras superfamily (Hall, 1990). When an activated mutant of Rho was injected into cells it caused the appearance of stress fibres (Paterson, et al., 1990). Injection of C3 inhibited serum-, LPA- and bombesin-induced focal adhesion formation (measured by vinculin redistribution) and stress fibre assembly in quiescent cells which was consistent with a role for rho in these processes (Ridley and Hall, 1992). Like Ras (see section 1.3.8.5), Rho is regulated by guanine nucleotides exchange factors (GEFs) and GTPase activating proteins (GAPs) (Takai, et al, 1995).

# 1.3.7.4. SRC

The first protein tyrosine kinase to be discovered was the v-src oncoprotein. There are many kinases closely related to src most showing restricted distribution to haematopoetic cells, only Src, Fyn, and Yes are widely expressed (Erpel and Courtneidge, 1995). Members of the src family are activated in response to a variety of stimuli in many cell types and are required for a mitogenic response to PDGF in fibroblasts (Erpel and Courtneidge, 1995). More recently thrombin and angiotensin have been shown to activate Src activity (Chen, *et al.*, 1994; Sadoshima and Izumo, 1996).

All members of the src family are associated with cellular membranes by myristic acid attachment and they contain an SH2, an SH3 and a catalytic domain, and C-terminal (tail) regulatory sequences (Erpel and Courtneidge, 1995; Superti-Furga and Courtneidge, 1995). The catalytic activity of Src is repressed by the <u>C</u>-terminal <u>Src</u> kinase (Csk) which phosphorylates a specific tyrosine residue in the tail. The current model suggests that in the inactivated state the phosphorylated tyrosine in the tail binds intramolecularly to the SH2 domain and that this interaction is stabilised by the SH3 domain. Activation of Src could occur by dephosphorylation of the inhibitory residue, or high affinity competition for the SH2 or SH3 domain. Activation by phosphatase activity appears to occur in lymphocytes with lymphocyte specific proteins, and in fibroblasts overexpression of protein tyrosine phosphatase  $\alpha$  will activate Src. An additional tyrosine is present within the catalytic domain of c-Src and

autophosphorylation of this residue stimulates kinase activity (Erpel and Courtneidge, 1995; Superti-Furga and Courtneidge, 1995).

## **1.3.7.5.** FOCAL ADHESION KINASE

In cells transformed with v-src it was noticed that src was associated with the cytoskeleton. When investigators looked for the substrates of src they found a major tyrosine phosphorylated protein of 120 kDa in v-src transformed chick embryo fibroblasts (Kanner, *et al.*, 1990). Upon cloning, the deduced amino acid sequence revealed it to be a distinctive protein tyrosine kinase (Schaller, *et al.*, 1992). Compared to other protein tyrosine kinases it had low amino acid identity in the catalytic domain and no homology outside the catalytic domain. In particular there are no SH2 or SH3 domains although there are two proline rich areas in the C-terminal half. In addition there are no hydrophobic regions indicative of membrane spanning potential and no acylation sites. Thus p125 is a cytosolic protein tyrosine kinase. The overall structure is unique in that the catalytic domain is flanked by large N- and C-terminal domains. Immunostaining of chick embryo fibroblasts show that p125 was localised with tensin and hence focal adhesions so it was called focal adhesion kinase (FAK). Localisation was shown to depend on a focal adhesion targeting (FAT) sequence at the C-terminal end of FAK (Hildebrand, *et al.*, 1993).

A protein of 120 kDa that was tyrosine phosphorylated in response to cross linking integrins or plating cells on fibronectin (Guan, *et al.*, 1991; Kornberg, *et al.*, 1991) was identified as FAK (Burridge, *et al.*, 1992; Lipfert, *et al.*, 1992). Also the 110 - 130 kDa band that was rapidly tyrosine phosphorylated in response to mitogenic neuropeptides was identified as containing FAK (Zachary and Rozengurt, 1992; Sinnett-Smith, *et al.*, 1993).

#### 1.3.7.6. p130<sup>CAS</sup>

p130 was also identified as a major tyrosine phosphorylated protein in v-src transformed cells (Kanner, *et al.*, 1990) and it was subsequently shown to be tyrosine phosphorylated in response to bombesin and other neuropeptides (Zachary, *et al.*, 1992). p130 was cloned by virtue of being a major v-crk associated substrate in v-crk transformed cells, thus p130<sup>cas</sup> is implicated as being a common mediator of v-crk and v-src (Sakai, *et al.*, 1994).

# **1.3.7.7. PAXILLIN**

Glenney and Zokas (1989) also made monoclonal antibodies to tyrosine phosphorylated proteins in v-src transformed cells. One antibody identified a protein which co-localised with vinculin to focal adhesions. This protein was affinity purified and shown to bind to vinculin but not to talin, filamin,  $\alpha$ -actinin or actin and was named paxillin (Turner, *et al.*, 1990). Tyrosine phosphorylation of paxillin, like FAK, had been linked to src activation and was subsequently shown to be concurrently tyrosine phosphorylated with FAK in integrin-mediated cell adhesion (Burridge, *et al.*, 1992), chick development (Turner, *et al.*, 1993) and upon stimulation of quiescent fibroblasts with mitogenic neuropeptides (Zachary, *et al.*, 1992; Zachary, *et al.*, 1993). The co-ordinate tyrosine phosphorylation of FAK and paxillin suggested that paxillin could be a substrate of FAK and this was substantiated by showing that FAK binds to and phosphorylates paxillin (Hildebrand, *et al.*, 1995; Schaller and Parsons, 1995). This association appears to be constitutive because the interaction can be detected in cells in suspension and therefore in the absence of focal adhesions (Hildebrand, *et al.*, 1995).

# **1.3.7.8. BOMBESIN-INDUCED FAK AND PAXILLIN TYROSINE**

#### **PHOSPHORYLATION**

Tyrosine phosphorylation of FAK and paxillin represented a new pathway stimulated by bombesin. This new pathway was especially interesting because it could be dissociated from both PKC activation and Ca<sup>2+</sup> mobilisation (Sinnett-Smith, *et al.*, 1993; Zachary, *et al.*, 1993). The lines of evidence were:

- 1 Bombesin-induced phosphorylation of FAK is consistently greater than the maximum effect elicited by PDB or Ca<sup>2+</sup> mobilising agents.
- 2 Bombesin stimulates FAK tyrosine phosphorylation more rapidly than PDB.
- 3 Bombesin stimulates FAK tyrosine phosphorylation at concentrations considerably lower than those required to activate PKC as judged by phosphorylation of 80K/MARCKS or to mobilise Ca<sup>2+</sup>.
- 4 Downregulation of PKC by chronic pretreatment of cells with PDB abrogates the subsequent effect of PDB but has no substantial effect on the response to either low or high concentrations of bombesin.
- 5 A highly selective inhibitor of PKC blocked the increase in FAK tyrosine phosphorylation induced by PDB but did not impair the response to bombesin.
- 6 Neither Ca<sup>2+</sup> ionophore nor thapsigargin caused a Ca<sup>2+</sup>-dependent increase in FAK phosphorylation
- 7 Depletion of the intracellular Ca<sup>2+</sup> pool by treating cells with thapsigargin blocked the mobilisation of Ca<sup>2+</sup> by a subsequent addition of bombesin but had no effect on FAK phosphorylation.

Because FAK localised to focal adhesions which are where stress fibres terminate it was thought that the actin cytoskeleton could be important in bombesininduced tyrosine phosphorylation. Using cytochalasin D to disrupt the actin network did indeed prevent bombesin-induced FAK phosphorylation (Sinnett-Smith, *et al.*, 1993). Because tyrosine phosphorylation was dependent on the actin cytoskeleton and Rho was upstream of actin stress fibre formation it was hypothesised that Rho was also upstream of neuropeptide-induced tyrosine phosphorylation. This was confirmed when C3 treated cells were stimulated with bombesin and endothelin and the resulting tyrosine phosphorylation of FAK and paxillin was inhibited (Rankin, *et al.*, 1994). Thus Rho was placed upstream of stress fibre assembly, focal adhesion formation and tyrosine phosphorylation of FAK and paxillin.

# 1.3.7.9. ROLE OF FAK AND PAXILLIN TYROSINE PHOSPHORYLATION

The above sections show that FAK and paxillin are implicated in integrating signals from cell adhesion, activated oncogenes and mitogenic neuropeptides (Zachary and Rozengurt, 1992). The formation of focal adhesions requires co-oordinated assembly of multiple proteins, thus the identification of proteins able to interact with FAK and paxillin has been the focus of much attention. Recently FAK has been demonstrated to bind multiple proteins *in vitro* and *in vivo*. These include p130<sup>cas</sup> (Polte and Hanks, 1995), talin (Chen, *et al.*, 1995), the SH2 domains of src and fyn (Cobb, *et al.*, 1994) and Csk (<u>C</u>-terminal <u>Src kinase</u>) (Bergman, *et al.*, 1995).

Similarly, studies with paxillin showed association with multiple proteins. Paxillin binds to the SH3 domain of Src (Weng, *et al.*, 1993) and when phosphorylated on tyrosine, paxillin will associate with the SH2 domain of v-Crk (Birge, *et al.*, 1993). It was subsequently shown that phosphorylation of paxillin by FAK creates the binding site for Crk (Schaller and Parsons, 1995). These investigators also demonstrated the ability of Src and Csk to phosphorylate paxillin *in vitro* and that the SH2 domains of Src and Csk can bind to phosphorylated paxillin.

Upon molecular cloning, the primary amino acid sequence of paxillin revealed the presence of one LIM domain (3) and 3 LIM-like (1,2,4) domains (Turner and Miller, 1994; Salgia, *et al.*, 1995) and see figure 1.9. LIM domains were identified as a cysteine rich motif found in the transcription factors *lin-11*, *isl-1* and *mec-3* (and also in *ceh-14*) where two copies were found in tandem (Freyd, *et al.*, 1990; Karlsson, *et al.*, 1990) and are thought to be involved in protein-protein interactions, similar to dimerisation of other transcription factors. Another gene, CRIP (cysteine rich intestinal peptide), contains just 1 LIM domain and no homeodomain, thus LIM domains can be found in proteins other than transcription factors (Freyd, *et al.*, 1990; Karlsson, *et al.*, 1990).

In addition paxillin has a proline rich region indicative of a potential SH3 binding domain, five tyrosines which have homology to SH2 binding motifs for Src, v-Crk, PLC $\gamma$  and p85, see figure 1.9. Paxillin also had consensus binding sites for PKC (Turner and Miller, 1994; Salgia, *et al.*, 1995).



#### Figure 1.9. Linear diagram of paxillin structure

Shown are potential tyrosine phosphorylation sites and the corresponding recognition by SH2 containing proteins, an SH3-binding region and LIM-like domains.

Thus FAK and paxillin interact with a large number of proteins but the physiological relevance of all these interactions is unknown. It has been suggested that paxillin functions as an adaptor or scaffold onto which many proteins can assemble. The sequential events in the assembly of the focal adhesion is also unknown. FAK kinase activity does not appear to be required for the assembly of focal adhesions (Wilson, *et al.*, 1995). However it does appear that FAK promotes the unstability of focal adhesions, because FAK-deficient mice have focal adhesions but the migration of cells during development is severely retarded (Ilic, *et al.*, 1995).

# **1.3.8.** THE MAPK PATHWAY

#### 1.3.8.1. OVERVIEW

The MAPK pathway is a point of convergence for a variety of mitogenic factors and the oncogene products v-Raf, v-Ras and v-Src. It has a central role in relaying extracellular signals to cellular responses. MAPKs are activated by a variety of mitogens (Rossomando, *et al.*, 1989) and by oncogenic ras (Leevers and Marshall, 1992; Wood, *et al.*, 1992). Because dominant negative N17ras (Feig and Cooper, 1988) abrogated MAPK activity stimulated by PDGF, insulin and NGF (de Vries-Smits, *et al.*, 1992; Robbins, *et al.*, 1992; Thomas, *et al.*, 1992; Wood, *et al.*, 1992), this suggested that MAPK was downstream of ras in a signalling pathway. MEK was identified as a MAPK activator (Gomez and Cohen, 1991; Kosako, *et al.*, 1992; Matsuda, *et al.*, 1992; Nakielny, *et al.*, 1992a; Nakielny, *et al.*, 1992b), and was activated by phosphorylation by a serine/threonine kinase. Thus Ras, MEK and MAPK were put on the same pathway. This was confirmed by reconstitution in a permeabilised cell system (Dent, *et al.*, 1993).

Raf-1 had also been linked to a Ras pathway through work with oncogenic forms of Ras. Antibodies against Ras or a dominant negative Ras did not inhibit v-raf

induced cell transformation (Smith, *et al.*, 1986; Feig and Cooper, 1988). On the contrary dominant negative or antisense Raf could block transformation by Ras (Kolch, *et al.*, 1991). Also, Ras was required for NGF-stimulated Raf phosphorylation and activation (Troppmair, *et al.*, 1992; Wood, *et al.*, 1992). This suggested that Raf-1 was downstream of Ras.

Thus Raf was a candidate for linking Ras and MEK and this was confirmed by showing that v-Raf activated MAPK independently of Ras, and both c-Raf and v-Raf could phosphorylate and activate MEK (Dent, *et al.*, 1992; Howe, *et al.*, 1992; Kyriakis, *et al.*, 1992). The MAPK pathway or MAPK cascade was put together as Ras-Raf-MEK-MAPK.

The importance of this pathway in mitogenic signalling is demonstrated by the ability of v-Ras (Barbacid, 1987) and v-Raf (Rapp, *et al.*, 1983) and constitutively active mutants of MEK (Cowley, *et al.*, 1994; Mansour, *et al.*, 1994) to transform fibroblasts. Also, interfering mutants of, or antibodies against, Ras (Smith, *et al.*, 1986; Feig and Cooper, 1988), Raf (Kolch, *et al.*, 1991), MEK (Cowley, *et al.*, 1994), and MAPK (Pages, *et al.*, 1993; Troppmair, *et al.*, 1994) inhibit cell proliferation in response to mitogenic stimuli such as oncogenes and growth factors. This pathway is not only involved in proliferation, prolonged stimulation of the MAPK cascade induces differentiation in PC12 cells (Cowley, *et al.*, 1994; Marshall, 1995). The activation of the individual components shall be described in detail.

# 1.3.8.2. MAPK

For over a decade it has been observed that bands of 42 and 44 kDa have been phosphorylated on tyrosine in response to polypeptide growth factors, such as PDGF, EGF and IGF; the proteases thrombin and trypsin and phorbol esters (Hunter and Cooper, 1985). Rossomando, *et al.* (1989) proposed this to be equivalent to a serine/threonine kinase, also ≈40 kDa and phosphorylated on tyrosine (Ray and Sturgill, 1988), that was activated by insulin, EGF, TPA and serum (Ray and Sturgill, 1987; Hoshi, *et al.*, 1988) and which phosphorylated MAP-2, myelin basic protein and S6 kinase II (also known as pp90<sup>rsk</sup> or RSK) (Ray and Sturgill, 1987; Sturgill, *et al.*, 1988; Erickson, *et al.*, 1990). It was called MAPK because it was a MAP2 kinase and a <u>mitogen activated protein kinase</u> (Rossomando, *et al.*, 1989).

The molecular cloning of MAPK revealed a family of protein kinases with homology to the yeast kinases FUS3 and KSS1 which mediate the yeast response to pheromones. This was evidence of an evolutionary conserved pathway which was regulated in response to extracellular signals to promote entry into the cell cycle. Therefore the proteins encoded by the cDNAs isolated were called <u>extracellular signal-</u> regulated <u>kinase</u>, ERK1 and ERK2, and these corresponded to pp44<sup>mapk</sup> and pp42<sup>mapk</sup> respectively (Boulton, *et al.*, 1990; Boulton, *et al.*, 1991). MAPK is phosphorylated on threonine and tyrosine residues in response to mitogens (Ray and Sturgill, 1988) and both of these residues are required to be phosphorylated for MAPK activity (Anderson, *et al.*, 1990). They were identified as Thr 183 and Tyr 185 (Payne, *et al.*, 1991).

# 1.3.8.3. MEK

MAPK activators were purified from fractions of stimulated cell lysates that caused the phosphorylation of MAPK on Thr and Tyr (Ahn, et al., 1991; Gomez and Cohen, 1991). Because recombinant MAPK could autophosphorylate at a very low level (Seger, et al., 1991) it was thought that MAPK activator could be enhancing the autophosphorylation activity. MAPK activator was shown to be a kinase itself by removing the kinase activity of MAPK by either a kinase inhibitor (5'-pfluorosulfonylbenzoyladenosine) (Adams and Parker, 1992) or by kinase inactive mutants (Nakielny, et al., 1992b; Seger, et al., 1992a). Kinase deficient MAPK was still phosphorylated demonstrating that the activator was a kinase itself. This was confirmed upon molecular cloning of MAPK kinase (Crews, et al., 1992; Seger, et al., 1992b) and the protein product was called MAPK/ERK kinase, MEK. MEK phosphorylates MAPK on thr and tyr residues demonstrating that it is a "dual specificity" kinase (Kosako, et al., 1992; Nakielny, et al., 1992a; Seger, et al., 1992a). MEK itself is a phosphoprotein and dephosphorylation by protein phosphatase 2A, a serine/threonine phosphatase, inactivates it (Gomez and Cohen, 1991; Matsuda, et al., 1992).

# 1.3.8.4. RAF

Identification of the MEK activator came from the observation that both Raf-1 and MAPK were downstream of Ras (see section 1.3.8.1). The activation of MAPK and MEK by Raf was looked for directly. MAPK and MEK activities were stimulated in cells infected with viruses containing v-Raf (Dent, *et al.*, 1992; Howe, *et al.*, 1992; Kyriakis, *et al.*, 1992) and also v-Raf activated MEK *in vitro* (Dent, *et al.*, 1992; Kyriakis, *et al.*, 1992). In addition immunoprecipitated Raf-1 from phorbol ester-stimulated cells activated MEK *in vitro* (Howe, *et al.*, 1992). Activation was shown to be due to phosphorylation of MEK by Raf-1 on two serine residues between subdomains VII and VIII of the kinase region (Huang, *et al.*, 1993; Alessi, *et al.*, 1994).

The activation of Raf-1 itself is not so well defined. (Moodie, *et al.*, 1993) were looking for proteins which bound to immobilised Ras and they identified Raf-1 and MEK which bound to Ras in a GTP-dependent manner. Similarly (Koide, *et al.*, 1993) added Ras.GTP to cell lysates and co-immunoprecipitated Raf-1. Because Raf-1 was localised to the plasma membrane in cells with active Ras, Leevers, *et al.* (1994) and Stokoe, *et al.* (1994) put a membrane localisation signal on Raf-1 thus bypassing the requirement for Ras. Membrane associated Raf-1 was activated independently of Ras,

but could also be stimulated further by treating the cells with EGF (Leevers, *et al.*, 1994). Thus active Ras functions by bringing Ras to the plasma membrane where other event(s) take place.

These activation event(s) appear to be phosphorylation of Raf. Morrison, *et al.* (1988) observed that in mitogen treated cells Raf-1 was phosphorylated on serine and threonine residues, and also on tyrosine residues in v-src transformed cells or after stimulation by PDGF. A more detailed analysis revealed that Ser 43, Ser 259 and Ser 621 were the major sites of serine phosphorylation in PDGF stimulated cells (Morrison, *et al.*, 1993) and phosphorylation of Tyr 340 and Tyr 341 in response to src, lck and the activated PDGF receptor were important for activity (Fabian, *et al.*, 1993; Marais, *et al.*, 1995). An *in vitro* study with plasma membranes from serum stimulated, v-ras or v-src transformed cells also showed that there was ras-dependent and ATP-dependent step(s) for Raf-1 activation (Dent and Sturgill, 1994).

PKC has been suggested to be the activator of Raf-1. Co-expression of cPKCs and Raf-1 in insect cells resulted in the phosphorylation and activation of Raf-1 (Sozeri, *et al.*, 1992). In NIH 3T3 cells PKC $\alpha$  induced the phosphorylation of Raf-1 on Ser 499. This serine was essential for activation by PKC $\alpha$  but not activation by Ras and the tyrosine kinase, lck (Kolch, *et al.*, 1993). Similarly Carroll and May (1994) showed PKC-mediated phosphorylation of Ser 497 and Ser 619. The relevance of the Ser 497/499 phosphorylation is unclear because these residues are not phosphorylated in PDGF stimulated cells (Morrison, *et al.*, 1993). Raf-1 activation could be dependent on tyrosine phosphorylation. Morrison, *et al.* (1988) was unable to detect tyrosine phosphorylation of Raf-1 from cells treated with phorbol ester, EGF, FGF or H-ras.

It has been suggested there are two mechanisms of activation of Raf-1. One is due to membrane association and another due to mitogenic stimulation (Leevers, *et al.*, 1994) or the tyrosine kinase v-src (Marais, *et al.*, 1995). Consistent with this is the suggestion that lipids play a role in regulating Raf-1 activity (Cai, *et al.*, 1993; Ghosh, *et al.*, 1996).

Other activators of Raf-1 could be the 14-3-3 proteins. These are a family,  $\alpha$ - $\eta$ , of proteins that were originally isolated by systematic analysis of brain proteins. Subsequently they have been shown to exist as dimers and are thought to be regulators in signal transduction or phosphorylation mechanisms (Aitken, *et al.*, 1992). Using yeast two hybrid screens or co-immunoprecipitation, 14-3-3 $\beta$  and 14-3-3 $\zeta$  were identified as Raf-1 binding proteins which also increased Raf-1 activity (Aitken, 1995; Burbelo and Hall, 1995). The crystal structure reveals that 14-3-3 are right angled structures and they dimerise to form a clamp or a channel. The amino-terminal helices of the two subunits contact one another and form the floor of the channel, while the carboxy-terminal helices form the walls. It is suggested that because

heterodimers form 14-3-3 can act as a scaffold to bring signalling molecules together (Bax and Jhoti, 1995). In this context it is interesting that 14-3-3 have been shown to regulate PKC activity (Aitken, *et al.*, 1992; Aitken, 1995) and to bind to PI3K 110kDa catalytic subunit (14-3-3 $\tau$ ) (Bonnefoy-Berard, *et al.*, 1995) and Bcr (14-3-3 $\beta$ ) (Braselmann and McCormick, 1995). Because Raf-1 does not interact with Bcr itself but does in the presence of 14-3-3, these investigators propose that 14-3-3 acts as a linker or scaffold. Consistent with this (Luo, *et al.*, 1995) describe that truncation of the 14-3-3 amino terminal domain does not prevent 14-3-3 interaction with Raf-1 but there is reduced dimerisation of 14-3-3, and Raf-1 bound to these monomeric C-terminal domains is inactive.

Thus there appears to be multiple mechanisms of Raf-1 activation, and this could reflect the variety of factors that signal through Raf-1.

There are three characterised mammalian Raf proteins, c-Raf or Raf-1, A-Raf and B-Raf. They share three highly conserved regions (CR1-3) surrounded by variable sequences, CR1 and CR2 play a regulatory role and CR3 contains the catalytic kinase domain (Daum, et al., 1994). CR1 consists of the binding domain for Ras and a cysteine rich region with a zinc finger motif. CR2 is rich in serine and threonine residues some of which are regulatory phosphorylation sites but these sites differ between the isoforms and can result in different regulatory mechanisms. For example in PC12 cells TPA transiently activates Raf-1 which is partially PKC-dependent, but gives a prolonged activation of A-Raf which is completely PKC-dependent (Bogoyevitch, et al., 1995; Erhardt, et al., 1995). Also it is well documented that there are Raf-1 independent mechanisms of ERK activation (Chao, et al., 1994; Vaillancourt, et al., 1994; Zheng, et al., 1994; Erhardt, et al., 1995; Faure and Bourne, 1995) There are other MEK activators which are not yet fully characterised. These include a 40-50 kDa MEK activator in NIH 3T3 cells which is stimulated by PDB, thrombin and PDGF (Reuter, et al., 1995) and a 45-50 kDa MEK activator in PC12 cells after NGF stimulation (Pang, et al., 1995).

# 1.3.8.5. RAS

The viral oncogenes of certain acute transforming retroviruses were the first ras genes to be identified. There are three functional ras genes: H-ras (v-H-ras of Harvey murine sarcoma virus), K-ras (of v-K-ras of Kirsten murine sarcoma virus) and N-ras (transforming gene of neuroblastoma cell line), reviewed by Lowy and Willumsen (1993). Ras undergoes a series of postranslational modifications including farnesylation and C-terminal methylation resulting in association with the inner face of the plasma membrane. Mutant proteins with alterations in this residue are cytosolic and transformation defective.  $p21^{ras}$  is a low molecular weight monomeric GTP binding protein. Similar to the Ga subunit of heterotrimeric G proteins, Ras is inactive when GDP is bound and is activated upon the binding of GTP. Ras can be activated in two ways:

1. The intrinsic rate of GDP dissociation is very low and this rate can be increased by guanine nucleotide exchange factors. This allows the more abundant GTP to bind and activate Ras. Examples of these proteins are the drosophila son of sevenless (SOS) and its mammalian counterpart (mSOS).

2. Ras possesses intrinsic GTPase activity that limits the life time of the active Ras.GTP species and this can be activated by GTPase activating proteins, such as p120ras-GAP, and neurofibromin 1. These proteins are therefore negative regulators of Ras and so Ras can be activated by the inhibition of GAPs (Lowy and Willumsen, 1993).

A point mutation at position 17 for asparagine (N17ras) results in a lower affinity for GTP. This Ras mutant acts in a dominant negative fashion and is often used to inhibit endogenous Ras activity by sequestering upstream regulatory proteins (Feig and Cooper, 1988).

Several groups showed that stimulating various cell types with PDGF or EGF or introducing v-src all increased the amount of GTP bound to Ras (Gibb, *et al.*, 1990; Satoh, *et al.*, 1990a; Satoh, *et al.*, 1990b). Although receptor tyrosine kinases bind and phosphorylate GAP (Molloy, *et al.*, 1989) they do not change GAP activity (Zhang, *et al.*, 1992; Buday and Downward, 1993b). Instead in EGF-stimulated cells there is an increase in the nucleotide exchange rate (Buday and Downward, 1993b; Medema, *et al.*, 1993). The clue for the mechanism of activation of Ras in fibroblasts by growth factors came from genetic analysis of eye development in *Drosophila melanogaster* and vulval development in *Caenorhabditis elegans*.

Son of Sevenless (Sos), a GEF that activated Ras, was identified downstream from the drosophila receptor tyrosine kinase, Sevenless (Simon, *et al.*, 1991; Bonfini, *et al.*, 1992). Then Lowenstein, *et al.* (1992), looking for EGF receptor binding proteins, isolated an adaptor molecule of 23 kDa which consisted of SH3-SH2-SH3 and called it growth factor receptor <u>b</u>ound protein <u>2</u> (Grb2). Grb2 bound to specific phosphotyrosine residues of activated PDGF and EGF receptors via its SH2 domain but it was not tyrosine phosphorylated itself. By analogy with the structurally similar Sem-5 from *C. elegans*, Grb2 was shown to potentiate Ras-induced DNA synthesis upon co-microinjection and mutants of Grb2 or isolated Grb2 domains abolished Ras induced DNA synthesis. This showed that intact Grb2 was important for Ras function (Lowenstein, *et al.*, 1992).

Another adaptor molecule, Shc (<u>s</u>rc-<u>h</u>omology and <u>c</u>ollagen), has a C-terminal SH2 domain and an adjacent proline rich motif with regions of homology with

collagen. Shc is tyrosine phosphorylated upon cellular stimulation by EGF (Pelicci, *et al.*, 1992), or transformation by v-src (McGlade, *et al.*, 1992). Co-immunoprecipitation showed that Grb2, via its SH2 domain, binds to tyrosine phosphorylated Shc. Overexpression of Shc is transforming in fibroblasts and induces Ras-dependent neurite outgrowth in PC12 cells. This suggested that a Shc/Grb2 complex regulates Ras activity (Rozakis-Adcock, *et al.*, 1992).

Grb2 was then shown to bind constitutively to Sos, or its mammalian counterpart mSos, by direct interactions between both SH3 domains of Grb2 and the proline-rich regions in Sos (Buday and Downward, 1993a; Chardin, *et al.*, 1993; Egan, *et al.*, 1993; Gale, *et al.*, 1993; Li, *et al.*, 1993; Rozakis-Adcock, *et al.*, 1993). In addition, upon growth factor stimulation, the GEF activity of Sos does not change but there is a translocation from the cytosol to the membrane (Buday and Downward, 1993a).

Thus, in the resting state the Grb2/mSos complex is present in the cytosol. Upon activation, growth factor receptors become tyrosine phosphorylated and this creates a specific binding site for Grb2. Grb2 associates with the receptor at the membrane creating a receptor/Grb2/mSos complex. In this manner, mSos is translocated to the membrane where its substrate, Ras, is localised (McCormick, 1993). In addition the tyrosine phosphorylation of Shc by the activated EGF receptor could also act as a docking site for Grb2/mSos and cause the activation of Ras (Egan, *et al.*, 1993; Rozakis-Adcock, *et al.*, 1993). This is summarised in figure 1.10.

#### **1.3.8.6.** EFFECTS OF MAPK

The activation of MAPK has pleiotrophic effects. Upon activation, ERK1 and ERK2 (but not MEK) enter the nucleus within 5 min (Chen, *et al.*, 1992; Lenormand, *et al.*, 1993). Substrates for ERK phosphorylation are transcription factors and nuclear proteins (Elk-1); upstream components of the MAPK pathway for feed back regulation such as the EGF receptor, SOS, Raf-1, MEK, and downstream kinases such as RSK and MAPKAP kinase 2; cytoskeletal elements such as the microtubule associated proteins (MAPs); and other signal transduction molecules such as PLA<sub>2</sub> (Malarkey, *et al.*, 1995).

# **1.3.8.7.** ACTIVATION OF THE MAPK PATHWAY BY G PROTEIN-COUPLED RECEPTORS

The MAPK pathway described above was elucidated for growth factors which signal through receptor tyrosine kinases such as EGF, PDGF, NGF and insulin, however ligands which stimulate seven transmembrane receptors also activate MAPK. Tyrosine phosphorylation of 40 - 45 kDa bands were first reported for thrombin (Hunter, *et al.*, 1985) and MAPK activity was shown for bombesin (Takuwa, *et al.*, 1991a) followed by endothelin (Wang, *et al.*, 1992), bradykinin (Ahn, *et al.*, 1992) and angiotensin (Duff, *et al.*, 1992; Ishida, *et al.*, 1992). Receptors coupled to G<sub>i</sub>, G<sub>s</sub> and G<sub>q</sub> have been shown to

#### **INTRODUCTION**

#### Chapter 1

stimulate MAPK activity but the mechanism of activation differs between the classes. Mechanisms of MAPK activation shall be described for G<sub>i</sub> and G<sub>q</sub>-coupled receptors.



#### Figure 1.10. Receptor tyrosine kinase activation of the MAPK pathway

A kinase cascade is initiated by growth factor activating their receptor tyrosine kinase or in cells transformed by tyrosine kinases such as v-src. GTP binds to ras which causes the association with Raf-1 at the plasma membrane where Raf-1 is activated by an unknown mechanism. Raf-1 phosphorylates and activates MEK which phosphorylates and activates MAPK.

#### Receptors coupled to G<sub>i</sub>

 $M_2$ -Ach-Ach and  $\alpha_1$ -AR receptors were shown to stimulate Raf-1, MEK and MAPK activity (Alblas, *et al.*, 1993; Winitz, *et al.*, 1993), and LPA and thrombin were shown to increase the proportion of GTP bound to Ras (Van Corven, *et al.*, 1993). In addition, thrombin-stimulated mitogenesis was blocked by microinjection of N17ras or anti-ras antibodies (LaMorte, *et al.*, 1993b). Activation of Ras, Raf-1 and MAPK were all shown to be pertussis toxin-sensitive consistent with G<sub>i</sub>-mediated signal transduction pathways.

However, this was not mediated by  $G\alpha_i$  as overexpression of activated mutants did not significantly enhance MAPK activity, instead overexpression of  $\beta$  and  $\gamma$ subunits together increased MAPK activity (Crespo, *et al.*, 1994b; Faure, *et al.*, 1994). Also endogenous  $\beta\gamma$  could be sequestered by introduction of  $G\alpha_t$  or the PH domain of  $\beta$ ARK and this inhibited  $\beta\gamma$  and agonist stimulation of MAPK (Crespo, *et al.*, 1994b;
Faure, *et al.*, 1994; Koch, *et al.*, 1994). N17ras inhibited agonist- and  $\beta\gamma$ -stimulated MAPK activity (Crespo, *et al.*, 1994b) and agonist-stimulated Ras.GTP loading was inhibited by  $\beta$ ARK (Koch, *et al.*, 1994). It can be concluded that receptors which couple to G<sub>i</sub> activate the MAPK pathway via  $\beta\gamma$ -mediated ras activation, followed by activation of Raf-1, MEK and MAPK. Thus G<sub>i</sub> coupled receptors use the same signalling pathways as receptor tyrosine kinases. This extends to the activation of Ras, disruption of the Grb2/Sos complex with truncated versions of Sos without GEF activity prevented agonist and  $\beta\gamma$  stimulated MAPK activity (van Biesen, *et al.*, 1995). It was also shown that  $\beta\gamma$  subunits caused the phosphorylation of Shc which co-immunoprecipitated with Grb2 and mSos (Touhara, *et al.*, 1995; van Biesen, *et al.*, 1995). Thus Shc may represent the link between  $\beta\gamma$  activation of a tyrosine kinase and Ras activation. This is summarised in figure 1.11.

#### Receptors coupled to G<sub>q</sub>

Receptors coupled to  $G_q$  do not generally activate the MAPK pathway through stimulation of Ras. Bombesin, endothelin and M<sub>1</sub>-Ach do not cause GTP loading of Ras in Rat-1 and Swiss 3T3 cells (Satoh, *et al.*, 1990b; Van Corven, *et al.*, 1993; Winitz, *et al.*, 1993) and  $\alpha_1$ -AR-stimulated MAPK is not inhibited by N17ras (Hawes, *et al.*, 1995).

Activation of the Raf-1 isoform is unclear: stimulation of CCK<sub>B</sub> and M<sub>1</sub>-Ach receptors activate Raf-1 in Rat-1 and COS 7 cells (Crespo, *et al.*, 1994a; Seufferlein, *et al.*, 1995) and  $\alpha_1$ -AR-stimulated MAPK is inhibited in COS 7 by a dominant negative Raf-1 (Hawes, *et al.*, 1995). In contrast, it has been shown that M<sub>1</sub>-Ach does not stimulate Raf-1 in Rat-1 cells (Winitz, *et al.*, 1993; Russell, *et al.*, 1994) and bombesin does not stimulate Raf-1 activity in Swiss 3T3 cells (Mitchell, *et al.*, 1995; Seufferlein, *et al.*, 1996).

Activation of seven transmembrane receptors that are coupled to  $G_q$  leads to stimulation of PKC activity (see section 1.3.3). Activation of MAPK is largely PKC-dependent for a variety of agonists such as bombesin in Swiss 3T3 cells (Pang, *et al.*, 1993; Seufferlein, *et al.*, 1996),  $\alpha_1$ -AR in COS 7 (Hawes, *et al.*, 1995), M<sub>1</sub>-Ach in NIH 3T3 cells (Crespo, *et al.*, 1994a), endothelin in mesangial cells (Wang, *et al.*, 1992), and M<sub>1</sub>-Ach in COS 7 (Crespo, *et al.*, 1994b).

Transfection of activated  $\alpha_q$  subunits into COS 7 cells activates MAPK (Faure, *et al.*, 1994). Consistent with this is the demonstration that  $\alpha_1$ -AR stimulated MAPK is not inhibited by the  $\beta$ ARK PH domain (Koch, *et al.*, 1994; Hawes, *et al.*, 1995) and bombesin-stimulated MAPK is not inhibited by G $\alpha_t$  (Faure, *et al.*, 1994).

Thus  $G_q$ -coupled receptors activate MAPK by an  $\alpha_q$ - and PKC-dependent but  $\beta_{\gamma}$ and Ras-independent pathway (Hawes, *et al.*, 1995) and see figure 1.11, although it is unclear whether the MAPK kinase kinase involved is Raf-1.

#### Chapter 1

#### **INTRODUCTION**

The mechanism by which PKC activates the MAPK pathway is still unclear. In section 1.3.8.4 it was described that phorbol esters cause the activation of Raf-1 (Morrison, *et al.*, 1988; Howe, *et al.*, 1992; Kyriakis, *et al.*, 1993) and PKC may activate Raf-1 directly (Sozeri, *et al.*, 1992; Kolch, *et al.*, 1993; Carroll and May, 1994). Studies with dominant negative Raf have showed that Raf-1 is necessary for phorbol esterstimulated MAPK activity and/or cell proliferation (Kolch, *et al.*, 1991; Troppmair, *et al.*, 1994). Therefore, in general, PKC can activate Raf-1 but there are exceptions, Chao, *et al.* (1994) showed that PKC activation of MAPK is not dependent on Raf-1. However there are multiple enzymes that function at the level of Raf-1. These are summarised in figure 1.12.



# Figure 1.11. Mechanism of MAPK activation by G protein coupled receptors

#### **1.3.8.8. BOMBESIN-INDUCED MAPK ACTIVATION**

The mechanism for MAPK activation by bombesin in Swiss 3T3 cells is consistent with the model for general  $G_q$ -coupled receptor activation. Addition of bombesin to Swiss 3T3 cells induces a rapid activation of MAPK that is maximum at 5 min and then declines to a plateau of a low level of MAPK activity which persists for several hours (Withers, *et al.*, 1995). After prolonged treatment with phorbol ester to down regulate phorbol ester-sensitive isoforms, bombesin-induced MAPK is abolished

73

showing complete dependence on PKC (Pang, *et al.*, 1993; Seufferlein, *et al.*, 1996). Bombesin does not increase Ras.GTP loading in Swiss 3T3 cells (Satoh, *et al.*, 1990b; Mitchell, *et al.*, 1995), nor does it activate Raf-1 (Mitchell, *et al.*, 1995; Seufferlein, *et al.*, 1996). Thus bombesin activates MAPK via a PKC-dependent pathway. Consistent with this is the observation that in COS 7 cells bombesin-stimulated MAPK was not dependent on  $\beta\gamma$  subunits (Faure, *et al.*, 1994).

#### **1.3.8.9.** CELLULAR CONTEXT

Some studies indicate that Ras could be involved in MAPK activation by  $G_q$ coupled receptors. Activation of the  $M_1$ -Ach, 5-HT<sub>2B</sub> or  $\alpha_{1B}$ -AR receptor stimulates Ras.GTP loading (Koch, *et al.*, 1994; Mattingley, *et al.*, 1994; Launay, *et al.*, 1996) and addition of cholecystokinin to rat pancreatic acinar cells activates Ras (Duan, *et al.*, 1995). Similarly,  $M_1$ -Ach-stimulated MAPK is inhibited by N17ras and G $\alpha_t$  (Crespo, *et al.*, 1994b). MAPK activation mediated by stimulation of the CCK<sub>B</sub> receptor is PKCindependent in Rat-1 cells (Seufferlein, *et al.*, 1995). This could describe a pathway that is PKC-independent and  $\beta_{\gamma}$ - and Ras-dependent (Crespo, *et al.*, 1994a; Crespo, *et al.*, 1994b).

These differences can be attributed to several factors such as receptor identity and experimental procedure. However an important aspect is the cellular context in which the MAPK pathway is activated. A study by Faure and Bourne (1995) demonstrates the variability in activating the MAPK cascade in three cell types, Swiss 3T3, COS-7 and Rat-1. These investigators show that EGF and LPA, which signal via receptor tyrosine kinase and G<sub>i</sub>-coupled receptor respectively, proportionally activate Raf-1, MEK and p42<sup>mapk</sup> to different extents between the three cell types. Also the effect of elevating cellular levels of cAMP inhibited these kinases to different levels such that MAPK and MEK were inhibited in Rat-1 cells but not in COS-7 or Swiss 3T3 cells.

Another study compared cellular context of  $G_q$ -coupled receptors by use of the  $M_1$ -Ach receptor in COS-7 cells versus CHO cells (van Biesen, *et al.*, 1996). It was demonstrated that in COS-7 cells  $M_1$ -Ach receptors activate MAPK via a PKC-dependent and  $\beta\gamma$ - and Ras-independent pathway that was pertussis toxin-insensitive whereas in CHO cells  $M_1$ -Ach activation of MAPK was sensitive to this toxin. This was attributed to  $M_1$ -Ach coupling to  $G\alpha_0$  in CHO cells. Interestingly, the  $\alpha_1$ -AR retains COS-7-like  $G_q$  coupling characteristics in CHO cells. To conclude, the mechanism by which  $G_q$ -linked receptors activate MAPK can vary according to receptor identity and cellular context.

#### **1.3.8.10.** FUNCTION OF $G\beta\gamma$

#### Activation of tyrosine kinases

G protein-coupled receptor activation of Ras requires the Grb2/mSos interaction (van Biesen, *et al.*, 1995). Also,  $\beta\gamma$  subunits (Touhara, *et al.*, 1995; van Biesen, *et al.*, 1995) and agonists for the receptors LPA,  $\alpha_{2A}$ -AR, angiotensin II, thrombin and gastrin (van Biesen, *et al.*, 1995; Chen, *et al.*, 1996; Sadoshima and Izumo, 1996; Seva, *et al.*, 1996) increase the phosphorylation of Shc on tyrosine. The model proposed is that Grb2 binds to the phosphotyrosine residue of Shc to bring mSos in contact with Ras (Touhara, *et al.*, 1995; van Biesen, *et al.*, 1995).

So what tyrosine kinases do  $\beta\gamma$  activate? It has been shown that  $\beta\gamma$  together with another membrane component activate the tyrosine kinases Btk and Tsk (Langhans-Rajasekaran, *et al.*, 1995) both of which contain PH domains. Btk and Tsk are found in B and myeloid cells and T cells respectively but there may be an analogous situation in fibroblasts. The possibility that Raf-1 or Shc were substrates of Tsk or Btk was not tested (Langhans-Rajasekaran, *et al.*, 1995). In the pheromone signalling pathway of *Saccharomyces cerevisiae* a serine kinase, Ste20p, has been identified to be activated by  $\beta\gamma$  (Wu, *et al.*, 1995a).

Other tyrosine kinases could be the EGF receptor or Src. Kinase competent EGF receptor chains were necessary for the activation of MAPK by thrombin, endothelin and LPA although  $\beta\gamma$  subunits were not tested on this model (Daub, *et al.*, 1996). Thrombin and angiotensin activate src family kinases (Chen, *et al.*, 1994; Sadoshima and Izumo, 1996) although activation of Src by  $\beta\gamma$  has not been demonstrated.

Recently  $\beta\gamma$  have been shown to activate novel isoforms of PI3K in platelets, neutrophils and U937 cells that are distinct from the type that is activated by receptor tyrosine kinases (Stephens, *et al.*, 1994; Thomason, *et al.*, 1994; Stoyanov, *et al.*, 1995) and  $\beta\gamma$  stimulation of Shc phosphorylation is inhibited by the PI3K inhibitor wortmannin (Touhara, *et al.*, 1995), although these investigators do point out that PI3K can associate with Src family kinases independently of agonist stimulation and the effect of wortmannin may be disrupting Src activity. This would imply that Src is involved in  $\beta\gamma$ -mediated Shc phosphorylation as suggested above. Hu, *et al.* (1996) also shows  $\alpha_1$ -AR activation of PI3K and wortmannin inhibition of Ras.GTP loading. Similarly wortmannin inhibits LPA and  $\alpha_2$ -AR stimulated Ras.GTP loading (Hawes, *et al.*, 1996). These observations have parallels to Ras activation described by Valius and Kazlauskas (1993). These investigators demonstrated that PDGF receptor mutants that only had the tyrosine residue specific for binding the p85 regulatory subunit of PI3K, but not Grb2, were able to activate Ras and induce DNA synthesis.

#### **Construction of multimeric complexes**

It was described in section 1.1.6.4 that a function of the  $\beta$  subunit could be to build multiprotein complexes.  $\beta\gamma$  appears to localise the tyrosine kinases Btk and Tsk to the membrane (Langhans-Rajasekaran, *et al.*, 1995) and initiate the formation of the Shc/Grb2/Sos complex by stimulating the tyrosine phosphorylation of Shc.

Raf is complexed with 14-3-3 and Bcr (Braselmann and McCormick, 1995) the latter of which has a PH domain and therefore could interact with  $\beta\gamma$ . Section 1.3.8.4 describes how 14-3-3 proteins could be involved in activation of Raf-1 by acting as linkers between signal transduction molecules. Thus 14-3-3 and/or  $\beta\gamma$  may contribute to building the complex required for Raf-1 activation at the membrane. Consistent with this is the demonstration that  $\beta\gamma$  are found in complexes with Ras, Raf and MEK (Wang and Durkin, 1995) and shown to bind Raf-1 (Pumiglia, *et al.*, 1995).

In conclusion G protein  $\beta\gamma$  subunits are implicated in many protein-protein interactions but further work is needed to clarify their role in activation of Ras and the MAPK pathway.

#### **1.3.8.11. STRESS-ACTIVATED PROTEIN KINASES**

The MAPK pathway described above is activated by mitogens, however there are other MAPK subtypes which are activated by homologous pathways in response to stress (Cano and Mahadevan, 1995; Cobb and Goldsmith, 1995). The first of these alternative cascades culminates in the activation of MAPKs called JNK (c-Jun <u>N</u>-terminal <u>kinase</u>) or SAPK (<u>stress-activated protein kinase</u>) and these are generally activated in response to UV irradiation and protein synthesis inhibitors. The other cascade culminates in the activation of p38/RK which shows the highest homology to yeast high-osmolarity glycerol response 1 (HOG1) suggesting it plays a part in mammalian osmotic sensors. The p38 cascade is also activated by heat shock. These cascades have their respective MAPK kinases and MAPK kinase kinases to make distinct MAPK modules although there does appear to be cross talk between the pathways at the level of MAPKKK (Cano and Mahadevan, 1995; Cobb and Goldsmith, 1995). These are summarised in figure 1.12.



Figure 1.12. Multiple mammalian MAPK modules

#### **1.3.9.** INDUCTION OF EARLY GENES

In addition to the events in the membrane and cytosol described above, bombesin rapidly and transiently induces the expression of the cellular oncogenes c*fos* and c-*myc* in quiescent fibroblasts (Palumbo, *et al.*, 1986; Rozengurt and Sinnett-Smith, 1987). Since these cellular oncogenes encode nuclear proteins, it is plausible that their transient expression may play a role in the transduction of the mitogenic signal to the nucleus (Lewin, 1991). The demonstration that the product of the protooncogene c-*jun*, identified as a major component of the transacting factor AP-1, forms a tight complex with Fos protein is consistent with a role of c-*fos* in the regulation of gene transcription (Lewin, 1991).

#### 1.3.9.1. AP-1

Activator protein-1 (AP-1) was first identified as a transcription factor that binds to an essential cis element of the human metallothionein  $II_A$  promoter (Angel and Karin, 1991). The binding site for AP-1 was recognised as the TPA response element (TRE) of several cellular genes including collagenase, stromelysin and interleukin-2 whose transcription is induced after 8 - 12 hours in response to treatment of cultures cells with phorbol ester. AP-1 consists of Jun proteins (c-Jun, JunB and JunD) and Fos proteins (c-Fos, FosB, Fra1 and Fra2). Jun proteins can form homo- and heterodimers whereas Fos proteins do not associate with each other. As dimerisation is a prerequisite for DNA binding, Fos does not stimulate transcription in the absence of Jun. However, Fos proteins can associate with any of the Jun proteins to generate stable heterodimers that have higher DNA binding activity than Jun homodimers. As well as being responsive to PKC, activation of the AP-1 complex serves as a nuclear target for signalling pathways elicited by mitogenic stimulation or acute expression of various oncoproteins such as Src, H-ras, Raf or Mos (Angel and Karin, 1991).

## **1.3.9.2.** INDUCTION OF AP-1 COMPONENTS

#### c-fos

c-*fos* transcription is increased within minutes in response to various stimuli such as serum, EGF, PDGF, phorbol ester, cAMP, neurotransmitters, TSH and Ca<sup>2+</sup> ionophore (Curran, *et al.*, 1985; Angel and Karin, 1991). c-*fos* is under the control of several upstream enhancer elements, the *sis* inducible element (SIE) at -345, the serum response element (SRE) at -300, and the cAMP response element (CRE) at -60 (see figure 1.13) (Janknecht, *et al.*, 1995). The SIE is responsive to cytokines and some growth factors such as PDGF (v-sis), but this shall not be described here. The SRE mediates serum induction of c-*fos* and also that of individual growth factors such as PDGF, EGF and phorbol ester. The SRE is recognised by the serum response factor (SRF) whose binding results in the recruitment of a ternary complex factor (TCF), which could be Elk-1 or SAP-1, which cannot bind to the SRE by itself. Following mitogenic stimulation, Elk-1 is rapidly phosphorylated by the ERK class of MAPKs which increase its transcriptional activity (Karin, 1995).

The CRE represents a binding site for CREB and related transcription factors. CREB is a well known substrate for PKA and phosphorylation increases its transactivation activity. Also CREB is shown to be phosphorylated by CaMKs activated by an increase in Ca<sup>2+</sup> (Janknecht, *et al.*, 1995).

It is unclear *in vivo* whether these elements act independently because signals which activate the SIE and CRE also activate the SRE. In Swiss 3T3 cells elevation of cAMP is a mitogenic signal but does not induces c-*fos*, similarly treatment with Ca<sup>2+</sup> ionophores does not induce this proto-oncogene. However both cAMP and Ca<sup>2+</sup> elevation can synergise with PKC activation to induce c-*fos* expression (Mehmet and Rozengurt, 1991).

#### c-jun

Unlike c-Fos, cells have a higher basal levels of c-Jun. Also the induction of c-Jun is not as transient as that of c-Fos (Angel and Karin, 1991). The c-*jun* promoter is simpler than the c-*fos* promoter in that it contains just one major cis acting element which is a modified TRE. c-Jun binds to this promoter upon dimerisation with ATF2 and this complex is activated when c-Jun and ATF2 becomes phosphorylated by active JNK (Karin, 1995).



#### Figure 1.13. Activation of AP-1 transcription factor

The MAPK pathway culminates in ERK-dependent phosphorylation of Elk-1 (TCF) which activates transcription of the *c-fos* gene. The JNK pathway phosphorylates and activates preexisting c-Jun and ATF2 which induce transcription of c-*jun.* c-Fos then dimerises with c-Jun to form the AP-1 complex which activates the transcription of early response genes. (TRE\*, modified TRE)

There is evidence implicating PKC activation in the sequence of events linking bombesin receptor occupancy and proto-oncogene induction (Rozengurt and Sinnett-Smith, 1988b). Accordingly, bombesin-induced oncogene expression is markedly reduced by down regulation of PKC. PKC activation leads to the activation of MAPK (see section 1.3.8) which directly phosphorylates transcription factors resulting in the increased expression of c-*fos*. However, neither direct activation of PKC by phorbol esters nor addition of vasopressin evoke a maximal increase in c-*fos* mRNA levels. It is likely that the induction of c-*fos* by bombesin is mediated by the coordinated effects of PKC activation, Ca<sup>2+</sup> mobilisation and an additional pathway dependent on arachidonic acid release (Rozengurt, 1991a).

## **1.4.** COUPLING TO MULTIPLE PATHWAYS

Signal transduction pathways mediated by the bombesin/GRP receptor is attributed to the  $G_q$ -mediated events of PIP<sub>2</sub> hydrolysis, PKC activation and Ca<sup>2+</sup> mobilisation. However, section 1.3.7.8 describes the evidence that protein tyrosine phosphorylation, stimulated by bombesin, is independent of PKC and Ca<sup>2+</sup>. The mechanism by which bombesin stimulates tyrosine phosphorylation is unknown particularly events immediately subsequent to receptor stimulation. Milligan (1993) describes three models for the mechanism for signal bifurcation by G protein coupled receptors:

- Model 1 Although pharmacologically currently indistinguishable, multiple receptor subtypes are co-expressed.
- Model 2 A single receptor activates multiple G proteins, each of which is able to regulate a separate effector.
- Model 3 A single receptor interacts with a single G protein. Bifurcation of signals is subsequently produced by the regulation of different effectors by the G protein  $\alpha$  and  $\beta\gamma$  subunits.

In the context of bombesin-induced tyrosine phosphorylation the first model has been proposed several times. Cirillo, *et al.* (1986) observed that bombesin stimulated the tyrosine phosphorylation of a protein of 115 kDa. This was attributed to the then unknown bombesin receptor being a novel receptor tyrosine kinase. The interpretation of these results is now judged to be incorrect and it has been convincingly demonstrated that the bombesin/GRP receptor has no phosphotyrosine activity (Zachary, *et al.*, 1991b). Recently Bold, *et al.* (1994) reported that bombesin antagonists which inhibited cytoplasmic Ca<sup>2+</sup> increases did not block cell proliferation and bombesin agonists which did increase Ca<sup>2+</sup> did not stimulate cell growth, however bombesin-induced cell proliferation was inhibited by a tyrosine kinase inhibitor, tyrophostin. They concluded that one bombesin receptor couples to Ca<sup>2+</sup> mobilisation and a distinct receptor couples to tyrosine phosphorylation and cell growth, see figure 1.14. However, no direct evidence was presented in support of this.

#### Chapter 1

#### **INTRODUCTION**

Recently a new bombesin-like receptor, BRS-3 has been identified by molecular biology techniques (Gorbulev, *et al.*, 1992; Fathi, *et al.*, 1993) although the physiological ligand is currently unknown. This raises the possibility that there could be other unidentified bombesin receptors.



# Figure 1.14. Possible mechanisms of multiple coupling for bombesin receptors

Two alternatives are shown. One receptor could couple to all the pathways or there could be two receptors.

Ligands such as thrombin and LPA have also been shown to induce multiple signalling pathways, including protein tyrosine phosphorylation and reinitiation of DNA synthesis. They have in common signals which are transduced by G<sub>q</sub> and G<sub>i</sub>.

Thrombin was first identified as a mitogen in 1975 when Chen was investigating mitogens present in blood and especially in serum which during its production simulates a wound response. Proteases of the blood clotting cascade were known to be present in serum and they could be obtained in a highly purified form. Addition of purified thrombin to chick embryo fibroblasts caused an increase in [<sup>3</sup>H]-thymidine incorporation, an increase in cell numbers over 4 days and increased cell motility (Chen and Buchanan, 1975).

81

LPA is the smallest and simplest phospholipid and has a wide range of biological and biochemical actions (Moolenaar, 1994). It is a platelet-derived serum factor which accumulates rapidly in thrombin-stimulated platelets and can be released from growth factor stimulated fibroblasts.

Cloning of the thrombin receptor revealed a seven transmembrane protein with a novel mechanism of action (reviewed in Grand, 1996). The N-terminal is cleaved to expose a "tethered ligand" which is then thought to bind to the receptor in the same fashion as other peptide ligands and in fact a synthetic peptide to this sequence can elicit similar responses to thrombin. These include pertussis toxin-sensitive and -insensitive activation of PLC $\beta$  (Murayama and Ui, 1985; Grand, *et al.*, 1996); pertussis toxin-sensitive inhibition of adenylyl cyclase and Ras activation (Van Corven, *et al.*, 1993; Grand, *et al.*, 1996); partially sensitive Src activation (Chen, *et al.*, 1994), and insensitive Shc (Chen, *et al.*, 1996) and FAK tyrosine phosphorylation and reorganisation of the actin cytoskeleton (Jalink and Moolenaar, 1992; Grand, *et al.*, 1996).

A similar set of signals are elicited upon addition of LPA to cells. These are phospholipid hydrolysis and arachidonic acid release which are pertussis toxin insensitive but potentiated by GTP<sub>Y</sub>S, and [<sup>3</sup>H]-thymidine incorporation and inhibition of cAMP production which are pertussis toxin sensitive (van Corven, *et al.*, 1989). Later it was shown that LPA stimulates the Ras-MAPK pathway in a pertussis toxinsensitive manner and it also activates Src (Jalink, *et al.*, 1993). In addition LPA stimulates stress fibre assembly and reorganisation of the actin cytoskeleton, and the tyrosine phosphorylation of FAK which are pertussis toxin insensitive (Ridley and Hall, 1992; Jalink, *et al.*, 1993; Hordijk, *et al.*, 1994).

Thus the action of thrombin and LPA involves two G proteins, a pertussis toxinsensitive  $G_{i/o}$  and -insensitive  $G_q$  (judged from pertussis toxin-insensitive phospholipid hydrolysis). The identity of the LPA receptor(s) coupling to these G proteins has remained elusive. Using a <sup>32</sup>P-labelled LPA analogue containing a photoreactive acyl chain, a putative receptor of 38 - 40 kDa was identified (Van der Bend, *et al.*, 1992), but unambiguous characterisation awaits the molecular cloning of the receptor. There are discrepancies between the actions of thrombin and thrombin receptor peptide and there does appear to be thrombin specific binding sites on the surface of cells which can elicit responses without the need for proteolytic action (Seiler, *et al.*, 1991; Vouret-Craviari, *et al.*, 1993). This has led to the suggestion that there is more than one thrombin receptor (Grand, *et al.*, 1996). In this context it has not been clearly defined by transfection of the known thrombin receptor which pathways are stimulated by thrombin and the effect on cellular proliferation. One study altered the thrombin cleavage site to that of enterokinase (Hung, *et al.*, 1992) to show that cleavage of this receptor activates coupling to  $G_q$  and  $G_i$  but this was not extended to cell proliferation and protein tyrosine phosphorylation. Similarly the exact identity of the bombesin receptor which couples to tyrosine phosphorylation has not been precisely demonstrated.

An approach to investigate how these ligands exert their multiple signalling effects and mitogenic capability is to transfect a defined receptor into an unresponsive cell line and examine the signalling pathways elicited. This approach was taken with the bombesin/GRP receptor and the results are presented in Chapter 3.

## **1.5. AIMS OF THESIS**

This thesis will be into three sections.

**1.5.1.** Bombesin elicits multiple signal transduction pathways including activation of PLC, tyrosine phosphorylation of FAK and paxillin and reinitiates DNA synthesis. In Swiss 3T3 cells tyrosine phosphorylation does not appear to be downstream of PLC-mediated events. It was proposed recently that a separate bombesin receptor is coupled to tyrosine phosphorylation than the one that is coupled to PLC-mediated hydrolysis of PIP<sub>2</sub>. The exact identity of the receptor which mediates tyrosine phosphorylation is an unresolved but fundamental issue. Therefore to address this, the bombesin/GRP receptor was transfected into a bombesin-unresponsive fibroblast cell line, Rat-1. The first set of questions posed are:

Does ligand activation of the bombesin/GRP receptor transfected into Rat-1 cells:

- a) Stimulate Ca<sup>2+</sup> mobilisation and 80K phosphorylation?
- b) Stimulate DNA synthesis and cell proliferation?
- c) Stimulate the tyrosine phosphorylation of multiple proteins including FAK and paxillin?

**1.5.2.** MAPK activation is common point of convergence of a variety of mitogenic factors. As described in section 1.3.8.7 receptors that couple to  $G_i$  activate MAPK via a Ras-dependent pathway. However, receptors that are coupled to  $G_q$  appear to activate MAPK via PKC-dependent and independent routes which could also be via activation of Ras and cellular context could be an important contributing factor. It is known that in

Swiss 3T3 cells bombesin activates the MAPK pathway via PKC-dependent routes and does not cause GTP-loading of Ras. In Rat-1 cells stimulation of the CCK<sub>B</sub> receptor causes the activation of Raf-1 independently of PKC (Seufferlein, *et al.*, 1995). This could mean that peptide receptors coupled to  $G_q$  in Rat-1 cells could activate MAPK via Ras. Therefore, by transfecting the bombesin/GRP receptor into Rat-1 cells this problem can be addressed. The second set of questions are:

- a) Does bombesin stimulate MAPK and Raf-1 activity in Rat-1 cells transfected with the bombesin/GRP receptor?
- b) Is the activation of the kinases in this cascade dependent on PKC activity?
- c) Does bombesin stimulate the GTP loading of Ras?

**1.5.3.** Bombesin and other mitogenic neuropeptides mobilise  $Ca^{2+}$  to cause a transient increase in  $[Ca^{2+}]_{cyt}$  followed by a persistent depletion of intracellular stores. However, the role of  $Ca^{2+}$  in initiating mitogenesis is subject to controversy. The  $Ca^{2+}ATP$  inhibitors thapsigargin and DBHQ offer a new approach to investigate the role of  $Ca^{2+}$  in DNA synthesis. The third set of questions are:

- a) What effect do the Ca<sup>2+</sup>ATPase inhibitors thapsigargin and DBHQ have on DNA synthesis?
- b) How are these effects mediated?

# **Chapter 2 – MATERIALS AND METHODS**

## 2.1. MATERIALS

The cDNA of the murine bombesin/GRP preferring receptor in pCD2-neo (BNR-pCD2-neo) was kindly provided by Dr. Jim Battey, Laboratory of Biological Chemistry, National Cancer Institute, Bethesda, Maryland, U.S.A).

The bombesin/GRP receptor-specific antagonist  $D-F_5$ -Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe was a kind gift of Dr. David H. Coy of Peptide Research Laboratories, Tulane University Medical Center, New Orleans, Louisiana, USA.

Anti-Tyr(P) mAb clone Py72 and anti-Ras mAb clone 259 was obtained from the hybridoma development unit, Imperial Cancer Research Fund, London, United Kingdom. Py20 anti-Tyr(P) mAb was from ICN, Buckinghamshire, UK. The 4G10 anti-Tyr(P) mAb was from Tissue Culture Supplies, Buckingham, UK. Anti FAK mAb was obtained from AFFINITI Research Products Ltd, Nottingham, UK. Anti-paxillin mAb 165 was from ICN, High Wycombe, UK. The anti-p74<sup>raf-1</sup> affinity purified rabbit polyclonal antibody was obtained from Santa Cruz Biotechnology Ltd., USA. p42<sup>mapk</sup> antisera were generous gifts from Dr. Johan Van Lint of The Katholieke Universiteit Leuven and Dr. Dominic J Withers of Department of Medicine, Hammersmith Hospital, London, UK.

GST-MEK and GST-p42<sup>mapk</sup> expression vectors were kind gifts from Professor C. Marshall, Chester Beattie Laboratory, Institute for Cancer Research, London, UK

Thapsigargin, GF 109203X and Fura-2/AME were obtained from Calbiochem. DBHQ was obtained from Aldrich. Dispase, Protein A-agarose conjugate and Protein G-Sepharose was obtained from Boehringer Mannheim, Germany. PDB, insulin, bombesin, econozole, GRP, BSA, litorin, NmB, bombesin fragment amino acids 8-14, endothelin, anti-mouse IgG (whole molecule) - agarose were obtained from Sigma. FBS was obtained from Gibco BRL, Life Technologies.

[*methyl*-<sup>3</sup>H]thymidine (25 Ci/mmol), *myo*-[2-<sup>3</sup>H]inositol (16 Ci/mmol), <sup>32</sup>P<sub>i</sub> (10 mCi/ml), <sup>125</sup>I-labelled sheep anti-mouse IgG (20  $\mu$ Ci/ $\mu$ g), <sup>125</sup>I-labelled Protein A (30 mCi/mg) and <sup>45</sup>CaCl<sub>2</sub> (10 - 40 mCi/mg calcium), (3-[<sup>125</sup>I]iodotyrosyl<sup>15</sup>)Gastrin releasing peptide (2000 Ci/mmole), and [ $\gamma$ -<sup>32</sup>P]ATP (5000 Ci/mmol) were obtained from Amersham International, U.K.

All other materials were of the purest grade commercially available.

# **2.2. Methods**

Throughout this section the principle is described first in *italics* followed by the method in plain text.

#### 2.2.1. TRANSFECTION OF RAT-1 CELLS

A 1.4 kb insert containing the complete cDNA coding region of the murine bombesin/GRP preferring receptor cloned into the mammalian expression vector pCD2-neo (BNR-pCD2-neo) and under the transcriptional control of the SV40 early region promoter, was kindly provided by Dr. Jim Battey. For transfection, Rat-1 cells were plated at a density of 1 x 10<sup>6</sup> cells/100 mm dish. The following day, cultures were washed twice in Dulbecco's Modified Eagle's Medium (DMEM) and 20  $\mu$ g DNA from BNR-pCD2-neo expression vector, mixed with 40  $\mu$ g lipofectin reagent, was added to the cultures in 5 ml DMEM at 37°C. After 14 h incubation, 5 ml of 20% foetal bovine serum (FBS) in DMEM were added and incubation extended for a further day. Then, cultures were trypsinised, divided into four and selected in DMEM supplemented with 1 mg/ml neomycin derivative G418. The medium was changed every 3 - 4 days. After 2 - 3 weeks, resistant colonies were cloned by ring isolation and propagated in DMEM supplemented with 5% FBS and 0.5 mg/ml G418.

#### **2.2.2.** CELL CULTURE

Only early passage cells were used which were replaced every 6 -8 weeks.

#### 2.2.2.1. RAT-1 CELLS

Stock cultures of Rat-1 cells were maintained as described previously (Higgins, *et al.*, 1992) except that transfected cell lines were propagated in the presence of 0.5 mg/ml G418. For experimental purposes,  $5 \times 10^4$  or  $3 \times 10^5$  cells were subcultured in 33 or 100 mm Nunc Petri dishes respectively with DMEM supplemented with 5% FBS and 0.5 mg/ml G418 as necessary and incubated in a humidified atmosphere of 10% CO<sub>2</sub>, 90% air at 37°C until they became 50 - 70% confluent (2 - 3 days), then switched down to 0.5% FBS for 3 - 4 days until they were quiescent.

#### 2.2.2.2. SWISS 3T3 CELLS

Stock cultures of Swiss 3T3 cells were propagated as described previously (Rozengurt and Sinnett-Smith, 1983). For experimental purposes, cells were

subcultured in 33- or 100- mm Nunc Petri dishes with DMEM supplemented with 10% FBS and incubated in a humidified atmosphere of 10% CO<sub>2</sub>, 90% air at 37°C until they became confluent and quiescent (6 - 8 days).

#### **2.2.3.**[125I]**GRPBINDINGASSAY**

[<sup>125</sup>I]GRP is incubated with intact cells at 37°C then binding is terminated by washing rapidly with PBS and solubilising the cells as described previously (Zachary and Rozengurt, 1985). Radioactivity associated with the whole cell is measured.

Confluent, quiescent, cultures of Rat-1 cells in 33 mm dishes were washed twice with DMEM and incubated for the indicated times at 37°C in DMEM with a mixture of [ $^{125}$ I]GRP and cold GRP, as described in the figure legends, in the presence of 1 mg/ml bovine serum albumin (BSA). For the competition studies various concentrations of competing peptides (bombesin, GRP, litorin, NmB and bombesin fragment 8-14) were also added to the binding medium. Non-specific binding was determined by adding 1  $\mu$ M bombesin. After the incubation, binding was terminated by placing cultures on ice where they were rapidly washed 4 times with ice cold phosphate buffered saline (PBS) containing 1 mg/ml BSA. Cultures were dissolved in 1 ml of 0.1 N NaOH/2% NaCO<sub>3</sub>/1% SDS and the radiation emitted was counted in a gamma counter. Non-specific binding, defined by 1  $\mu$ M bombesin, was subtracted from total binding to calculate specific binding. Results are expressed as specific binding per 10<sup>6</sup> cells. Specific binding in the transfected cells was 60 - 80% of the total binding. No statistically significant binding was observed in untransfected cells in this assay which is able to detect as little as 200 receptors per cell.

#### 2.2.4. MEASUREMENT OF TOTAL INOSITOL PHOSPHATES

Cells are labelled overnight with [<sup>3</sup>H]inositol and total inositol phosphates are collected on an anion exchange column then eluted off with ammonium formate as described previously by Murphy and Rozengurt (1992).

Cultures of Swiss 3T3 cells in 33 mm dishes were labelled for 16 - 18 h with *myo*-[2-<sup>3</sup>H]inositol (15  $\mu$ Ci/ml). Then they were washed twice with DMEM/Waymouth's medium (1:1 [vol/vol]) and incubated in this medium with additions as described in the table legend. LiCl was added to a final concentration of 20 mM for the last 30 min of the incubation to inhibit breakdown of inositol phosphates, see section 1.3.1. Inositol phosphates were extracted by replacing the medium with 1 ml ice-cold 3% HClO<sub>4</sub>. After 20 min at 4°C the extract was neutralised with 1 M KOH containing 50

mM HEPES, 5 mM EGTA and 0.005% phenol red. Precipitated KClO<sub>4</sub> was removed by centrifugation. Samples were diluted to 10 ml with water then loaded onto 1 ml of Dowex AG1-X8 (200 - 400 mesh, HCOO<sup>-</sup> form) in Bio-Rad Econo-columns. Columns were washed 3 x 10 ml H<sub>2</sub>O and 2 x 10 ml of 60 mM NH<sub>4</sub>COOH, 5 mM Na<sub>2</sub>B<sub>4</sub>O<sub>7</sub> and inositol phosphates eluted with 6 ml 1 M NH<sub>4</sub>COOH, 0.1 M HCOOH. An aliquot (1 ml) of eluate was counted in 10 ml Ultima Gold.

#### 2.2.5. MEASUREMENT OF Ca<sup>2+</sup>

Two methods have been used in this thesis. The fluorescent  $Ca^{2+}$  indicator, fura-2, allows continuous measurement of the cytoplasmic concentration, whereas the  ${}^{45}Ca^{2+}$  isotope allows measurement of the total cellular  $Ca^{2+}$ .

#### 2.2.5.1. USING THE FLUORESCENT Ca<sup>2+</sup> INDICATOR, FURA-2

Fura-2 is a fluorescent derivative of EGTA (Grynkiewicz, et al., 1985) with four carboxylate groups that make it very membrane-impermeable. This problem has been addressed by masking the carboxyl groups with acetoxymethyl ester (AME) groups. The resulting fura-2/AME is hydrophobic, penetrates the plasma membrane and accumulates in the cytosol. Once inside the cell, the AME are hydrolysed by esterases present in the cytosol and fura-2 is again membrane-impermeable and is trapped inside the cell (Borle and Snowdowne, 1987).  $[Ca^{2+}]_{cyt}$  was measured with the fluorescent  $Ca^{2+}$  indicator fura-2/AME using procedures previously described (Millar and Rozengurt, 1990b).

#### Rat-1 Cells

Confluent, quiescent, cultures of transfected Rat-1 cells in 100 mm dishes were washed twice with DMEM and incubated for 10 min at 37°C in 2.5 mg/ml dispase/DMEM to digest the intercellular matrix. Cells (4 - 5 x10<sup>6</sup>) in single cell suspension were then incubated at 37°C, 10 min, in 5 ml of DMEM containing 1  $\mu$ M fura-2/AME freshly prepared from a 1 mM stock maintained in DMSO. Cells were collected by centrifugation and resuspended in 2 ml electrolyte solution (150 mM NaCl, 0.5 mM KCl, 0.9 mM MgCl<sub>2</sub>, 1.8 mM CaCl<sub>2</sub>, 25 mM glucose, 16 mM HEPES, the same amino acid composition as DMEM, pH 7.1).

#### Swiss 3T3 Cells

Cells on 100 mm dishes were washed twice with DMEM and incubated for 10 min in 5 ml of DMEM containing 1  $\mu$ M fura-2/AME freshly prepared from a 1 mM stock

maintained in DMSO. Cells were washed twice with PBS then gently scraped and resuspended in 2 ml of electrolyte solution (see above).

Both Rat-1 and Swiss 3T3 cells in electrolyte solution were then transferred to a 1 cm<sup>2</sup> quartz cuvette which was kept at 37°C and stirred continuously throughout the experiment. Fluorescence was monitored in a Perkin-Elmer LS-5 luminescence spectrometer with an excitation wavelength of 336 nm and an emission wavelength of 510 nm.  $[Ca^{2+}]_{cyt}$  was calculated after sequential addition of Triton X-100 to 0.05% and EGTA to 100 mM to obtain the maximum ( $F_{max}$ ) and minimum ( $F_{min}$ ) fluorescence respectively, using the formula:  $[Ca^{2+}]_{cyt}$  in nM =  $K (F - F_{min})/(F_{max} - F)$ , where F is the fluorescence at the unknown  $[Ca^{2+}]_{cyt}$  and the value of K is 220 nM for fura-2/AME.

#### 2.2.5.2. MEASUREMENT OF TOTAL <sup>45</sup>Ca<sup>2+</sup> CONTENT

 $^{45}Ca^{2+}$  is equilibrated with total cellular  $Ca^{2+}$  overnight. As described in section 1.3.2.1  $Ca^{2+}$  in the intracellular stores can reach nearly mM concentrations whereas the  $Ca^{2+}$  in the cytoplasm is sub-micromolar. This method therefore is more a measure of stored  $Ca^{2+}$  rather than cytoplasmic  $Ca^{2+}$ .

Confluent, quiescent Swiss 3T3 cells in 33 mm dishes were loaded to equilibrium with 4  $\mu$ Ci <sup>45</sup>Ca<sup>2+</sup> per ml over 16 h. Cells were then washed twice with DMEM and incubated with factors for 10 min at 37°C. Cells were rapidly washed 2 x 5 ml DMEM with 2 mM EGTA, then 6 x 5 ml DMEM. Cells were dissolved in 1 ml of 2% Na<sub>2</sub>CO<sub>3</sub>, 0.1 N NaOH, 1% SDS and the total amount of <sup>45</sup>Ca<sup>2+</sup> in the cells was measured by liquid scintillation counting in 10 ml Ultima Gold.

### 2.2.6. MEASURING PKC ACTIVATION BY 80K/MARCKS PHOSPHORYLATION

# Cells are labelled overnight with ${}^{32}P_i$ and 80K/MARCKS is immunoprecipitated. Phosphorylated 80K/MARCKS is visualised by autoradiography.

Confluent, quiescent, cultures of cells in 33 mm dishes were washed twice with DMEM without P<sub>i</sub> and labelled overnight in this medium with 50  $\mu$ Ci/ml <sup>32</sup>P<sub>i</sub>. Cells were treated with various factors as indicated then lysed in 250  $\mu$ l of ice cold lysis buffer (10 mM Tris/HCl, 5 mM EDTA, 50 mM NaCl, 30 mM sodium pyrophosphate, 50 mM NaF, 100  $\mu$ M Na<sub>3</sub>VO<sub>4</sub>, 1% Triton X-100 and 1 mM phenylmethylsulphonyl fluoride (PMSF), pH 7.6). Lysates were clarified by centrifugation at 15,000 g for 10 min. 80K/MARCKS was immunoprecipitated for 3 h at 4°C with 80K/MARCKS antiserum (Brooks, *et al.*, 1991) then 1 h with Protein A - agarose. Immunoprecipitates were washed three times with lysis buffer, extracted in 2x sample buffer (200 mM Tris-HCl,

6% SDS, 2 mM EDTA, 4% 2-mercaptoethanol, 10% glycerol, pH 6.8) and then resolved by one-dimensional SDS-polyacrylamide gel electrophoresis (SDS-PAGE) (8% acrylamide) (Laemmli, 1970). Gels were fixed in 10% acetic acid and 25% methanol then dried. Phosphorylated 80K/MARCKS was visualised by autoradiography.

#### **2.2.7.** MAPK ACTIVATION

This has been measured by 2 approaches. *In vitro* immunecomplex kinase assays measure the enzymatic activity of p42<sup>mapk</sup> whereas p42<sup>mapk</sup> bandshift is an indication of the phosphorylation state of the enzyme and hence activation.

#### 2.2.7.1. IMMUNE COMPLEX ASSAY FOR p42<sup>mapk</sup> ACTIVATION

 $p42^{mapk}$  is immunoprecipitated and the immune complex is incubated with an MBP peptide substrate and  $\gamma$ -<sup>32</sup>P ATP. The peptide is collected on anion exchange paper as described in Withers, et al. (1995).

Quiescent Rat-1 cells in 100 mm dishes were treated with factors and lysed at 4°C in 1 ml of lysis buffer. Lysates were clarified by centrifugation at 15,000 x g for 10 min at 4°C. Immunoprecipitation was performed using a polyclonal anti-p42<sup>mapk</sup> antibody raised against the C-terminal 9 amino acids of  $p42^{mapk}$ , and incubating the samples on a rotating wheel for 2 h. Washed protein A-agarose beads (50 µl 1:1 slurry) were added for the second hour. Immune complexes were collected by centrifugation and washed twice in lysis buffer without PMSF and twice in kinase buffer (15 mM Tris-HCl, 15 mM  $MgCl_2$ ). The kinase reaction was performed by resuspending the pellet in 25  $\mu$ l of kinase assay cocktail containing kinase buffer with 0.5 mM EGTA, 1 mg/ml MBP-peptide (APRTPGGRR), 50  $\mu$ M ATP, 50  $\mu$ Ci/ml of [ $\gamma$ - $^{32}$ P]ATP. Incubations were performed under linear assay conditions (10 min) at 30°C and terminated by spotting 25  $\mu$ l of the supernatant onto Whatman P81 cation exchanger chromatography paper. Free  $[\gamma^{-32}P]$ ATP was removed by washing filters four times for 5 min in 0.5% orthophosphoric acid. Filters were immersed in acetone to aid drying before Cerenkov counting. The average radioactivity of two blank samples containing no immune complex was subtracted from the result of each sample. The specific activity of  $[\gamma^{-32}P]$  ATP used was 900 - 1200 cpm/pmol.

#### 2.2.7.2. p42<sup>mapk</sup> BANDSHIFT

Active MAPK is phosphorylated on threonine and tyrosine residues and this phosphorylated form of MAPK exhibits retarded mobility in SDS-PAGE. Therefore the appearance of a slower migrating isoform, visualised by Western blotting with anti-

MAPK antibodies, is an indication of the activation state of the cellular pool of MAPK (Leevers and Marshall, 1992).

Confluent, quiescent dishes of Swiss 3T3 cells were treated with factors as described in the figure legend for 5 min and lysed in 150  $\mu$ l of 2x sample buffer. Samples were boiled for 10 min, loaded onto a 10% polyacrylamide gel. Resolved proteins were transferred to Immobilon-P (PVDF) membranes in transfer buffer containing 25 mM Tris base, 0.2 M glycine and 20% methanol for 2.5 h at 100 V (current limited to 0.4 A). Immobilon membranes were blocked in 5% non-fat milk and incubated for 2 h with the polyclonal anti-p42<sup>mapk</sup> antibody in 3% non-fat milk, 0.02% azide. Bands were visualised by incubating for 1 h with <sup>125</sup>I-labelled Protein A followed by autoradiography.

#### 2.2.8. RAF-1 KINASE ASSAY

Raf-1 activity was assayed using a modification of the method described by Alessi, et al. (1994) using a GST-fusion proteins of MEK and  $p42^{mapk}$  as substrates for sequential activation, initiated by active Raf-1. This is followed by MAPK phosphorylation of MBP peptide which is then collected on cation exchanger paper as above.

#### **2.2.8.1. PREPARATION OF SUBSTRATES**

Overnight cultures of *Escherichia coli* strain BL21 DE3 transformed with GST-p42<sup>mapk</sup> and GST-MEK expression vectors (pGEX-2T) were diluted 1 in 10 and grown for 1 h. GST-p42<sup>mapk</sup> was induced with 1 mM IPTG for 4 h at 37°C and GST-MEK induced with 30  $\mu$ M IPTG at 27°C overnight. The cells were then pelleted, freeze thawed and lysed. Cell debris was removed by centrifugation and GST-fusion proteins were purified by adding 0.5 ml glutathione-Sepharose beads and rotating for 30 min at 4°C. The GST-p42<sup>mapk</sup> was cleaved from the GST in thrombin buffer while the GST-MEK was eluted from the glutathione-Sepharose beads with 5 mM glutathione in 50 mM Tris pH 8. Both preparations were then dialysed and concentrated. The purity of each preparation was checked by subjecting the proteins to SDS-PAGE and staining the gels with Coomassie blue. For the p42<sup>mapk</sup> typical yields were 10  $\mu$ g/ml of culture, with a purity of >95%.

#### 2.2.8.2. RAF-1 KINASE ASSAY

Quiescent cells were treated as indicated in the figure legends and lysed in lysis buffer as above. Raf-1 immunoprecipitation was performed with an affinity purified rabbit polyclonal antibody for 2 h with protein A-agarose added for the second hour. Immune complexes were collected by centrifugation and then washed twice in lysis buffer with no PMSF and twice with buffer A (50 mM Tris-HCl pH 7.5, 0.1 mM EGTA, 0.5 mM Na<sub>3</sub>VO<sub>4</sub> and 0.1% β-mercaptoethanol). Pellets were then resuspended in 30 µl buffer B (30 mM Tris-HCl pH 7.5, 0.1 mM EGTA, 0.1% β-mercaptoethanol, 6.5 µg/ml GST-MEK, 100 µg/ml p42<sup>mapk</sup>, 0.03% Brij-35, 10 mM Mg-ATP and 20 mM N-octyl-β-Dglucopyranoside) and incubated at 30°C for 30 min. The reaction was then terminated by diluting the supernatant in 40 µl of buffer A with 1 mg/ml of BSA and, after mixing, 10 µl of the supernatant was removed to a fresh tube.  $p42^{mapk}$  activation was then measured using the MBP peptide phosphorylation assay essentially as above.

#### 2.2.9. GTP LOADING OF RAS

Cells are metabolically labelled with  ${}^{32}P_{i}$ , then the cell lysate is partitioned and Ras with bound nucleotide is immunoprecipitated from the detergent phase. The bound nucleotides are eluted, separated by thin layer chromatography and visualised by autoradiography as described in Basu, et al, (1992).

Confluent, quiescent cells in 100 mm dishes were washed twice in DMEMphosphate and incubated in 5 ml DMEM-phosphate with 20 mM HEPES pH 7.5 and 1 mg/ml tissue culture grade BSA with 1.5 mCi of  ${}^{32}P_i$  and labelled for 16 - 18 h. Cells were then stimulated with factors as described in the figure legend for 2.5 min at 37°C then rinsed with cold PBS and lysed in Ras lysis buffer (50 mM Tris pH 7.5, 100 mM NaCl, 1 mM EGTA, 10 mM benzamidine, 10 µg/ml trypsin inhibitor, 10 µg/ml aprotinin, 10 µg/ml leupeptin, 1 mM DTT, 1 mg/ml BSA) with 5 mM MgCl<sub>2</sub> and 1% Triton X-114. After removal of cell debris by centrifugation, phases were partitioned in the presence of 0.5 M NaCl for 2 min at 37°C. The detergent gel containing Ras was recovered by spinning down for two minutes at 14,000 g at 20°C and redissolved in Ras lysis buffer containing 1% Triton X-100, 0.5% sodium deoxycholate, 0.05% SDS and 0.5 M NaCl. Ras was immunoprecipitated for 2 h with 25 µl [1:1 slurry] anti-Ras mAb 259 pre-coupled to Protein G - Sepharose (0.1 mg 259 mAb per 1 ml [1:1 slurry] Protein G - Sepharose). All immunoprecipitations were carried out in duplicate along with non-specific controls. Immunoprecipitates were washed with 6 x 1 ml of 50 mM HEPES, pH 7.4, 0.5 M NaCl, 5 mM MgCl<sub>2</sub>, 0.1% Triton X-100, 0.005% SDS and the nucleotide was eluted with 2 mM EDTA, 2 mM DTT, 0.2% SDS, 0.5 mM GTP, 0.5 mM GDP, at 68°C for 20 min. Eluted nucleotides were separated by thin layer chromatograpy on PEI-cellulose plates (MERCK) run in 1.2 M ammonium formate, 0.8 M HCl. Results were quantitated on a phosphorimager and analysed by ImageQuant (Molecular Dynamics).

#### 2.2.10. ANALYSIS OF TYROSINE PHOSPHORYLATED PROTEINS

The tyrosine phosphorylation of multiple proteins has been analysed by immunoprecipitation of specific proteins and analysis by Western blotting.

#### 2.2.10.1. TYROSINE PHOSPHORYLATION OF MULTIPLE PROTEINS

Proteins are both immunoprecipitated and Western Blotted with antiphosphotyrosine (Tyr(P)) mAbs.

Confluent, quiescent cultures of Rat-1 cells in 33 mm dishes (2.5 x  $10^6$  cells total) or Swiss 3T3 cells in 100 mm dishes were washed twice with DMEM, treated with factors in 5 ml of DMEM as indicated and lysed at 4°C in 1 ml lysis buffer as used for 80K/MARCKS immunoprecipitation. Lysates were clarified by centrifugation at 15,000g for 10 min. Supernatants were incubated with anti-Tyr(P) mAb Py72 for 4 h at 4°C then phosphotyrosine immune complexes were precipitated with anti-mouse IgG (whole molecule) - agarose for 1 h at 4°C. Immunoprecipitates were washed three times with lysis buffer, extracted in 2x sample buffer (200 mM Tris-HCl, 6% SDS, 2 mM EDTA, 4% 2-mercaptoethanol, 10% glycerol, pH 6.8) and then resolved by SDS-PAGE. Proteins were then transferred to Immobilon transfer membranes in transfer buffer containing 50 mM Tris base, 0.4 mM glycine, 0.1% SDS and 20% methanol for 2 h at 35 V then 2 h at 70 V. Membranes were blocked using 5% non-fat dried milk in PBS, pH 7.2, and incubated for 2 h with anti-Tyr(P) mAb (4G10 1 µg/ml) in PBS containing 3% non-fat dried milk. Immunoreactive bands were visualised by autoradiography using [<sup>125</sup>I]-labelled sheep anti-mouse IgG (1:1000).

#### 2.2.10.2. TYROSINE PHOSPHORYLATION OF FAK

*Tyrosine phosphorylated protein are immunoprecipitated and the presence of FAK in these immune complexes is analysed by Western blotting with anti-FAK mAb.* 

Cells were stimulated and phosphotyrosyl containing proteins were immunoprecipitated and transferred to Immobilon membranes as above. Membranes were blocked using 5% non-fat dried milk in PBS, pH 7.2, and incubated for 2 h with anti-p125<sup>FAK</sup> mAb (1:1000) in PBS containing 3% non-fat dried milk. Immunoreactive bands were visualised by autoradiography using [<sup>125</sup>I]-labelled sheep anti-mouse IgG (1:1000).

#### 2.2.10.3. TYROSINE PHOSPHORYLATION OF PAXILLIN

The tyrosine phosphorylation of paxillin is analysed by immunoprecipitating paxillin and Western blotting with anti-Tyr(P) mAb.

Confluent, quiescent cultures of Rat-1 cells in 33 mm dishes (2.5 x  $10^6$  cells total) were washed twice with DMEM, treated with factors in 5 ml of DMEM as indicated and lysed at 4°C in 1 ml lysis buffer. Lysates were clarified by centrifugation at 15,000g for 10 min then incubated with anti-paxillin mAb 165 for 2 h at 4°C. Paxillin immune complexes were precipitated with anti-mouse IgG (whole molecule) - agarose for 1 h at 4°C. Immunoprecipitates were washed three times with lysis buffer, extracted in 2x sample buffer and then resolved by SDS-PAGE and transferred to Immobilon transfer membranes as above. Membranes were blocked using 5% non-fat dried milk in PBS, pH 7.2, and incubated for 2 h with anti-Tyr(P) mAb (4G10 1 µg/ml) in PBS containing 3% non-fat dried milk. Immunoreactive bands were visualised by autoradiography using [<sup>125</sup>I]-labelled sheep anti-mouse IgG (1:1000).

Autoradiograms were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area.

#### 2.2.11. MEASUREMENTS OF CELLULAR PROLIFERATION

Two basic methods have been used in this thesis. The first involves measuring the incorporation of the DNA-specific base thymidine into stimulated cells as a measure of re-entry into S-phase and DNA synthesis. The second method involves looking for an increase in cell numbers as a measure of cell proliferation.

#### 2.2.11.1. [<sup>3</sup>H]THYMIDINE INCORPORATION ASSAY

Determinations of DNA synthesis were performed as previously described (Dicker and Rozengurt, 1980). Cells are stimulated in the presence of [<sup>3</sup>H]thymidine and the amount of radioactivity incorporated into acid-insoluble pools is measured.

#### **Rat-1** Cells

Confluent, quiescent cultures of Rat-1 cells in 33 mm dishes were washed twice with DMEM and incubated at 37°C with DMEM/Waymouth's medium (1:1[vol/vol]) and various additions as described in the figure legends. After times specified [<sup>3</sup>H]thymidine (0.25  $\mu$ Ci/ml, 1  $\mu$ M) was added for 3 h.

#### Swiss 3T3 Cells

Confluent, quiescent cultures of Swiss 3T3 cells in 33 mm dishes were washed twice with DMEM and incubated with DMEM/Waymouth's medium (1:1[vol/vol]) containing [<sup>3</sup>H]thymidine (0.25  $\mu$ Ci/ml, 1  $\mu$ M) and various additions as described in the figure legends for 40 h at 37°C.

Cultures were then washed twice with PBS and incubated in 5% trichloroacetic acid at  $4^{\circ}$ C for 30 min to remove acid-soluble radioactivity. Cultures were washed with ethanol, solubilised in 1 ml of 2% Na<sub>2</sub>CO<sub>3</sub>, 0.1 M NaOH, 1% SDS and the radioactivity in the acid-insoluble pools determined by scintillation counting in 6 ml Ultima Gold (Packard).

#### 2.2.11.2. AUTORADIOGRAPHY OF LABELLED NUCLEI

Labelled nuclei were determined by autoradiography as described previously (Dicker and Rozengurt, 1980). This method of analysing the incorporation of [<sup>3</sup>H]thymidine reveals if any differences seen in looking at [<sup>3</sup>H]thymidine incorporation into acid insoluble pools are reflected in the proportion of cells that are labelled. This method therefore controls for alterations in the endogenous cellular thymidine pool or in the transport and processing of [<sup>3</sup>H]thymidine.

Cultures were washed twice with DMEM and incubated with DMEM/Waymouth's medium (1:1 [vol/vol]) containing [<sup>3</sup>H]thymidine (2.5  $\mu$ Ci/ml) and various additions as described in the figure legends. After 40 h, the medium was removed and the cultures fixed in formol saline (1:9 [vol/vol]). The cultures were washed twice with Tris-saline pH 7.4 at 4°C, fixed twice (for 5 and 2 min) with 5% trichloroacetic acid at 4°C, and washed three times with 70% ethanol. Autoradiographic film was put on the cultures and developed after a 2 week exposure. After counterstaining with Giemsa, the labelled nuclei were counted.

#### 2.2.11.3. MEASUREMENT OF CELL PROLIFERATION

Confluent, quiescent cultures of Rat-1 cells in 33 mm dishes were washed twice with DMEM and incubated at 37°C in DMEM/Waymouth's with additions as specified in the figure legends. After various times cell number was determined by removing the cells from the dish with a trypsin solution (0.05% trypsin in Ca<sup>2+</sup> and Mg<sup>2+</sup> free PBS with EDTA) and counting a portion of the resulting cell suspension in a Coulter counter.

#### **2.2.12. PRETREATMENTS**

Pretreatments with thapsigargin and DBHQ were carried out in conditioned medium which is the medium on 6 - 8 day old cells that have attained quiescence and is thus essentially devoid of growth promoting activity.

Stock solutions of thapsigargin were made up at 1  $\mu$ M in DMSO and stored at -20°C. DBHQ was dissolved fresh each time to 100  $\mu$ M in DMSO. Dilutions of compounds in DMSO, and DMSO controls, were added to the medium at 1  $\mu$ l/ml. This concentration of DMSO had no measurable effects on our cells.

# RESULTS

# and DISCUSSIONS

# Chapter 3 – Signalling Pathways Coupled to The Bombesin/GRP Receptor

At present three mammalian members of the bombesin-like receptor family have been cloned and shown to belong to the G protein-linked receptor superfamily. They are the bombesin/GRP-preferring (Spindel, *et al.*, 1990; Battey, *et al.*, 1991), the neuromedin B- (NmB) preferring (Corjay, *et al.*, 1991; Wada, *et al.*, 1991) and BRS-3 (Gorbulev, *et al.*, 1992; Fathi, *et al.*, 1993), a bombesin-like receptor whose physiological ligand is not yet known. It has been generally assumed that all the early signals and the subsequent mitogenic response elicited by bombesin are mediated by occupancy of the bombesin/GRP receptor. This assumption has recently been questioned. Bold, *et al.* (1994) reported that antagonists that inhibit bombesininduced Ca<sup>2+</sup> mobilisation do not prevent bombesin-induced cell growth in human gastric adenocarcinoma cells. However tyrophostin blocked bombesin-induced proliferation of these cells. Bold *et al.* suggested that a bombesin receptor subtype couples to Ca<sup>2+</sup> mobilisation and a different bombesin receptor subtype, not yet identified, couples to the protein tyrosine phosphorylation pathway and cell growth. However, no direct evidence supporting this hypothesis has been provided.

The experiments presented here were designed to elucidate whether the bombesin/GRP-preferring receptor could couple to both PLC activation as well as to FAK and paxillin tyrosine phosphorylation. Initially, it is shown that expression of the bombesin/GRP preferring receptor in Rat-1 cells, a cell line that can be reversibly arrested in the  $G_0/G_1$  phase of the cell cycle, confers a proliferative response to bombesin in the receptor transfected cells. Using this model system, it is demonstrated that agonist activation of the bombesin/GRP receptor elicits both PLC activation as well as tyrosine phosphorylation of multiple proteins including FAK and paxillin. These results demonstrate, for the first time, that the bombesin/GRP receptor couples to both signalling pathways and to cell proliferation.

### 3.1. BINDING CHARACTERISTICS OF THE EXPRESSED BOMBESIN/GRP RECEPTOR

#### 3.1.1. EXPRESSION OF THE BOMBESIN/GRP RECEPTOR IN RAT-1 CELLS.

The mouse bombesin/GRP-preferring receptor cDNA cloned into the mammalian expression vector pCD2, and under the transcriptional control of the SV40 early region promoter, was used for transfection experiments. Rat-1 fibroblasts were

chosen as the recipient cell line because these cells can be reversibly arrested in the  $G_0/G_1$  phase of the cell cycle (Higgins, *et al.*, 1992) and do not express endogenous functional bombesin/GRP receptors (see below). After transfection, 21 G418 resistant colonies were isolated and propagated; six of these cell lines expressed the bombesin/GRP-preferring receptor compared with the control, untransfected Rat-1 cells.

The functional expression of bombesin/GRP receptors in the transfected Rat-1 cells was verified using [<sup>125</sup>I]-labelled GRP. Figure 3.1 shows that the transfectants have specific binding sites for [<sup>125</sup>I]GRP. Parental Rat-1 cells did not specifically bind [<sup>125</sup>I]GRP. The apparent affinity of [<sup>125</sup>I]GRP for the transfected receptor in BOR 5 and BOR 15 cells (K<sub>d</sub> = 1 - 2 nM) was similar to the endogenous receptor expressed in Swiss 3T3 cells (K<sub>d</sub> = 0.8 nM). The number of receptors on the transfected cells was 30 - 60% lower than that of Swiss 3T3 cells, about 10<sup>5</sup> per cell (Zachary and Rozengurt, 1985), under identical experimental conditions, see figure 3.2.

#### 3.1.2. TIMECOURSE OF [125I]GRP BINDING TO TRANSFECTED RAT-1 CELLS

When BOR 5 cells were incubated with [<sup>125</sup>I]GRP, cell associated radioactivity increased rapidly reaching a maximum value after 30 min and decreasing afterwards (figure 3.3). This is consistent with internalisation and degradation of the peptide as reported previously in Swiss 3T3 cells (Zachary and Rozengurt, 1987) and in transformed rat kidney cells (Grady, *et al.*, 1995).

#### 3.1.3. LIGAND RECOGNITION PROFILE OF THE TRANSFECTED BOMBESIN/GRP RECEPTOR

Addition of unlabelled GRP, bombesin, litorin, the structurally related neuromedin B (NmB), or the bombesin C-terminal fragment (amino acids 8-14) inhibited specific [ $^{125}I$ ]GRP binding to BOR 5 cells in a concentration dependent manner (figure 3.4). The IC<sub>50</sub> values for bombesin, GRP and litorin were approximately 1 - 3 nM. In contrast NmB inhibited binding with much lower affinity (IC<sub>50</sub> = 500 nM). The 8-14 bombesin fragment is the C-terminal heptapeptide of bombesin which is the minimum sequence required to show activity, however full activity is seen with the C-terminal nonapeptide (Erspamer, *et al.*, 1988). Consistent with this the 8-14 bombesin peptide did show binding to the transfected bombesin/GRP receptor but with low affinity.

These results show that BOR 5 has specific binding sites with high affinity for bombesin/GRP and low affinity for NmB demonstrating that the transfected receptor in Rat-1 cells showed the characteristic ligand recognition of the bombesin/GRP-preferring receptor (Zachary and Rozengurt, 1985; Sinnett-Smith, *et al.*, 1990).



#### Figure 3.1. Specific [<sup>125</sup>I]GRP binding to different clones

Confluent, quiescent cultures of transfected and parental (WT) Rat-1 cells were washed twice with DMEM and incubated for 30 min at 37°C in DMEM containing [<sup>125</sup>I]GRP (0.05 nM, 0.01  $\mu$ Ci and 0.95 nM cold GRP) and 1 mg/ml BSA. Cultures were then placed on ice and rapidly washed 4 x with PBS containing 1 mg/ml BSA. Cultures were dissolved in 0.1 N NaOH/2% NaCO<sub>3</sub>/1% SDS and counted in a gamma counter. Results are expressed as mean specific binding per 10<sup>6</sup> cells ± SE from 3 independent experiments performed in duplicate.

Non-specific binding was determined in the presence of 1  $\mu$ M bombesin. Specific binding was 60 - 80% of the total binding.



#### Figure 3.2. Binding of [<sup>125</sup>I]-GRP as a function of GRP concentration

Confluent, quiescent cultures of transfected Rat-1 cells and Swiss 3T3 cells were washed twice with DMEM and incubated for 30 min at 37°C in DMEM containing various amounts of [<sup>125</sup>I]GRP and 1 mg/ml BSA. The maximum concentration was 4 nM GRP (0.2 nM, 0.04  $\mu$ Ci and 3.8 nM cold GRP) which was serial diluted 1:1 in DMEM with 1 mg/ml BSA. Cultures were then placed on ice and rapidly washed 4 x with PBS containing 1 mg/ml BSA. Cultures were dissolved in 0.1 N NaOH/2% NaCO<sub>3</sub>/1% SDS and counted in a gamma counter. Results are expressed as specific binding per 10<sup>6</sup> cells from one experiment performed in duplicate. Non-specific binding was determined in the presence of 1  $\mu$ M bombesin.



#### Figure 3.3. Time-course of [<sup>125</sup>]GRP binding to BOR 5 cells

Confluent, quiescent cultures of BOR 5 cells were washed twice with DMEM and incubated at  $37^{\circ}$ C in DMEM with [<sup>125</sup>I]GRP (as above) in the presence of 1 mg/ml BSA (closed circles). Non-specific binding was determined by adding 1  $\mu$ M bombesin (open circles). The binding of [<sup>125</sup>I]GRP to BOR 5 cells was terminated after various times as indicated by rapid washing as in figure 3.1. Cultures were dissolved and counted in a gamma counter. Results are expressed as total binding per 10<sup>6</sup> cells from one experiment performed in duplicate.

101



#### Figure 3.4. Competition Binding of [<sup>125</sup>I]GRP to BOR 5 cells

Confluent, quiescent cultures of BOR 5 cells were washed twice with DMEM and incubated for 30 min at 37°C in DMEM with [<sup>125</sup>I]GRP (as above) and various concentrations of competing peptides in the presence of 1 mg/ml BSA. Peptides were bombesin (open squares), GRP (open circles), litorin (open triangles), NmB (closed squares) and bombesin fragment 8-14 (closed circles). Cultures were then placed on ice and washed and counted as in figure 3.1. Results are expressed as a percentage of the total specific binding of [<sup>125</sup>I]GRP from one experiment performed in duplicate.

## 3.2. THE TRANSFECTED BOMBESIN/GRP RECEPTOR FUNCTIONALLY COUPLES TO PLC-ACTIVATED SIGNALLING PATHWAYS.

In order to verify that the transfected bombesin/GRP receptor expressed in Rat-1 cells could couple to PLC, the effect of bombesin on the intracellular calcium concentration and on the phosphorylation of the prominent PKC substrate 80K/MARCKS (Rodriguez-Pena and Rozengurt, 1986; Brooks, *et al.*, 1991) was measured.

#### 3.2.1. BOMBESIN STIMULATES INTRACELLULAR Ca<sup>2+</sup> MOBILISATION

Bombesin increased  $[Ca^{2+}]_{cyt}$  in a concentration dependent manner in BOR 5 cells (figure 3.5). The maximum increase, elicited by 100 nM bombesin, was 4.0 fold over the basal level (186 nM ± 5.5, n = 62). In contrast bombesin caused no increase in  $[Ca^{2+}]_{cyt}$  in parental Rat-1 cells even up to 1  $\mu$ M (data not shown). Addition of the specific bombesin/GRP receptor antagonist, D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe (Coy, *et al.*, 1992), inhibited bombesin-induced increases in  $[Ca^{2+}]_{cyt}$  (figure 3.6) but had no effect on the increase in  $[Ca^{2+}]_{cyt}$  induced by endothelin which binds to receptors endogenously expressed in Rat-1 cells. Studies of Ca<sup>2+</sup> mobilisation with cultures of BOR 15 cells produced exactly the same results.

#### 3.2.2. BOMBESIN STIMULATES 80K/MARCKS PHOSPHORYLATION

Addition of bombesin to cultures of BOR 5 cells prelabelled with <sup>32</sup>P<sub>i</sub> markedly increased the phosphorylation of the prominent PKC substrate, 80K/MARCKS, in a concentration dependent manner (figure 3.7). The level of phosphorylation was comparable to that induced by 200 nM PDB, a direct activator of PKC. In contrast 80K/MARCKS was not phosphorylated in response to bombesin (up to 100 nM) in parental Rat-1 cells.

These results show that the transfected bombesin/GRP receptor is functionally coupled to intracellular signalling pathways initiated by PLC activation.



#### Figure 3.5. Bombesin-induced increase in [Ca<sup>2+</sup>]<sub>cyt</sub> in BOR 5 cells

Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and detached from the dishes by a 10 min incubation in DMEM with 2.5  $\mu$ g/ml dispase. Cells were loaded with 1  $\mu$ M fura-2/AME while in suspension. Loaded cells were pelleted and resuspended in 2 ml electrolyte solution and transferred to a quartz cuvette. Cells were kept at 37°C and stirred continuously while fluorescence was monitored. After a 1 min control period different concentrations of bombesin were added. Results are expressed as the difference in [Ca<sup>2+</sup>]<sub>cyt</sub> between unstimulated and stimulated levels from one representative experiment that was performed in duplicate. Similar results were obtained in 3 independent experiments.



# Figure 3.6. Antagonist inhibition of bombesin-induced increase in $[Ca^{2+}]_{cyt}$

Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and detached from the dishes by a 10 min incubation in DMEM with 2.5  $\mu$ g/ml dispase. Cells were loaded with 1  $\mu$ M fura-2/AME while in suspension. Loaded cells were pelleted and resuspended in 2 ml electrolyte solution and transferred to a quartz cuvette. Cells were kept at 37°C and stirred continuously while fluorescence was monitored. After 3 min of monitoring fluorescence bombesin was added to 1 nM (Bom) and at 5 min endothelin was added to 50 nM (En). At 1 min the bombesin antagonist D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe was added to 10 nM as indicated (Ant). Shown are the traces from one representative experiment with [Ca<sup>2+</sup>]<sub>cyt</sub> in nM on the right. Similar results were seen in two independent experiments.

105

80K -

0.1 1 10 PDB Bom (nM)

#### Figure 3.7. Bombesin-induced phosphorylation of 80K/MARCKS

Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM without phosphate and incubated in this medium with 5  $\mu$ Ci carrier-free <sup>32</sup>P<sub>1</sub> at 37°C for 18 h. Various concentrations of bombesin (0.1 nM, 1 nM, 10 nM), or 200 nM PDB were added for 10 min. Cultures were put on ice and rapidly washed twice with cold DMEM without phosphate and lysed with 250  $\mu$ l lysis buffer. 80K/MARCKS was immunoprecipitated and resolved on a 8% polyacrylamide gel. <sup>32</sup>P-labelled 80K was visualised by autoradiography. Results shown are from one representative experiment that was performed three times with similar results.

# **3.3.** EXPRESSION OF THE BOMBESIN/GRP RECEPTOR CONFERS A GROWTH RESPONSE TO BOMBESIN IN RAT-1 CELLS.

#### **3.3.1. BOMBESIN STIMULATES DNA SYNTHESIS**

Next, it was determined whether agonist activation of the\_transfected bombesin/GRP receptor leads to reinitiation of DNA synthesis in Rat-1 cells. Quiescent cultures of BOR 5 cells were incubated with bombesin and at various times after stimulation the cells were labelled with [<sup>3</sup>H]thymidine for 3 h. The rate of [<sup>3</sup>H]thymidine incorporation reached a maximum between 18 h - 24 h after the addition of bombesin (figure 3.8). This is consistent with previous results of bombesin-induced DNA synthesis in Swiss 3T3 cells (Rozengurt and Sinnett-Smith, 1983). DNA synthesis induced by bombesin in cultures of BOR 5 cells was inhibited by the specific bombesin/GRP receptor antagonist, D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe, (figure 3.9) with half maximum and maximum inhibition at 1 nM and 30 nM respectively. When BOR 5 cultures were incubated with various concentrations of bombesin, maximum [<sup>3</sup>H]thymidine incorporation was elicited at 1 nM (data not shown). These results substantiate that bombesin-stimulated reinitiation of DNA synthesis in BOR 5 cells was elicited through a bombesin/GRP preferring receptor.

#### **3.3.2. BOMBESIN INDUCES CELL PROLIFERATION**

The mitogenic activity of bombesin in cultures of BOR 5 cells could also be demonstrated by monitoring cell number (rather than [<sup>3</sup>H]thymidine incorporation into acid precipitable material) over a period of several days (figure 3.10).

The addition of bombesin to BOR 5 in serum-free medium increased cell number compared with control cultures as early as 1 day of incubation and reached a maximum after 3 days of bombesin addition. In other experiments it was also verified that bombesin induced DNA synthesis and cell proliferation in cultures of BOR 15 cells. Thus, the transfected bombesin/GRP receptor is coupled to signal transduction pathways which stimulate DNA synthesis and subsequently cell proliferation.


### Figure 3.8. Kinetics of [<sup>3</sup>H]thymidine incorporation induced by bombesin

Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at  $37^{\circ}$ C in DMEM/Waymouth's [1:1] with (solid bars) or without (open bars) 2 nM bombesin. After the times indicated below each bar 0.25  $\mu$ Ci [<sup>3</sup>H]thymidine per mI was added for 3 h and the incorporation was stopped as described in Materials and Methods. DNA synthesis was assessed by measuring the level of [<sup>3</sup>H]thymidine incorporated into half of the acid precipitable material by liquid scintillation counting. Results are expressed as the mean of two determinations from one representative experiment that was performed twice with similar results.



# Figure 3.9. Effect of antagonist on bombesin-induced [<sup>3</sup>H]thymidine incorporation.

Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at 37°C in DMEM/Waymouth's [1:1] with various concentrations of antagonist D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe in the presence (closed squares) or absence (open squares) of 1 nM bombesin. [<sup>3</sup>H]thymidine was added after 18 h and its incorporation was stopped after 6 h of additional incubation as described in Materials and Methods. Results are expressed as the mean cpm from two dishes from a representative experiment that was performed twice with similar results.



Figure 3.10. Cell proliferation induced by bombesin in BOR 5 cells

Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at  $37^{\circ}$ C in DMEM/Waymouth's [1:1] with (closed squares) or without (open squares) 1 nM bombesin. Cells were trypsinised off the dishes and counted with a Coulter counter at 0, 24, 48 and 72 hours after stimulation. Each value is the mean  $\pm$  SE of 5 cultures. Where error bar is not shown, it is smaller than the symbol.

110

## **3.4. BOMBESIN STIMULATES TYROSINE PHOSPHORYLATION IN** RAT-1 CELLS TRANSFECTED WITH THE BOMBESIN/GRP RECEPTOR

In view of the preceding results that demonstrated that bombesin stimulates  $Ca^{2+}$  mobilisation, 80K/MARCKS phosphorylation and cell proliferation, it was next examined whether bombesin can also stimulate tyrosine phosphorylation in Rat-1 cells transfected with the bombesin/GRP receptor. Tyrosine phosphorylation has been implicated in the intracellular signalling of bombesin and other neuropeptides (Rozengurt, 1995). Bombesin stimulates tyrosine phosphorylation of multiple proteins in Swiss 3T3 cells, including broad bands migrating with the apparent molecular masses of 110,000 - 130,000 and 70,000 - 80,000 (Zachary, et al., 1991a; Zachary, et al., 1991b). More recently the focal adhesion associated proteins FAK and paxillin have been identified as prominent tyrosine phosphorylated proteins in Swiss 3T3 cells after stimulation by bombesin (Zachary, et al., 1992; Sinnett-Smith, et al., 1993; Zachary, et al., 1993). FAK is a novel cytosolic tyrosine kinase that lacks Src Homology 2 (SH2) and SH3 domains (Hanks, et al., 1992; Schaller, et al., 1992) but associates with signalling proteins including Src (Cobb, et al., 1994) and paxillin (Hildebrand, et al., 1995). Recent molecular cloning of paxillin revealed a multidomain protein that may function as an adaptor capable of associating with FAK, Crk, Src and vinculin (Turner and Miller, 1994; Salgia, et al., 1995).

# 3.4.1. BOMBESIN INDUCES TYROSINE PHOSPHORYLATION OF MULTIPLE PROTEINS IN **BOR 5** AND **BOR 15** CELLS.

Lysates of quiescent cultures of BOR 5 and BOR 15 cells treated with bombesin were immunoprecipitated with the anti-phosphotyrosine (anti-Tyr(P)) monoclonal antibody (mAb) Py72 and the immunoprecipitates were analysed by Western blotting with anti-Tyr(P) mAb 4G10. As shown in figures 3.11 bombesin induced a dose-dependent increase in the phosphotyrosine content of multiple components including bands migrating with an apparent  $M_r$  of 110,000 - 130,000 and 70,000 - 80,000. Scanning densitometry of these bands revealed that half maximum and maximum tyrosine phosphorylation was achieved at bombesin concentrations of 0.1 - 0.3 nM and 1 - 3 nM, respectively (figure 3.12). In contrast, parental Rat-1 cells did not show any increase in tyrosine phosphorylation in response to bombesin (100 nM) in these and the following experiments.



# Figure 3.11. Bombesin-induced tyrosine phosphorylation of multiple proteins (I)

Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with various concentrations of bombesin for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with anti-Tyr(P) mAb Py72. Immunoprecipitates were resolved by SDS-PAGE and Western blotted with 4G10 anti-Tyr(P) mAb followed by <sup>125</sup>I-labelled anti-mouse IgG. Bands were visualised by autoradiography. Similar results were obtained in at least 3 independent experiments.



# Figure 3.12. Bombesin-induced tyrosine phosphorylation of multiple proteins (II)

Autoradiograms from the previous figure (3.11) were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area. The 70,000 - 80,000 band is depicted by closed circles and the 110,000 - 130,000 band by open circles. Results shown are the average of three independent experiments. Values are expressed as a percentage of the maximum tyrosine phosphorylation after subtraction of the control.



# Figure 3.13. Kinetics of bombesin induction of tyrosine phosphorylation of multiple proteins (I)

Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with 10 nM bombesin for various times. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 anti-Tyr(P) mAb. Immunoprecipitates were resolved by SDS-PAGE and Western blotted with 4G10 anti-Tyr(P) mAb. Similar results were obtained in at least three independent experiments.



# Figure 3.14. Kinetics of bombesin induction of tyrosine phosphoryiation of multiple proteins (II)

Autoradiograms from the previous figure (3.13) were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area. Results shown are of the 70,000 - 80,000 band from a representative experiment of BOR 15 which was repeated at least three times with similar results. Values expressed as a percentage of the maximal tyrosine phosphorylation after subtraction of the control.

RESULTS

The time course of tyrosine phosphorylation induced by bombesin in Rat-1 cells transfected with the bombesin/GRP receptor is shown in figures 3.13 and 3.14. The increase in tyrosine phosphorylation of the  $M_r$  110,000 - 130,000 and the  $M_r$  70,000 - 80,000 bands was detectable within seconds and reached a maximum after 1 min of incubation with 10 nM bombesin in cultures of both BOR 5 and BOR 15 cells. Increased protein tyrosine phosphorylation persisted for at least 3 h.

## **3.4.2.** BOMBESIN-INDUCED TYROSINE PHOSPHORYLATION IS INHIBITED BY THE SPECIFIC BOMBESIN RECEPTOR ANTAGONIST

To further substantiate the involvement of the bombesin/GRP receptor in tyrosine phosphorylation, the bombesin/GRP receptor-specific antagonist D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe was used. This antagonist has already been shown to inhibit bombesin-induced Ca<sup>2+</sup> mobilisation (figure 3.6) and DNA synthesis (figure 3.9). We therefore examined whether it can also inhibit bombesin-induced tyrosine phosphorylation. Cultures of BOR 5 and BOR 15 cells were pretreated with the bombesin/GRP receptor-specific antagonist for 10 min then stimulated by bombesin or endothelin and cell lysates were analysed for tyrosine phosphorylated proteins. Figure 3.15 shows that the antagonist completely blocked bombesin-induced tyrosine phosphorylation of the M<sub>r</sub> 110,000 - 130,000 and the M<sub>r</sub> 70,000 - 80,000 bands. In contrast, the antagonist did not interfere with the stimulation of tyrosine phosphorylation of similar bands induced by endothelin.

## **3.4.3. BOMBESIN-INDUCED TYROSINE PHOSPHORYLATION IS DEPENDENT ON** THE INTEGRITY OF THE ACTIN CYTOSKELETON

It has been shown that bombesin induces tyrosine phosphorylation in Swiss 3T3 cells through a PKC-independent pathway that requires the integrity of the actin cytoskeleton (Sinnett-Smith, *et al.*, 1993). It was examined whether the stimulation of tyrosine phosphorylation by agonist activation of the transfected bombesin/GRP receptor exhibited similar signalling characteristics. The effect of cytochalasin D, which selectively disrupts the network of actin filaments, on bombesin-induced tyrosine phosphorylation was determined. Figure 3.16 shows that pretreating BOR 5 and BOR 15 cultures with cytochalasin D for 2 h, dramatically inhibited the tyrosine phosphorylation of all bands stimulated by bombesin. In contrast, cytochalasin D did not inhibit EGF-induced tyrosine phosphorylation of its receptor (data not shown) indicating that this drug specifically inhibits the tyrosine phosphorylation of bands M<sub>r</sub> 110,000 - 130,000 and 70,000 - 80,000. Thus, bombesin-induced tyrosine phosphorylation in BOR 5 and BOR 15 cells is dependent on the integrity of the actin cytoskeleton.



# Figure 3.15. Effect of antagonist on bombesin-induced tyrosine phosphorylation of multiple proteins

「大学」を行う

Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with (+) or without (-) 100 nM D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe (Ant) for 10 min. Then 10 nM bombesin (Bom) or 50 nM endothelin (En) was added as indicated for 10 min. Cells were lysed and tyrosine phosphorylated proteins immuno-precipitated with Py72 anti-Tyr(P) mAb and Western blotted with 4G10 anti-Tyr(P) mAb. Bands were visualised by autoradiography. Similar results were obtained in two independent experiments.



# Figure 3.16. Effect of Cytochalasin D on bombesin-induced tyrosine phosphorylation of multiple proteins

Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with (+) or without (-) 2  $\mu$ M cytochalasin D (Cyt D) for 2 h. Then 10 nM bombesin (Bom) was added as indicated for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 anti-Tyr(P) mAb and Western blotted with 4G10 anti-Tyr(P) mAb. Bands were visualised by autoradiography. Similar results were obtained in two independent experiments.

118



# Figure 3.17. Effect of GF 109203X on bombesin-induced tyrosine phosphorylation of multiple proteins

Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at  $37^{\circ}$ C in DMEM with or without (-)  $3.5 \mu$ M GF 109203X for 1 h. Then 0.1, 1 or 10 nM bombesin or 200 nM PDB was added as indicated for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 anti-Tyr(P) mAb and Western blotted with 4G10 anti-Tyr(P) mAb. Bands were visualised by autoradiography. Similar results were obtained in three independent experiments.

## 3.4.4. TYROSINE PHOSPHORYLATION INDUCED BY BOMBESIN IS INDEPENDENT OF PKC

We also verified that protein tyrosine phosphorylation was not downstream of PKC. Although the selective PKC inhibitor bisindolylmaleimide (GF 109203X) (Toullec, *et al.*, 1991) did prevent PDB-induced tyrosine phosphorylation of  $M_r$  110,000 - 130,000 and 70,000 - 80,000 bands, it did not prevent bombesin-induced tyrosine phosphorylation in the transfected Rat-1 cells, figure 3.17.

# 3.4.5. BOMBESIN STIMULATES TYROSINE PHOSPHORYLATION OF FAK AND PAXILLIN IN BOR 5 AND BOR 15 CELLS

The cytosolic tyrosine kinase FAK and the adaptor protein paxillin which localise to focal adhesion plaques have been identified as prominent tyrosine phosphorylated proteins in bombesin-stimulated Swiss 3T3 cells. It was important to therefore assess whether agonist activation of the transfected bombesin/GRP receptor stimulates tyrosine phosphorylation of FAK and paxillin in Rat-1 cells. Lysates of bombesin-treated cultures of BOR 5 and BOR 15 were immunoprecipitated with anti-Tyr(P) mAb and the immunoprecipitates were analysed by Western blotting with anti-FAK mAb. Figure 3.18 and 3.19 show that bombesin induced tyrosine phosphorylation of FAK in a concentration dependent manner; half maximum and maximum effects were achieved at 0.3 nM and 3 nM respectively.

Parallel cultures of BOR 5 and BOR 15 cells treated with various concentrations of bombesin were lysed and the lysates were immunoprecipitated with anti-paxillin mAb. The immunoprecipitates were analysed by Western blotting with anti-Tyr(P) mAb. Figure 3.18 and 3.19 show that bombesin caused tyrosine phosphorylation of paxillin in a concentration dependent manner. The concentrations of bombesin that induced half-maximum and maximum stimulation were identical to those required to induce tyrosine phosphorylation of FAK (i.e. 0.3 and 3 nM respectively).



# Figure 3.18. Bombesin-induced tyrosine phosphorylation of FAK and Paxillin (I)

**FAK.** Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with various concentrations of bombesin for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 mAb. Immunoprecipitates were resolved by SDS-PAGE and Western blotted with anti-p125<sup>FAK</sup> mAb. **Paxillin.** Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with various concentrations of bombesin for 10 min. Cells were lysed and paxillin immunoprecipitated with anti-paxillin mAb 165 and Western blotted with 4G10 anti-Tyr(P) mAb.

Similar results were obtained in three independent experiments.

「「「「「「「」」」



# Figure 3.19. Bombesin-induced tyrosine phosphorylation of FAK and Paxillin (II)

Autoradiograms from the previous figure (3.18) were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area. BOR 5 is represented by squares and BOR 15 by circles. Results shown are from a representative experiment which was repeated three times with similar results. Values expressed as a percentage of the maximum tyrosine phosphorylation after subtraction of the control.

15 and find meaning 0.05 - 0.3 and (Lamory et al., 1902) Symmetrican in et al. 952: Lechary, et al. 1957) both Swise (F3 ) wile (Lachery et al., 1951) a and meadedred Rect calls show similar take expendence of symmetry becastory jamos structure are defected initial seconds of southeau addition, teach a meathem of a natural result element for hours. Symphotic structure of synosic photophory jamos inform 7 and 1638 (Cheff), as in teach 3 (Cheff) in an preventer of the specific PSC inform CP (Duble), sharing that this patients is an development of PSC. Take and

## **3.5. SUMMARY AND DISCUSSION**

#### **3.5.1. MULTIPLE COUPLING FROM ONE RECEPTOR**

The experiments presented here were designed to determine whether the bombesin/GRP preferring receptor couples to both PLC activation and tyrosine kinase pathway. To examine this possibility, the cDNA encoding the bombesin/GRP receptor has been expressed in Rat-1 fibroblasts, a cell line that can be reversibly arrested in the  $G_0/G_1$  phase of the cell cycle. The relative abilities of bombesin related peptides to displace <sup>125</sup>I-GRP in these transfected Rat-1 cells are consistent with the binding properties of endogenous bombesin/GRP receptors found in Swiss 3T3 fibroblasts and other cell types. Furthermore the apparent affinity of GRP for the transfected receptor is similar to that for the endogenous receptor. It should be pointed out that the Rat-1 cell lines transfected with the bombesin/GRP receptor used in this study (BOR 5 and BOR 15) expressed fewer receptors per cell than Swiss 3T3 cells. This is of importance in our analysis of cellular responses because overexpression of G protein linked receptors could overcome the specificity of coupling to effector systems (Ashkenazi, *et al.*, 1987). Thus, the clones selected in this study, BOR 5 and BOR 15, provide a useful system to examine the signalling pathways activated by the bombesin/GRP receptor.

Our results demonstrate that agonist binding to the transfected bombesin/GRP receptor elicits multiple responses in Rat-1 cells: 1) mobilisation of Ca<sup>2+</sup> from internal stores leading to a rapid increase in  $[Ca^{2+}]_{cyt}$  and increased phosphorylation of the 80K/MARCKS protein mediated by PKC, 2) reinitiation of DNA synthesis and cell proliferation in serum free medium and 3) tyrosine phosphorylation of multiple proteins including broad bands of M<sub>r</sub> 110,000 - 130,000 and 70,000 - 90,000. We identified two major substrates in the transfected cells as FAK and paxillin. None of these responses were elicited by bombesin in the parental Rat-1 cells.

The characteristics of protein tyrosine phosphorylation induced by agonist binding to the bombesin/GRP receptor transfected into Rat-1 cells were examined in detail in two clones, BOR 5 and BOR 15. An increase in the tyrosine phosphorylation of multiple proteins including FAK and paxillin occurred at low concentrations of bombesin, the half maximum being 0.3 nM. This is similar to that observed in Swiss 3T3 cells (half maximum 0.08 - 0.3 nM) (Zachary, *et al.*, 1992; Sinnett-Smith, *et al.*, 1993; Zachary, *et al.*, 1993). Both Swiss 3T3 cells (Zachary, *et al.*, 1991a) and transfected Rat-1 cells show similar time dependence of tyrosine phosphorylation, increases are detected within seconds of bombesin addition, reach a maximum at 1 min and remain elevated for hours. Bombesin stimulation of tyrosine phosphorylation in BOR 5 and BOR 15 cells, as in Swiss 3T3 cells, is not prevented by the specific PKC inhibitor GF 109203X showing that this pathway is not downstream of PKC. FAK and paxillin are localised in the focal adhesions which form at the termini of actin stress fibres and disruption of the actin cytoskeleton by cytochalasin D prevents tyrosine phosphorylation of these proteins in Swiss 3T3 cells (Sinnett-Smith, *et al.*, 1993; Zachary, *et al.*, 1993). In BOR 5 and BOR 15 cells bombesin-induced tyrosine phosphorylation of multiple proteins was also prevented by cytochalasin D. Therefore the protein tyrosine phosphorylation pathway activated by the transfected bombesin/GRP receptor in Rat-1 cells shows the same characteristics as that induced by bombesin stimulation of Swiss 3T3 cells.

In the transfected Rat-1 cells the increases in  $Ca^{2+}$  mobilisation, protein tyrosine phosphorylation and DNA synthesis were inhibited by the specific bombesin/GRP receptor antagonist, D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe (Coy, *et al.*, 1992), further substantiating that all these responses emanate from the same receptor. This is the first report demonstrating that ligand activation of a transfected bombesin/GRP receptor elicits multiple signalling pathways and leads to cell proliferation. This is also the first time that a G protein linked receptor retains the endogenous ability to activate different pathways and cell proliferation.

The results presented here raise important questions regarding the mechanism(s) by which a single seven transmembrane receptor subtype can couple to multiple signalling pathways. While the mechanism by which bombesin/GRP receptor and other seven-transmembrane domain receptors couple to PLC through heterotrimeric G proteins of the  $G_q$  subfamily is increasingly understood (Lee and Rhee, 1995), little is known about the pathways coupling the bombesin/GRP receptor to tyrosine phosphorylation of FAK and paxillin.

Recent work has implicated activation of the monomeric G protein of the p21<sup>rho</sup> subfamily as one step in the signalling pathway leading to tyrosine phosphorylation of these proteins (Rankin, *et al.*, 1994; Seckl, *et al.*, 1995). Thus, our results suggest that the bombesin/GRP receptor couples to both heterotrimeric ( $G_q$ ) and monomeric (p21<sup>rho</sup>) G proteins.

### **3.5.2. MULTIPLE COUPLING VIA** $\beta\gamma$ **SUBUNITS**

The demonstration that certain seven transmembrane domain receptors couple to  $p21^{ras}$  via  $\beta\gamma$  subunits of the heterotrimeric G protein of the G<sub>i</sub> subfamily (see section 1.3.8.7) suggests a possible coupling mechanism between the bombesin/GRP receptor and  $p21^{rho}$ . It is thought that  $\beta\gamma$  from G<sub>i</sub> activate Ras by stimulating the phosphorylation of the adaptor protein Shc which causes the formation of the Shc/Grb2/mSos complex. Sos then exchanges GDP for GTP on ras. Other GEFs (C3G, Vav) can associate with other adaptors (Crk, Crk-l) (Smit, *et al.*, 1996) and  $p21^{rho}$  is also regulated by GEFs and GAPs (Takai, *et al.*, 1995). Thus by analogy, the  $\beta\gamma$  subunits generated by activation of  $G_q/G_{11}$  could be preferentially coupled to activation of  $p21^{rho}$  and thereby to the tyrosine phosphorylation pathway, whereas the  $\alpha$  subunit of  $G_q$  stimulates PLC (Lee and Rhee, 1995). This would imply specificity of  $\beta\gamma$  complexes coupling to different  $G\alpha$  and activating different effectors.

Section 1.1.6.4 presents some evidence for  $\beta\gamma$  specificity but this is not well documented at the functional level. In the context of activation of p21<sup>rho</sup>, a limited selection of combinations were tested which did not induce stress fibre formation, see table 1.IV. Intriguingly, specificity in localisation is demonstrated for  $\gamma_5$ , which has been shown by immunostaining to localise to focal adhesions and stress fibres. In contrast  $\gamma_3$  had a more diffuse cellular distribution (Hansen, *et al.*, 1994). It is unclear if this of functional importance.

### **3.5.3.** MULTIPLE COUPLING VIA A DISTINCT $G\alpha$ SUBUNIT

Alternatively bombesin-induced tyrosine phosphorylation of FAK and paxillin could be mediated by a  $G\alpha$  subunit. There is evidence which suggests the involvement of  $\alpha_q$ . For example, agonists which cause tyrosine phosphorylation of FAK including bombesin, vasopressin, bradykinin, endothelin, angiotensin, carbachol (M1, M3 receptor subtypes) and norepinephrine ( $\alpha_1$  receptor subtypes) (Huckle, *et al.*, 1990; Force, *et al.*, 1991; Leeb-Lundberg and Song, 1991; Tsuda, et al., 1991; Zachary, et al., 1991a; Gutkind and Robbins, 1992), bind to receptors which have been shown to couple to  $G_{a}$ (Gutowski, et al., 1991; Berstein, et al., 1992; Offermanns, et al., 1994a). Although thrombin and LPA couple to the Ras/MAPK through a pertussis toxin-sensitive G<sub>i</sub> signalling pathway (Van Corven, et al., 1993; Hordijk, et al., 1994) they can activate PLCβ via pertussis toxin-insensitive pathways (Murayama and Ui, 1985; van Corven, et *al.*, 1989; Grand, *et al.*, 1996). Receptors which couple solely to  $G_i$ , for example M<sub>2</sub>-Ach and M<sub>4</sub>-Ach receptors, are not associated with tyrosine phosphorylation of FAK (Gutkind and Robbins, 1992) and adrenomedullin, which elevates cAMP, also does not stimulate protein tyrosine phosphorylation (Withers, et al., 1996). Thus tyrosine phosphorylation is associated with receptors coupled to  $G_{\alpha}$ . Also it has been recognised for some time that Ca<sup>2+</sup> and PIP<sub>2</sub> have important roles to play in cell cytoskeletal arrangements (Janmey, 1994). 80K/MARCKS crosslinks actin and this is inhibited by Ca<sup>2+</sup>/CaM and PKC-dependent phosphorylation. Aderem (1992), proposed that when cells are stimulated with mitogenic peptides the ability of 80K/MARCKS to crosslink actin is reduced and actin around the cell membrane becomes destabilised. This could allow cellular shape changes required during the mitotic cycle to occur. Interestingly  $G\alpha_{q}$  and  $G\alpha_{11}$  have been shown to associate with actin although it is not known if this is of any functional consequence (Ibarrondo, et al., 1995).

It is established that the bombesin/GRP receptor couples to  $G_q$  (Offermanns, *et al.*, 1994a). However, it has been convincingly shown that bombesin-induced tyrosine

phosphorylation of FAK is PKC- and Ca<sup>2+</sup>-independent (see section 1.3.7.8). Another explanation could be that one domain of the bombesin/GRP receptor may couple to  $G_q$  and thereby to PLC, whereas a separate domain could lead to activation of p21<sup>rho</sup> and to the tyrosine phosphorylation of FAK and paxillin through interaction with a different heterotrimeric G-protein.

The concept of a single receptor subtype coupling to multiple G proteins is not without president. Multiple coupling can be observed when components are ectopically expressed and these results must be interpreted with caution. In one study the  $\alpha_2$ -AR was shown to couple to  $G_s$ ,  $G_i$  and  $G_q$  (Chabre, *et al.*, 1994), however, both receptors and G proteins were transfected, potentially compromising specificity. Also coupling of the  $M_2$  receptor to PLC was shown to be dependent on the number of receptors expressed (Ashkenazi, *et al.*, 1987). It has already been stressed that the bombesin/GRP receptor in the present study was not over-expressed. Table 3.I shows receptors for which multiple G protein coupling has been demonstrated. These are single receptor subtypes defined by transfection. Several methods have been used to show *bona fida* coupling to more than one G protein-coupled receptor which couple specifically to only one pathway in the same cell system; the use of pertussis toxin if the  $G_i$  family was involved, and more recently photolabelling with  $[\alpha^{-32}P]$ GTP azidoanilide and specific immunoprecipitation of  $\alpha$  subunits.

Receptor	Gα	comment	ref.
., .,			
parathyroid	G <sub>s</sub> , G <sub>q</sub>	compared to other GPCR	a
thrombin	G <sub>i</sub> , G <sub>q</sub>	PTX sensitivities	b
LH	$G_s, G_a$	compare to other GPCR	С
Calcitonin	$G_{s}, G_{q}$	compare to other GPCR	d
$\alpha_2$ -AR	$G_i, G_s$	PTX, CTX, immunoprecipitation	e
TSH	$G_{s(s,l)}, G_{(q,11)}$	photolabel	f
A <sub>3</sub> adenosine	G <sub>i</sub> , G <sub>o</sub>	photolabel	g
TSH	G <sub>i</sub> , G <sub>12,13</sub> , G <sub>q</sub>	photolabel	ĥ

Table 3.I. Seven transmembrane receptors that couple to multiple G proteins

Refs.: **a** (Abou-Samra, *et al.*, 1992); **b** (Hung, *et al.*, 1992); **c** (Gudermann, *et al.*, 1992); **d** (Chabre, *et al.*, 1992); **e** (Eason, *et al.*, 1992); **f** (Allgeier, *et al.*, 1994); **g** (Palmer, *et al.*, 1995); **h** (Laugwitz, *et al.*, 1996). GPCR, G protein-coupled receptor; PTX, pertussis toxin; CTX, cholera toxin.

Where it has been measured, the  $EC_{50}$  of the receptor agonist for the activation of the G $\alpha$  effector is shown in table 3.II. Interestingly different pathways are activated at different concentrations of agonist with  $G\alpha_q$  mediated events occurring at higher ligand concentration.

Receptor	EC <sub>50</sub> of receptor agonist for G protein activation				
LH Calcitonin α <sub>2</sub> -AR TSH	$\begin{array}{l} G_{s}=0.8 \ nM\\ G_{s}=0.16 \ nM\\ G_{i}=0.16 \ \mu M\\ G_{i}=3 \ nM \end{array}$	$G_q = 2.4 \text{ nM}$ $G_q = 3.7 \text{ nM}$ $G_s = 17 \mu M$ $G_{12,13} = 10 \text{ nM}$	G <sub>q</sub> = 30 nM		

# Table 3.II. Comparison of receptor-ligand $EC_{50}$ values between different $G\alpha$ mediated signals

Refs. as in previous table.

If the effects of bombesin/GRP on cellular events measured in this laboratory in Swiss 3T3 cells are examined more carefully an interesting pattern emerges (see table 3.III). Effects which are dependent on  $G_q$  mediated effects such as PKC and Ca<sup>2+</sup> have an  $EC_{50} \approx 1$  nM, whereas effects which are PKC-independent occur at  $EC_{50} \approx 0.3$ nM. This difference in  $EC_{50}$  values could reflect coupling to a different  $G_{\alpha}$  subunit. Thus the bombesin/GRP receptor could couple to rho and tyrosine phosphorylation of FAK and paxillin via different heterotrimeric G-proteins than those which couple it to PLC.

It has already been documented that G protein-coupled receptors are designed to interact with more than one G protein. Mutational studies within the third intracellular loop show that different G $\alpha$  require different stretches of amino acids on the receptor to bind. For example in the  $\alpha_{2A}$ -AR-C10 amino acids 218-228 are required for G<sub>s</sub> coupling but they do not affect G<sub>i</sub> coupling (Eason and Liggett, 1995). Also in the  $\alpha_1$ -AR distinct areas are required for coupling to different members of the G<sub>q</sub> family (Wu, *et al.*, 1995b) see figure 3.20.

Effector	EC <sub>50</sub> [Bom] nM	PKC/Ca <sup>2+</sup>	ref.
PKC activation	0.4 - 0.6	+	а
Ca <sup>2+</sup> mobilisation	1	+	b
inositol accumulation	>2.5	+	С
AA release	4	N/D	d
cAMP enhancement	1	+	е
fos induction	1	+	f
DNA synthesis	1	+	g
DNA companya i Inc	0.3	_	h
Not /Ut ovchongo	0.5	- -	11 i
Na / II exchange	0.4	± +	i
general Tyr(D)	0.3	_	յ Ն
Tyr kingeo activity	0.1	_	I I
FAK-Tvr(P)	0.08 - 0.3		m
paxillin-Tyr(P)	0.09	-	n

### Table 3.III. Comparison of EC<sub>50</sub> values of cellular effects induced by bombesin

Refs.: **a** (Zachary, *et al.*, 1986); **b** (Mendoza, *et al.*, 1986; Lopez-Rivas, *et al.*, 1987); **c** see table 5.II, this thesis; **d** (Millar and Rozengurt, 1990a); **e** (Millar and Rozengurt, 1988); **f** (Rozengurt and Sinnett-Smith, 1987; Mehmet, *et al.*, 1990); **g** and **h** (Rozengurt and Sinnett-Smith, 1987; Mehmet, *et al.*, 1990); **g** and **h** (Rozengurt and Sinnett-Smith, 1983; Rozengurt and Sinnett-Smith, 1987); **i** (Mendoza, *et al.*, 1986); **j** (Mendoza, *et al.*, 1986); **k** (Zachary, *et al.*, 1991a); **I** (Zachary, *et al.*, 1991b); **m** (Zachary, *et al.*, 1992; Sinnett-Smith, *et al.*, 1993); **n** (Zachary, *et al.*, 1993). N/D, not done.



Figure 3.20. Different areas in the 3rd cytoplasmic loop of  $\alpha_{2A}$ -AR-c10 are required for coupling to  $G\alpha_{14}$  vs  $G\alpha_{q/11}$ 

If the bombesin/GRP receptor is coupling to a second G $\alpha$  then the recent demonstration that microinjected G $\alpha_{12}$  and G $\alpha_{13}$ , but not G $\alpha_q$ , induce Rho-dependent stress fibre assembly is a potential clue (Buhl, *et al.*, 1995). G<sub>12</sub> and G<sub>13</sub> were originally identified by homology cloning (Strathmann and Simon, 1991). They comprise a fourth class of  $\alpha$  subunit and, like G<sub>q</sub>, are unable to be modified by pertussis toxin. Expression of activated mutants of G<sub>q</sub>, G<sub>12</sub> and G<sub>13</sub> increases Na<sup>+</sup>/H<sup>+</sup> exchange causing cellular alkalisation (Dhanasekaran, *et al.*, 1994; Voyno-Yasenetskaya, *et al.*, 1994a). The effect of  $\alpha_q$  and  $\alpha_{12}$  on Na<sup>+</sup>/H<sup>+</sup> exchange is PKC-dependent but that of  $\alpha_{13}$  is not (Dhanasekaran, *et al.*, 1994). Interestingly, table 3.III shows that in Swiss 3T3 cells bombesin stimulates tyrosine phosphorylation and Na<sup>+</sup>/H<sup>+</sup> exchange with a similar low (sub-nanomolar) EC<sub>50</sub>. Also bombesin-induced tyrosine phosphorylation is PKCindependent and Na<sup>+</sup>/H<sup>+</sup> exchange is partially PKC-independent. These observations are consistent with bombesin coupling to a G<sub>12/13</sub>-like G protein.

Agonists which have already been shown to activate  $G_{12}$  and  $G_{13}$  by  $[\alpha^{-32}P]$ GTP azidoanilide photolabelling are thromboxane  $A_2$ , thrombin and TSH. In each case they also cause GTP labelling of  $G_q$  (Offermanns, *et al.*, 1994b; Laugwitz, *et al.*, 1996). However it was not shown categorically that there was only one receptor involved. It is assumed that there is only one TSH receptor because the transfected receptor couples to both cAMP and PLC regulation (van Sande, *et al.*, 1990), but Offermanns, *et al.* (1994b) points out that in their study multiple receptors for thromboxane  $A_2$  and thrombin cannot be excluded. Indeed the possibility of multiple thrombin receptors is debated (reviewed in Grand, *et al.*, 1996). This controversy stems from the differing intracellular events induced by thrombin and the thrombin receptor agonist peptide (Seiler, *et al.*, 1991; Vouret-Craviari, *et al.*, 1993), and see section 1.4. In addition recent experiments with cells derived from a thrombin receptor knockout mouse support the existence of additional thrombin receptors (Connolly, *et al.*, 1996).

In summary, this is the first time that a seven transmembrane receptor, defined by transfection, has been shown to couple to multiple signal transduction pathways and induce cell growth.

## Chapter 4 – The MAPK Pathway Stimulated By Bombesin in Rat-1 Cells

MAPK is a common point of convergence for most mitogenic signalling pathways but there are exceptions, see Withers, *et al.* (1995). The mechanism by which G protein-coupled receptors activate the MAPK kinase pathway is beginning to be unravelled. It is clear that receptors coupled to  $G_i$  such as  $M_2$ -Ach, thrombin, LPA and  $\alpha_2$ -AR activate Ras via  $\beta\gamma$  subunits, but the situation for receptors linked to  $G_q$  is less clear. Some reports document  $G_q$ -coupled receptors activating MAPK via PKC dependent routes, however there are reports of these receptors utilising ras sometimes via  $\beta\gamma$  subunits similar to  $G_i$ -coupled receptors. Section 1.3.8.9 described how controversial the literature is regarding  $G_q$ -coupled receptors especially regarding Ras versus PKC routes to MAPK and some of this can be explained by cellular context.

In Swiss 3T3 cells bombesin-induced MAPK is completely dependent on PKC (Pang, *et al.*, 1993) and does not increase Ras.GTP loading (Satoh, *et al.*, 1990b; Mitchell, *et al.*, 1995) Neither does it activate Raf-1 (Mitchell, *et al.*, 1995; Seufferlein, *et al.*, 1996). Thus bombesin activates MAPK via PKC and not Ras. Consistent with this is the observation that in COS 7 cells bombesin-stimulated MAPK is not dependent on  $\beta\gamma$  subunits (Faure, *et al.*, 1994). It remains to be demonstrated that bombesin can activate ras.

In Rat-1 cells stimulation of the CCK<sub>B</sub> receptor causes the activation of Raf-1 via a PKC-independent mechanism (Seufferlein, *et al.*, 1995). This could mean that peptide receptors coupled to  $G_q$  in Rat-1 cells could activate MAPK via Ras. The transfected bombesin/GRP receptor confers a mitogenic response to bombesin in Rat-1 cells, so the effect of bombesin on the activation of MAPK was examined. The results in this chapter show that bombesin can activate MAPK and Raf-1 activity in a PKC-independent manner. In addition it is shown that bombesin activates Ras.GTP loading in Rat-1 cells transfected with the bombesin/GRP receptor.

## 4.1. BOMBESIN STIMULATION OF MAPK ACTIVITY IN RAT-1 CELLS TRANSFECTED WITH THE BOMBESIN/GRP RECEPTOR

### 4.1.1. BOMBESIN STIMULATES MAPK ACTIVITY

First it was established that bombesin did stimulate MAPK in Rat-1 cells transfected with the bombesin/GRP receptor. Quiescent BOR 15 cells were treated with various factors for 5 min then the were cells lysed. p42<sup>mapk</sup> was

immunoprecipitated and subjected to an *in vitro* immunecomplex kinase assay, using MBP peptide as a substrate. The results in figure 4.1 show that bombesin activated MAPK to the same level as EGF, which was approximately 15-fold over background levels. In contrast PDB activated MAPK to a level that was 30% of that induced by bombesin and EGF.

## 4.1.2. DOSE RESPONSE OF BOMBESIN-INDUCED MAPK ACTIVITY IN BOR 15 CELLS

Bombesin stimulation of MAPK was then examined more closely. BOR 15 cells were treated with increasing concentrations of bombesin for 5 min then MAPK was immunoprecipitated and kinase activity measured as above. Figure 4.2 shows that bombesin induced a concentration dependent increase in MAPK activity the half maximum and maximum concentration being 0.3 and 10 nM respectively. Thus bombesin stimulated a robust increase in MAPK activation in the transfected Rat-1 cells.

## 4.1.3. BOMBESIN-STIMULATED MAPK ACTIVITY IS INDEPENDENT OF PKC IN BOR 15

In view of the role of PKC in  $G_q$ -mediated MAPK activation in some cell types, it was important to assess the contribution of PKC to the MAPK response induced by bombesin in bombesin/GRP receptor transfected Rat-cells. This was examined both by down regulation of phorbol ester-sensitive isoforms of PKC by prolonged treatment with 800 nM PDB, and by use of the specific PKC inhibitor GF 109203X. The specific PKC inhibitor, GF 109203X, is a bisindolylmaleimide which selectively inhibits PKC (assayed on isoforms  $\alpha$ ,  $\beta$ I,  $\beta$ II and  $\gamma$ ) by competing with ATP for the ATP binding site (Toullec, *et al.*, 1991). Subsequently it has also been shown to inhibit other DAGregulated PKC isoforms  $\delta$  and  $\varepsilon$  (Yeo and Exton, 1995). The results presented in figure 4.3 show that phorbol ester-mediated MAPK activation was significantly attenuated by GF 109203X (66% inhibition) and completely abolished by chronic pretreatment with PDB. In contrast EGF, which does not activate PKC in Rat-1 cells was not significantly inhibited by PKC inhibition and only slightly by PKC down regulation. Similarly bombesin-induced MAPK was not significantly inhibited by GF 109203X or pretreatment with PDB.

Thus the results in figures 4.1 - 4.3 demonstrate that bombesin induces a strong stimulation of MAPK activity in Rat-1 cells transfected with the bombesin GRP receptor and this is a PKC-independent event.



### Figure 4.1. p42<sup>mapk</sup> activation induced by various factors

BOR 15 cells in 100 mm dishes were washed twice with DMEM and incubated in 5 ml DMEM without (–) or with 1 nM Bombesin (Bom), 200 nM PDB or 5 ng/ml EGF for 5 min at 37°C. Cells were lysed and p42<sup>mapk</sup> was immunoprecipitated with polyclonal anti-p42<sup>mapk</sup> antibodies. MAPK activity was measured by an *in vitro* kinase assay using MBP peptide as a substrate as described in materials and methods. Results shown are from one experiment performed in duplicate. Similar results were obtained in two independent experiments.



Figure 4.2. Bombesin-induced increase in p42<sup>mapk</sup> activity

BOR 15 cells in 100 mm dishes were washed twice in DMEM and incubated in 5 ml DMEM with increasing concentrations of bombesin for 5 min at 37°C. p42<sup>mapk</sup> was immunoprecipitated followed by an *in vitro* kinase reaction as above. Results shown are the mean of two independent experiments that were performed in duplicate.



## Figure 4.3. Effect of PKC inhibition and down regulation on bombesinstimulated MAPK activity

5 day-old cells were pretreated with (black bars) or without 800 nM PDB for 48 h. Then all the cells were washed twice in DMEM and incubated in 5 ml DMEM for 1 h with (grey bars) or without 3.5  $\mu$ M GF 109203X. The cells were stimulated without (–) or with 100 nM PDB, 1 nM bombesin (Bom) or 5 ng/ml EGF for 5 min at 37°C. The cells were lysed and p42<sup>mapk</sup> was immunoprecipitated and kinase activity assayed as above. Results shown are from one representative experiment that was performed in duplicate. Similar results were obtained in two independent experiments.

## 4.2. BOMBESIN STIMULATION OF RAF-1 ACTIVITY IN TRANSFECTED RAT-1 CELLS

Next, the upstream signals were investigated. The prototypical MEK activator is Raf-1, although other Raf and non-Raf MEK activators exist. In Rat-1 cells the  $G_{q}$ -coupled M<sub>1</sub>-Ach receptor does not activate Raf-1 (Russell, *et al.*, 1994) and in Swiss 3T3 cells bombesin does not activate Raf-1 (Mitchell, *et al.*, 1995). This would suggest that bombesin/GRP receptor transfected into Rat-1 cells would not activate Raf-1. However receptors that are coupled to  $G_q$  have been shown to activate Raf-1 (Seufferlein, *et al.*, 1995) or require functional Raf-1 to activate MAPK (Hawes, *et al.*, 1995). The ability of bombesin to stimulate Raf-1 activity in transfected Rat-1 cells was determined.

### 4.2.1. BOMBESIN STIMULATES RAF-1 ACTIVATION IN BOR 15 CELLS

Quiescent BOR 15 cells were stimulated with 1 nM bombesin for increasing periods of time. The cells were lysed and Raf-1 was immunoprecipitated. Raf-1 immunocomplexes were incubated with GST-MEK and recombinant ERK2 proteins in the presence of ATP to reconstitute the MAPK cascade *in vitro*. Recombinant ERK2 was then collected and used to phosphorylate MBP peptide in the presence of  $\gamma$ -<sup>32</sup>P-ATP as above. The results are shown in figure 4.4. Bombesin induced a rapid and transient activation of Raf-1 kinase activity. The maximum activation occurred 3 min after bombesin stimulation and was nearly all over by 10 min.

### 4.2.2. BOMBESIN-INDUCED RAF-1 ACTIVATION IS NOT DEPENDENT ON PKC

Receptors coupled to  $G_q$  activate PKC and PKC has been proposed to directly activate Raf-1. In addition phorbol esters can cause activation of Raf-1 in at least some cell types. However, in Rat-1 cells the CCK<sub>B</sub> receptor activated Raf-1 in a PKCindependent manner (Seufferlein, *et al.*, 1995). Therefore the effect of GF 109203X was examined on bombesin stimulation of Raf-1 activity. Quiescent BOR 15 cells were pretreated with 3.5 µM GF 109203X for 1 h prior to stimulation by various factors for 3 min. The results are shown in figure 4.5. In Rat-1 cells PDB activates MAPK 3-7 fold above unstimulated levels (figures 4.1 and 4.3). However figure 4.5 shows that PDB does not stimulate Raf-1 activity in Rat-1 cells. Raf-1 activity stimulated by EGF, which does not stimulate PKC in Rat-1 cells, was not inhibited by the specific PKC inhibitor GF 109203X. In fact, a small stimulation was noticed in every case suggesting that GF 109203X eliminates a constitutive feed-back loop in Rat-1 cells. Similarly, bombesininduced Raf-1 activity was not affected by the PKC inhibitor demonstrating that bombesin activates Raf-1 via a PKC-independent route.



Figure 4.4. Time course of bombesin-induced Raf-1 activity

BOR 15 cells in 100 mm dishes were washed twice in DMEM and incubated in 5 ml DMEM with 1 nM bombesin for increasing times at 37°C. Cells were lysed and Raf-1 was immunoprecipitated with polyclonal C12 anti-Raf-1 antibody. Raf-1 activity was measured by a two stage *in vitro* immunecomplex kinase assay as described in materials and methods. Results shown are the mean of two independent experiments performed in duplicate.

136



# Figure 4.5. Effect of PKC inhibition on Raf-1 activity induced by various mitogens.

BOR 15 cells in 100 mm dishes were washed twice with DMEM and incubated in 5 ml DMEM for 1 h with (closed bars) or without (open bars) 3.5  $\mu$ M GF 109203X. Cells were then stimulated for 3 min at 37°C without (–) or with 1 nM bombesin (Bom), 200 nM PDB or 5 ng/ml EGF. Raf-1 was immunoprecipitated and kinase activity measured as above. Results shown are from one representative experiment performed in duplicate. Similar results were obtained in two independent experiments.

## 4.2.3. PERTUSSIS TOXIN DOES NOT INHIBIT BOMBESIN-INDUCED RAF-1 ACTIVITY

G protein-coupled receptors that do not activate MAPK via PKC-dependent mechanisms are often coupled to G<sub>i</sub> and in the previous chapter evidence of coupling to multiple G-proteins was presented. Coupling to G<sub>i</sub> can be identified with the use of pertussis toxin which will ADP-ribosylate a specific arginine residue in  $G\alpha_i$  thus preventing its activation, see section 1.1.6.5. This also means that the  $\alpha_i$  does not dissociate from the receptor or its  $\beta\gamma$  subunit, and hence pertussis toxin can block both  $\alpha_i$  and  $\beta_\gamma$  signalling. To assess whether bombesin-induced activation of Raf-1 was mediated by a G protein from the G<sub>i</sub> family, BOR 15 cells were pretreated for 3 h with 30 ng/ml pertussis toxin and then stimulated with bombesin, LPA or EGF for 3 min. LPA stimulates the MAPK pathway and Raf-1 activity in a pertussis toxin-sensitive fashion (Hordijk, et al., 1994; Seufferlein, et al., 1995). In agreement with this, treatment of BOR 15 cells with pertussis toxin inhibited LPA-induced Raf-1 activity (figure 4.6). In contrast EGF, which signals through a receptor tyrosine kinase that does not use G<sub>i</sub> proteins, was not affected by pertussis toxin pretreatment. Similarly bombesin-induced Raf-1 activation was not inhibited by pretreatment with pertussis toxin.

The results in figures 4.4 - 4.6 demonstrate that ligand activation of the bombesin/GRP receptor transfected into Rat-1 fibroblasts induces Raf-1 kinase activity which is pertussis toxin-insensitive and PKC-independent.

## 4.3. BOMBESIN STIMULATION OF RAS.GTP LOADING IN TRANSFECTED RAT-1 CELLS

Because bombesin-induced stimulation of Raf-1 activity was PKC-independent in Rat-1 transfected with the bombesin/GRP receptor cells the possibility that it could activate Ras was investigated. Quiescent BOR 15 cells were labelled overnight with <sup>32</sup>Porthophosphate and then stimulated for 2.5 min with various factors. Ras was immunoprecipitated and the bound guanine nucleotides were eluted and separated by thin layer chromatography. The results are shown in figures 4.7 and 4.8. Bombesin increased the proportion of Ras in the GTP bound form from 28% to 43% (a 50% increase). This was half the level of EGF-induced stimulation which increased Ras.GTP from 28% to 61% (a 115% increase). Thus activation of the transfected bombesin/GRP receptor stimulated the GTP loading of Ras.



# Figure 4.6. Effect of pertussis toxin on Raf-1 activity induced by various mitogens.

BOR 15 cells in 100 mm dishes were washed twice in DMEM and incubated in 5 ml DMEM with (closed bars) or without (open bars) 30 ng/ml pertussis toxin for 4 h. Cells were then stimulated for 3 min at 37°C without (–) or with 1 nM bombesin (Bom), 2  $\mu$ M LPA or 5 ng/ml EGF. Raf-1 was immunoprecipitated and kinase activity measured as above. Results shown are from one experiment that was performed in duplicate.



### Figure 4.7. Bombesin stimulation of loading (I)

BOR 15 cells in 100 mm dishes were washed twice in DMEM without phosphate and incubated overnight in 5 ml DMEM without phosphate supplemented with 1 mg/ml BSA, 20 mM HEPES and 1.6 mCi orthophosphate. Cells were stimulated without (–) or with 1 nM bombesin (Bom) or 5 ng/ml EGF for 2.5 min at 37°C. The cells were lysed and ras was immunoprecipitated with mAb 259 in duplicate (+) with non-specific controls (–) from the detergent phase as described in materials and methods. The guanine nucleotides were eluted from ras and resolved by thin layer chromatography on PEI cellulose plates as described in materials and methods and visualised by autoradiography. Results shown are from one representative experiment that was performed with duplicate immunoprecipitations from duplicate cultures. Similar results were obtained in 2 independent experiments.





Labelled nucleotides from the previous figure (4.7.) were quantified in terms of total volume with a Phosphorimager using ImageQuant from Molecular Dynamics. The results are presented as the mean percent of Ras in GTP bound form (GTP/(GTP + GDP) x100)  $\pm$  SE, of duplicate immunoprecipitations from duplicate cultures.

## 4.4. SUMMARY AND DISCUSSION

### 4.4.1. MULTIPLE MEK ACTIVATORS

The increasing recognition of the importance of cell context in determining the signal transduction pathways leading to MAPK prompted the investigation of the effect of bombesin on MAPK activation in Rat-1 cells. These results demonstrate that MAPK, Raf-1 and Ras are activated by bombesin in Rat-1 fibroblasts.

Although direct stimulation of PKC by phorbol esters can activate MAPK and bombesin also stimulates PKC (see section 1.3.8), in BOR 15 cells the activation of MAPK by bombesin is PKC-independent. It cannot be discounted that there is a component of bombesin-induced MAPK activation that is PKC-dependent but if there is, it is compensated for when PKC is inhibited or down regulated. On the contrary, direct stimulation of PKC by phorbol esters does not activate Raf-1. It could be envisaged that PKC could activate Raf-1 but only in synergy with another signal elicited by bombesin. However inhibition of PKC with the specific inhibitor GF 109203X did not affect bombesin-stimulated Raf-1 activity. Thus PKC does not contribute to Raf-1 activation that is stimulated by bombesin. Because both bombesinstimulated MAPK and Raf-1 activation were PKC-independent this suggests that Raf-1 is the upstream activator of MAPK in bombesin-stimulated cells. In contrast Raf-1 does not appear to be the upstream activator of MAPK in PDB stimulated cells. Raf-1independent stimulation of MAPK by phorbol ester has also been reported by (Chao, *et al.*, 1994).

Bombesin activates MAPK kinase to the same level as EGF but is much poorer at stimulating Raf-1. This is not a new observation. Faure and Bourne (1995) show that compared to EGF, LPA stimulates little Raf-1 and MEK activity but a comparable amount of MAPK activity in Rat-1 and Swiss 3T3 cells. But they do point out that low levels of Raf-1 activity may still saturate MAPK activation. Thus in BOR 15 cells the smaller Raf-1 activation stimulated by bombesin could saturate MAPK activation and a larger stimulation of Raf-1 by EGF is not translated to further stimulation of MAPK.

PDB could activate MAPK via a different MEK activator. For example in PC12 cells TPA transiently activates Raf-1 which is partially PKC-dependent, but gives a prolonged activation of A-Raf which is completely PKC-dependent (Bogoyevitch, *et al.*, 1995; Erhardt, *et al.*, 1995). Also it is well documented that there are Raf-1-independent mechanisms of ERK activation (Chao, *et al.*, 1994; Vaillancourt, *et al.*, 1994; Zheng, *et al.*, 1994; Erhardt, *et al.*, 1995; Faure and Bourne, 1995) and also uncharacterised MEK activators that non-Raf isoforms (Pang, *et al.*, 1995; Reuter, *et al.*, 1995).

### 4.4.2. BOMBESIN ACTIVATION OF RAS. MEDIATION BY $G\beta\gamma$ ?

How does bombesin activate Ras in BOR 15 cells? In Swiss 3T3 cells the bombesin receptor couples to  $G_q$  and is completely dependent on PKC for MAPK activation. So when PKC is removed by inhibition or down regulation why is MAPK activation not compensated for by  $\beta\gamma$ -mediated Ras activation? It could be argued that  $\beta\gamma$  from  $G_q$  are different, however, the evidence presented in section 1.1.6.4 suggests that there is little specificity between  $\beta\gamma$  subunit combinations when activating effectors although it is realised that expression levels could be an important part of specificity.

Another explanation could be that the levels of expression of  $G\alpha_{\alpha}$  may not be high enough to support the release of enough  $\beta\gamma$  upon activation of the receptor. For example in  $\beta\gamma$ -mediated activation of adenylyl cyclase type II, Taussig and Gilman (1995) describes that this is not sensitive to  $\beta\gamma$  combinations, but rather to the concentration.  $G\alpha_s$  activates adenylyl cyclase type II at picomolar concentrations whereas  $G\beta\gamma$  activates at nanomolar concentrations. Therefore  $\beta\gamma$  stimulation of adenylyl cyclase type II occurs from a much more abundant receptor-G-protein complex. Similarly  $G\alpha_q$  activates PLC  $EC_{50} = 0.6$  nM whereas  $\beta\gamma$  activates  $EC_{50} = 25$  nM (Park, *et al.*, 1993). Exton (1994) notes that G proteins that are the source of  $\beta\gamma$  that stimulate PLC are more abundant than  $G_q$  or  $G_{11}$ , i.e. they may be released from  $G_i$  or G<sub>o</sub> subtypes. If the level of G protein expression is important this could explain why the more abundant G<sub>i</sub> utilises a signalling pathway that is well agreed upon. If G<sub>q</sub> is less abundant then thresholds for activation of  $\beta\gamma$  effectors which lead to the activation of Ras may or may not be reached. Also if the receptor couples to multiple G proteins (see Chapter 3) then the sum of the multiple  $G\alpha$  could be important. Alternatively other components of the pathway downstream of  $\beta\gamma$  (discussed in section 1.3.8.10) could be expressed at higher levels in Rat-1 than in Swiss 3T3.

### 4.4.3. CELLULAR CONTEXT

To conclude this chapter it has been shown that in Rat-1 cells bombesin stimulates the MAPK pathway via a different mechanism than has been reported previously in Swiss 3T3 cells, which involves the activation of Ras. The regulation of the MAPK pathway is different in some specialised cells such as neurons where voltage dependent Ca<sup>2+</sup> influx will activate Src and Ras (Rosen, *et al.*, 1994; Farnsworth, *et al.*, 1995; Rusanescu, *et al.*, 1995), and in B cells where leukocyte specific tyrosine kinases such as Lyn and Syk are necessary for G protein-coupled receptor MAPK activation (Wan, *et al.*, 1996). However, among fibroblasts there are differences and these have been described in earlier sections. In the context of bombesin, in Swiss 3T3 cells bombesin does not activate Ras (Satoh, *et al.*, 1990b and unpublished observations from this laboratory) and the activation of MAPK is completely PKC-dependent (Pang,
*et al.*, 1993; Seufferlein, *et al.*, 1996). However the results presented here have shown that in Rat-1 cells bombesin-induced MAPK activation is independent of PKC and most likely is dependent on the activation of Ras. This has important implications. Raf-1 is not the only downstream effector of Ras. PI3K has been suggested to be downstream of Ras (Rodriguez-Viciana, *et al.*, 1994) and a number of other molecules have been shown to bind to Ras in a GTP-dependent manner potentially stimulating diverse signalling pathways (Marshall, 1996). Also if  $\beta\gamma$  are stimulating protein tyrosine kinases (EGF receptor, Src, Tsk, Btk), serine kinases homologous to the yeast Ste20p, or lipid kinases such as PI3K (see section 1.3.8.10) then more signalling pathways could branch out from these points.

In conclusion,  $G_q$  coupled receptors have the potential to activate MAPK via multiple pathways such as a PKC route and a Ras route, the latter of which is probably mediated by  $\beta\gamma$  subunits. The exact route to MAPK activation in a given cell may depend on the exact repertoire and expression levels of the individual components of the multiple pathways.

### Chapter 5 - The Calcium Signal

As discussed in section 1.3.2.5, the possibility that Ca<sup>2+</sup> fluxes could play a role in the transduction of mitogenic signals has attracted intense interest. Neuropeptides such as bombesin, vasopressin, endothelin and bradykinin, which are potent mitogens for Swiss 3T3 cells (Rozengurt, 1986), induce a rapid and transient increase in the cytoplasmic concentration of Ca<sup>2+</sup> followed by a persistent depletion of Ca<sup>2+</sup> from stores in the ER (Lopez-Rivas and Rozengurt, 1984; Mendoza, et al., 1986; Lopez-Rivas, et al., 1987; Nanberg and Rozengurt, 1988). This effect is mediated by the second messenger  $Ins(1,4,5)P_3$  (Berridge, 1993) which is generated in response to receptormediated activation of  $G_q$ , the GTP-binding protein that couples neuropeptide receptors to PLC $\beta$  (Berstein, et al., 1992; Lee, et al., 1992). However, hydrolysis of PIP<sub>2</sub> also generates DAG which activates PKC, a well known pathway leading to gene expression and cell proliferation (Rozengurt, et al., 1984; Nishizuka, 1986; Nishizuka, 1988; Rozengurt and Sinnett-Smith, 1988a). Furthermore, insulin and EGF can stimulate DNA synthesis in Swiss 3T3 cells without inducing Ca<sup>2+</sup> mobilisation (Rozengurt, et al., 1983b; Vara and Rozengurt, 1985). Similarly, mutant receptors for platelet-derived growth factor and fibroblast growth factor defective in PLC<sub> $\gamma$ </sub> activation have been shown to retain mitogenic activity upon ligand activation (Mohammadi, et al., 1992; Peters, et al., 1992; Valius, et al., 1993; Valius and Kazlauskas, 1993). Thus, the contribution of Ca<sup>2+</sup> to mitogenic signalling remains unclear.

An approach for assessing the role of  $Ca^{2+}$  in the transition from  $G_0$  into DNA synthesis, is to use of specific inhibitors of the  $Ca^{2+}$  ATPase pump of the ER, which is responsible for the accumulation of  $Ca^{2+}$  into these stores. The tumour promoter thapsigargin and the structurally distinct compound DBHQ have been identified as inhibitors of  $Ca^{2+}$  accumulation in the ER (Moore, *et al.*, 1987; Takemura, *et al.*, 1989; Thastrup, *et al.*, 1989; Thastrup, *et al.*, 1990; Bian, *et al.*, 1991; Sagara and Inesi, 1991). Addition of these compounds to intact cells induces mobilisation of  $Ca^{2+}$  from internal stores, bypassing PLC-mediated formation of  $Ins(1,4,5)P_3$ . Previous studies have shown that the addition of either thapsigargin or DBHQ, at micromolar concentrations, profoundly inhibited cell proliferation or induced cytotoxic effects (Ghosh, *et al.*, 1991; Short, *et al.*, 1993). These findings are difficult to reconcile with the fact that mitogenic neuropeptides and other growth promoting factors also cause persistent depletion of  $Ca^{2+}$  from the ER and thapsigargin has been shown to be a tumour promoter.

The experiments presented here were designed to assess the effect of gradual depletion of  $Ca^{2+}$  from the ER on the ability of quiescent cells to exit from  $G_0$  and enter DNA synthesis.

### 5.1. EFFECT OF THAPSIGARGIN AND DBHQ ON PDB-STIMULATED DNA SYNTHESIS

Initial experiments confirmed that thapsigargin (2 - 10 nM) inhibited DNA synthesis induced by a variety of growth promoting factors in Swiss 3T3 cells. However, when the dose-dependency of this effect was carefully examined, a stimulatory action was discovered at low concentrations of thapsigargin. This prompted a detailed analysis of the effects of Ca<sup>2+</sup>ATPase pump inhibitors on cellular DNA synthesis.

# 5.1.1. THAPSIGARGIN AND DBHQ STIMULATE [<sup>3</sup>H]THYMIDINE INCORPORATION IN SYNERGY WITH PDB

Quiescent cultures of Swiss 3T3 cells were preincubated with a range of concentrations of DBHQ or thapsigargin for 6 h. The cultures were then exposed to fresh medium containing PDB, the same concentrations of thapsigargin or DBHQ and [<sup>3</sup>H]thymidine. Cumulative [<sup>3</sup>H]thymidine incorporation was measured after 40 h of incubation. Figure 5.1 shows that both DBHQ and thapsigargin caused a striking stimulation of DNA synthesis in the presence of PDB. The optimum effects were observed at 7.5  $\mu$ M DBHQ and 0.5 nM thapsigargin where they reached 63% and 62%, respectively, of the maximum response induced by FBS. The stimulation of [<sup>3</sup>H]thymidine incorporation by these agents was induced within a narrow concentration range. At higher concentrations of DBHQ or thapsigargin cell toxicity (detachment) was clearly evident after 40 h of incubation. DBHQ and thapsigargin added alone did not induce any [<sup>3</sup>H]thymidine incorporation in Swiss 3T3 cells.

To examine the extent to which the pretreatment with 0.5 nM thapsigargin and 7.5  $\mu$ M DBHQ affected the resulting stimulation of [<sup>3</sup>H]thymidine incorporation, confluent quiescent cultures of Swiss 3T3 cells were preincubated with these agents for various times. Cultures were then washed, thapsigargin or DBHQ was replaced and PDB added. Cumulative [<sup>3</sup>H]thymidine incorporation was measured after 40 h. The results in figure 5.2 reveal that DBHQ or thapsigargin synergised with PDB in stimulating [<sup>3</sup>H]thymidine incorporation without preincubation but the effect was enhanced after a pretreatment of 6 h. Similar enhancement of [<sup>3</sup>H]thymidine incorporation was seen when the cells were preincubated with thapsigargin or DBHQ for 8 h instead of 6 h.

To further substantiate the observed synergy between PDB and DBHQ or thapsigargin, the effect of a range of concentrations of PDB in the presence of a fixed and optimum concentration of either DBHQ (7.5  $\mu$ M) or thapsigargin (0.5 nM) was examined.



# Figure 5.1. Dose response curve for the stimulation of DNA synthesis by DBHQ or thapsigargin

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated with various concentrations of DBHQ (upper), or thapsigargin (lower) for 6 h. The cells were washed twice with DMEM and incubated at 37°C in 2 ml of 1:1 (vol/vol) DMEM/Waymouth's medium containing 0.25  $\mu$ Ci [<sup>3</sup>H]thymidine per ml and various concentrations of DBHQ or thapsigargin in presence (closed circles) or absence (open circles) of 80 nM PDB. After 40 h, DNA synthesis was assessed by measuring the level of [<sup>3</sup>H]thymidine incorporated into half of the acid precipitable material by liquid scintillation counting. Each point is the mean of two determinations from one representative experiment. 10% FBS gave an incorporation of 197 x 10<sup>3</sup> cpm (upper) or 189 x 10<sup>3</sup> cpm (lower). Similar results were obtained in 17 - 22 independent experiments.



Pretreatment Time (h)

### Figure 5.2. Effect of the length of preincubation with DBHQ and thapsigargin on the subsequent stimulation of DNA synthesis by these compounds in synergy with PDB

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated without (-, open bars) or with (closed bars) 7.5  $\mu$ M DBHQ (upper) or 0.5 nM thapsigargin (lower) for 0, 2, 6, or 24 h. Then (as above), the cells were washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine, 7.5  $\mu$ M DBHQ (upper) or 0.5 nM thapsigargin (lower) and 20 nM (upper) or 80 nM PDB (lower). DNA synthesis is expressed as percent of maximum. Each bar is the mean of 3 - 8 determinations. Error bars are standard error of the mean (SEM).



### Figure 5.3. Dose response curve for the stimulation of DNA synthesis by PDB

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 - 8 h without (open circles) or with (closed circles) 7.5  $\mu$ M DBHQ (upper) or 0.5 nM thapsigargin (lower). Then, as above, the cells were washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine and various concentrations of PDB, in the absence (open circles) or presence (closed circles) of 7.5  $\mu$ M DBHQ (upper) or 0.5 nM thapsigargin (lower). 10% FBS gave 164 x 10<sup>3</sup> cpm (upper) and 168 x 10<sup>3</sup> cpm (lower). Similar results were obtained in 3 independent experiments.

The results show striking synergistic stimulation of [<sup>3</sup>H]thymidine incorporation by PDB in the presence of either DBHQ or thapsigargin (figure 5.3). Maximum effects were produced at a concentration of 80 nM PDB. PDB alone did not induce any significant [<sup>3</sup>H]thymidine incorporation in Swiss 3T3 cells, in agreement with previous results (Dicker and Rozengurt, 1978; Dicker and Rozengurt, 1980).

### 5.1.2. KINETICS OF THAPSIGARGIN AND DBHQ STIMULATED [3H]THYMIDINE INCORPORATION IS CONSISTENT WITH ENTRY TO S-PHASE

To verify that the increase in [<sup>3</sup>H]thymidine incorporation was due to DNA synthesis, the incorporation of [<sup>3</sup>H]thymidine by the combination of PDB with either DBHQ or thapsigargin was also assessed as a function of time (figure 5.4). These compounds stimulated [<sup>3</sup>H]thymidine incorporation after a lag period of 20 - 22 h. This is consistent with the duration of the transition from  $G_0/G_1$  to DNA synthesis stimulated by neuropeptide growth factors (e.g. bombesin, vasopressin) in Swiss 3T3 cells (Dicker and Rozengurt, 1980; Rozengurt and Sinnett-Smith, 1983).

### 5.1.3. THAPSIGARGIN AND DBHQ INCREASE THE PROPORTION OF CELLS THAT REINITIATE DNA SYNTHESIS

To assess whether PDB and thapsigargin or DBHQ interact synergistically in eliciting initiation of DNA synthesis rather than in changing the specific activity of the [<sup>3</sup>H]-thymidine precursor pool, quiescent cultures of Swiss 3T3 cells were treated with PDB and thapsigargin or DBHQ and incorporation of [<sup>3</sup>H]thymidine into DNA was quantified by autoradiography of labelled nuclei. As shown in figure 5.5 (left panel), thapsigargin and DBHQ caused a marked enhancement of the labelling index (from 7% in cultures treated with PDB alone to 44% and 54% in cultures treated also with 0.5 nM thapsigargin and 7.5  $\mu$ M DBHQ respectively). Thus, PDB in combination with either thapsigargin or DBHQ synergistically enhances the proportion of cells that enter into DNA synthesis. This was in very good agreement with the cumulative [<sup>3</sup>H]thymidine incorporation into parallel cultures expressed as a percentage of the response induced by 10% FBS (figure 5.5, right panel).

The results shown in figures 5.1 - 5.5 demonstrate that DBHQ and thapsigargin, at low concentrations, promote synergistic stimulation of DNA synthesis in combination with PDB.



# Figure 5.4. Time course for incorporation of [<sup>3</sup>H]thymidine in the presence of thapsigargin and DBHQ

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 - 8 h without (open circles) or with 0.5 nM thapsigargin (squares) or 7.5  $\mu$ M DBHQ (closed circles). Then the cells were washed twice with DMEM and incubated for various times (from 16 h to 44 h) in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine, 80 nM PDB and in the absence (open circles) or presence of 0.5 nM thapsigargin (squares) or 7.5  $\mu$ M DBHQ (closed circles). A similar lag period prior to DNA synthesis was obtained in 3 independent experiments.



### Figure 5.5. Effect of thapsigargin and DBHQ on DNA synthesis assessed by autoradiography and liquid scintillation counting of [<sup>3</sup>H]thymidine incorporation

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 - 8 h without (open bars) or with (closed bars) 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ. Then, the cells were washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with 2.5  $\mu$ Ci [<sup>3</sup>H]thymidine per ml (left panel) or 0.25  $\mu$ Ci/ml (right panel), and with 20 nM PDB either in the absence (-, open bars) or presence (closed bars) of 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ. For autoradiography the experiment was stopped and cells fixed by replacing medium with formol saline. Labelled nuclei were visualised by *in situ* autoradiography using Kodak autoradiographic stripping plate AR10. Films were then developed *in situ*. Values on the left are the percent of nuclei which were labelled, counted from eight microscopic fields (total of 65 - 80 cells per field) per dish. 10% FBS labelled 93% of nuclei. Values on the right are expressed as percentage of the values obtained with 10% FBS and are presented for comparison with the autoradiography results. For both left and right panels values are the mean of 2 dishes with the error bars showing the range.

### **5.2. EFFECT OF DBHQ WITH OTHER MITOGENS**

Next, it was examined whether DBHQ could act synergistically with signalling pathways other than those mediated by PKC. As shown in table 5.I, the effect of DBHQ on DNA synthesis in the presence of various growth promoting factors was determined.

MITOGEN(S)	[ <sup>3</sup> H]Thymidine Incorporation (% of FBS)	
		+ DBHQ
_	$1.3 \pm 0.9$	$0.8 \pm 0.5$
INS	$10.4 \pm 3.3$	$17.3 \pm 3.8$
EGF	$6.5 \pm 1.3$	$8.0 \pm 2.5$
FOR + IBMX	$4.2 \pm 2.3$	$4.5 \pm 1.0$
BK	$4.1 \pm 1.3$	$8.8 \pm 4.3$
BOM	$9.1 \pm 3.5$	$36.7 \pm 6.5$
PDB	$7.1 \pm 0.7$	$43.7 \pm 3.5$

Table 5.I.	Effect of DBHQ on D	NA synthesis induce	d by various mitogens
------------	---------------------	---------------------	-----------------------

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 8 h without (-) or with (+) 7.5  $\mu$ M DBHQ. Then, as in Fig. 5.1, the cells were washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine and various mitogens: 1  $\mu$ g/ml insulin (INS), 5 ng/ml EGF, 50  $\mu$ M forskolin (FOR), 25  $\mu$ M IBMX, 50 nM bradykinin (BK), 2.5 nM bombesin (BOM) and 20 nM PDB, in the presence (+) or absence (-) of 7.5  $\mu$ M DBHQ. Values are expressed as percent of [<sup>3</sup>H]thymidine incorporation obtained with 10% FBS and are the mean, with the error shown as SE, of 4 independent experiments performed in duplicate, or 27 experiments for PDB.

The polypeptides EGF and insulin bind to receptors endowed with intrinsic, ligand dependent, tyrosine kinase activity which are not linked to PLC $\gamma$  and hence to PKC activation in Swiss 3T3 cells (Rozengurt, *et al.*, 1983b; Vara and Rozengurt, 1985). These polypeptides failed to act synergistically with DBHQ. Similarly, the cAMP elevating agents forskolin and IBMX which are potent mitogens for Swiss 3T3 cultures in the presence of insulin (Rozengurt, *et al.*, 1981b), did not increase DNA synthesis in the presence of DBHQ. Interestingly, bradykinin which causes a rapid but transient activation of PKC (Issandou and Rozengurt, 1990), also failed to act synergistically with DBHQ.

# 5.2.1. SUB-MAXIMAL CONCENTRATIONS OF BOMBESIN SYNERGISE WITH THAPSIGARGIN AND DBHQ

Bombesin, a potent mitogen for Swiss 3T3 cells, induces persistent stimulation of PKC even at concentrations sub-maximal for inducing DNA synthesis. As shown in table 5.I and figure 5.6 (right panel), 2.5 nM bombesin (a concentration which was submaximal) acted synergistically with DBHQ in promoting DNA synthesis in Swiss 3T3 cells. This synergistic effect was also demonstrated when DNA synthesis was determined by autoradiography of [<sup>3</sup>H]thymidine labelled nuclei (figure 5.6, left panel). Bombesin (2.5 nM) was almost as effective as PDB in promoting DNA synthesis in the presence of DBHQ. This synergy was not evident when saturating doses of bombesin (10 nM) were used (data not shown). These results strongly suggest that DBHQmediated Ca<sup>2+</sup> mobilisation acts synergistically with agents that induce persistent activation of PKC. Similar results were obtained when thapsigargin was added instead of DBHQ.

### 5.3. EXAMINATION OF THE SYNERGISM BETWEEN PKC AND THAPSIGARGIN OR DBHQ

# 5.3.1. THE SYNERGISTIC STIMULATION OF DNA SYNTHESIS REQUIRES FUNCTIONAL PKC

The preceding results prompted a verification that the synergistic effects between DBHQ and either PDB or bombesin on stimulation of DNA synthesis are dependent on PKC. It is known that treatment with saturating doses of PDB (800 nM) causes down regulation of all phorbol ester-sensitive isoforms of PKC present in Swiss 3T3 cells (Rodriguez-Pena and Rozengurt, 1984), i.e. PKCs  $\alpha$ ,  $\delta$  and  $\varepsilon$  (Olivier and Parker, 1992). As shown in figure 5.7, down regulation of phorbol ester sensitive PKCs strikingly decreased the [<sup>3</sup>H]thymidine incorporation induced by DBHQ and PDB or DBHQ and bombesin.

In order to substantiate the requirement of PKC activity for synergistic stimulation of DNA synthesis by DBHQ and PDB or bombesin, we used the inhibitor GF 109203X which selectively inhibits PKC isoforms  $\alpha$ ,  $\beta$ I,  $\beta$ II,  $\gamma$ ,  $\delta$  and  $\varepsilon$  in Swiss 3T3 cells (Toullec, *et al.*, 1991; Yeo and Exton, 1995).. Pretreatment of Swiss 3T3 cells with 3.5  $\mu$ M GF 109203X for 1 h before addition of PDB or bombesin, conditions which have been previously shown to effectively inhibit PKC activity in intact Swiss 3T3 cells (Sinnett-Smith, *et al.*, 1993), also strikingly decreased the [<sup>3</sup>H]thymidine incorporation induced by DBHQ and PDB or DBHQ and bombesin (figure 5.8). The results shown in figures 5.7 and 5.8 demonstrate that PKC is necessary for the synergistic stimulation of DNA synthesis induced by DBHQ and either PDB or bombesin.



# Figure 5.6. Effect of DBHQ on DNA synthesis induced by bombesin assessed both by autoradiography and liquid scintillation counting of [<sup>3</sup>H]thymidine incorporation as comparison

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 - 8 h without (open bars) or with 7.5  $\mu$ M DBHQ (closed bars). Cells were then washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with 2.5  $\mu$ Ci [<sup>3</sup>H]thymidine per ml (left) or 0.25  $\mu$ Ci/ml (right), and with 2.5 nM bombesin, in the absence (open bars) or presence of 7.5  $\mu$ M DBHQ (closed bars). Values in left panel are the percent of nuclei which were labelled, counted from eight microscopic fields (total of 65 - 80 cells per field) per dish. 10% FBS labelled 93% of nuclei. Values in right panel are expressed as mean percentage of the values obtained with 10% FBS, with error as SEM.



# Figure 5.7. Effect of down regulating PKC on [<sup>3</sup>H]thymidine incorporation induced by DBHQ and either PDB or bombesin

Confluent, quiescent cultures of Swiss 3T3 cells were treated in the absence (open bars) or presence (striped bars) 800 nM PDB for 48 h to down regulate PKC. Then, the cells were pretreated without (-) or with 7.5  $\mu$ M DBHQ (+). After 6 h, the cells were washed twice with DMEM and incubated in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine in the absence (-) or presence of 80 nM PDB or 1 nM bombesin (Bom), and without (-) or with (+) 7.5  $\mu$ M DBHQ. Results shown are the mean of two determinations with the error bar showing the range. Similar results were obtained in two independent experiments.



# Figure 5.8. Effect of inhibiting PKC on [<sup>3</sup>H]thymidine incorporation induced by DBHQ and either PDB or bombesin

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated without (-) or with (+) 7.5  $\mu$ M DBHQ. After 6 h, the cells were washed twice with DMEM, incubated in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine and DBHQ was replaced. Then, cells were treated without (open bars) or with 3.5  $\mu$ M GF 109203X (striped bars) for 1 h before the addition of nothing (-), 80 nM PDB or 1 nM bombesin (Bom). Results shown are the mean of two determinations with the error bar showing the range. Similar results were obtained in two independent experiments.

# 5.3.2. THAPSIGARGIN AND DBHQ DO NOT POTENTIATE PDB-MEDIATED SIGNAL TRANSDUCTION

Some PKC isoforms are Ca<sup>2+</sup>-dependent see section 1.3.3.1 and of these Swiss 3T3 cells contain PKC $\alpha$  (Olivier and Parker, 1992; Neri, *et al.*, 1994). Therefore treatment with thapsigargin or DBHQ could enhance PDB-mediated signals, or alternatively, converge with them at a downstream locus further into G<sub>1</sub>. To distinguish between these possibilities we determined whether prior exposure of Swiss 3T3 cells to 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ for 6 h had any effect on PDB-mediated events.

#### 80K/MARCKS

Cells that were metabolically labelled with  ${}^{32}P_i$  overnight were stimulated and 80K/MARCKS was immunoprecipitated. Figure 5.9 (upper panel) shows that preincubation with thapsigargin or DBHQ did not affect the ability of PDB to stimulate the phosphorylation of 80K/MARCKS, a prominent and specific substrate of PKC in intact cells (Rozengurt, *et al.*, 1983b).

#### **MAPK** activation

Phorbol esters activate MAPK, see section 1.3.8 resulting in phosphorylation on tyrosine and threonine residues and a lowered electrophoretic mobility on SDS-PAGE (Leevers and Marshall, 1992). To assess whether thapsigargin or DBHQ increased PDB-stimulated MAPK activity, whole cell lysates were resolved by SDS-PAGE and MAPK was visualised by Western blotting with anti-p42<sup>mapk</sup> polyclonal antibody. Figure 5.9 (lower panel) demonstrates that thapsigargin or DBHQ did not affect the activation of MAPK by exposure to PDB. The activation of MAPK by bombesin shows a biphasic profile, there is a large transient activation that peaks at 5 - 10 min followed by a lower sustained plateau that lasts for several hours (Withers, *et al.*, 1995). It was verified that pretreating cells with thapsigargin or DBHQ did not alter the timecourse of PDB-induced MAPK activation by incubating cells for 5 or 60 min (instead of 20 min) with PDB. There was no difference between pretreated cells and control cells.

#### Tyrosine phosphorylation

Recently, it has been shown that PDB stimulates the tyrosine phosphorylation of multiple substrates including bands of  $M_r$  110,000 - 130,000 and  $M_r$  70,000 - 80,000 (Sinnett-Smith, *et al.*, 1993). The novel cytosolic tyrosine kinase FAK has been identified as a prominent tyrosine phosphorylated protein of the  $M_r$  110,000 - 130,000 band whereas paxillin corresponds to the  $M_r$  70,000 - 80,000 heterogeneous band (Sinnett-Smith, *et al.*, 1993; Zachary, *et al.*, 1993).



# Figure 5.9. Effect of thapsigargin and DBHQ on PDB-mediated signal transduction (I)

#### Upper panel. 80K/MARCKS phosphorylation

Confluent, quiescent cultures of Swiss 3T3 cells were loaded with <sup>32</sup>P<sub>1</sub> overnight then pretreated for 6 h without or with 0.5 nM thapsigargin or 7.5 μM DBHQ as indicated. Cells were incubated 10 min, 37°C, with or without 80 nM PDB as indicated. Cells were lysed and immunoprecipitated with 80K/MARCKS antiserum and resolved by SDS-PAGE. <sup>32</sup>P-labelled bands were visualised by autoradiography. Similar results were obtained in two independent experiments.

#### Lower panel. MAP kinase activation

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 h with or without 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ as indicated. Cells were then washed twice with DMEM and incubated in DMEM at 37°C, without (-) or with 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ in the presence or absence of 80 nM PDB as indicated for 5, 20 or 60 min. Cells were lysed with 2x sample buffer and proteins separated by SDS-PAGE. Proteins were transferred to Immobilon membranes and Western blotted with MAP kinase antiserum. Bands were visualised by autoradiography. Results shown are after 20 min of stimulation. Similar results were obtained when PDB was added for 5 or 60 min instead of 20 min (results not shown). Similar results were obtained in 3 independent experiments.





# Figure 5.10. Effect of thapsigargin and DBHQ on PDB-mediated signal transduction (II)

### Tyrosine phosphorylation of proteins Mr 110,000 - 130,000 and Mr 70,000 - 80,000

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 h with or without 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ as indicated. Then, cells were washed twice with DMEM and incubated in DMEM for 10 min, 37°C, with or without (-) 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ and in the presence or absence of 80 nM PDB as indicated. Cells were lysed and immunoprecipitated with agarose-linked anti-Tyr(P) mAb Py72 and resolved by SDS-PAGE. Phosphotyrosyl proteins were transferred to Immobilon membranes and Western blotted with 4G10 and Py20 anti-Tyr(P) mAbs. Bands were visualised by autoradiography. Similar results were obtained in two independent experiments.

As shown in figure 5.10 (lower panel), pretreatment with thapsigargin or DBHQ did not change the basal level of tyrosine phosphorylation or affect the PDB induced increase in tyrosine phosphorylation of the  $M_r$  110,000 - 130,000 or  $M_r$  70,000 - 80,000 bands. In accordance with all these results, treatment of Swiss 3T3 cells with either DBHQ or thapsigargin did not alter the extent to which [<sup>3</sup>H]PDB bound to intact cells

(data not shown). These results indicate that thapsigargin and DBHQ do not directly potentiate PKC-mediated signals.

### 5.4. EFFECT OF THAPSIGARGIN AND DBHQ ON INTRACELLULAR CALCIUM STORES

Thapsigargin inhibits the activity of the ER Ca<sup>2+</sup>ATPase leading to a depletion of the intracellular Ca<sup>2+</sup> stores (Moore, *et al.*, 1987; Takemura, *et al.*, 1989; Thastrup, *et al.*, 1990; Bian, *et al.*, 1991; Sagara and Inesi, 1991). However, most previous studies on the effects of thapsigargin on cell proliferation and Ca<sup>2+</sup> sequestering in the ER have used high concentrations of this compound (0.1 - 10  $\mu$ M). Here, we found synergistic stimulation of DNA synthesis at sub-nanomolar concentrations. Consequently, it was important to examine the effects of low concentrations of thapsigargin and DBHQ on intracellular Ca<sup>2+</sup>.

### 5.4.1. THAPSIGARGIN AND DBHQ INHIBIT BOMBESIN-INDUCED Ca<sup>2+</sup> MOBILISATION

Specifically, if treatment with mitogenic concentrations of thapsigargin or DBHQ inhibits the accumulation of Ca<sup>2+</sup> into the Ins(1,4,5)P<sub>3</sub>-sensitive pool, it should prevent the increase in  $[Ca^{2+}]_{cyt}$  induced by subsequent addition of Ca<sup>2+</sup>-mobilising agents. Figure 5.11 shows that treatment with 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ markedly reduced the ability of 2.5 nM bombesin to induce a rapid and transient increase in the  $[Ca^{2+}]_{cyt}$  from Swiss 3T3 cells as measured by the fluorescent Ca<sup>2+</sup> indicator fura-2. Preincubation with 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ for 6 - 8 h inhibited bombesin-induced increase in  $[Ca^{2+}]_{cyt}$  by 79 ± 2% (n = 7) and 90 ± 1% (n = 9), respectively (Figure 5.12).

### 5.4.2. KINETICS OF THAPSIGARGIN AND DBHQ INHIBITION OF BOMBESIN-INDUCED Ca<sup>2+</sup> MOBILISATION

The reduction in bombesin-induced  $Ca^{2+}$  mobilisation by thapsigargin or DBHQ as a function of length of pretreatment time was examined. Figure 5.13 demonstrates that maximum inhibition of the increase in  $[Ca^{2+}]_{cyt}$  was reached within 1 h of exposure to thapsigargin or DBHQ which was maintained over a period of several hours.



# Figure 5.11. Reduction of bombesin induced Ca<sup>2+</sup> mobilisation by thapsigargin and DBHQ

Confluent, quiescent Swiss 3T3 cells in 100 mm Nunc dishes were incubated at 37°C for 8 h without (-) or with 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ. The cells were then washed twice with DMEM and incubated with DMEM containing 1  $\mu$ M Fura-2/AME in the absence (-) or presence of 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ for 10 min at 37°C. The cells were washed twice with PBS and gently scraped into 2 ml electrolyte solution containing 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ as necessary and transferred to a quartz cuvette which was kept at 37°C and stirred continuously throughout the experiment. Fluorescence was monitored. After 1 min control time bombesin was added to 2.5 nM (shown by arrow). Values on the left hand side of each tracing (A) are [Ca<sup>2+</sup>]<sub>cyt</sub> in nM.



# Figure 5.12. The mean increase in $[Ca^{2+}]_{cyt}$ induced by bombesin after pretreatment with thapsigargin or DBHQ

Confluent, quiescent Swiss 3T3 cells in 100 mm Nunc dishes were incubated at 37°C for 8 h without (-) or with 0.5 nM thapsigargin (TG) or 7.5  $\mu$ M DBHQ. The cells were then washed twice with DMEM and incubated with DMEM containing 1  $\mu$ M Fura-2/AME and fluorescence was monitored as in the previous figure. After 1 min control time bombesin was added to 2.5 nM (shown by arrow). Values on the left hand side of each tracing (A) are  $[Ca^{2+}]_{cyt}$  in nM. This figure shows the mean increase in  $[Ca^{2+}]_{cyt}$  induced by 2.5 nM bombesin in control cells (-, open bars) or in cells pretreated (closed bars) with thapsigargin (TG) or DBHQ. Errors are

presented as SEM. The number of samples are n = 17 (-), n = 7 (TG), n = 9 (DBHQ).



# Figure 5.13. Time course of depletion of bombesin induced calcium response by thapsigargin and DBHQ

Confluent, quiescent Swiss 3T3 cells were incubated at 37°C for various times with 0.5 nM thapsigargin (squares) or 7.5  $\mu$ M DBHQ (circles). The cells were then washed twice with DMEM and loaded with Fura-2/AME, as above, in the presence of 0.5 nM thapsigargin (squares) or 7.5  $\mu$ M DBHQ (circles). The cells were then washed twice in PBS and scraped into 2 mI electrolyte solution containing 0.5 nM thapsigargin (squares) or 7.5  $\mu$ M DBHQ (circles). After 1 min 2.5 nM bombesin was added. Values are expressed as a mean percentage of the difference in [Ca<sup>2+</sup>]<sub>cyt</sub> induced by 2.5 nM bombesin in untreated cells, with errors as SE from 2 - 3 experiments.

### 5.4.3. THAPSIGARGIN AND DBHQ DO NOT PREVENT BOMBESIN-INDUCED INOSITOL PHOSPHATE ACCUMULATION

It was verified that treatment with these agents did not inhibit bombesininduced Ca<sup>2+</sup> mobilisation by preventing the production of inositol phosphates. This was assessed by collecting inositol phosphates on an anion exchange column from cells that had been labelled overnight with [<sup>3</sup>H]inositol. Table 5.II shows that pretreatment with thapsigargin or DBHQ did not prevent the production of inositol phosphates, so thapsigargin and DBHQ inhibit Ca<sup>2+</sup> release by a different mechanism.

# Table 5.II. Effect of thapsigargin on total inositol phosphates generated by various mitogens

MITOGEN	- (cpm)	+ TG (cpm)
-	$209 \pm 21$	$263 \pm 26$
20 nM PDB	$194 \pm 6$	$199 \pm 10$
2.5 nM BOM	$482 \pm 21$	$560 \pm 15$
10 nM BOM	822 ± 21	800 ± 48

Confluent, quiescent cultures of Swiss 3T3 cells were equilibrated overnight with 15  $\mu$ Ci/ml *myo*-[2-<sup>3</sup>H]inositol then pretreated for 8 h without (-) or with 0.5 nM thapsigargin (+ TG). Cells were washed twice, incubated in DMEM/Waymouth's for 10 min, 37°C, with either 20 nM PDB, 2.5 nM or 10 nM bombesin (BOM), in the absence (-) or presence of 0.5 nM thapsigargin (+ TG). Total inositol phosphates were extracted and collected on Dowex AG1-X8 columns, HCOO<sup>-</sup> form, and eluted with 1 M NH4COOH, 0.1 M HCOOH as described in materials and methods. Values are expressed as cpm. The experiment was performed in triplicate, with the error shown as SE.

# 5.4.4. THAPSIGARGIN AND DBHQ CAUSE PROLONGED DEPLETION OF TOTAL CELLULAR Ca<sup>2+</sup>

To demonstrate that the decrease in bombesin-induced  $Ca^{2+}$  mobilisation was due to depletion of  $Ca^{2+}$  stores, cells were loaded with  ${}^{45}Ca^{2+}$  and the total  $Ca^{2+}$ remaining in the cells was determined after various treatments. This method measures  $Ca^{2+}$  in intracellular stores which is accumulated to millimolar concentrations rather than cytoplasmic  $Ca^{2+}$  which is maintained at nanomolar concentrations as described in section 1.3.2.1.



### Figure 5.14. Effect of Thapsigargin and DBHQ on total intracellular calcium measured by <sup>45</sup>Ca<sup>2+</sup>

Confluent, quiescent Swiss 3T3 cells loaded with  ${}^{45}Ca^{2+}$  were then pretreated for 8 h without (-) or with 0.5 nM thapsigargin (TG 8h), 7.5  $\mu$ M DBHQ (DBHQ 8h) 10 nM bombesin (Bom 8h), or for 30 min with 30 nM thapsigargin (TG 30). Cells were then washed twice with DMEM, the same factors added back as for the pretreatment and then incubated for 10 min at 37°C in the presence (striped bars) or absence (open bars) of 10 nM bombesin. In the case of bombesin pretreatment, due to down regulation of the bombesin response, further stimulation was obtained with 10 min of 20 nM vasopressin (reverse striped bar). Residual  ${}^{45}Ca^{2+}$  in the cells was measured as described in materials and methods. Values are mean of three determinations from one experiment with errors shown as SE.

The results in figure 5.14 demonstrate that 8 h pretreatment by either 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ, conditions which are optimum for DNA synthesis, reduced total cellular <sup>45</sup>Ca<sup>2+</sup> content to a level comparable to that reached after 10 min exposure to 10 nM bombesin (a maximum concentration). When thapsigargin or DBHQ treated cells were further challenged with 10 nM bombesin for 10 min a small amount of additional <sup>45</sup>Ca<sup>2+</sup> was released which is in agreement with the measurements of [Ca<sup>2+</sup>]<sub>cyt</sub> shown in figure 5.11. Complete depletion of the Ins(1,4,5)P<sub>3</sub> responsive pool was obtained by exposure of cells to 30 nM thapsigargin for 30 min since subsequent exposure to bombesin caused no further reduction in cellular <sup>45</sup>Ca<sup>2+</sup>. For comparison, an 8 h treatment with 10 nM bombesin induced a depletion in <sup>45</sup>Ca<sup>2+</sup> content similar to that achieved by 8 h pretreatment with 0.5 nM thapsigargin. The results in figures 5.11 - 5.14 demonstrate that treatment of Swiss 3T3 cells with thapsigargin or DBHQ, at concentrations that induce synergistic stimulation of DNA synthesis with PDB, cause a partial and persistent depletion in Ca<sup>2+</sup> from the Ins(1,4,5)P<sub>3</sub> responsive Ca<sup>2+</sup> pool.

#### 5.5. EFFECT OF EXTRACELLULAR CALCIUM ON DNA SYNTHESIS

The Ca<sup>2+</sup> response to agents that mobilise Ca<sup>2+</sup> through formation of  $Ins(1,4,5)P_3$  is typically biphasic. The first event is a large transient increase in  $[Ca^{2+}]_{cvt}$ originating from intracellular stores and the second is a sustained influx of Ca<sup>2+</sup> through the plasma membrane which causes a prolonged elevation of cytoplasmic  $Ca^{2+}$  (Putney and Bird, 1993). Because this sustained  $Ca^{2+}$  influx through the plasma membrane is dependent on extracellular Ca<sup>2+</sup>, the effect of extracellular Ca<sup>2+</sup> concentration on DNA synthesis induced by a variety of growth promoting factors was examined. The variety of factors will also distinguish between regulatory signals and obligatory events. The results in figure 5.15 show that DBHQ and PDB induced  $[^{3}H]$ thymidine incorporation is strikingly sensitive to extracellular Ca<sup>2+</sup>. The  $[Ca^{2+}]_{out}$ that is required to achieve 50% of the maximum response (ED<sub>50</sub>) induced by PDB and DBHQ was 410  $\mu$ M. A similar result was obtained for thapsigargin. In sharp contrast, the ED<sub>50</sub> for PDB and insulin, a combination that induces DNA synthesis but does not stimulate  $Ca^{2+}$  mobilisation, was 6  $\mu$ M. In order to substantiate the involvement of Ca<sup>2+</sup> influx in the synergistic stimulation of DNA synthesis, the inhibitor econozole was used. This compound inhibits Ca<sup>2+</sup> influx in several cell types (Alvarez, et al., 1991; Breittmayer, et al., 1993; Randriamampita and Tsien, 1993; Vostal and Fratantoni, 1993) through a mechanism independent of cytochrome P-450 inhibition (Vostal and Fratantoni, 1993).

The results in figure 5.16 show that DNA synthesis stimulated by PDB and DBHQ was reduced by 55% in the presence of 1  $\mu$ M econozole (a concentration which has been shown to inhibit capacitative Ca<sup>2+</sup> influx, (Alvarez, *et al.*, 1991). In contrast, DNA synthesis induced by PDB and insulin was only reduced by 13%. The results in figures 5.15 and 5.16 demonstrate that the synergistic stimulation of DNA synthesis by DBHQ with PDB requires capacitative Ca<sup>2+</sup> influx.

Because bombesin is a well known mitogen for Swiss 3T3 cells that mobilises Ca<sup>2+</sup> (Rozengurt and Sinnett-Smith, 1983; Mendoza, *et al.*, 1986) the contribution of Ca<sup>2+</sup> influx to bombesin-induced DNA synthesis was examined. This was compared to the combination of bombesin with insulin because insulin elicits intracellular signals other than Ca<sup>2+</sup>. Figure 5.17 demonstrates that bombesin-induced DNA synthesis is more sensitive to  $[Ca^{2+}]_{out}$  (ED<sub>50</sub> = 20 µM) than in the presence of insulin (ED<sub>50</sub> = 0.9 µM).



# Figure 5.15. Effect of extracellular calcium on [<sup>3</sup>H]thymidine incorporation induced by different mitogens

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 - 8 h with 7.5  $\mu$ M DBHQ (closed circles). Cells were then washed twice with DMEM and incubated in DMEM/Waymouth's medium containing [<sup>3</sup>H]thymidine with various amounts of Ca<sup>2+</sup> buffered by 2 mM EGTA in the presence of 80 nM PDB (open circles), 80 nM PDB and 7.5  $\mu$ M DBHQ (closed circles), or 80 nM PDB and 1  $\mu$ g/ml insulin (closed triangles). Similar results were obtained in 4 independent experiments.



# Figure 5.16. Effect of econozole on [<sup>3</sup>H]thymidine incorporation induced by different mitogens

Confluent, quiescent cultures of Swiss 3T3 cells were pretreated for 6 h with or without 7.5  $\mu$ M DBHQ. Cells were then washed twice with DMEM and incubated in DMEM/Waymouth's medium containing [<sup>3</sup>H]thymidine. Cultures were stimulated with either 80 nM PDB (-), 80 nM PDB and 7.5  $\mu$ M DBHQ (DBHQ), or with 80 nM PDB and 1  $\mu$ g/ml insulin (INS) in the absence (open bars) or presence (striped bars) of 1  $\mu$ M econozole. Similar results were obtained in two independent experiments.



# Figure 5.17. Effect of extracellular calcium on [<sup>3</sup>H]thymidine incorporation induced by bombesin

Confluent, quiescent cultures of Swiss 3T3 cells were washed twice with DMEM and incubated in DMEM/Waymouth's medium containing [<sup>3</sup>H]thymidine with various amounts of Ca<sup>2+</sup> buffered by 2 mM EGTA in the presence of 10 nM bombesin and 1  $\mu$ g/ml insulin (open circles) or 10 nM bombesin (closed circles). Similar results were obtained in 3 independent experiments.

### 5.6. SUMMARY AND DISCUSSION

The findings presented here demonstrate that thapsigargin and DBHQ synergistically stimulate DNA synthesis with PDB or bombesin in cultures of Swiss 3T3 cells. At the low concentrations that induce DNA synthesis, thapsigargin and DBHQ cause a slow but persistent depletion of the intracellular Ca<sup>2+</sup> stores The stores remain depleted for at least 8 h and importantly bombesin, a physiological agonist, induces a prolonged depletion for more than 8 h also. This extends earlier measurements which showed that vasopressin caused Ca<sup>2+</sup> depletion for upto 1 h (Lopez-Rivas and Rozengurt, 1984). This is the first time that either thapsigargin or DBHQ have been shown to stimulate the reinitiation of DNA synthesis in any target cell.

#### 5.6.1. CAPACITATIVE Ca<sup>2+</sup> ENTRY

It is recognised that depletion of ER Ca<sup>2+</sup> stores e.g. by thapsigargin, DBHQ or neuropeptides, activates Ca<sup>2+</sup> influx from the extracellular medium of many cell types (Putney and Bird, 1993). This has been termed capacitative Ca<sup>2+</sup> entry (Putney, 1986) and is thought to be activated by a diffusible messenger released from the ER when the Ca<sup>2+</sup> is depleted (Montero, et al., 1992; Parekh, et al., 1993; Randriamampita and Tsien, 1993). How does depletion of intracellular Ca<sup>2+</sup> stores act in signalling DNA synthesis? Two alternative mechanisms could be envisaged. The diffusible messenger released from the depleted  $Ca^{2+}$  stores could, as well as activating  $Ca^{2+}$  entry, interact with other signalling components to promote synergistic stimulation of cell cycle entry from  $G_0$ . Alternatively, capacitative  $Ca^{2+}$  influx may lead to a persistent increase in  $[Ca^{2+}]_{cvt}$ . An attempt was made to distinguish between these possibilities. If the stimulation of DNA synthesis elicited by thapsigargin and DBHQ depends on Ca<sup>2+</sup> influx it was reasoned that DNA synthesis should be critically dependent on  $[Ca^{2+}]_{out}$ . Indeed this was the case for the mitogenic combination of DBHQ and PDB. The  $ED_{50}$  of  $[Ca^{2+}]_{out}$  required to support DNA synthesis stimulated by DBHQ and PDB was 100-fold higher than that required for DNA synthesis induced by insulin and PDB. The high  $[Ca^{2+}]_{out}$  in DBHQ-treated cells is consistent with the apparent K<sub>m</sub> (1 mM) of the channel that mediates capacitative Ca<sup>2+</sup> influx (Mauger, et al., 1984). In support of this, econozole, which inhibits Ca<sup>2+</sup> influx (Alvarez, et al., 1991; Breittmayer, et al., 1993; Randriamampita and Tsien, 1993; Vostal and Fratantoni, 1993), inhibited DNA synthesis induced by PDB and DBHQ to a much greater extent than DNA synthesis induced by PDB and insulin. Taking these data together it can be concluded that the induction of DNA synthesis by DBHQ and thapsigargin in the presence of PDB is dependent on the activity of the capacitative Ca<sup>2+</sup> influx pathway.

The results presented in this chapter contrast with the earlier studies on removal of Ca<sup>2+</sup> from the growth medium (see section 1.3.2.5) which demonstrated that Ca<sup>2+</sup> was essential, but could not show whether it had a stimulatory role in reinitiating DNA synthesis. The results presented in figures 5.15 - 5.17 clearly demonstrate that  $[Ca^{2+}]_{out}$  is differentially required by different combinations of mitogens, defining Ca<sup>2+</sup> fluxes as a regulatory signal rather than an obligatory event. PDB and either thapsigargin or DBHQ, induce only a subset of potential cellular signalling pathways, i.e. PKC activation and Ca<sup>2+</sup> mobilisation, and are the most dependent on  $[Ca^{2+}]_{out}$ . Bombesin, which in addition activates PKC-independent pathways, is less sensitive to  $[Ca^{2+}]_{out}$ . Insulin induces different signalling pathways such as activation of PI3K (White and Kahn, 1994) and thus insulin together with PDB or bombesin are the least sensitive combinations to  $[Ca^{2+}]_{out}$ . The inhibition of DNA synthesis seen with all combinations at 0.2  $\mu$ M  $[Ca^{2+}]_{out}$  represents the essential role of Ca<sup>2+</sup> in cell viability.

Interestingly, signals generated by thapsigargin- or DBHQ-depleted  $Ca^{2+}$  stores synergise with submaximal concentrations bombesin even though the bombesininduced  $Ca^{2+}$  spike is attenuated. This would argue against the large transient increase in  $[Ca^{2+}]_{cyt}$  being the mitogenic signal and suggest it is the consequence of the depleted stores that stimulates DNA synthesis. It has been shown in T-lymphocytes that it is the sustained influx rather than the intracellular  $Ca^{2+}$  spike which is important for IL-2 secretion (Gelfand, *et al.*, 1988). Similarly it is the influx which is necessary for catecholamine release in adrenal chromaffin cells (Cheek, *et al.*, 1993).

#### 5.6.2. SYNERGY WITH PKC

In theory, treatment with thapsigargin or DBHQ leading to persistent depletion of Ca<sup>2+</sup> stores could enhance the ability of PDB to induce cellular responses through cPKC which contain CalB domains. Several lines of evidence indicate that this is not the case. Specifically, treatment with either thapsigargin or DBHQ did not enhance the ability of PDB to induce a variety of responses such as direct PKC-mediated phosphorylation of 80K/MARCKS, stimulation of the kinase cascade leading to MAP kinase activation and tyrosine phosphorylation of multiple bands identified previously as FAK and paxillin (Sinnett-Smith, *et al.*, 1993; Zachary, *et al.*, 1993). Therefore, it can be concluded that Ca<sup>2+</sup> and PKC signals act in parallel and converge downstream to induce synergistic stimulation of DNA synthesis.

The effect of  $[Ca^{2+}]_{out}$  on bombesin-stimulated signalling has been investigated by Takuwa, *et al.* (1991a). It was shown that S6 kinase activity, MAPK activity, protein phosphorylation and c-*fos* induction were not inhibited by low  $[Ca^{2+}]_{out}$ . This also indicates that Ca<sup>2+</sup> influx does not contribute to early events that are activated by PKC. The only event measured that was shown to be inhibited by low Ca<sup>2+</sup> was AA release, see below.

#### 5.6.3. CYTOTOXIC EFFECTS

Our results contrast with previous studies that demonstrate that thapsigargin and DBHQ profoundly inhibit cell proliferation or induce cytotoxic effects (Ghosh, *et al.*, 1991; Marks, *et al.*, 1991; Short, *et al.*, 1993). Although different cell types were used, an obvious difference between the present and previous studies is the concentration of thapsigargin employed. In fact, inhibition of DNA synthesis and cytotoxic effects in Swiss 3T3 cells at higher concentrations of thapsigargin and DBHQ were readily obtained. The existence of stimulatory and inhibitory effects could explain the bell-shaped dose-response of these agents on DNA synthesis found in this study. The inhibitory effects are likely to be caused by extensive depletion of Ca<sup>2+</sup> from the ER which, as well as being a store for hormonally releasable Ca<sup>2+</sup>, plays an important role in the production of secretory, lysosomal and membrane bound proteins.

Draining of Ca<sup>2+</sup> from stores by oscillations in  $[Ca^{2+}]_{cyt}$  is suggested to induce controlled waves of secretion of proteins from distinct regions of the heterogeneous ER. However, mass perturbation of these stores by excessive amounts of ionophore could lead to indiscriminate vesiculation (Sambrook, 1990).

Sequestered  $Ca^{2+}$  rather than  $[Ca^{2+}]_{cyt}$  is proposed to control protein synthesis (Brostrom and Brostrom, 1990). It is therefore not suprising that the initial effect of high concentrations of thapsigargin is to inhibit protein synthesis (Price, et al., 1992; Wong, et al., 1993). One of the earliest proteins to be synthesised 2 - 5 h after treatment with thapsigargin or A23187 is Grp78 (Benton, et al., 1989; Price, et al., 1992; Wong, et al., 1993). Grp78 belongs to a family of glucose regulated proteins (Grp) and their induction is stimulated by a variety of stressful conditions such as glucose starvation, blockage of cellular glycosylation, reducing agents, Ca<sup>2+</sup> ionophores and chelating agents and low extracellular pH (Lee, 1987). Other members of the Grp family are Grp94 (endoplasmin), ERp72 (has sequence identity with PDI) and Grp58 (Lee, 1992). Endoplasmin contains low affinity, high capacity Ca<sup>2+</sup> binding sites and may bind to malfolded proteins retained in the ER. In the stress response the levels of Grp78 are elevated the most. Grp78 binds to a wide variety of proteins in the ER and assists in an ATP-dependent manner in the translocation, folding and assembly of oligomeric proteins. In normal cells Grp78 is found primarily in an aggregated oligomeric state where it is modified by phosphorylation or ADPribosylation. In stress-induced cells where there is an accumulation of nontransportable proteins, Grp78 is predominately monomeric and unmodified. Upon release from associated proteins Grp78 can be reversibly modified (Lee, 1992).

It is believed that reduction of  $Ca^{2+}$  in the ER inhibits protein glycosylation and it is the accumulation of underglycosylated proteins that triggers the induction of Grp78 (Lee, 1987). Consistent with this is the demonstration that the glycosylation of  $\alpha$ 1-antitrypsin was blocked at an early high mannose step when cells were treated with  $Ca^{2+}$  ionophore resulting in decreased secretion, whereas the secretion of albumin which is not glycosylated was affected to a lesser extent (Kuznetsov, *et al.*, 1992). The high mannose state was probably due to inhibition of  $\alpha$ 1,2-mannosidase, which requires  $Ca^{2+}$  for activity (Schutzbach and Forsee, 1990).

Control of the filling state of the ER by physiological agonists is carefully balanced but the excessive depletion caused by ionophores or high concentrations of thapsigargin could lead to serious disturbances of protein synthesis, processing and trafficking. This could explain the inhibition of cell proliferation induced by thapsigargin and DBHQ at higher concentrations than those used in the present study. In this respect Ghosh, *et al.* (1991), Marks, *et al.* (1991) and Short, *et al.* (1993) have identified the permissive requirement for Ca<sup>2+</sup> rather than a regulatory role.

It is hypothesised that thapsigargin and DBHQ, at optimum mitogenic concentrations, reduce the Ca<sup>2+</sup> content from internal stores in Swiss 3T3 cells to a critical level that leads to entry into DNA synthesis but does not impair essential ER functions.

### 5.6.4. POSSIBLE PATHWAYS ACTIVATED

#### Preparation of the ER

The two roles of the ER described above (hormonally-responsive  $Ca^{2+}$  store and protein processing organelle) are not mutually exclusive but very much interlinked. It was mentioned above that draining of  $Ca^{2+}$  from ER stores by oscillations in  $[Ca^{2+}]_{cyt}$  is suggested to induce waves of secretion of proteins (Sambrook, 1990). Consistent with this, decreased  $Ca^{2+}$  in the ER results in the dissociation of T cell receptor chain from BiP and subsequent expression on the cell surface (Suzuki, *et al.*, 1991). Similarly Milner, *et al.* (1992) describe that  $Ca^{2+}$  regulation of KDEL on the KDEL receptor could regulate retention and secretion of ER luminal proteins.

In Swiss 3T3 cells the mitogens PMT, bombesin and PDGF all reduce the ADPribosylation of Grp78 (Staddon, *et al.*, 1992) hence converting it from an inactive to an active conformation. This was attributed to the preparation of the ER for the increase in protein traffic associated with re-entry into the cell-cycle that would require the activity of BiP. This further illustrates the interplay between protein processing in the ER and stimulation by mitogenic agonists that mobilise Ca<sup>2+</sup>. Interestingly Grp78 is also induced by treatment with insulin (Lee, 1987) and this may also be in preparation for increased protein synthesis, processing and trafficking.

#### Activation of signal transduction molecules

Ca<sup>2+</sup> influx is crucial for DNA synthesis induced by thapsigargin or DBHQ with PDB, but the possible pathways activated have not been elucidated. Whatever these pathways are it is evident they cannot initiate DNA synthesis alone but require a synergistic signal from PKC. Clues could come from proteins which bind Ca<sup>2+</sup>. As already described in section 1.3.2.4, Ca<sup>2+</sup> is a very localised second messenger, giving rise to Ca<sup>2+</sup> micro-gradients and possibly nano-gradients around Ca<sup>2+</sup> channels in membranes. In this respect Petersen, *et al.* (1994) point out that low affinity Ca<sup>2+</sup> binding proteins such as those with CalB domains including cPLA<sub>2</sub>, PKC<sub>γ</sub> and PLC<sub>γ</sub>, are generally associated with membranes whereas proteins which bind Ca<sup>2+</sup> with high affinity are in the cytosol. So could treatment with thapsigargin and DBHQ be causing the activation of membrane bound proteins? The results in Chapter 5 showed that thapsigargin and DBHQ did not potentiate PKC-stimulated signals so it is unlikely that species of PLC or PKC contribute to the observed synergy. In contrast there is evidence that suggests the involvement of PLA<sub>2</sub> and AA.

cPLA<sub>2</sub> is regulated by Ca<sup>2+</sup> and MAPK (Clark, *et al.*, 1991; Piomelli, 1993). Also A23187 can induce AA release in Swiss 3T3 cells (Millar and Rozengurt, 1990a). In addition there are many reports that link extracellular Ca<sup>2+</sup> and/or Ca<sup>2+</sup> influx to subsequent AA release (Toyoshima, *et al.*, 1982; Balsinde, *et al.*, 1990; Takuwa, *et al.*, 1991a; Takuwa, *et al.*, 1991b; Liu, *et al.*, 1993; Sharma, 1993; Arkinstall, *et al.*, 1994; Tornquist, *et al.*, 1994). Of particular interest is the observation that bombesin-induced AA release requires high  $[Ca^{2+}]_{out}$  (Takuwa, *et al.*, 1991b). It is thought that bombesin-induced release of AA in Swiss 3T3 cells results in production of PGE<sub>1</sub> which acts on a G<sub>s</sub>-coupled receptor to increase cAMP (Millar and Rozengurt, 1988; Millar and Rozengurt, 1990a). It is of interest to note that cAMP does not increase 80K phosphorylation (Rodriguez-Pena and Rozengurt, 1986), MAPK activation (Withers, *et al.*, 1995) or tyrosine phosphorylation of bands 70,000 - 80,000 and 110,000 - 130,000 (Zachary, *et al.*, 1991a).

An unusual aspect of DNA synthesis induced by thapsigargin and DBHQ is their inability to synergise with insulin. Characterisation of a plethora of growth promoting agents, both physiological and pharmacological, has identified insulin as a factor that is not able to stimulate DNA synthesis by itself but can synergise with almost all other compounds tested including cAMP elevating agents (Rozengurt, 1986). If thapsigargin and DBHQ were acting to increase cAMP levels via AA release and prostaglandin synthesis then it would have been expected that they would synergise with insulin. However,  $Ca^{2+}$  influx alone is not enough to fully stimulate PLA<sub>2</sub>. Sustained activation of PLA<sub>2</sub> also requires PKC activity because down regulation of PKC by prolonged treatment with PDB abolishes the sustained bombesin-induced PLA<sub>2</sub> activity (Takuwa, *et al.*, 1991a). Hence thapsigargin and DBHQ could not have synergised with insulin in the absence of PDB. It would then be difficult to show synergism between insulin and PDB in combination with thapsigargin or DBHQ because PDB and insulin is a potent mitogenic combination itself. This illustrates that PKC signals and Ca<sup>2+</sup> signals could converge on one single enzyme.

CaM and multifunctional CaMK are also potential targets for a Ca<sup>2+</sup> stimulated signal transduction pathway and it has been demonstrated that bombesin induces CaMK activity (Duan, *et al.*, 1994). CaMK has been proposed to act as frequency decoder for intracellular Ca<sup>2+</sup> oscillations (Hanson, *et al.*, 1994). Interestingly one of the few studies that has also used 0.5 nM thapsigargin demonstrated that partial inhibition of Ca<sup>2+</sup>ATPase pumps increased intracellular spiking frequency (Petersen, *et al.*, 1993) suggesting a mechanism whereby treatment with low concentrations of thapsigargin could increase intracellular Ca<sup>2+</sup> oscillations to a threshold frequency thus activating CaMK.

### **Chapter 6 – SUMMARY AND PERSPECTIVES**

It has been recognised for over a decade that the activation of multiple signal transduction pathways is required for a full mitogenic effect (Rozengurt, 1986). For instance activation of PKC by phorbol ester or the addition of insulin to Swiss 3T3 cells will not stimulate DNA synthesis but phorbol ester together with insulin is a potent mitogenic combination (Dicker and Rozengurt, 1978; Dicker and Rozengurt, 1980). Similarly in Swiss 3T3 cells increasing cAMP does not induce DNA synthesis but elevation of cAMP together with activation of PKC does induce DNA synthesis (Rozengurt, 1986). This identifies the co-operation between two signalling pathways.

This principle can be extended to physiological growth factors. PDGF and bombesin are sole mitogens for Swiss 3T3 cells whereas EGF, vasopressin and bradykinin are not. Thus it was hypothesised that PDGF and bombesin can stimulate multiple, synergistic signalling pathways (Rozengurt, 1986). The mechanisms by which receptor tyrosine kinases can elicit multiple signals involves phosphorylated tyrosine residues which create multiple docking sites for signal transduction molecules. The mechanism by which G protein-coupled receptors can elicit multiple signal transduction pathways is more elusive. Because bombesin is a sole mitogen much attention has been given to elucidating its cellular mechanisms of activation.

Bombesin stimulates multiple signal transduction pathways including PLCmediated hydrolysis of PIP<sub>2</sub>, PKC activation, Ca<sup>2+</sup> mobilisation, Na<sup>+</sup>/H<sup>+</sup> exchange activity and cytoplasmic alkalisation, arachidonic acid release, enhancement of cAMP elevation, MAPK activation and tyrosine phosphorylation of multiple substrates including FAK and paxillin. All of the above events are totally or partially dependent on PKC, this is summarised in table 3.III. However tyrosine phosphorylation was unusual in that it had no PKC or  $Ca^{2+}$  dependency. This raised questions concerning the mechanisms by which bombesin could couple to multiple pathways. At least three possible mechanisms can be envisaged: 1) the different pathways could be elicited by distinct receptors; 2) distinct Ga subunits could couple to the same receptor; 3)  $\beta\gamma$ subunits of heterotrimeric G proteins could elicit distinct pathways from  $G\alpha$ . It had been suggested that a separate bombesin receptor mediated tyrosine phosphorylation and mitogenesis but no direct evidence was presented (Bold et. al., 1995). Therefore we aimed to characterise signalling pathways elicited by the bombesin/GRP receptor which was transfected into Rat-1 fibroblasts, a cell line that is unresponsive to bombesin.

#### The bombesin receptor couples to multiple pathways

The results presented in Chapter 3 of this thesis provide compelling evidence that ligand activation of the transfected bombesin/GRP receptor activates tyrosine phosphorylation of FAK, paxillin and other proteins as well as couple to PLC-mediated events and induces cell proliferation. Thus the full signalling and mitogenic potential of bombesin can be mediated by the bombesin/GRP receptor without the need to propose further receptor subtypes. This is the first time that a single subtype of G protein-coupled receptor, defined by transfection, has been shown to couple to PLCmediated events, tyrosine phosphorylation of FAK and paxillin and stimulate DNA synthesis. This contrasts with the other well known mitogens LPA and thrombin for which receptor identity has not been clearly established. The possible mechanism of coupling to multiple pathways was also discussed.

It had already been shown that tyrosine phosphorylation induced by bombesin and other neuropeptides was unlikely to be downstream of  $\alpha_q$  (Sinnett-Smith *et. al.*, 1993) and this is supported by the demonstration that microinjection of activated  $\alpha_q$ did not enhance the formation of stress fibres (Buhl, *et al.*, 1995). The possibility of signalling via  $\beta_{\gamma}$  subunits is also unlikely because there is very little evidence for the specificity of  $\beta_{\gamma}$  combinations on effector activation, see section 1.1.6.4 and microinjection of several  $\beta_{\gamma}$  combinations had no effect on stress fibre formation (Buhl, et. al., 1995). The most likely mechanism of bombesin-induced tyrosine phosphorylation of multiple substrates including FAK and paxillin is the coupling of the bombesin/GRP receptor to multiple G $\alpha$ , potential candidates being  $\alpha_{12}$  or  $\alpha_{13}$ .

Further experimental work is required to test which  $G\alpha$  the bombesin/GRP receptor couples to. This could be, for example, by using  $[\alpha^{-32}P]$ GTP azidoanilide to photolabel the  $\alpha$  subunits which are activated. Alternatively specific  $G\alpha$  could be inhibited by microinjection of antibodies against the  $\alpha$  subunits and then bombesin-induced stress fibre assembly and tyrosine phosphorylation could be examined by fluorescence microscopy. Antisense vectors against different  $G\alpha$  followed by phosphotyrosine analysis could determine which  $G\alpha$  are necessary for bombesin-induced tyrosine phosphorylation. Also mutational analysis of the bombesin/GRP receptor could identify sites of multiple G protein interaction.

#### Bombesin stimulates the GTP binding of Ras

It is being increasingly recognised that activation of the MAPK cascade is important on receptor identity and especially on cellular context. This is in contrast to tyrosine phosphorylation which displays similar characteristics when stimulated by a variety of receptors coupled to  $G_q$  in various cells. Receptors coupled to  $G_q$  generally activate the MAPK pathway in a PKC-dependent fashion. Consistent with this, MAPK activation induced by bombesin in Swiss 3T3 cells is dependent on PKC. However
other reports imply that, like  $G_i$ -coupled receptors,  $G_q$ -coupled receptors activate MAPK in a Ras dependent fashion. Therefore characteristics of MAPK stimulation were investigated in Rat-1 cells that were transfected by the bombesin/GRP receptor.

Chapter 4 showed that in contrast to earlier studies in Swiss 3T3 cells, bombesin-stimulation of the MAPK cascade was independent of PKC activation in Rat-1 cells. In addition it was shown for the first time that bombesin can stimulate Raf-1 activity and this also occurred in a PKC-independent and pertussis toxin-insensitive fashion. These results led to the examination of Ras activation in Rat-1 cells transfected with the bombesin/GRP receptor. Addition of bombesin to BOR 15 cells caused an increase in the amount of GTP bound to Ras, implicating Ras in the activation of MAPK induced by bombesin in Rat-1 cells. This demonstrates that the variability of bombesin/GRP receptor coupling depends on the cell type that expresses the receptor. The possible upstream pathway leading to Ras activation was discussed. This includes activation of protein tyrosine kinases or lipid kinases and these may have other cellular consequences in addition to the activation of Ras. Similarly Ras has multiple downstream effectors further expanding the repertoire of potential signalling pathways.

## Calcium

The first and second parts of the thesis investigated the multiple pathways that bombesin induced and the mechanisms by which bombesin could induce multiple pathways. One of the earliest events induced by bombesin is the mobilisation of  $Ca^{2+}$ , and before this work was started it had not been established if  $Ca^{2+}$  fluxes initiated another signal transduction pathway that was capable of synergising with the signalling pathways already documented to reinitiate DNA synthesis. Addition of bombesin to Swiss 3T3 cells causes a transient increase in  $Ca^{2+}$  followed by a depletion of total cellular  $Ca^{2+}$  that lasts at least 1 h. To investigate the effect of  $Ca^{2+}$ mobilisation on DNA synthesis we used new tools which were thapsigargin and DBHQ. These agents inhibit the ER  $Ca^{2+}ATPase$  causing a depletion of  $Ca^{2+}$  from hormonal responsive stores.

Chapter 5 shows that depletion of intracellular  $Ca^{2+}$  by low concentrations of thapsigargin or DBHQ, in a manner similar to that induced by bombesin, has profound effects on [<sup>3</sup>H]thymidine incorporation, shown to be due to reinitiation of DNA synthesis. This contrasted with earlier studies which showed that thapsigargin and DBHQ inhibited cell growth and induced cytotoxic effects and this was attributed to gross perturbation of ER Ca<sup>2+</sup> stores which are necessary for protein synthesis, processing and trafficking. The presence of functional PKC was crucial but the early events stimulated by PKC were not potentiated after pretreatment with thapsigarin or DBHQ suggesting that PKC signals and Ca<sup>2+</sup>-induced signals converge further into G<sub>1</sub>. The remarkable sensitivity to  $[Ca^{2+}]_{out}$  on DNA synthesis induced by PDB with either thapsigargin or DBHQ suggested that  $Ca^{2+}$  influx was crucial. Bombesin was also partially dependent on extracellular  $Ca^{2+}$  but in the presence of insulin the requirements for  $[Ca^{2+}]_{out}$  were bypassed. Thus  $Ca^{2+}$  could be an important signalling pathway participating in bombesin regulated cell proliferation. The observation that different combinations of mitogenic factors are dependent on  $[Ca^{2+}]_{out}$  to varying degrees identifies  $Ca^{2+}$  influx induced by DBHQ as a regulatory signal rather than an obligatory event. The demonstration that  $Ca^{2+}$  can stimulate DNA synthesis is also important for receptor tyrosine kinases which bind to and activate PLC<sub>Y</sub>.

The pathways activated by thapsigargin and DBHQ are unknown. Identifying the signal transduction pathways would help to further understand the role of  $Ca^{2+}$  in initiating DNA synthesis. Starting points could be measurements of CaMK activity or arachidonic acid release. It would also be interesting to monitor  $[Ca^{2+}]_{cyt}$  by fluorescence microscopy to see if there is increased  $Ca^{2+}$  around the periphery of the cell close to the influx channels or if there is an increase in  $Ca^{2+}$  spiking frequency in response to treatment of cells with thapsigargin or DBHQ.

## Summary

The results presented in this thesis contribute to a new scheme of signalling pathways stimulated by the bombesin/GRP receptor. This is shown in figure 6.1. Thus bombesin may couple to FAK and paxillin phosphorylation via  $G_{12/13}$ , couple to p21<sup>ras</sup> and MAPK activation in Rat-1 cells via  $\beta\gamma$  subunits and couple to PLC $\beta$ -mediated activation of PKC and Ca<sup>2+</sup> mobilisation the latter of which has been demonstrated to contribute signals to the mitogenic response.

These results further support the concept of multiple synergistic signalling pathways. It has been demonstrated at the molecular level that a single species of receptor can elicit all pathways induced by bombesin, and also at a cellular level that the interaction between PKC and Ca<sup>2+</sup> is a potent synergistic combination.

## Implications

The demonstration that bombesin can couple to multiple pathways that can vary from cell type to cell type, raises important questions to how bombesin signals in *in vivo* rather than in cultured cells. For example, bombesin has long been implicated in lung development and recently it was shown more specifically to induce foetal lung branching (Li, *et al.*, 1994; King, *et al.*, 1995). Bombesin is also implicated in small cell lung cancer (Sethi, *et al.*, 1992). The pathways activated by bombesin in these situations have not been fully characterised, although it has been shown in small cell lung cancer cell lines that bombesin mobilises Ca<sup>2+</sup> and as demonstrated in Chapter 5, Ca<sup>2+</sup> fluxes could have an important contribution in reinitiating DNA synthesis. However, the effect on Ras in these cell lines has not been studied. Thus the work presented in this thesis may be relevant to bombesin/GRP signalling *in vivo*, both in normal and pathological conditions.



Figure 6.1. The bombesin/GRP receptor couples to multiple signal transduction pathways.

## REFERENCES

- Abdel-Baset H, Bozovic V, Szyf M & Albert PR. (1992) Conditional transformation meditated via a pertussis toxin-sensitive receptor signalling pathway. *Mol. Endocrinol.* 6: 730-740
- Abou-Samra A-B, Juppner H, Force T, Freeman MW, Kong X-F, Schipani E, Urena P, Richards J, Bonventre JV, Potts JT Jr, Kronenberg HM & Segre GV. (1992) Expression cloning of a common receptor for parathyroid hormone and parathyroid hormone-related peptide from rat osteoblast-like cells: A single receptor stimulates intracellular accumulation of both cAMP and inositol trisphosphates and increases intracellular free calcium. *Proc. Natl. Acad. Sci. USA* **89:** 2732-2736
- Adams PD & Parker PJ. (1992) Activation of mitogen-activated protein (MAP) kinase by a MAP kinase-kinase. *J. Biol. Chem.* **267:** 13135-13137
- Aderem A. (1992) Signal transduction and the actin cytoskeleton: the roles of MARCKS and profilin. *Trends Biochem. Sci.* 17: 438-443
- Ahn NG, Robbins DJ, Haycock JW, Seger R, Cobb MH & Krebs EG. (1992) Identification of an activator of the microtubule-associated protein 2 kinases ERK1 and ERK2 in PC12 cells stimulated with nerve growth factor or bradykinin. *J. Neurochem.* **59**: 147-156
- Ahn NG, Seger R, Bratlien RL, Diltz CD, Tonks NK & Krebs ED. (1991) Multiple components in an epidermal growth factor-stimulated protein kinase cascade. *In vitro* activation of a myelin basic protein/microtubule-associated protein 2 kinase. *J. Biol. chem.* **266**: 4220-4227
- Aitken A. (1995) 14-3-3 proteins on the MAP. Trends Biochem. Sci. 20: 95-97
- Aitken A, Collinge DB, van Heusden BPH, Isobe T, Roseboom PH, Rosenfield G & Soll J. (1992) 14-3-3 proteins: a highly conserved, widespread family of eukaryotic proteins. *Trends Biochem. Sci.* 17: 498-501
- Alblas J, van Corven EJ, Hordijk PL, Milligan G & Moolenaar WH. (1993) G<sub>i</sub>-mediated activation of the p21ras-mitogen-activated protein kinase pathway by alpha 2-adrenergic receptors expressed in fibroblasts. *J. Biol. Chem.* **268**: 22235-22238
- Alessi DR, Saito Y, Campbell DG, Cohen P, Sithanandam G, Rapp U, Ashworth A, Marshall CJ & Cowley S. (1994) Identification of the sites in MAP kinase kinase-1 phosphorylation by p74<sup>*raf-1*</sup>. *EMBO J.* **13**: 1610-1619
- Ali H, Brogger Christensen S, Foreman JC, Pearce FL, Piotrowski W & Thastrup O. (1985) The ability of thapsigargin and thapsigargicin to activate cells involved in the inflammatory response. *Br. J. Pharmac.* **85**: 705-712
- Allbritton NL, Meyer T & Stryer L. (1992) Range of messenger action of calcium ion and inositol 1,4,5-trisphosphate. *Science* **258**: 1812-1815

- Allen LF, Lefkowitz RJ, Caron MG & Cotecchia S. (1991) G-protein-coupled receptor genes as protooncogenes: constitutively activating mutation of the  $\alpha_{1B}$ -adrenergic receptor enhances mitogenesis and tumorigenicity. *Proc. Natl. Acad. Sci. USA* 88: 11354-11358
- Allgeier A, Offermanns S, Van Sande J, Spicher K, Schultz G & Dumont JE. (1994) The human thyrotrophin receptor activates G-proteins Gs and Gq/11. *J. Biol. Chem.* **269:** 13733-13735
- **Alvarez J, Montero M & Garcia-Sancho J. (1991)** Cytochrome *P*-450 may link intracellular Ca<sup>2+</sup> stores with plasma membrane Ca<sup>2+</sup> influx. *Biochem. J.* **274:** 193-197
- **Anastasi A, Erspamer V & Bucci M. (1971)** Isolation and structure of bombesin and alytesin, two analogous active peptides from the skin of the European amphibians Bombina and Alytes. *Experimentia* **27:** 166-167
- Anderson NG, Maller JL, Tonks NK & Sturgill TW. (1990) Requirement for integration of signals from two distinct phosphorylation pathways for activation of MAP kinase. *Nature* 343: 651-653
- **Angel P & Karin M. (1991)** The role of Jun, Fos, and the AP-1 complex in cellproliferation and transformation. *Biochem. Biophys. Acta* **1072:** 129-157
- Arkinstall S, Emergy I, Church D, Chollet A & Kawashima E. (1994) Calcium influx and protein kinase  $C\alpha$  activation mediate arachidonic acid mobilization by the human NK-2 receptor expressed in Chinese Hamster Ovary cells. *FEBS Lett.* **338**: 75-80
- Ashkenazi M, Winslow JW, Peralta EG, Peterson GL, Schimerlik MI, Capon DJ & Ramachandran J. (1987) An M2 muscarinic receptor subtype coupled to both adenylyl cyclase and phosphoinositide turnover. *Science* 238: 672-675
- Avis I, Jett M, Kasprzyk PG, Cuttitta F, Treston AM, Maneckjee R & Mulshine JL. (1993) Effect of gastrin-releasing peptide on the pancreatic tumor cell line (Capan). *Mol. Carcinogen.* 8: 214-220
- **Baldwin JM. (1993)** The probable arrangement of the helices in G protein-coupled receptors. *EMBO J.* **12:** 1693-1703
- Balk SD. (1971) Calcium as a regulator of the proliferation of normal, but not of transformed, chicken fibroblasts in a plasma-containing medium. *Proc. Natl. Acad. Sci. USA* 68: 271-275
- Balk SD, Whitfield JF, Youdale T & Braun AC. (1973) Roles of calcium, serum, plasma, and folic acid in the control of proliferation of normal and Rous Sarcoma virus-infected chicken fibroblasts. *Proc. Natl. Acad. Sci. USA* 70: 675-679
- Balsinde J, Fernandez B & Diez E. (1990) Regulation of arachidonic acid release in mouse peritoneal macrophages. The role of extracellular calcium and protein kinase C. J. Immunol. 144: 4298-4304

Barbacid M. (1987) ras genes. Ann. Rev. Biochem. 56: 779-827

- **Baserga R. (1976)** Multiplication and Division in Mammalian Cells. Dekker, New York, 239-250
- Basu TN, Gutmann DH, Fletcher JA, Glover TW, Collins FS & Downward J. (1992) Aberrant regulation of *ras* proteins in malignant tumour cells from type 1 neurofibromatosis patients. *Nature* **356**: 713-715
- Battey JF, Way JM, Corjay MH, Shapira H, Kusano K, Harkins R, Wu JM, Slattery T, Mann E & Feldman RI. (1991) Molecular cloning of the bombesin/gastrin-releasing peptide receptor from Swiss 3T3 cells. *Proc. Natl. Acad. Sci. USA.* 88: 395-399
- Bax B & Jhoti H. (1995) Putting the pieces together. Curr. Biol. 5: 1119-1121
- Ben-Ze'ev A. (1985) The cytoskeleton in cancer cells. *Biochem. Biophys. Act.* 780: 197-212
- **Benton HP, Jackson TR & Hanley MR. (1989)** Identification of a novel inflammatory stimulant of chondrocytes. Early events in cell activation by bradykinin receptors on pig articular chondrocytes. *Biochem. J.* **258:** 861-867
- **Bergman M, Joukov V, Virtanen I & Alitalo K. (1995)** Overexpressed Csk tyrosine kinase is localized in focal adhesions, causes reorganization of alpha v beta 5 integrin, and interferes with HeLa cell spreading. *Mol. Cell. Biol.* **15:** 711-722
- **Berridge MJ. (1984)** Inositol triphosphate and diacylglycerol as second messengers. *Biochem. J.* **220:** 345-360
- **Berridge MJ. (1993)** Inositol trisphosphate and calcium signalling. *Nature* **361:** 315-325
- Berridge MJ. (1995) Calcium signalling and cell proliferation. BioEssays 17: 491-500
- **Berridge MJ & Irvine RF. (1989)** Inositol phosphates and cell signalling. *Nature* **341:** 197-204
- Berstein G, Blank JL, Jhon D-Y, Exton JH, Rhee SG & Ross EM. (1992) Phospholipase C- $\beta$ 1 is a GTPase-activating protein for G<sub>q/11</sub>, its physiologic regulator. *Cell* **70**: 411-418
- **Berven LA, Crouch MF, Katsis F, Kemp BE, Harland LM & Barritt GJ. (1995)** Evidence that the pertussis toxin-sensitive trimeric GTP-binding protein G<sub>i2</sub> is required for agonist- and store-activated Ca<sup>2+</sup> inflow in hepatocytes. *J. Biol. Chem.* **270:** 25893-25897
- **Bian J, Ghosh TK, Wang J-C & Gill DL. (1991)** Identification of intracellular calcium pools: selective modification by thapsigargin. *J. Biol. Chem.* **266:** 8801-8806
- **Bianchini L & Pouyssegur J. (1994)** Molecular structure and regulation of vertebrate Na<sup>+</sup>/H<sup>+</sup> exchangers. *J. Exp. Biol.* **196:** 337-345
- **Bird GSJ & Putney JW Jr. (1993)** Inhibition of thapsigargin-induced calcium entry by microinjected guanine nucleotides analogues. *J. Biol. Chem.* **268**: 21486-21488
- Birge RB, Fajardo JE, Reichman C, Shoelson SE, Songyang Z, Cantley LC & Hanafusa H. (1993) Identification and characterization of a high affinity interaction between v-

Crk and tyrosine phosphorylated paxillin in CT10 transformed fibroblasts. *Mol. Cell. Biol.* **13**: 4648-4656

- Bishop JM. (1991) Molecular themes in oncogenesis. Cell 64: 235-48
- **Bode H-P & Goke B. (1994)** Protein kinase C activates capacitative calcium entry in the insulin secreting cell line RINm5F. *FEBS Lett.* **339:** 307-311
- **Bogoyevitch MA, Marshall CJ & Sugden PH. (1995)** Hypertrophic agonists stimulate the activities of the protein kinases c-Raf and A-Raf in cultured ventricular myocytes. *J. Biol. Chem.* **270:** 26303-26310
- Bold RJ, Lowry PS, Ishizuka J, Battey JF, Townsend CM Jr. & Thompson JC. (1994) Bombesin stimulates the *in vitro* growth of a human gastric cancer cell line. *J. Cell. Physiol.* **161:** 519-525
- **Bonfini L, Karlovich CA, Dasgupta C & Banerjee U. (1992)** The Son of sevenless gene product: a putative activator of Ras. *Science* **255:** 603-606
- Bonnefoy-Berard N, Liu Y-C, von Willerbrand M, Sung A, Elly C, Mustelin T, Yoshida H, Ishizaka K & Altman A. (1995) Inhibition of phosphatidylinositol 3-kinase activity by association with 14-3-3 proteins in T cells. *Proc. Natl. Acad. Sci. USA* 92: 10142-10146
- **Borle AB & Snowdowne KW. (1987)** Methods for the measurement of intracellular ionized calcium in mammalian cells: Comparison of four classes of Ca<sup>2+</sup> indicators. (In Calcium and cell function) Academic Press, Inc., VII: 159-200
- Bottenstein J, Hayashi J, Hutchings S, Masui H, Mather J, McClure DB, Ohash S, Rizzino A, Sato G, Serrero G, Wolfe R & Wu R. (1979) The growth of cells in serum-free hormone supplemented media. *Meth. Enzymol.* 58: 94-109
- Boulton TG, Nye SH, Robbins DJ, Ip NY, Radziejewska E, Morgenbesser SD, DePinho RA, Panayotatos N, Cobb MH & Yancopoulos GD. (1991) ERKs: A family of proteinserine/threonine kinases that are activated and tyrosine phosphorylated in response to insulin and NGF. *Cell* 65: 663-675
- Boulton TG, Yancopoulos GD, Gregory JS, Slaughter C, Moomaw C, Hsu J & Cobb MH. (1990) An insulin-stimulated protein kinase similar to yeast kinases involved in cell cycle control. *Science* 249: 64-67
- **Boynton AL, Whitfield JF, Isaacs RJ & Tremblay R. (1977)** The control of human WI-38 cell proliferation by extracellular calcium and its elimination by SV-40 virusinduced proliferative transformation. *J. Cell. Physiol.* **92:** 241-248
- Braselmann S & McCormick F. (1995) BCR and RAF form a complex *in vivo* via 14-3-3 proteins. *EMBO J.* 14: 4839-4848
- Braun AP & Schulman H. (1995) The multifunctional calcium/calmodulin-dependent protein kinase: from form to function. *Annu. Rev. Physiol.* 57: 417-445

- **Breittmayer J-P, Berthe P, Cousin J-L, Bernard A & Aussel C. (1993)** CD3 monoclonal antibodies evoke the same cytochrome P450-regulated capacitative entry of calcium as thapsigargin in Jurkat T cells. *Cell. Immunol.* **152:** 143-151
- **Brooks SF, Hergert T, Broad S & Rozengurt E. (1992)** The expression of 80K/MARCKS, a major substrate of PKC, is down-regulated through both PKC-dependent and -independent pathways: effects of bombesin, PDGF and cAMP. *J. Biol. Chem.* **267**: 14212-14218
- **Brooks SF, Herget T, Erusalimsky JD & Rozengurt E. (1991)** Protein kinase C activation potently down-regulates the expression of its major substrate, 80K, in Swiss 3T3 cells. *EMBO J.* **10**: 2497-2505
- **Brooks SF, Erusalimsky JD, Totty NF & Rozengurt E. (1990)** Purification and internal amino acid sequence of the 80kDa protein kinase C substrate from Swiss 3T3 fibroblasts. Homology with substrates from brain. *FEBS Lett.* **268:** 291-295
- **Brostrom CO & Brostrom MA. (1990)** Calcium-dependent regulation of protein synthesis in intact mammalian cells. *Ann. Rev. Physiol.* **52:** 577-590
- Brown KD & Littlewood CJ. (1989) Endothelin stimulates DNA synthesis in Swiss 3T3 cells. *Biochem. J.* 263: 977-980
- **Buday L & Downward J. (1993a)** Epidermal growth factor regulates p21ras through the formation of a complex of receptor, Grb2 adaptor protein, and Sos nucleotide exchange factor. *Cell* **73:** 611-620
- **Buday L & Downward J. (1993b)** Epidermal growth factor regulates the exchange rate of guanine nucleotides on p21<sup>ras</sup> in fibroblasts. *Mol. Cell. Biol.* **13:** 1903-1910
- **Buhl AM, Johnson NL, Dhanasekaran N & Johnson GL. (1995)**  $G\alpha_{12}$  and  $G\alpha_{13}$  stimulate Rho-dependent stress fibre formation and focal adhesion assembly. *J. Biol. Chem.* **270:** 24631-24634
- Burbelo PD & Hall A. (1995) Hot numbers in signal transduction. Curr. Biol. 5: 95-96
- **Burridge K, Turner CE & Romer LH. (1992)** Tyrosine phosphorylation of paxillin and pp125<sup>FAK</sup> accompanies cell adhesion to extracellular matrix: A role in cytoskeletal assembly. *J. Cell Biol.* **119:** 893-903
- Cai H, Erhardt P, Troppmair J, Diaz-Meco MT, Sithanandam G, Rapp UR, Moscat J & Cooper GM. (1993) Hydrolysis of phosphatidylcholine couples ras to activation of Raf protein kinase during mitogenic signal transduction. *Mol. Cell. Biol.* 13: 7645-7651
- **Cali JJ, Balcueva EA, Rybalkin I & Robishaw JD. (1992)** Selective tissue distribution of G protein  $\gamma$  subunits, including a new form of the  $\gamma$  subunits identified by cDNA cloning. *J. Biol. Chem.* **267:** 24023-24027
- **Cano E & Mahadevan LC. (1995)** Parallel signal processing among mammalian MAPKs. *Trends Biochem. Sci.* **20:** 117-122
- Carafoli E. (1987) Intracellular calcium homeostasis. Ann. Rev. Biochem. 56: 395-433

- **Carafoli E. (1992)** The Ca<sup>2+</sup> pump of the plasma membrane. *J. Biol. Chem.* **267:** 2115-2118
- **Carney DN, Cuttitta F, Moody TW & Minna JD. (1987)** Selective stimulation of small cell lung cancer clonal growth by bombesin and gastrin-releasing peptide. *Cancer Res.* **47:** 821-825
- Carpenter G. (1985) Epidermal growth factor: Biology and receptor metabolism. *J. Cell Sci. Suppl.* 3: 1-9
- Carpenter G & Cohen S. (1979) Epidermal Growth Factor. Ann. Rev. Biochem. 48: 193-216
- **Carroll MP & May WS. (1994)** Protein kinase C-mediated serine phosphorylation directly activates Raf-1 in murine hematopoietic cells. *J. Biol. Chem.* **269:** 1249-1256
- Cassel D & Selinger Z. (1977) Mechanism of adenylate cyclase activation by cholera toxin: Inhibition of GTP hydrolysis at the regulatory site. *Proc. Natl. Acad. Sci. USA* 74: 3307-3311
- **Chabre O, Conklin BR, Brandon S, Bourne HR & Limbird LE. (1994)** Coupling of the  $\alpha_{2A}$ -adrenergic receptor to multiple G-proteins. A simple approach for estimating receptor-G-protein coupling efficiency in a transient expression system. *J. Biol. Chem.* **269:** 5730-5734
- Chabre O, Conklin BR, Lin HY, Lodish HF, Wilson E, Ives HE, Catanzariti L, Hemmings BA & Bourne HR. (1992) A recombinant calcitonin receptor independently stimulates 3',5'-cyclic adenosine monophosphate and Ca<sup>2+</sup>/inositol phosphate signalling pathways.
- **Chao T-SO, Foster DA, Rapp UR & Rosner MR. (1994)** Differential Raf requirement for activation of mitogen-activated protein kinase by growth factors, phorbol esters and calcium. *J. Biol. Chem.* **269:** 7337-7341
- Chardin P, Boquet P, Madaule P, Popoff MR, Rubin EJ & Gill DM. (1989) The mammalian G protein *rhoC* is ADP-ribosylated by *Clostridium botulinum* exoenzyme C3 and affects actin microfilaments in Vero cells. *EMBO J.* **8**: 1087-1092
- Chardin P, Camonis JH, Gale NW, van Aelst L, Schlessinger J, Wigler MH & Bar-SagiD. (1993) Human Sos1: A guanine nucleotide exchange factor for ras that binds toGRB2. *Science* 260: 1338-1343
- **Cheek TR, Morgan A, O'Sullivan AJ, Moreton RB, Berridge MJ & Burgoyne RD. (1993)** Spatial localization of agonist-induced Ca<sup>2+</sup> entry in bovine adrenal chromaffin cells. Different patterns induced by histamine and angiotensin II, and relationship to catecholamine release. *J. Cell Sci.* **105:** 913-921
- Chen HC, Appeddu PA, Parsons JT, Hildebrand JD, Schaller MD & Guan J-L. (1995) Interaction of focal adhesion kinase with cytoskeletal protein talin. *J. Biol. Chem.* 270: 16995-16999

- Chen LB & Buchanan JM. (1975) Mitogenic activity of blood components. I. Thrombin and prothrombin. *Proc. Natl. Acad. Sci.* 72: 131-135
- Chen RH, Sarnecki C & Blenis J. (1992) Nuclear localization and regulation of *erk-* and *rsk-*encoded protein kinases. *Mol. Cell. Biol.* **12:** 915-927
- Chen Y, Grall D, Salcini AE, Pelicci PG, Pouyssegur J & van Obberghen-Schilling E. (1996) Shc adaptor proteins are key transducers of mitogenic signaling mediated by the G protein-coupled thrombin receptor. *EMBO J.* **15**: 1037-1044
- Chen Y-H, Pouyssegur J, Courtneidge SA & van Obberghen-Schilling E. (1994) Activation of Src family kinase activity by the G protein-coupled thrombin receptor in growth-responsive fibroblasts. *J. Biol. Chem.* **269**: 27372-27377
- **Ciapa B, Pesando D, Wilding M & Whitaker M. (1994)** Cell-cycle calcium transients driven by cyclic changes in inositol trisphosphate levels. *Nature* **368:** 875-878
- **Cirillo DM, Gaudino G, Naldini L & Comoglio PM. (1986)** Receptor for bombesin with associated tyrosine kinase activity. *Mol. Cell. Biol.* **6**: 4641-4649
- Clapham DE. (1996) The G-protein nanomachine. Nature 379: 297-299
- **Clapham DE & Neer EJ. (1993)** New roles for G-protein  $\beta\gamma$ -dimers in transmembrane signalling. *Nature* **365:** 403-406
- Clapham DE & Sneyd J. (1995) Intracellular calcium waves. Adv. Second Messenger Phosphoprotein Res. 30: 1-24
- Clark JD, Lin L-L, Kriz RW, Ramesha CS, Sultzman LA, Lin AY, Milona N & Knopf JL. (1991) A novel arachidonic acid-selective cytosolic PLA<sub>2</sub> contains a Ca<sup>2+</sup> -dependent translocation domain with homology to PKC and GAP. *Cell* 65: 1043-1051
- **Cobb BS, Schaller MD, Leu T-H & Parsons JT. (1994)** Stable association of pp60<sup>src</sup> and pp59<sup>fyn</sup> with the focal adhesion-associated protein tyrosine kinase, pp125<sup>FAK</sup>. *Mol. Cell. Biol.* **14:** 147-155
- Cobb MH & Goldsmith EJ. (1995) How MAP kinases are regulated. *J. Biol. Chem.* 270: 14843-14846
- **Cohen GB, Ren R & Baltimore D. (1995)** Modular binding domains in signal transduction proteins. *Cell* **80**: 237-248
- **Collins MKL & Rozengurt E. (1982a)** Binding of phorbol esters to high affinity sites on murine fibroblastic cells elicits a mitogenic response. *J. Cell Physiol.* **112:** 42-50
- **Collins MKL & Rozengurt E. (1982b)** Stimulation of DNA synthesis in murine fibroblasts by the tumour promoter teleocidin: relationship to phorbol esters and vasopressin. *Biochem. Biophys. Res. Commun.* **104:** 1159-1166
- **Collins MKL & Rozengurt E. (1984)** Homologous and heterologous mitogenic desensitization of Swiss 3T3 cells to phorbol esters and vasopressin: role of receptor and postreceptor steps. *J. Cell. Physiol.* **118**: 133-142

- **Connolly AJ, Ishihara H, Kahn ML, Farese RV Jr & Coughlin SR. (1996)** Role of the thrombin receptor in development and evidence for a second receptor. *Nature* **381**: 516-519
- **Conroy LA, Merritt JE & Hallam TJ. (1994)** Regulation of T-cell-receptor-stimulated bivalent-cation entry in Jurkat E6 cells: role of protein kinase C. *Biochem. J.* **303:** 671-677
- Corjay MH, Dobrzamski DJ, Way JM, Vallet J, Shapira H, Worland P, Sausville EA & Battey JF. (1991) Two distinct bombesin receptor subtypes are expressed and functional in Human Lung carcinoma cells. *J. Biol. Chem.* 266: 18771-18779
- **Cowley S, Paterson H, Kemp P & Marshall CJ. (1994)** Activation of MAP kinase kinase is necessary and sufficient for PC12 differentiation and for transformation of NIH 3T3 cells. *Cell* **77:** 841-852
- **Coy D, Mungan Z, Rossowski W, Cheng B, Lin JT, Mrozinski J & Jensen R. (1992)** Development of a potent bombesin antagonist with a prolonged *in vivo* inhibitory activity on bombesin-stimulated amylase and protein release in rat. *Peptides* **13:** 775-781
- Crabtree GR. (1989) Contingent genetic regulatory events in T lymphocyte activation. *Science* 243: 355-361
- Crespo P, Xu N, Daniotti JL, Troppmair J, Rapp UR & Gutkind JS. (1994a) Signaling through transforming G protein-coupled receptors in NIH 3T3 cells involves c-Raf activation. Evidence for a protein kinase C-independent pathway. *J. Biol. Chem.* 269: 21103-21109
- **Crespo P, Xu N, Simonds WF & Gutkind JS. (1994b)** Ras-dependent activation of MAP kinase pathway mediated by G-protein βγ subunits. *Nature* **369:** 418-420
- **Crews CM, Alessandrini A & Erikson RL. (1992)** The primary structure of MEK, a protein kinase that phosphorylates the *ERK* gene product. *Science* **258**: 478-480
- **Cross M & Dexter TM. (1991)** Growth factors in development, transformation and tumorigenesis. *Cell* **64:** 271-280
- **Curran T, Bravo R & Muller R. (1985)** Transient induction of c-*fos* and c-*myc* is an immediate consequence of growth factor stimulation. *Cancer Surv.* **4**: 655-681
- Cuttitta F, Carney DN, Mulshine J, Moody TW, Fedorko J, Fischler A & Minna JD. (1985) Bombesin like peptides can function as autocrine growth factors in human small cell lung cancer. *Nature* **316**: 823-826
- **Daub H, Weiss FU, Wallasch C & Ullrich A. (1996)** Role of transactivation of the EGF receptor in signalling by G-protein-coupled receptors. *Nature* **379**: 557-560
- Daum G, Eisenmann-Tappe I, Fries H-W, Troppmair J & Rapp UR. (1994) The ins and outs of Raf kinases. *Trends Biochem. Sci.* 19: 474-480

- de Vries-Smits AMM, Burgering BMT, Leevers SJ, Marshall CJ & Bos JL. (1992) Involvement of p21<sup>ras</sup> in activation of extracellular signal-regulated kinase 2. *Nature* 357: 602-604
- **Dekker LV & Parker PJ. (1994)** Protein kinase C a question of specificity. *Trends Biochem. Sci.* **19:** 73-77
- **Demaurex N, Lew DP & Krause K-H. (1992)** Cyclopiazonic acid depletes intracellular Ca<sup>2+</sup> stores and activates an influx pathway for divalent cations in HL-60 cells. *J. Biol. Chem.* **267:** 2318-2324
- Dent P, Haser W, Haystead TAJ, Vincent LA, Roberts TM & Sturgill TW. (1992) Activation of mitogen-activated protein kinase kinase by v-Raf in NIH 3T3 cells and *in vitro. Science* **257**: 1404-1407
- **Dent P & Sturgill TW. (1994)** Activation of (His)<sub>6</sub>-Raf-1 *in vitro* by partially purified plasma membranes from v-Ras-transformed and serum-stimulated fibroblasts. *Proc. Natl. Acad. Sci. USA* **91:** 9544-9548
- Dent P, Wu J, Romero G, Vincent LA, Castle D & Sturgill TW. (1993) Activation of the mitogen-activated protein kinase pathway in Triton X-100 disrupted NIH-3T3 cells by p21 ras and *in vitro* by plasma membranes from NIH 3T3 cells. *Mol. Biol. Cell* **4**: 483-493
- Dhanasekaran N, Prasad MVVSV, Wadsworth SJ, Dermott JM & von Rossum G. (1994) Protein kinase C-dependent and -independent activation of Na<sup>+</sup>/H<sup>+</sup> exchanger by Gα12 class of G proteins. *J. Biol. Chem.* **269**: 11802-11806
- **Dicker P & Rozengurt E. (1978)** Stimulation of DNA synthesis by tumour promoter and pure mitogenic factor. *Nature* **276:** 723-726
- Dicker P & Rozengurt E. (1979) Retinoids enhance mitogenesis by tumour promoter and polypeptide growth factors. *Biochem. Biophys. Res. Commun.* 91: 1203-1210
- **Dicker P & Rozengurt E. (1980)** Phorbol esters and vasopressin stimulate DNA synthesis by a common mechanism. *Nature* **287:** 607-612
- **Dicker P & Rozengurt E. (1981)** Stimulation of DNA synthesis by transient exposure of cell cultures to TPA or polypeptide mitogens: induction of competence or incomplete removal. *J. Cell. Physiol.* **109**: 99-109
- Divecha N & Irvine RF. (1995) Phospholipid Signaling. Cell 80: 269-278
- Domin J & Rozengurt E. (1993) Platelet-derived growth factor stimulates a biphasic mobilication of arachidonic acid in Swiss 3T3 cells. The role of phospholipase A<sub>2</sub>. J. Biol. Chem. 268: 8927-8934
- Downward J, Yarden Y, Mayes E, Scrace G, Totty N, Stockwell P, Ullrich A, Schlessinger J & Waterfield MD. (1984) Close similarity of epidermal growth factor receptor and v-*erb*-B oncogene protein sequences. *Nature* **307**: 521-527

- **Duan R-D, Guo Y-J & Williams JA. (1994)** Conversion to Ca<sup>2+</sup>-independent form of Ca<sup>2+</sup>/calmodulin protein kinase II in rat pancreatic acini. *Biochem. Biophys. Res. Comm.* **199:** 368-373
- **Duan RD, Zheng CF, Guan KL & Williams JA. (1995)** Activation of MAP kinase kinase (MEK) and ras by cholecystokinin in rat pancreatic acini. *Am. J. Physiol.* **268:** G1060-G1065
- **Duff JL, Berk BC & Corson MA. (1992)** Angiotensin II stimulates the pp44 and pp42 mitogen-activated protein kinases in cultured rat aortic smooth muscle cells. *Biochem. Biophys. Res. Commun.* **188**: 257-264
- Earp HS, Huckle WR, Dawson TL, Li X, Graves LM & Dy R. (1995) Angiotensin II activates at least two tyrosine kinases in rat liver epithelial cells. Separation of the major calcium-regulated tyrosine kinase p125<sup>FAK</sup>. *J. Biol. Chem.* **270**: 28440-28447
- Eason MG, Kurose H, Holt BD, Raymond JR & Liggett SB. (1992) Simultaneous coupling of  $\alpha_2$ -adrenergic receptors to two G-proteins with opposing effects: Subtype-selective coupling of  $\alpha_2$ C10,  $\alpha_2$ C4, and  $\alpha_2$ C2 adrenergic receptors to G<sub>i</sub> and G<sub>s</sub>. *J. Biol. Chem.* 267: 15795-15801

**Eason MG & Liggett SB. (1995)** Identification of a  $G_s$  coupling domain in the amino terminus of the third intracellular loop of the  $g_{0,s}$ -adrenergic recentor: Evidence for

- terminus of the third intracellular loop of the  $\alpha_{2A}$ -adrenergic receptor: Evidence for distinct structural determinants that confer G<sub>s</sub> versus G<sub>i</sub> coupling. J. Biol. Chem. 270: 24753-24760
- **Egan SE, Giddings BW, Brooks MW, Buday L, Sizeland AM & Weinberg RA. (1993)** Association of Sos Ras exchange protein with Grb2 is implicated in tyrosine kinase signal transduction and transformation. *Nature* **363**: 45-51
- **Erhardt P, Troppmair J, Rapp UR & Cooper GM. (1995)** Differential regulation of Raf-1 and B-Raf and Ras-dependent activation of Mitogen-Activated Protein Kinase by cyclic AMP in PC12 cells. *Mol. Cell. Biol.* **15:** 5524-5530
- Erickson AK, Payne DM, martino PA, Rossomando AJ, Shabanowitz J, Weber MJ, Hunt DF & Sturgill TW. (1990) Identification by mass spectrometry of threonine 97 in bovine myelin basic protein as a specific phosphorylation site for mitogenactivated protein kinase. *J. Biol. Chem.* **265:** 19728-19735
- Erisman MD, Linnoila RI, Hernandez O, DiAugustine RP & Lazarus LH. (1982) Human lung small-cell carcinoma contains bombesin. *Proc. Natl. Acad. Sci.* **79**: 2379-2383
- **Erpel T & Courtneidge SA. (1995)** Src family protein tyrosine kinases and cellular signal transduction pathways. *Curr. Opin. Cell Biol.* **7:** 176-182
- Erspamer GF, Severini C, Erspamer V, Melchiorri P, Fave GD & Nakajima T. (1988) Parallel bioassay of 27 bombesin-like peptides on 9 smooth muscle preperations. Structure-activity relationships and bombesin receptor subtypes. *Reg. Peptides* 21: 1-11

- **Erusalimsky JD, Brooks SF, Herget T, Morris C & Rozengurt E. (1991)** Molecular cloning and characterization of the acidic 80-kDa protein kinase C substrate from rat brain. Identification as a glycoprotein. *J. Biol. Chem.* **266:** 7073-7080
- **Erusalimsky JD, Friedberg I & Rozengurt E. (1988)** Bombesin, diacylglycerols, and phorbol esters rapidly stimulate the phosphorylation of an Mr = 80,000 protein kinase C substrate in permeabilized 3T3 cells. Effect of guanine nucleotides. *J. Biol. Chem.* **263:** 19188-19194
- **Exton JH. (1994)** Phosphoinositide phospholipases and G proteins in hormone action. *Annu. Rev. Physiol.* **56:** 349-369
- Fabian JR, Daar IO & Morrison DK. (1993) Critical tyrosine residues regulate the enymatic and biological activity of Raf-1 kinase. *Mol. Cell. Biol.* 13: 7170-7179
- **Fabregat I & Rozengurt E. (1990)** Vasoactive intestinal contractor, a novel peptide, shares a common receptor with endothelin-1 and stimulates Ca<sup>2+</sup> mobilization and DNA synthesis in Swiss 3T3 cells. *Biochem. Biophys. Res. Commun.* **167:** 161-167
- **Fantl WJ, Johnson DE & Williams LT. (1993)** Signalling by receptor tyrosine kinases. *Annu. Rev. Biochem.* **62:** 453-481
- Farnsworth CL, Freshney NW, Rosen LB, Ghosh A, Greenberg ME & Feig LA. (1995) Calcium activation of ras mediated by neuronal exchange factor Ras-GRF. *Nature* 376: 524-527
- Fasolato C, Hoth M & Penner R. (1993) A GTP-dependent step in the activation mechanism of capacitive calcium influx. *J. Biol. Chem.* **268:** 20737-20740
- Fathi Z, Corjay MH, Shapira H, Wada E, Benya R, Jensen R, Viallet J, Sausville EA & Battey JF. (1993) BRS-3: A novel bombesin receptor subtype selectively expressed in testis and lung carcinoma cells. *J. Biol. Chem.* 268: 5979-5984
- **Faure M & Bourne HR. (1995)** Differential effects of cAMP on the MAP kinase cascade: Evidence for a cAMP-insenstive step that can bypass Raf-1. *Mol. Biol. Cell* **6:** 1025-1035
- **Faure M, Voyno-Yasenetskaya TA & Bourne HR. (1994)** cAMP and βγ subunits of heterotrimeric G proteins stimulate the mitogen-activated protein kinase pathway in COS-7 cells. *J. Biol. Chem.* **269:** 7851-7854
- Feig LA & Cooper GM. (1988) Inhibition of NIH 3T3 cell proliferation by a mutant *ras* protein with preferential affinity for GDP. *Mol. Cell. Biol.* 8: 3235-3243
- **Fernando KC & Barritt GJ. (1994)** Evidence from studies with hepatocyte suspensions that store-operated Ca<sup>2+</sup> inflow requires a pertussis toxin-sensitive trimeric G-protein. *Biochem. J.* **303:** 351-356
- Fischer JB & Schonbrunn A. (1988) The bombesin receptor is coupled to a guanine nucleotide-binding protein which is insensitive to pertussis and cholera toxins. *J. Biol. Chem.* 263: 2808-2816

- Force T, Kyriakis JM, Avruch J & Bonventre JV. (1991) Endothelin, vasopressin, and angiotensin II enhance tyrosine phosphorylation by protein kinase C-dependent and -independent pathways in glomerular mesangial cells. *J. Biol. Chem.* **266**: 6650-6656
- **Freyd G, Kim SK & Horvitz HR. (1990)** Novel cysteine-rich motif and homeodomain in the product of the *Caenorhabditis elegans* cell lineage gene *lin-11*. *Nature* **344**: 876-879
- Friedkin M & Rozengurt E. (1981) The role of cytoplasmic microtubules in the regulation of activity of peptide growth factors. *Adv. Enzyme Regul.* 19: 39-59
- Froesch ER, Zapf J, Rinderkhecht E, Morell B, Schoenle E & Humbel RE. (1979) Insulin-like growth factor (IGF-NSILA): Structure, function and physiology. (In Hormones and Cell Culture) eds. GH Satoh. & R Ross. Cold Spring Harbour Conferences on Cell Proliferation Cold spring Harbour Lab., 61-77
- Gale NW, Kaplan S, Lowenstein EJ, Schlessinger J & Bar-Sagi D. (1993) Grb2 mediates the EGF-dependent activation of guanine nucleotide exchange on ras. *Nature* 363: 88-92
- Gamberucci A, Innocenti B, Fulceri R, Banhegyi G, Giunti R, Pozzan T & Benedetti A. (1994) Modulation of Ca<sup>2+</sup> influx dependent on store depletion by intracellular adenine-guanine nucleotide levels. *J. Biol. Chem.* **269**: 23597-23602
- Gelehrter D & Rozengurt E. (1980) Stimulation of monovalent ion fluxes and DNA synthesis in 3T3 cells by melittin and vasopressin is not mediated by phospholipid deacylation. *Biochem. Biophys. Res. Commun.* 97: 716-724
- **Gelfand EW, Cheung RK, Mills GB & Grinstein S. (1988)** Uptake of extracellular Ca<sup>2+</sup> and not recruitment from internal stores is essential for T-lymphocyte proliferation. *Eur. J. Immunol.* **18:** 917-922
- Ghatei MA, Jung RT, Stevenson JC, Hillyard CJ, Adrian TE, Lee YC, Christofides ND, Sarson DL, Mashiter K, MacIntyre I & Bloom SR. (1982) Bombesin: Action on gut hormones and calcium in man. *J. Clin. Endocrinol. Metab.* 54:
- **Ghosh S, Strum JC, Sciorra VA, Daniel L & Bell RM. (1996)** Raf-1 kinase possesses distinct binding domains for phosphatidylserine and phosphatidic acid: Phosphatidic acid regulates the translocation of Raf-1 in 12-*O*-tetradecanoylphorbol-13-acetate-stimulated Madin-Darby canine kidney cells. *J. Biol. Chem.* **271**: 8472-8480
- Ghosh TK, Bian J, Short AD, Rybak SL & Gill DL. (1991) Persistent intracellular calcium pool depletion by thapsigargin and its influence on cell growth. *J. Biol. Chem.* 266: 24690-24697
- Gibb JB, Marshall MS, Scolnick EM, Dixon RAF & Vogel US. (1990) Modulation of guanine nucleotides bound to ras in NIH 3T3 cells by oncogenes, growth factors, and the GTPase activating protein (GAP). *J. Biol. Chem.* **265**: 2037-20442
- Gibson TJ, Hyvonen M, Musacchio A & Saraste M. (1994) PH domain: the first anniversary. *Trends Biochem. Sci.* 19: 349-353

- Gil J, Higgins T & Rozengurt E. (1991) Mastoparan, a novel mitogen for Swiss 3T3 cells, stimulates pertussis toxin-sensitive arachidonic acid release without inositol phosphate accumulation. *J. Cell Biol.* **113**: 943-50
- Gill DL, Ghosh TK, Bian J, Short AD, Waldron RT & Rybak SL. (1992) Function and organisation of the inositol 1,4,5-trisphosphate-sensitive calcium pool. *Adv. Second Messenger Phosphoprotein Res.* 26: 265-308
- Gilon P, Bird GSJ, Bian X, Yakel JL & Putney JW Jr. (1995) The Ca<sup>2+</sup>-mobilizing actions of a Jurkat cell extract on mammalian cells and *Xenopus laevis* oocytes. *J. Biol. Chem.* 270: 8050-8055
- **Glaser KB, Mobilio D, Chang JY & Senko N. (1993)** Phospholipase A<sub>2</sub> enzymes: regulation and inhibition. *Trends Pharmacol. Sci.* **14**: 92-98
- **Glenney JR & Zokas L. (1989)** Novel tyrosine kinase substrates from Rous sarcoma virus-transformed cells are present in the membrane skeleton. *J. Cell Biol.* **108:** 2401-2408
- **Gnegy ME. (1993)** Calmodulin in neurotransmitter and hormone action. *Ann. Rev. Pharmacol. Toxicol.* **32:** 45-70
- **Gomez N & Cohen P. (1991)** Dissection of the protein kinase cascade by which nerve growth factor activates MAP kinases. *Nature* **353**: 170-173
- Gorbulev V, Akhundova A, Buchner H & Farrenholz F. (1992) Molecular cloning of a new bombesin receptor subtype expressed in uterus during pregnacy. *Eur. J. Biochem* 208: 405-410
- Grady EF, Slice LW, Brant WO, Walsh JH, Payan DG & Bunnett NW. (1995) Direct observation of endocytosis of gastrin releasing peptide and its receptor. *J. Biol. Chem.* **270**: 4603-4611
- Grand RJA, Turnell AS & Grabham PW. (1996) Cellular consequences of thrombinreceptor activation. *Biochem. J.* **313:** 353-368
- Greco FA & Oldham RK. (1979) Small-cell lung cancer. N. Eng. J. Med. 301: 355-358
- **Grynkiewicz G, Peonie M & Tsien RY. (1985)** A new generation of Ca<sup>2+</sup> indicators with greatly improved fluorescence properties. *J. Biol. Chem.* **260:** 3440-3450
- **Guan J-L, Trevithick JE & Hynes RO. (1991)** Fibronectin/integrin interaction induces tyrosine phosphorylation of a 120-kDa protein. *Cell Regul.* **2:** 951-964
- **Gudermann T, Birnbaumer M & Birnbaumer L. (1992)** Evidence for dual coupling of the murine luteinizing hormone receptor to adenylyl cyclase and phosphoinositide breakdown and Ca<sup>2+</sup> mobilisation. Studies with the cloned murine luteinizing hormone receptor expressed in L cells. *J. Biol. Chem.* **267:** 4479-4488
- Gupta SK, Gallego C, Lowndes JM, Pleiman CM, Sable C, Eisfelder BJ & Johnson GL. (1992) Analysis of the fibroblast transformation potential of GTPase-deficient *gip*2 oncogenes. *Mol. Cell. Biol.* **12**: 190-197

- Gutkind JS, Novotny EA, Brann MR & Robbins KC. (1991) Muscarinic acetylcholine receptor subtypes as agonist-dependent oncogenes. *Proc. Natl. Acad. Sci. U.S.A.* 88: 4703-7
- **Gutkind JS & Robbins KC. (1992)** Activation of transforming G protein-coupled receptors induces rapid tyrosine phosphorylation of cellular proteins including p125<sup>FAK</sup> and the p130 v-*src* substrate. *Biochem. Biophys. Res. Comm.* **188:** 155-161
- Gutowski S, Smrcka AV, Nowak L, Wu D, Simon M & Sternweis PC. (1991) Antibodies to the  $\alpha q$  subfamily of guanine nucleotide-binding regulatory protein  $\alpha$  subunits attenuate activation of phosphatidylinositol 4,5-bisphosphate hydrolysis by hormones. *J. Biol. Chem.* **266**: 20519-20524
- Hakii H, Fujiki H, Suganuma M, Nakayasu M, Tahira T, Sugimura T, Scheuer PJ & Christensen SB. (1986) Thapsigargin, a histamine secretagogue, is a non-12-O-tetradecanoylphorbol-13-acetate (TPA) type tumor promoter in two-stage mouse skin carcinogenesis. *J. Cancer Res. Clin. Oncol.* **111**: 177-181
- Hall A. (1990) The cellular functions of small GTP-binding proteins. *Science* 249: 635-640
- Hamm HE & Gilchrist A. (1996) Heterotrimeric G proteins. *Curr. Opin. Cell Biol.* 8: 189-196
- Hanks SK, Calalb MB, Harper MC & Patel SK. (1992) Focal adhesion protein-tyrosine kinase phosphorylated in response to cell attachment to fibronectin. *Proc. Natl. Acad. Sci. U.S.A.* 89: 8487-8491
- Hansen CA, Schroering AG, Carey DJ & Robishaw JD. (1994) Localisation of a heterotrimeric G protein γ subunit to focal adhesions and associated stress fibres. *J. Cell Biol.* **126**: 811-819
- Hanson PI, Meyer T, Stryer L & Schulman H. (1994) Dual role of calmodulin in autophosphorylation of multifunctional CaM kinase may underlie decoding of calcium signals. *Neuron* 12: 943-956
- Harlan JE, Hajduk PJ, Yoon HS & Fesik SW. (1994) Pleckstrin homology domains bind to phosphatidylinositol-4,5-bisphosphate. *Nature* **371:** 168-170
- Hashii M, Nozawa Y & Higashida H. (1993) Bradykinin-induced cytosolic Ca<sup>2+</sup> oscillation and inositol tetrakisphosphate-induced Ca<sup>2+</sup> influx in voltage-clamped *ras*-transformed NIH/3T3 fibroblasts. *J. Biol. Chem.* **268**: 19403-19410
- Hawes BE, Luttrell LM, van Biesen T & Lefkowitz RJ. (1996) Phosphatidylinositol 3kinase is an early intermediate in the  $G\beta\gamma$ -mediated mitogen-activated protein kinase signalling pathway. *J. Biol. Chem.* **271:** 12133-12136
- Hawes BE, van Biesen T, Koch WJ, Luttrell LM & Lefkowitz RJ. (1995) Distinct pathways of G<sub>i</sub>- and G<sub>q</sub>-mediated Mitogen Activated Protein Kinase activation. *J. Biol. Chem.* 270: 17148-17153

- Heldin C-H & Westermark B. (1984) Growth factors: mechanism of action and relation to oncogenes. *Cell* 37: 9-20
- Hepler JR & Gilman AG. (1992) G proteins. Trends Biochem. Sci. 17: 383-387
- Herget T, Brooks SF, Broad S & Rozengurt E. (1992) Relationship between the major protein kinase C substrates acidic 80-kDa protein-kinase-C substrate (80K) and myristoylated alanine-rich C-kinase substrate (MARCKS). *Eur. J. Biochem.* 209: 7-14
- Herget T, Brooks SF, Broad S & Rozengurt E. (1993) Expression of the major protein kinase C substrate, the acidic 80-kilodalton myristoylated alanine-rich C kinase substrate, increases sharply when Swiss 3T3 cells move out of cycle and enter G<sub>0</sub>. *Proc. Natl. Acad. Sci. USA* 909: 2945-2949
- **Herget T & Rozengurt E. (1994)** Bombesin, endothelin, and platelet-derived growth factor induce rapid translocation of the myristoylated alanine-rich C-kinase substrate in Swiss 3T3 cells. *Eur. J. Biochem.* **225:** 539-548
- Higgins TE, Murphy AC, Staddon JM, Lax AJ & Rozengurt E. (1992) Pasteurella multocida toxin is a potent inducer of anchorage-independent cell growth. *Proc. Natl. Acad. Sci. USA* 89: 4240-4244
- Hildebrand JD, Schaller MD & Parsons JT. (1993) Identification of sequences required for the efficient localization of the focal adhesion kinase, pp125<sup>FAK</sup>, to cellular focal adhesions. *J. Cell Biol.* **123**: 993-1005
- Hildebrand JD, Schaller MD & Parsons JT. (1995) Paxillin, a tyrosine phosphorylated focal adhesion-associated protein binds to the carboxyl terminal domain of focal adhesion kinase. *Mol. Biol. Cell* 6: 637-647
- **Holley RW. (1975)** Control of growth of mammalian cells in culture. *Nature* **258:** 487-490
- Hordijk PL, Verlaan I, van Corven EJ & Moolenaar WH. (1994) Protein tyrosine phosphorylation induced by lysophosphatidic acid in Rat-1 fibroblasts. Evidence that phosphorylation of map kinase is mediated by the G<sub>i</sub>-p21<sup>ras</sup> pathway. *J. Biol. Chem.* 269: 645-651
- Hoshi M, Nishida E & Sakai H. (1988) Activation of a Ca<sup>2+</sup>-inhibitable protein kinase that phosphorylates microtubule-associated protein 2 *in vitro* by growth factors, phorbol esters, and serum in quiescent cultured human fibroblasts. *J. Biol. Chem.* 263: 5396-5401
- Howe LR, Leevers SJ, Gomez N, Nakielny S, Cohen P & Marshall CJ. (1992) Activation of the MAP kinase pathway by the protein kinase raf. *Cell* **71**: 335-342
- Hu Z-W, Shi X-Y, Lin RZ & Hoffman BB. (1996)  $\alpha_1$  adrenergic receptors activate phosphatidylinositol 3-kinase in human vascular smooth muscle cells. Role in mitogenesis. *J. Biol. Chem.* 271: 8977-8982

- Huang W, Alessandrini A, Crews CM & Erikson RL. (1993) Raf-1 forms a stable complex with Mek1 and activates Mek1 by serine phosphorylation. *Proc. Natl. Acad. Sci. USA* 90: 10947-10951
- Huckle WR, Dy RC & Earp HS. (1992) Calcium-dependent increase in tyrosine kinase activity stimulated by angiotensin. *Proc. Natl. Acad. Sci. U.S.A.* **89**: 8837-8841
- Huckle WR, Prokop CA, Dy RC, Herman B & Earp S. (1990) Angiotensin II stimulates protein-tyrosine phosphorylation in a calcium dependent manner. *Mol. Cell. Biol.* 10: 6290-6298
- Hung DT, Wong YH, Vu T-K & Coughlin SR. (1992) The cloned platelet thrombin receptor couples to at least two distinct effectors to stimulate phosphoinositide hydrolysis and inhibit adenylyl cyclase. *J. Biol. Chem.* **267**: 20831-20834
- Hunter T, Alexander CB & Cooper JA. (1985) Protein phosphorylation and growth control. (In Growth factors in biology and medicine) eds. *Ciba Foundation Symposium* Pitman, London, **116**: 188-204
- Hunter T & Cooper JA. (1985) Protein-tyrosine kinases. Ann. Rev. Biochem 54: 897-930
- **Hynes RO. (1992)** Integrins: versatility, modulation, and signaling in cell adhesion. *Cell* **69:** 11-25
- Ibarrondo J, Joubert D, Dufour MN, Cohen-Solal A, Homburger V, Jard S & Guillon G. (1995) Close association of the  $\alpha$  subunits of  $G_q$  and  $G_{11}$  proteins with actin filaments in WRK<sub>1</sub> cells: Relation to G protein-mediated phospholipase C activation. *Proc. Natl. Acad. Sci. USA* 92: 8413-8417
- Ilic D, Furuta Y, Kanazawa S, Takeda N, Sobue K, Nakatsuji J, Nomura S, Fujimoto J, Okada M, Yamamoto T & Aizawa S. (1995) Reduced cell motility and enhanced focal adhesion contact formation in cells from FAK-deficient mice. *Nature* **377**: 539-544
- Inglese J, Freedman NJ, Koch WJ & Lefkowitz RJ. (1993) Structure and mechanism of the G protein-coupled receptor kinases. *J. Biol. Chem.* 268: 23735-23738
- **Inglese J, Koch WJ, Touhara K & Lefkowitz RJ. (1995)** Gβγ interactions with PH domains and Ras-MAPK signaling pathways. *Trends Biochem. Sci.* **20**: 151-156
- **Iniguez-Lluhi, Simon MI, Robishaw JD & Gilman AG. (1992)** G Protein  $\beta\gamma$  subunits synthesised in Sf9 cells. Functional characterization and the significance of prenylation of  $\gamma$ . *J. Biol. Chem.* **2267:** 23409-23417
- Irvine RF & Moor RM. (1986) Micro-injection of inositol 1,3,4,5-tetrakisphosphate activates sea urchin eggs by a mechanism dependent on external Ca<sup>2+</sup>. *Biochem. J.* 240: 917-920
- Irvine RF & Moor RM. (1987) Inositol(1,3,4,5,)tetrakisphosphate-induced activation of sea urchin eggs requires the presence of inositol trisphosphate. *Biochem. Biophys. Res. Comm.* 146: 284-290

- Ishida Y, Kawahara Y, Tsuda T, Koide M & Yokoyama M. (1992) Involvement of MAP kinase activators in angiotensin II-induced activation of MAP kinase in cultured vascular smooth muscle cells. *FEBS Lett.* **310**: 41-45
- **Issandou M & Rozengurt E. (1990)** Bradykinin transiently activates protein kinase C in Swiss 3T3 cells. Distinction from activation by bombesin and vasopressin. *J. Biol. Chem.* **265:** 11890-11896
- Jackson TR, Patterson SI, Thastrup O & Hanley MR. (1988) A novel tumour promoter, thapsigargin, transiently increases cytoplasmic free Ca<sup>2+</sup> without generation of inositol phosphates in NG115-401L neuronal cells. *Biochem. J.* **253:** 81-86
- Jaken S. (1996) Protein kinase C isozymes and substrates. *Curr. Opin. Cell Biol.* 8: 168-173
- Jalink K, Eichholtz T, Postma FR, van Corven EJ & Moolenaar WH. (1993) Lysophosphatidic acid induces neuronal shape changes via a novel, receptormediated signaling pathway: similarity to thrombin action. *Cell Growth Diff.* 4: 247-255
- Jalink K & Moolenaar WH. (1992) Thrombin receptor activation causes rapid neural cell rounding and neurite retraction independent of classic second messengers. *J. Cell Biol.* **118**: 411-419
- James R & Bradshaw RA. (1984) Polypeptide growth factors. Ann. Rev. Biochem. 53: 259-292
- Janknecht R, Cahill MA & Nordheim A. (1995) Signal integration at the c-*fos* promoter. *Carcinogenesis* 16: 443-450
- Janmey PA. (1994) Phosphoinositides and calcium as regulators of cellular actin assembly and disassembly. *Annu. Rev. Physiol.* 56: 169-191
- Jensen RT, Moody T, Pert C, Rivier JE & Gardner JD. (1978) Interaction of bombesin and litorin with specific membrane receptors on pancreatic acinar cells. *Proc. Natl. Acad. Sci.* 75: 6139-6143
- Jockusch BM, Bubeck P, Giehl K, Kroemker M, Moschner J, Rothkegel M, Rudiger M, Schluter K, Stanke G & Winkler J. (1995) The molecular architecture of focal adhesions. *Annu. Rev. Cell Dev. Biol.* 11: 379-416
- Julius D, Livelli TJ, Jessell TM & Axel R. (1989) Ectopic expression of the serotonin 1c receptor and the triggering of malignant transformation. *Science* 244: 1057-1062
- Kalinec G, Nazarali Aj, Hermouet S, Xu N & Gutkind JS. (1992) Mutated α subunit of the Gq protein induces malignant transformation in NIH 3T3 Cells. *Mol. Cell. Biol.* 12: 4687-4693
- Kanner SB, Reynolds AB, Vines RR & Parsons JT. (1990) Monoclonal antibodies to individual tyrosine-phosphorylated protein substrates of oncogene-encoded tyrosine kinases. *Proc. Natl. Acad. Sci. USA* 87: 3328-3332

- Karin M. (1995) The regulation of AP-1 activity by mitogen-activated protein kinases. *J. Biol. Chem.* 270: 16483-16486
- **Karin M & Hunter T. (1995)** Transcriptional control by protein phosphorylation: signal transmission from the cell surface to the nucleus. *Curr. Biol.* **5**: 747-757
- Karlsson O, Thor S, Norberg T, Ohlsson H & Edlund T. (1990) Insulin gene enhancer binding protein Isl-1 is a member of a novel class of proteins containing both a homeo- and a Cys-His domain. *Nature* 344: 879-882
- Kass GEN, Duddy SK, Moore GA & Orrenius S. (1989) 2,5-Di-(*tert*-butyl)-1,4benzohydroquinone rapidly elevates cytosolic Ca<sup>2+</sup> concentration by mobilizing the inositol 1,4,5-trisphosphate-sensitive Ca<sup>2+</sup> pool. *J. Biol. Chem.* **264:** 15192-15198
- Kim HK, Kim JW, Zilberstein A, Margolis B, Kim JG, Schesslinger J & Rhee SG. (1991) PDGF stimulation of inositol phosphate hydrolysis requires PLC-γ1 phosphorylation on tyrosine residues 783 and 1254. *Cell* **65**: 435-441
- Kim HY, Thomas D & Hanley MR. (1995) Chromatographic resolution of an intracellular calcium influx factor from thapsigargin-activated Jurkat cells. Evidence for multiple activities influencing calcium elevation in *Xenopus* oocytes. *J. Biol. Chem.* 270: 9706-9708
- **King KA, Torday JS & Sunday ME. (1995)** Bombesin and [Leu<sup>8</sup>]phyllolitorin promote fetal mouse lung branching morphogenesis via a receptor-mediated mechanism. *Proc. Natl. Acad. Sci.* **92:** 4357-4361
- **Kleuss C, Hescheler J, Ewel C, Rosenthal W, Schultz G & Wittig B. (1991)** Assignment of G-protein subtypes to specific receptors inducing inhibition of calcium currents. *Nature* **353:** 43-48
- **Kleuss C, Scherubl H, Hescheler J, Schultz G & Wittig B. (1992)** Different β-subunits determine G-protein interaction with transmembrane receptors. *Nature* **358**: 424-426
- Kleuss C, Scherubl H, Hescheler J, Schultz G & Wittig B. (1993) Selectivity in signal transduction determined by  $\gamma$  subunits of heterotrimeric G proteins. *Science* 259: 832-834
- Koch GLE. (1990) The endoplasmic reticulum and calcium storage. *BioEssays* 12: 527-531
- **Koch WJ, Hawes BE, Allen LF & Lefkowitz RJ. (1994)** Direct evidence that G<sub>i</sub>-coupled receptor stimulation of mitogen-activated protein kinase is mediated by Gβ<sub>γ</sub> activation of p21<sup>ras</sup>. *Proc. Natl. Acad. Sci. USA* **91:** 12706-12710
- Kohn EC, Reed E, Sarosy G, Christian M, Link CJ, Cole K, Figg WD, Davis PA, Jacob J, Goldspiel B & Liotta LA. (1996) Clinical investigation of a cytostatic calcium influx inhibitor in patients with refractory cancers. *Cancer Res.* 56: 569-573
- Koide H, Satoh T, Nakafuku M & Kaziro Y. (1993) GTP-dependent association of Raf-1 with Ha-ras: Identification of Raf as a target downstream of ras in mammalian cells. *Proc. Natl. Acad. Sci. USA* 90: 8683-8686

- Kolch W, Heidecker G, Kochs G, Hummel R, Vahidi H, Mischak H, Finkenzeller G, Marme D & Rapp UR. (1993) Protein kinase C alpha activates Raf-1 by direct phosphorylation. *Nature* 364: 249-252
- Kolch W, Heidecker G, Lloyd P & Rapp UR. (1991) Raf-1 protein kinase is required for growth of induced NIH/3T3 cells. *Nature* **349**: 426-428
- Kornberg LJ, Earp HS, Turner CE, Prockop C & Juliano RL. (1991) Signal transduction by integrins: increased protein tyrosine phosphorylation caused by clustering of  $\beta_1$ integrins. *Proc. Natl. Acad. Sci. USA* 88: 8392-8396
- Kosako H, Gotoh Y, Matsuda S, Ishikawa M & Nishida E. (1992) *Xenopus* MAP kinase activator is a serine/threonine/tyrosine activated by threonine phosphorylation. *1992* **11**: 2903-2908
- Kuang Y, Wu Y, Smrcka A, Jiang H & Wu D. (1996) Identification of a phospholipase C  $\beta_2$  region that interacts with G $\beta_\gamma$ . *Proc. Natl. Acad. Sci. USA* 93: 2964-2968
- **Kuznetsov G, Brostrom MA & Brostrom CO. (1992)** Demonstration of a calcium requirement for secretory protein processing and export. *J. Biol. Chem* **267:** 3932-3939
- Kyriakis JM, App H, Zhang X-f, Banerjee P, Brautigan DL, Rapp UR & Avruch J. (1992) Raf-1 activates MAP kinase-kinase. *Nature* **358**: 417-421
- **Kyriakis JM, Force TL, Rapp UR, Bonventre JV & Avruch J. (1993)** Mitogen regulation of c-Raf-1 protein kinase activity toward mitogen-activated protein kinase kinase. *J. Biol. Chem.* **268:** 16009-16019
- **Laemmli UK. (1970)** Cleavage of structural proteins during the assembly of the head of bacteriophage T4. *Nature* **227:** 680-685
- Lambright DG, Sondek J, Bohm A, Skiba NP, Hamm HE & Sigler PB. (1996) The 2.0 Å crystal structure of a heterotrimeric G protein. *Nature* **379**: 311-319
- **LaMorte VJ, Harootunian AT, Spiegel AM, Tsien RY & Feramisco JR. (1993a)** Mediation of growth factor induced DNA synthesis and calcium mobilization by G<sub>q</sub> and G<sub>i2</sub>. *J. Cell Biol.* **121**: 91-99
- LaMorte VJ, Kennedy ED, Collins LR, Goldstein D, Harootunian AT, Brown JH & Feramisco JR. (1993b) A requirement for ras protein function in thrombinstimulated mitogenesis in astrocytoma cells. *J. Biol. Chem.* **268**: 19411-19415
- **Langhans-Rajasekaran SA, Wan Y & Huang X-Y. (1995)** Activation of Tsk and Btk tyrosine kinases by G protein βγ subunits. *Proc. Natl. Acad. Sci. USA* **92:** 8601-8605
- Laugwitz K-L, Allgeier A, Offermanns S, Spicher K, van Sande J, Dumont JE & Schultz G. (1996) The human thyrotropin receptor: A heptahelical receptor capable of stimulating members of all four G protein families. *J. Biol. Chem.* **93**: 116-120
- Launay J-M, Birraux G, Bondoux D, Callebert J, Choi D-S, Loric S & Maroteaux L. (1996) Ras involvement in signal transduction by the serotonin 5-HT<sub>2B</sub> receptor. *J. Biol. Chem.* 271: 3141-3147

- Lee AS. (1987) Coordinated regulation of a set of genes by glucose and calcium ionophores in mammalian cells. *Trends Biochem. Sci.* 12: 20-23
- Lee AS. (1992) Mammalian stress response: induction of the glucose-regulated protein family. *Curr. Opin. Cell Biol.* 4: 267-273
- **Lee CH, Park D, Wu D, Rhee S-G & Simon MI. (1992)** Members of the Gq α subunit gene family activate phospholipase C β isozymes. *J. Biol. Chem.* **267:** 16044-16047
- Lee K-M, Toscas K & Villereal ML. (1993a) Inhibition of bradykinin- and thapsigargininduced Ca<sup>2+</sup> entry by tyrosine kinase inhibitors. *J. Biol. Chem.* **268**: 9945-9948
- Lee SB & Rhee SG. (1995) Significance of PIP<sub>2</sub> hydrolysis and regulation of phospholipase C isozymes. *Curr. Opin. Cell Biol.* 7: 183-189
- Lee SB, Shin SH, Helper JR, Gilman AG & Rhee SG. (1993b) Activation of phospholipase C- $\beta$ 2 mutants by G protein  $\alpha_q$  and  $\beta\gamma$  subunits. *J. Biol. Chem.* 268: 25952-25957
- Leeb-Lundberg LMF & Song X-H. (1991) Bradykinin and bombesin rapidly stimulate tyrosine phosphorylation of a 120-kDa group of proteins in Swiss 3T3 cells. *J. Biol. Chem.* **266:** 7746-7749
- **Leevers SJ & Marshall CJ. (1992)** Activation of extracellular signal-regulated kinase, ERK2, by p21*ras* oncoprotein. *EMBO J.* **11:** 569-574
- Leevers SJ, Paterson HF & Marshall CJ. (1994) Requirement for ras in raf activation is overcome by targeting raf to the plasma membrane. *Nature* 369: 411-414
- Lefkowitz RJ, Cotecchia S, Samama P & Costa T. (1993) Constitutive activity of receptors coupled to guanine nucleotide regulatory proteins. *Trends Pharmacol. Sci.* 14: 303-307
- **Lenormand P, Sardet C, Pages G, L'Allemain G, Brunet A & Pouyssegur J. (1993)** Growth factors induce nuclear translocation of MAP kinases (p42<sup>mapk</sup> and p44<sup>mapk</sup>) but not their activator MAP kinase kinase (p45<sup>mapkk</sup>) in fibroblasts. *J. Cell Biol.*
- Lewin B. (1991) Oncogenic conversion by regulatory changes in transcription factors. *Cell* 64: 303-312
- Li K, Nagalla SR & Spindel ER. (1994) A rhesus monkey model to characterize the role of gastrin-releasing peptide (GRP) in lung development. *J. Clin. Invest.* 94: 1605-1615
- Li N, Batzer A, Daly R, Yajnik V, Skolnik E, Chardin P, Bar-Sagi D, Margolis B & Schlessinger J. (1993) Guanine-nucleotide-releasing factor hSos1 binds to Grb2 and links receptor tyrosine kinase to ras signalling. *Nature* 363: 85-88
- Lin L-L, Wartmann M, Lin AY, Knopf JL, Seth A & Davis RJ. (1993) cPLA<sub>2</sub> is phosphorylated and activated by MAP kinase. *Cell* **72**: 269-278
- Lipfert L, Haimovich B, Schaller MD, Cobb BS, Parsons JT & Brugge JS. (1992) Integrin-dependent phosphorylation and activation of the protein tyrosine kinase pp125<sup>FAK</sup> in platelets. *J. Cell Biol.* **119**: 905-912

Liu C & Herman TE. (1978) Characterisation of ionomycin as a calcium ionophore. *J. Biol. Chem.* 253: 5892-5894

Liu J-P. (1996) Protein kinase C and its substrates. Mol. Cell. Endocrinol. 116: 1-29

**Liu P, Wen M, Sun L & Hayashi J. (1993)** Activation of phospholipase  $A_2$  and stimulation of prostaglandin  $E_2$  production by transforming growth factor- $\alpha$  in rat thymic epithelial cells requires influx of calcium. *Biochem. J.* **293:** 109-113

- Lopez-Rivas A, Mendoza SA, Nanberg E, Sinnett-Smith J & Rozengurt E. (1987) Ca<sup>2+-</sup> mobilizing actions of platelet-derived growth factor differ from those of bombesin and vasopressin in Swiss 3T3 mouse cells. *Proc. Natl. Acad. Sci. USA.* **84**: 5768-5772
- Lopez-Rivas A & Rozengurt E. (1983) Serum rapidly mobilizes calcium from an intracellular pool in quiescent fibroblastic cells. *Biochem. Biophys. Res. Commun.* 114: 240-247
- **Lopez-Rivas A & Rozengurt E. (1984)** Vasopressin rapidly stimulates Ca<sup>2+</sup> efflux from intracellular pool in quiescent Swiss 3T3 cells. *Am. J. Physiol.* **247:** C156-C162
- Lopez-Rivas A, Stroobant P, Waterfield MD & Rozengurt E. (1984) Ionic responses rapidly elicited by porcine PDGF in Swiss 3T3 cells. *EMBO J.* **3**: 937-944
- Lowenstein EJ, Daly RJ, Batzer AG, Li W, Margolis B, Lammers R, Ullrich A, Skolnik EY, Bar-Sagi D & Schlessinger J. (1992) The SH2 and SH3 domain-containing protein GRB2 links receptor tyrosine kinases to ras signalling. *Cell* **70**: 431-442
- Lowy DR & Willumsen BM. (1993) Function and regulation of ras. *Annu. Rev. Biochem.* 62: 851-891
- **Luckhoff A & Clapham DE. (1992)** Inositol 1,3,4,5-tetrakisphosphate activates an endothelial Ca<sup>2+</sup>-permeable channel. *Nature* **355:** 356-358
- **Luo Z-J, Zhang X-F, Rapp U & Avruch J. (1995)** Identification of the 14.3.3 ζ domains important for self-association and Raf binding. *J. Biol. Chem.* **270:** 23681-23687
- Luttrell LM, Hawes BE, Touhara K, van Biesen T, Koch WJ & Lefkowitz RJ. (1995) Effect of cellular expression of pleckstrin homology domains on G<sub>i</sub>-coupled receptor signaling. *J. Biol. Chem.* **270**: 12984-12989
- Lytton J & Nigam SK. (1992) Intracellular calcium: molecules and pools. *Curr. Opin. Cell. Biol.* 4: 220-226
- Mahmoud S, Staley J, Taylor J, Bogden A, Moreau JP, Coy D, Avis I, Cuttitta F, Mulshine JL & Moody TW. (1991)  $[\psi^{13,14}]$ -Bombesin analogues inhibit growth of small cell lung cancer *in vitro* and *in vivo*. *Cancer Res.* **51**: 1798-1802
- Majerus PW. (1992) Inositol phosphate biochemistry. Annu. Rev. Biochem. 61: 225-250
- Malarkey K, Belham CM, Paul A, Graham A, McLees A, Scott PH & Plevin R. (1995) The regulation of tyrosine kinase signalling pathways by growth factor and Gprotein-coupled receptors. *Biochem. J.* **309:** 361-375

- Mansour SJ, Matten WT, Hermann AS, Candia JM, Rong S, Fukasawa K, Vande Woude GF & Ahn NG. (1994) Transformation of mammalian cells by constitutively active MAP kinase kinase. *Science* 265: 966-970
- Marais R, Light Y, Paterson HF & Marshall CJ. (1995) Ras recruits Raf-1 to the plasma membrane for activation by tyrosine phosphorylation. *EMBO J.* 14: 3136-3145
- Marks F, Hanke B, Thastrup O & Furstenberger G. (1991) Stimulatory effect of thapsigargin, a non-TPA-type tumor promoter, on arachidonic acid metabolism in the murine keratinocyte line HEL30 and on epidermal cell proliferation *in vivo* as compared to the effects of phorbol ester TPA. *Carcinogenesis* 12: 1491-1497
- Marshall CJ. (1995) Specificifty of receptor tyrosine kinase signaling: transient versus sustained extracellular signal-regulated kinase action. *Cell* **80**: 179-185
- Marshall CJ. (1996) Ras effectors. Curr. Opin. Cell Biol. 8: 197-204
- Matsuda S, Kosako H, Takenaka K, Moriyama K, Sakai H, Akiyama T, Gotoh Y & Nishida E. (1992) *Xenopus* MAP kinase activator: identification and function as a key intermediate in the phosphorylation cascade. *EMBO J.* **11**: 973-982
- Mattingley RR, Sorisky A, Brann MR & Macara IG. (1994) Muscarinic receptors transform NIH 3T3 cells through a ras-dependent signalling pathway inhibited by the Ras-GTPase-activating protein SH3 domain. *Mol. Cell. Biol.* 14: 7943-7952
- Mauger J-P, Poggioli J, Guesdon F & Claret M. (1984) Noradrenaline, vasopressin and angiotensin increase Ca<sup>2+</sup> influx by opening a common pool of Ca<sup>2+</sup> channels in isolated rat liver cells. *Biochem. J.* 221: 121-127
- McCormick F. (1993) How receptors turn ras on. Nature 363: 15-16
- McDonald TJ, Jornvall H, Nilsson G, Vagne M, Ghatei M, Bloom SR & Mutt V. (1979) Characterisation of a gastrin releasing peptide from porcine non-antral gastric tissue. *Biochem. Biophys. Res. Com.* **90:** 227-233
- McGlade J, Cheng A, Pelicci G, Pelicci PG & Pawson T. (1992) Shc proteins are phosphorylated and regulated by the v-src and v-fps protein-tyrosine kinases. *Proc. Natl. Acad. Sci. USA* **89:** 8869-8873
- Means AR. (1994) Calcium, calmodulin and cell cycle regulation. FEBS Lett. 347: 1-4
- Medema RH, de Vries-Smits AMM, van der Zon GCM, Maassen JA & Bos JL. (1993) Ras activation by insulin and epidermal growth factor through enhanced exchange of guanine nucleotides on p21<sup>ras</sup>. *Mol. Cell. Biol.* **13**: 155-162
- Mehmet H, Millar JB, Lehmann W, Higgins T & Rozengurt E. (1990) Bombesin stimulation of c-fos expression and mitogenesis in Swiss 3T3 cells: the role of prostaglandin E<sub>2</sub>-mediated cyclic AMP accumulation. *Exp. Cell Res.* **190**: 265-270
- Mehmet H & Rozengurt E. (1991) Regulation of c-*fos* expression in Swiss 3T3 cells: An interplay of multiple signal transduction pathways. *Brit. Med. Bull.* 47: 76-86

- Meldolesi J, Madeddu L & Pozzan T. (1990) Intracellular Ca<sup>2+</sup> storage organelles in non-muscle cells: heterogeneity and functional assignment. *Biochim. Biophys. Acta* 1055: 130-140
- Mendoza SA, Schneider JA, Lopez-Rivas A, Sinnett-Smith J & Rozengurt E. (1986) Early events elicited by bombesin and structurally related peptides in quiescent Swiss 3T3 cells. II. Changes in Na<sup>+</sup> and Ca<sup>2+</sup> fluxes, Na<sup>+</sup>/K<sup>+</sup> pump activity, and intracellular pH. *J. Cell Biol.* **102**: 2223-2233
- **Menniti FS, Oliver KG, Putney JW Jr & Shears SB. (1993)** Inositol phosphates and cell signaling: new views of Ins*P*<sub>5</sub> and Ins*P*<sub>6</sub>. *Trends Biochem. Sci.* **18**: 53-56
- Mikoshiba K. (1993) Inositol 1,4,5-trisphosphate receptor. *Trends Pharmacol. Sci.* 14: 86-89
- Millar JB & Rozengurt E. (1988) Bombesin enhancement of cAMP accumulation in Swiss 3T3 cells: evidence of a dual mechanism of action. *J. Cell. Physiol.* 137: 214-222
- Millar JB & Rozengurt E. (1990a) Arachidonic acid release by bombesin: a novel postreceptor target for heterologous mitogenic desensitization. *J. Biol. Chem.* 265: 19973-19979
- Millar JBA & Rozengurt E. (1990b) Chronic desensitization to bombesin by progressive down-regulation of bombesin receptors in Swiss 3T3 cells: distinction from acute desensitization. *J. Biol. Chem.* 265: 12052-12058
- **Milligan G. (1993)** Mechanisms of multifunctional signalling by G protein-linked receptors. *Trends Pharmacol. Sci.* **14**: 239-244
- Milner RE, Famulski KS & Michalak M. (1992) Calcium binding proteins in the sarcoplasmic/endoplasmic reticulum of muscle and nonmuscle cells. *Mol. Cell. Biochem.* **112:** 1-13
- Minamino N, Kangawa K & Matsuo H. (1983) Neuromedin B: A novel bombesin-like peptide identified in porcine spinal cord. *Biochem. Biophys. Res. Com.* 114: 541-548
- Minamino N, Kangawa K & Matsuo H. (1984) Neuromedin C: A bombesin-like peptide identified in porcine spinal cord. *Biochem. Biophys. Res. Com.* 119: 14-20
- Minamino N, Kangawa K & Matsuo H. (1988) Neuromedin B and neuromedin C. Two mammalian bombesin-like peptides identified in porcine spinal cord and brain. *Ann. N.Y. Acad. Sci.* **547:** 373-390
- Mitchell FM, Heasley LE, Qian N-X, Zamarripa J & Johnson GL. (1995) Differential modulation of bombesin-stimulated phospholipase Cβ and mitogen-activated protein kinase activity by [D-Arg<sup>1</sup>,D-Phe<sup>5</sup>,D-Trp<sup>7,9</sup>,Leu<sup>11</sup>]substance P. *J. Biol. Chem.* **270**: 8623-8628
- Mohammadi M, Dionne CA, Li W, Li N, Spivak T, Honegger AM, Jaye M & Schlessinger J. (1992) Point mutation in FGF receptor eliminates phosphatidylinositol hydrolysis without affecting mitogenesis. *Nature* 358: 681-684

<u>\_\_\_</u>

- Molloy CJ, Bottaro DP, Flemming TP, Marshall MS, Gibbs JB & Aaronson SA. (1989) PDGF induction of tyrosine phosphorylation of GTPase activating protein. *Nature* **342:** 711-714
- **Montero M, Alvarez J & Garcia-Sancho J. (1992)** Control of plasma-membrane Ca<sup>2+</sup> entry by the intracellular Ca<sup>2+</sup> stores. *Biochem. J.* **288:** 519-525
- **Montero M, Garcia-Sancho J & Alvarez J. (1993a)** Comparitive effects of cytochrome *P*-450 inhibitors on Ca<sup>2+</sup> and Mn<sup>2+</sup> entry induced by agonists or by emptying the Ca<sup>2+</sup> stores of human neutrophils. *Biochem. Biophys. Acta* **1177**: 127-133
- Montero M, Garcia-Sancho J & Alvarez J. (1993b) Transient inhibition by chemotactic peptide of a store-operated Ca<sup>2+</sup> entry pathway in human neutrophils. *J. Biol. Chem.* 268: 13055-13061
- Moodie SA, Willumsen BM, Weber MJ & Wolfman A. (1993) Complexes of Ras.GTP with Raf-1 and mitogen-activated protein kinase kinase. *Science* 260: 1658-1661
- Moody TW & Korman LY. (1988) The release of bombesin-like peptides from small cell lung cancer cells. (In Bombesin-like peptides in health and disease) eds. Y. Tache, P. Melchiorri and L. Negri. *Ann. N. Y. Acad. Sci.* New York Acadamy of Sciences, New York, 547: 351-359
- Moody TW, Pert CB, Gazdar AF, Carney DN & Minna JD. (1981) High levels of intracellular bombesin characterize human small-cell lung carcinoma. *Science* 214: 1246-1248
- **Moolenaar WH. (1994)** LPA: a novel lipid mediator with diverse biological actions. *Trends Cell Biol.* **4:** 213-219
- **Moolenaar WH, Tsien RY, VanderSaag PT & DeLaat SW. (1983)** Na<sup>+</sup>/H<sup>+</sup> exchange and cytoplasmic pH in the action of growth factors in human fibroblasts. *Nature* **304**: 645-648
- **Moore GA, McConkey DJ, Kass GEN, O'Brien PJ & Orrenius S. (1987)** 2,5-Di(*tert*butyl)-1,4-benzohydroquinone - a novel inhibitor of liver microsomal Ca<sup>2+</sup> sequestration. *FEBS Lett.* **224:** 331-336
- Morrison DK, Heidecker G, Rapp UR & Copeland TD. (1993) Identification of the major phosphorylation sites of the Raf-1 kinase. *J. Biol. Chem.* 268: 17309-17316
- Morrison DK, Kaplan DR, Rapp U & Roberts TM. (1988) Signal transduction from membrane to cytoplasm: growth factors and membrane bound oncogene products increase Raf-1 phosphorylation and associated protein kinase activity. *Proc. Natl. Acad. Sci. USA* 85: 8855-8859
- **Murayama T & Ui M. (1985)** Receptor-mediated inhibition of adenylate cyclase and stimulation of arachidonic acid release in 3T3 fibroblasts. Selective susceptibility to islet-activating protein, pertussis toxin. *J. Biol. Chem.* **260:** 7226-7233

- Murphy AC & Rozengurt E. (1992) *Pasteurella multocida* toxin selectively facilitates phosphatidylinositol 4,5-bisphosphate hydrolysis by bombesin, vasopressin and endothelin. *J. Biol. Chem.* 267: 25296-25303
- Nakielny S, Campbell DG & Cohen P. (1992a) MAP kinase kinase from rabbit skeletal muscle. A novel dual specificity enzyme showing homology to yeast protein kinases involved in pheromone-dependent signal transduction. *FEBS Lett.* **308**: 183-189
- Nakielny S, Cohen P, Wu J & Sturgill T. (1992b) MAP kinase activator from insulinstimulated skeletal muscle is a protein threonine/tyrosine kinase. *EMBO J.* 11: 2123-2129
- **Nanberg E & Rozengurt E. (1988)** Temporal relationship between inositol polyphosphate formation and increases in cytosolic Ca<sup>2+</sup> in quiescent 3T3 cells stimulated by platelet-derived growth factor, bombesin and vasopressin. *EMBO J.* **7**: 2741-2747
- **Neer EJ. (1995)** Heterotrimeric G proteins: Organizers of transmembrane signals. *Cell* **80:** 249-257
- **Neer EJ, Schmidt CJ, Nambudripad R & Smith TF. (1994)** The ancient regulatoryprotein family of WD-repeat proteins. *Nature* **371:** 297-300
- Neher E. (1992) Controls on calcium influx. Nature 355: 298-299
- Neri LM, Billi AM, Manzoli L, Rubbini S, Gilmour RS, Cocco L & Martelli AM. (1994) Selective nuclear translocation of protein kinase C  $\alpha$  in Swiss 3T3 cells treated with IGF-I, PDGF and EGF. *FEBS Lett.* **347:** 63-68
- Newton AC. (1995) Protein kinase C: Structure, function, and regulation. *J. Biol. Chem.* 270: 28495-28498
- Nicholson NB, Chen S, Blanck G & Pollack R. (1984) SV40 transformation of Swiss 3T3 cells can cause a stable reduction in the calcium requirement for growth. *J. Cell. Biol.* **99:** 2314-2321
- Nishizuka Y. (1984) The role of PKC in cell surface signal transduction and tumour promotion. *Nature* 308: 693-698
- Nishizuka Y. (1986) Studies and perspectives of protein kinase C. *Science* 233: 305-312
- Nishizuka Y. (1988) The molecular heterogeneity of protein kinase C and its implications for cellular regulation. *Nature* 334: 661-665
- Nishizuka Y. (1995) Protein kinase C and lipid signaling for sustained cellular responses. *FASEB J.* 9: 484-496
- **Offermanns S, Heiler E, Spicher K & Schultz G. (1994a)** G<sub>q</sub> and G<sub>11</sub> are concurrently activated by bombesin and vasopressin in Swiss 3T3 cells. *FEBS Lett.* **349:** 201-204
- **Offermanns S, Laugwitz K-L, Spicher K & Schultz G. (1994b)** G proteins of the G<sub>12</sub> family are activated via thromboxane A<sub>2</sub> and thrombin receptors in human platelets. *Proc. Natl. Acad. Sci. USA* **91:** 504-508

- **Olivier AR & Parker PJ. (1992)** Identification of multiple isoforms in Swiss 3T3 cells: Differential down-regulation by phorbol ester. *J. Cell Physiol.* **152:** 240-244
- **Pace AM, Wong YH & Bourne HR. (1991)** A mutant α subunit of G<sub>i2</sub> induces neoplastic transformation of Rat-1 cells. *Proc. Natl. Acad. Sci. USA* **88:** 7031-7035
- Pages G, Lenormand P, L'Allemain G, Chambard J-C, Meloche S & Pouyssegur J. (1993) Mitogen-activated protein kinases p42<sup>mapk</sup> and p44<sup>mapk</sup> are required for fibroblast proliferation. *Proc. Natl. Acad. Sci. USA* 90: 8319-8323
- **Palmer TM, Gettys TW & Stiles GL. (1995)** Differential interaction with and regulation of multiple G-proteins by the rat A3 adenosine receptor. *J. Biol. Chem.* **270:** 16895-16902
- Palumbo AP, Rossino P & Comoglio PM. (1986) Bombesin stimulation of c-fos and cmyc gene expression in cultures of Swiss 3T3 cells. *Exp. Cell. Res.* 167: 276-280
- Pang L, Decker SJ & Saltiel AR. (1993) Bombesin and epidermal growth factor stimulate the mitogen-activated protein kinase through different pathways in Swiss 3T3 cells. *Biochem. J.* 289: 283-287
- Pang L, Zheng C-F, Guan K-L & Saltiel AR. (1995) Nerve growth factor stimulates a novel protein kinase in PC-12 cells that phosphorylates and activates mitogenactivated protein kinase kinase (MEK). *Biochem. J.* 307: 513-519
- Pardee AB. (1974) A restriction point for control of normal animal cell proliferation. *Proc. Natl. Acad. Sci. USA* 71: 1286-1290
- Pardee AB. (1989) G1 events and regulation of cell proliferation. Science 246: 603-608
- Pardee AB, Dubrow R, Hamlin JL & Kletzien RF. (1978) Animal cell cycle. Ann. Rev. Biochem. 47: 715-750
- Pardee AB & Rozengurt E. (1975) Role of the surface in production of new cells. (In Biochemistry of cell walls and membranes) eds. C. Fox. *MTP International Review of Science Biochemistry Series One* Butterworth, London, 2: 155-185
- Parekh AB, Terlau H & Stuhmer W. (1993) Depletion of InsP<sub>3</sub> stores activates a Ca<sup>2+</sup> and K<sup>+</sup> current by means of a phosphatase and a diffusible messenger. *Nature* 364: 814-818
- **Park D, Jhon D-Y, Lee C-W, Lee K-H & Rhee S-G. (1993)** Activation of phospholipase C isozymes by G protein βγ subunits. *J. Biol. Chem.* **268:** 4573-4576
- Parma J, Duprez L, Van Sande J, Cochaux P, Gervy C, Mockel J, Dumont J & Vassart G. (1993) Somatic mutations in the thyrotropin receptor gene cause hyperfunctioning thyroid adenomas. *Nature* 365: 649-651
- Paterson HF, Self AJ, Garrett MD, Just I, Aktories K & Hall A. (1990) Microinjection of recombinant p21<sup>rho</sup> induces rapid changes in cell morphology. *J. Cell Biol.* 111: 1001-1007
- Paul D & Ristow HJ. (1979) Cell cycle control by Ca<sup>++</sup>-ions in mouse 3T3 cells and in transformed 3T3 cells. J. Cell. Physiol. 98: 31-40

- Payne DM, Rossomando AJ, Martino P, Erickson AK, Her J-H, Shabanowitz J, Hunt DF, Weber MJ & Sturgill TW. (1991) Identification of the regulatory phosphorylation sites in pp42/mitogen-activated protein kinase (MAP kinase). *EMBO J.* 10: 885-892
- Pelicci G, Lanfrancone L, Grignani F, McGlade J, Cavallo F, Forni G, Nicoletti I, Grignani F, Pawson T & Pelicci PG. (1992) A novel transforming protein (SHC) with an SH2 domain is implicated in mitogenic signal transduction. *Cell* **70**: 93-104
- Peters KG, Marie J, Wilson E, Ives HE, Escobedo J, Del Rosario M, Mirda D & Williams LT. (1992) Point mutation of an FGF receptor abolishes phosphatidylinositol turnover and Ca<sup>2+</sup> flux but not mitogenesis. *Nature* **358**: 678-681
- **Petersen CCH, Petersen OH & Berridge MJ. (1993)** The role of endoplasmic reticulum calcium pumps during cytosolic calcium spiking in pancreatic acinar cells. *J. Biol. Chem.* **268:** 22262-22264
- Petersen OH, Petersen CCH & Kasai H. (1994) Calcium and hormone action. *Ann. Rev. Physiol.* 56: 297-319

Piomelli D. (1993) Arachidonic acid in cell signaling. Curr. Opin. Cell Biol. 5: 274-280

- Pittet D, Lew DP, Mayr GW, Monod A & Schlegel W. (1989) Chemoattractant Receptor Promotion of Ca<sup>2+</sup> Influx across the Plasma Membrane of HL-60 Cells: A role for cytosolic free calcium elevations and inositol 1,3,4,5-tetrakisphosphate production. *J. Biol. Chem.* 264: 7251-7261
- **Poggioli J, Mauger J-P, Guesdon F & Claret M. (1985)** A regulatory calcium-binding site for calcium channel in isolated rat hepatocytes. *J. Biol. Chem.* **260**: 3289-3294
- **Polte TR & Hanks SK. (1995)** Interaction between focal adhesion kinase and Crkassociated tyrosine kinase substrate p130<sup>Cas</sup>. *Proc. Natl. Acad. Sci. USA* **92:** 10678-10682
- **Price BD, Mannheim-Rodman LA & Calderwood SK. (1992)** Brefeldin A, thaspigargin, and  $AlF_4^-$  stimulate the accumulation of GRP78 mRNA in a cycloheximide dependent manner, whilst induction by hypoxia Is independent of protein synthesis. *J. Cell. Physiol.* **152:** 545-552
- **Pronin AN & Gautam N. (1992)** Interaction between G-protein  $\beta$  and  $\gamma$  subunit types is selective. *Proc. Natl. Acad. Sci. USA* **89:** 6220-6224
- Pumiglia KM, LeVine H, Haske T, Habib T, Jove R & Decker SJ. (1995) A direct interaction between G-protein βγ subunits and the Raf-1 protein kinase. *J. Biol. Chem.* 270: 14251-14254
- **Putney JW Jr. (1986)** A model for receptor-regulated calcium entry. *Cell Calcium* 7: 1-12
- **Putney JW Jr. & Bird GSJ. (1993)** The inositol phosphate-calcium signaling system in nonexitable cells. *Endocrine Rev.* **14:** 610-631
- **Putney JW Jr., Poggioli J & Weiss SJ. (1981)** Receptor regulation of calcium release and calcium permeability in parotid gland cells. *Phil. Trans. R. Soc. Lond.* **296:** 37-45

- Qin Y, Halmos G, Cai R-Z, Szoke B, Ertl T & Schally AV. (1994) Bombesin antagonists inhibit *in vitro* and *in vivo* growth of human gastric cancer and binding of bombesin to its receptor. *J. Cancer Res. Clin. Oncol.* **120**: 519-528
- **Randriamampita C & Tsien RY. (1993)** Emptying of intracellular Ca<sup>2+</sup> stores releases a novel small messenger that Stimulates Ca<sup>2+</sup> influx. *Nature* **364**: 809-814
- **Rankin S, Morii N, Narumiya S & Rozengurt E. (1994)** Botulinum C3 exoenzyme blocks the tyrosine phosphorylation of p125<sup>FAK</sup> and paxillin induced by bombesin and endothelin. *FEBS Lett.* **354:** 315-319
- Rapp UR, Goldsborough MD, Mark GE, Bonner TI, Groffen J, Reynolds FH Jr. & Stephenson JR. (1983) Structure and biological activity of v-*raf*, a unique oncogene transduced by a retrovirus. *Proc. Natl. Acad. Sci. USA* 80: 4218-4222
- Rasmussen U, Broogger-Christensen S & Sandberg F. (1978) Thapsigargine and thapsigargicine, two new histamine liberators from *Thapsia garganica* L. *Acta Pharm. Suec.* **15**: 133-140
- Ray LB & Sturgill TW. (1987) Rapid stimulation by insulin of a serine/threonine kinase in 3T3-L1 adipocytes that phosphorylates microtubule-associated protein 2 *in vitro*. *Proc. Natl. Acad. Sci. USA* 84: 1502-1506
- **Ray LB & Sturgill TW. (1988)** Insulin-stimulated microtubule-associated protein kinase is phosphorylated on tyrosine and threonine *in vivo. Proc. Natl. Acad. Sci. USA* **85**: 3753-3757
- **Reed PW & Lardy HA. (1972)** A23187: a divalent cation ionophore. *J. Biol. Chem.* **247**: 6970-6977
- **Reuter CWM, Catling AD, Jelinek T & Weber MJ. (1995)** Biochemical analysis of MEK activation in NIH3T3 fibroblasts. Identification of B-Raf and other activators. *J. Biol. Chem.* **270:** 7644-7655
- **Ridley AJ & Hall A. (1992)** The small GTP-binding protein rho regulates the assembly of actin stress fibers in response to growth factors. *Cell* **70:** 389-399
- Robbins DJ, Cheng M, Zhen E, Vanderbilt CA, Feig LA & Cobb MH. (1992) Evidence for a ras-dependent extracellular signal-regulated protein kinase (ERK) cascade. *Proc. Natl. Acad. Sci. USA* **89:** 6924-6928
- **Robinson IM, Cheek TR & Burgoyne RD. (1992)** Ca<sup>2+</sup> influx induced by the Ca<sup>2+</sup>-ATPase inhibitors 2,5-di-(t-butyl)-1,4-benzohydroquinone and thapsigargin in bovine adrenal chromaffin cells. *Biochem. J.* **288:** 457-463
- **Rodriguez-Pena** A & **Rozengurt E. (1984)** Disappearance of Ca<sup>2+</sup>-sensitive, phospholipid-dependent PKC activity in phorbol ester-treated 3T3 cells. *Biochem. Biophys. Res. Commun.* **120:** 1053-1059
- **Rodriguez-Pena A & Rozengurt E. (1986)** Phosphorylation of an acidic mol. wt. 80 000 cellular protein in a cell-free system and intact Swiss 3T3 cells: a specific marker of protein kinase C activity. *EMBO J.* **5:** 77-83

- Rodriguez-Viciana P, Warne PH, Dhand R, Vanhaesebroeke B, Gout I, Fry MJ, Waterfield MD & Downward J. (1994) Phosphatidylinositol-3-OH kinase as a direct target of ras. *Nature* **370**: 527-532
- **Rosen LB, Ginty DD, Weber MJ & Greenberg ME. (1994)** Membrane depolarization and calcium influx stimulate MEK and MAP kinase via activation of ras. *Neuron* **12**: 1207-1221
- **Ross R, Raines EW & Bowen-Pope DF. (1986)** The biology of Platelet-Derived Growth Factor. *Cell* **46**: 155-169
- Rossomando AJ, Payne DM, Weber MJ & Sturgill TW. (1989) Evidence that pp42, a major tyrosine kinase target protein, is a mitogen-activated serine/threonine protein kinase. *Proc. Natl. Acad. Sci. USA* 86: 6940-6943
- **Rozakis-Adcock M, Fernly R, Wade J, Pawson T & Bowtell D. (1993)** The SH2 and SH3 domains of mammalian Grb2 couple the EGF receptor to the ras activator mSos1. *Nature* **363:** 83-85
- Rozakis-Adcock M, McGlade J, Mbamalu G, Pelicci G, Daly R, Li W, Batzer A, Thomas S, Brugge J, Pelicci PG, Sclessinger J & Pawson T. (1992) Association of the Shc and Grb2/Sem5 SH2-containing proteins is implicated in activation of the Ras pathway by tyrosine kinases. *Nature* **360**: 689-692
- **Rozengurt E. (1980)** Stimulation of DNA synthesis in quiescent cultured cells: Exogenous agents, internal signals and early events. *Curr. Top. Cell. Regul.* **17:** 59-88
- **Rozengurt E. (1981)** cAMP: a growth promoting signal for 3T3 cells. *Adv. Cyclic Nucleotide Res.* **14**: 429-442
- **Rozengurt E. (1982a)** Adenosine receptor activation in quiescent Swiss 3T3 cells. *Exp. Cell. Res.* **139:** 71-78
- **Rozengurt E. (1982b)** Synergistic stimulation of DNA synthesis by cAMP derivatives and growth factors in mouse 3T3 cells. *J. Cell. Physiol.* **112**: 243-250
- Rozengurt E. (1986) Early signals in the mitogenic response. Science 234: 161-166
- **Rozengurt E. (1991a)** Neuropeptides as cellular growth factors. *Eur. J. Clin. Invest.* **21**: 123-134
- **Rozengurt E. (1991b)** Neuropeptides as cellular growth factors: role of multiple signalling pathways. *Eur. J. Clin. Invest.* **21**: 123-134
- **Rozengurt E. (1995)** Convergent signalling in the action of Integrins, Neuropeptides, Growth Factors and Oncogenes. *Cancer Surveys* **24**: 81-96
- **Rozengurt E, Brown KD & Pettican P. (1981a)** Vasopressin inhibition of EGF binding to cultured mouse cells. *J. Biol. Chem.* **256:** 716-722
- **Rozengurt E & Collins M. (1983)** Molecular aspects of growth factor action: receptors and intracellular signals. *J. Pathology* **141:** 309-331
- **Rozengurt E, Collins MKL & Keehan M. (1983a)** Mitogenic effect of prostglandin E<sub>1</sub> in Swiss 3T3 cells: role of cAMP. *J. Cell. Physiol.* **116**: 379-384

- **Rozengurt E & Heppel L. (1975)** Serum rapidly stimulates oubain-sensitive <sup>86</sup>Rb uptake in quiescent 3T3 cells. *Proc. Natl. Acad. Sci. USA* **72:** 4492-4495
- Rozengurt E, Higgins T, Chanter N, Lax AJ & Staddon JM. (1990) Pasteurella multocida toxin: Potent mitogen for cultured fibroblasts. *Proc. Natl. Acad. Sci. USA* 87: 123-127
- **Rozengurt E, Legg A & Pettican P. (1979)** Vasopressin stimulation of mouse 3T3 cell growth. *Proc. Natl. Acad. Sci. USA* **76:** 1284-1287
- Rozengurt E, Legg A, Strang G & Courtenay-Luck N. (1981b) cAMP: a mitogenic signal for Swiss 3T3 cells. *Proc. Natl. Acad. Sci. USA* 78: 4392-4396
- **Rozengurt E & Mendoza S. (1980)** Monovalent ion fluxes and the control of cell proliferation in cultured fibroblasts. *Ann. New York Acad. Sci.* **339:** 175-190
- **Rozengurt E & Ober SS. (1988)** Monovalent ion fluxes and growth stimulation: Early activation of the Na<sup>+</sup>/H<sup>+</sup> antiport and the Na<sup>+</sup>-K<sup>+</sup> pump in fibroblast mitogenesis. (In Monovalent cations in biological systems) eds. C. A. Pasternak. CRC Press, Inc, Boca Raton, Florida, 303-319
- **Rozengurt E, Rodriguez-Pena A, Coombs M & Sinnett-Smith J. (1984)** Diacylglycerol stimulates DNA synthesis and cell division in mouse 3T3 cells. *Proc. Natl. Acad. Sci. USA* **81:** 5748-5752
- **Rozengurt E, Rodriguez-Pena A & Smith KA. (1983b)** Phorbol esters, phospholipase C, and growth factors rapidly stimulate the phosphorylation of a Mr 80,000 protein in intact quiescent 3T3 cells. *Proc. Natl. Acad. Sci. USA.* **80:** 7224-7248
- **Rozengurt E & Sinnett-Smith J. (1983)** Bombesin stimulation of DNA synthesis and cell division in cultures of Swiss 3T3 cells. *Proc. Natl. Acad. Sci. USA* **80**: 2936-2940
- Rozengurt E & Sinnett-Smith J. (1987) Bombesin induction of c-fos and c-myc protooncogenes in Swiss 3T3 cells: significance for the mitogenic response. *J. Cell. Physiol.* 131: 218-225
- **Rozengurt E & Sinnett-Smith J. (1988a)** Early signals underlying the induction of the cfos and c-myc genes in quiescent fibroblasts: studies with bombesin and other growth factors. *Prog. Nucleic Acid Res. Mol. Biol.* **35**: 261-295
- **Rozengurt E & Sinnett-Smith J. (1988b)** Early signals underlying the induction of the proto-oncogenes c-fos and c-myc in quiescent fibroblasts: studies with peptides of the bombesin family and other growth factors. *Prog. Nucleic Acid Res. Mol. Biol.* **35**: 261-295
- **Rozengurt E, Stroobant P, Waterfield MD, Deuel TF & Keehan M. (1983c)** PDGF elicits cAMP accumulation in Swiss 3T3 cells. *Cell* **34**: 265-272
- **Rubin EJ, Gill DM, Boquet P & Popoff MR. (1988)** Functional modification of a 21kilodalton G protein when ADP-ribosylated by exoenzyme C3 of *Clostridium botulinum. Mol. Cell. Biol.* **8:** 418-426

- **Rusanescu G, Qi H, Thomas SM, Brugge JS & Halegoua S. (1995)** Calcium influx induces neurite growth through a Src-Ras signaling cassette. *Neuron* **15**: 1415-1425
- **Russell M, Winitz S & Johnson G. (1994)** Acetylcholine muscarinic m1 receptor regulation of cyclic AMP synthesis controls growth factor stimulation of Raf activity. *Mol. Cell. Biol.* **14:** 2343-2351
- Sadoshima J & Izumo S. (1996) The heterotrimeric G<sub>q</sub> protein-coupled angiotensin II receptor activates p21<sup>ras</sup> via the tyrosine kinase-Shc-Grb2-Sos pathway in cardiac myocytes. *EMBO J.* 15: 775-787
- **Sagara Y & Inesi G. (1991)** Inhibition of the sarcoplasmic reticulum Ca<sup>2+</sup> transport ATPase by thapsigargin at subnanomolar concentrations. *J. Biol. Chem.* **266:** 13503-13506
- **Sager R. (1989)** Tumour suppressor genes: the puzzle and the promise. *Science* **246**: 406-412
- Sakai R, Iwamatsu A, Hirano N, Ogawa S, Tanaka T, Mano H, Yazaki Y & Hirai H. (1994) A novel signaling molecule, p130, forms stable complexes *in vivo* with v-Crk and v-Src in a tyrosine phosphorylation dependent manner. *EMBO J.* **13**: 3748-3756
- Salgia R, Li J-L, Lo SH, Brunkhorst B, Kansas GS, Sobhany ES, Sun Y, Pisick E, Hallek M, Ernst T, Tantravahi R, Chen LB & Griffin JD. (1995) Molecular cloning of human paxillin, a focal adhesion protein phosphorylated by p210<sup>BCR/ABL</sup>. *J. Biol. Chem.* 270: 5039-5047
- **Sambrook JF. (1990)** The involvement of calcium in transport of secretory proteins from the endoplasmic reticulum. *Cell* **61:** 197-199
- Satoh T, Endo M, Nakafuku M, Akiyama T, Yamamoto T & Kaziro Y. (1990a) Accumulation of p21<sup>ras</sup>GTP in response to stimulation with epidermal growth factor and oncogene products with tyrosine kinase activity. *Proc. Natl. Acad. Sci. USA* 87: 7926-7929
- Satoh T, Endo M, Nakafuku M, Nakamura S & Kaziro Y. (1990b) Platelet-derived growth factor stimulates formation of active p21<sup>ras</sup>GTP complex in Swiss mouse 3T3 cells. *Proc. Natl. Acad. Sci. USA* 87: 5993-5997
- Schaller MD, Borgman CA, Cobb BS, Vines RR, Reynolds AB & Parsons JT. (1992) pp125<sup>FAK</sup>, a structurally distinctive protein-tyrosine kinase associated with focal adhesions. *Proc. Natl. Acad. Sci. USA* 89: 5192-5196
- **Schaller MD & Parsons JT. (1995)** pp125<sup>FAK</sup> -dependent tyrosine phosphorylation of paxillin creates a high-affinity binding site for Crk. *Mol. Cell. Biol.* **15:** 2635-2645
- Schertler GFX, Villa C & Henderson R. (1993) Projection structure of rhodopsin. *Nature* 362: 770-772
- **Schilling WP, Cabello OA & Rajan L. (1992)** Depletion of the inositol 1,4,5trisphosphate-sensitive intracellular Ca<sup>2+</sup> store in vascular endothelial cells activates the agonist-sensitive Ca<sup>2+</sup>-influx pathway. *Biochem. J.* **284:** 521-530

- **Schmidt CJ, Thomas TC, Levine MA & Neer EJ. (1992)** Specificity of G protein β and γ subunit interactions. *J. Biol. Chem.* **267:** 13807-13810
- Schuldiner S & Rozengurt E. (1982) The Na<sup>+</sup>-H<sup>+</sup> antiport in Swiss 3T3 cells: mitogenic stimulation leads to cytoplasmic alkalinization. *Proc. Natl. Acad. Sci. USA* **79:** 7778-7782
- **Schutzbach JS & Forsee WT. (1990)** Calcium ion activation of rabbit liver α1,2mannosidase. *J. Biol. Chem.* **265:** 2546-2549
- **Seckl MJ, Morii N, Narumiya S & Rozengurt E. (1995)** Guanosine 5'-3-0-(thio)triphosphate stimulates tyrosine phosphorylation of p125<sup>FAK</sup> and paxillin in permeabilized Swiss 3T3 cells. *J. Biol. Chem.* **270**: 6984-6990
- Seger R, Ahn NG, Boulton TG, Yancopoulos GD, Panayotatos N, Radziejewska E, Ericsson L, Bratlien RL, Cobb MH & Krebs EG. (1991) Microtubule-associated protein 2 kinases, ERK1 and ERK2, undergo autophosphorylation on both tyrosine and threonine residues: Implications for their mechanism of activation. *Proc. Natl. Acad. Sci. USA* 88: 6142-6146
- Seger R, Ahn NG, Posada J, Munar ES, Jensen AM, Cooper JA, Cobb MH & Krebs EG. (1992a) Purification and characterization of mitogen-activated protein kinase activator(s) from epidermal growth factor-stimulated A431 cells. *J. Biol. Chem.* 267: 14373-14381
- Seger R, Seger D, Lozeman FJ, Ahn NG, Graves LM, Campbell JS, Ericsson L, Harrylock M, Jensen AM & Krebs EG. (1992b) Human T-cell mitogen-activated protein kinase kinases are related to yeast signal transduction kinases. *J. Biol. Chem.* 267: 25628-25631
- Seiler SM, Goldenberg HJ, Michel IM, Hunt JT & Zavoico GB. (1991) Multiple pathways of thrombin-induced platelet activation differentiated by desensitization and a thrombin exosite inhibitor. *Biochem. Biophys. Res. Commun.* **181**: 636-643
- Sethi T, Herget T, Wu SV, Walsh JH & Rozengurt E. (1993) CCK<sub>A</sub> and CCK<sub>B</sub> receptors are expressed in small cell lung cancer lines and mediate Ca<sup>2+</sup> mobilization and clonal growth. *Cancer Res.* **53**: 5208-5213
- Sethi T, Langdon S, Smyth J & Rozengurt E. (1992) Growth of small cell lung cancer cells: stimulation by multiple neuropeptides and inhibition by a broad spectrum of antagonists. *Cancer Res.* 52: 2737s-2742s
- **Sethi T & Rozengurt E. (1991a)** Galanin stimulates Ca<sup>2+</sup> mobilization, inositol phosphate accumulation, and clonal growth in small cell lung cancer cells. *Cancer Res.* **51:** 1674-9
- Sethi T & Rozengurt E. (1991b) Multiple neuropeptides stimulate clonal growth of small cell lung cancer: effects of bradykinin, vasopressin, cholecystokinin, galanin and neurotensin. *Cancer Res.* 51:

- **Sethi T & Rozengurt E. (1992)** Gastrin stimulates Ca<sup>2+</sup> mobilization and clonal growth in small cell lung cancer cells. *Cancer Res.* **52:** 6031-6035
- **Seufferlein T, Withers D, Broad S, Herget T, Walsh J & Rozengurt E. (1995)** The human CCK<sub>B</sub>/gastrin receptor transfected into Rat1-fibroblasts mediates activation of MAP kinase, p74<sup>raf-1</sup> kinase and mitogenesis. *Cell Growth Diff.* **6:** 383-393
- **Seufferlein T, Withers DJ & Rozengurt E. (1996)** Reduced requirement of MAPK activity for entry into the S phase of the cell cycle in Swiss 3T3 fibroblasts stimulated by bombesin and insulin. *J. Biol. Chem.* **271:** in press
- Seva C, Kowalski-Chauvel A, Blanchet J-S, Vaysse N & Pradayrol L. (1996) Gastrin induces tyrosine phosphorylation of Shc proteins and their association with the Grb2/Sos complex. *FEBS Lett.* **378:** 74-78
- **Sharma SV. (1993)** Melittin-induced hyperactivation of phospholipase A<sub>2</sub> activity and calcium influx in ras-transformed cells. *Oncogene* **8:** 939-947
- Shears SB. (1992) Metabolism of inositol phosphates. Adv. Second Messenger Phosphoprotein Res. 26: 63-92
- Shipley GP & Ham RG. (1981) Improved medium and culture conditions for clonal growth with minimal serum protein and for enhanced serum-free survival of Swiss 3T3 cells. *In Vitro* 17: 656-670
- Short AD, Bian J, Ghosh TK, Waldron RT, Rybak SL & Gill DL. (1993) Intracellular Ca<sup>2+</sup> pool content is linked to control of cell growth. *Proc. Natl. Acad. Sci. USA* 90: 4986-4990
- Simon MA, Bowtell DD, Dodson GS, Laverty TR & Rubin GM. (1991) Ras1 and a putative guanine nucleotide exchange factor perform crucial steps in signaling by the sevenless protein tyrosine kinase. *Cell* 67: 701-716
- Sinnett-Smith J, Lehmann W & Rozengurt E. (1990) Bombesin receptor in membranes from Swiss 3T3 cells. Binding characteristics, affinity labelling and modulation by guanine nucleotides. *Biochem. J.* **265:** 485-493
- Sinnett-Smith J, Zachary I & Rozengurt E. (1988) Characterization of a bombesin receptor on Swiss mouse 3T3 cells by affinity cross-linking. *J. Cell. Biochem.* 38: 237-249
- **Sinnett-Smith J, Zachary I, Valverde AM & Rozengurt E. (1993)** Bombesin stimulation of p125 focal adhesion kinase tyrosine phosphorylation: role of protein kinase C, Ca<sup>2+</sup> mobilization and the actin cytoskeleton. *J. Biol. Chem.* **268:** 14261-14268
- Smit L, van der Horst G & Borst J. (1996) Sos, Vav, and C3G participate in B cell receptor-induced signaling pathways and differentially associate with Shc-Grb2, Crk, and Crk-L adaptors. *J. Biol. Chem.* 271: 8564-8569
- Smith JB & Rozengurt E. (1978) Lithium transport by fibroblastic mouse cells: characterization and stimulation by serum and growth factors in quiescent cultures. *J. Cell. Physiol.* 97: 441-449
- Smith MR, DeGudicibus SJ & Stacey DW. (1986) Requirement for c-*ras* proteins during viral oncogene transformation. *Nature* **320**: 540-543
- Smith MR, Ryu S-H, Suh P-G, Rhee S-G & Kung H-F. (1989) S-phase induction and transformation of quiescent NIH 3T3 cells by microinjection of phospholipase C. *Proc. Natl. Acad. Sci. USA* 86: 3659-3663
- Solcia E, Buffa R, Gini A, Capella C, Rindi G & Polak JM. (1988) Bombesin-related peptides in the diffuse neuroendocrine system. (In Bombesin-like peptides in health and disease) eds. Y. Tache, P. Melchiorri and L. Negri. *Ann. N. Y. Acad. Sci.* New York Acadamy of Sciences, New York, 547: 83-94
- **Sondek J, Bohm A, Lambright DG, Hamm HE & Sigler PB. (1996)** Crystal structure of a G protein βγ dimer at 2.1 Å resolution. *Nature* **379:** 369-374
- Sozeri O, Vollmer K, Liyanage M, Frith D, Kour G, Mark GE & Stabel S. (1992) Activation of the c-Raf protein kinase by protein kinase C phosphorylation. *Oncogene* 7: 2259-2262
- Spiegel AM. (1988) Guanine nucleotide binding proteins and signal transduction. *Vit. Horm.* 44: 47-101
- Spindel ER, Giladi E, Brehm P, Goodman RJ & Segerson TP. (1990) Cloning and functional characterisation of complementary DNA encoding the murine fibroblast bombesin/gastrin-releasing peptide receptor. *Mol. Endocrinol.* 4: 1956-1963
- **Staddon JM, Bouzyk MM & Rozengurt E. (1992)** Interconversion of GRP78/BiP. A novel event in the action of Pasteurella multocida toxin, bombesin, and platelet derived growth factor. *J. Biol. Chem.* **267:** 25239-25245
- Stephens L, Smrcka A, Cooke FT, Jackson TR, Sternwweis PC & Hawkins PT. (1994) A novel phosphoinositide 3 kinase activity in myleloid-derived cells is activated by G protein βγ subunits. *Cell* **77**: 83-93
- **Steranka LR, Farmer SG & Burch RM. (1989)** Antagonists of B2 bradykinin receptors. *FASEB J.* **3:** 2019-2025
- Stoker M, O'Neill C, Berryman S & Waxman V. (1968) Anchorage and growth regulation in normal and virus-transformed cells. *Int. J. Cancer* **3**: 683-693
- **Stokoe D, Macdonald SG, Cadwallader K, Symons M & Hancock JF. (1994)** Activation of Raf as a result of recruitment to the plasma membrane. *Science* **264**: 1463-1467
- Stoyanov B, Volinia S, Hanck T, Rubio I, Vanhaesebroeck M, Dhand R, Nurnberg B, Gierschik P, Seedorf K, Hsuan JJ, Waterfield MD & Wetzker R. (1995) Cloning and characterisation of a G protein-activated human phosphoinositide-3 kinase. *Science* 269: 690-693
- Strader CD, Fong TM, Graziano MP & Tota MR. (1995) The family of G-proteincoupled receptors. *FASEB J.* 9: 745-754
- **Strader CD, Fong TM, Tota MR, Underwood D & Dixon RAF. (1994)** Structure and function of G protein-coupled receptors. *Ann. Rev. Biochem.* **63:** 101-132

- **Strathmann MP & Simon MI. (1991)**  $G\alpha_{12}$  and  $G\alpha_{13}$  subunits define a fourth class of G protein  $\alpha$  subunits. *Proc. Natl. Acad. Sci. U.S.A.* **88:** 5582-5586
- Stumpo DJ, Bock CB, Tuttle JS & Blackshear PJ. (1995) MARCKS deficiency in mice leads to abnormal brain development and perinatal death. *Proc. Natl. Acad. Sci. USA* 92: 944-948
- Sturgill TW, Ray LB, Erikson E & Maller JL. (1988) Insulin-stimulated MAP-2 kinase phosphorylates and activates ribosomal protein S6 kinase II. *Nature* 334: 715-718
- Sunday ME. (1988) Tissue-specific expression of the mammalian bombesin gene. (In Bombesin-like peptides in health and disease) eds. Y. Tache, P. Melchiorri and L. Negri. Ann. N. Y. Acad. Sci. New York Acadamy of Sciences, New York, 547: 95-113
- **Superti-Furga G & Courtneidge SA. (1995)** Structure-function relationships in Src family and related protein tyrosine kinases. *BioEssays* **17:** 321-330
- Suzuki CK, Bonifacino JS, Lin AY, Davis MM & Klausner RD. (1991) Regulating the retention of T-cell receptor  $\alpha$  chain variants within the endoplasmic reticulum: Ca<sup>2+-</sup> dependent association with BiP. *J. Cell Biol.* **114**: 189-205
- Tache Y, Melchiorri P & Negri L. (1988) Bombesin-like peptides in health and disease. (In *Ann. N. Y. Acad. Sci.*). New York Acadamy of Sciences, New York, **547:** 1-541
- Takai Y, Sasaki T, Tanaka K & Nakanishi H. (1995) Rho as a regulator of the cytoskeleton. *Trends Biochem. Sci.* 20: 227-231
- Takemura H, Hughes AR, Thastrup O & Putney JW Jr. (1989) Activation of calcium entry by the tumor promoter thapsigargin in parotid acinar cells. *J. Biol. Chem.* 264: 12266-12271
- Takemura H & Putney JW Jr. (1989) Capacitative calcium entry in parotid acinar cells. *Biochem. J.* 258: 409-412
- Takuwa N, Iwamoto A, Kumada M, Yamashita K & Takuwa Y. (1991a) Role of Ca<sup>2+</sup> influx in bombesin-induced mitogenesis in Swiss 3T3 fibroblasts. *J. Biol. Chem.* 266: 1403-1409
- Takuwa N, Kumada M, Yamashita K & Takuwa Y. (1991b) Mechanisms of bombesininduced arachidonate mobilization in Swiss 3T3 fibroblasts. *J. Biol. Chem.* 266: 14237-14243
- Takuwa N, Takuwa Y, Yanagisawa M, Yamashita K & Masaki T. (1989) A novel vasoactive peptide endothelin stimulates mitogenesis through inositol lipid turnover in Swiss 3T3 fibroblasts. *J. Biol. Chem.* **264:** 7856-61
- **Takuwa N, Zhou W & Takuwa Y. (1995)** Calcium, calmodulin and cell cycle progression. *Cell. Signal.* **7:** 93-104
  - Taussig R & Gilman AG. (1995) Mammalian membrane-bound adenylyl cyclases. J. Biol. Chem. 270: 1-4
  - **Taylor CW & Marshall ICB. (1992)** Calcium and inositol 1,4,5-trisphosphate receptors: a complex relationship. *Trends Biochem. Sci.* **17:** 403-407

- Temin HM, Pearson RW & Dulak NC. (1972) The role of serum in the control of multiplication of avian and mammalian cells in culture. *In: Growth, Nutrition and Metabolism of Cells in Culture.* eds. G.H. Rothblat and V.J. Crisfalo, Academic Press. pp. 50-81
- **Tepel M, Kuhnapfel S, Theilmeier G, Teupe C, Schlotmann R & Zidek W. (1994)** Filling state of intracellular Ca<sup>2+</sup> pools triggers trans plasma membrane Na<sup>+</sup> and Ca<sup>2+</sup> influx by a tyrosine kinase-dependent pathway. *J. Biol. Chem.* **269:** 26239-26242
- Tepikin AV & Petersen OH. (1992) Mechanisms of cellular calcium oscillations in secretory cells. *Biochim. Biophys. Acta* 1137: 197-207
- Thastrup O, Cullen PJ, Drobak BK, Hanley MR & Dawson AP. (1990) Thapsigargin, a tumor promoter, discharges intracellular Ca<sup>2+</sup> stores by specific inhibition of the endoplasmic reticulum Ca<sup>2+</sup>-ATPase. *Proc. Natl. Acad. Sci. USA* **87:** 2466-2470
- Thastrup O, Dawson AP, Scharff O, Foder B, Cullen PJ, Drobak BK, Bjerrum PJ, Christensen SB & Hanley MR. (1989) Thapsigargin, a novel molecular probe for studying intracellular calcium release and storage. *Agents Actions* 27: 17-23
- Thomas D & Hanley MR. (1994) Pharmacological tools for perturbing intracellular calcium storage. *Methods Cell Biol.* 40: 65-89
- Thomas D & Hanley MR. (1995) Evaluation of calcium influx factors from stimulated Jurkat T-lymphocytes by microinjection into *Xenopus* oocytes. *J. Biol. Chem.* 270: 6429-6432
- Thomas SM, DeMarco M, D'Arcangelo G, Halegoua S & Brugge JS. (1992) Ras is essential for nerve growth factor- and phorbol ester-induced tyrosine phosphorylation of MAP kinases. *Cell* 68: 1031-1040
- Thomason PA, James SR, Casey PJ & Downes CP. (1994) A G-protein  $\beta\gamma$ -subunitresponsive phosphoinositide 3-kinase activity in human platelet cytosol. *J. Biol. Chem.*
- **Todaro GJ & Green H. (1963)** Quantitative studies of the growth of mouse embryo cells in culture and their development into established cell lines. *J. Cell. Biol.* **17:** 299-313
- Tornquist K, Ekokoski E & Forss L. (1994) Thapsigargin-induced calcium entry in FRTL-5 cells: Possible dependence on phospholipase A<sub>2</sub> activation. *J. Cell. Physiol.* 160: 40-46
- Touhara K, Hawes BE, van Biesen T & Lefkowitz RJ. (1995) G protein  $\beta\gamma$  subunits stimulate phosphorylation of shc adapter protein. *Proc. Natl. Acad. Sci. USA* 92: 9284-9287
- Toullec D, Pianetti P, Coste H, Bellevergue P, Grand-Perret T, Ajakane M, Baudet V, Boissin P, Boursier E, Loriolle F, Duhamel L, Charon D & Kirilovsky J. (1991) The bisindolylmaleimide GF 109203X is a potent and selective inhibitor of protein kinase C. J. Biol. Chem. 266: 15771-15781

- Toyoshima S, Hirata F, Axelrod J, Beppu M, Osawa T & Waxdal MJ. (1982) The relationship between phospholipid methylation and calcium influx in murine lymphocytes stimulated with native and modified Con A. *Mol. Immunol.* **19**: 229-234
- Troppmair J, Bruder JT, App H, Cai H, Liptak L, Szeberenyi J, Cooper GM & Rapp UR. (1992) Ras controls coupling of growth factor receptors and protein kinase C in the membrane to Raf-1 and B-Raf protein serine kinases in the cytosol. *Oncogene* 7: 1867-1873
- Troppmair J, Bruder JT, Munoz H, Lloyd PA, Kyriakis J, Banerjee P, Avruch J & Rapp UR. (1994) Mitogen-Activated Protein Kinase/Extracellular signal-Regulated protein Kinase activation by oncogenes, serum, and 12-*O*-tetradecanoylphorbol-13-acetate requires Raf and is necessary for transformation. *J. Biol. Chem.* **269**: 7030-7035
- Tsuda T, Kawahara Y, Shii K, Koide M, Ishida Y & Yokoyama M. (1991) Vasoconstrictor-induced protein-tyrosine phosphorylation in cultured vascular smooth muscle cells. *FEBS Lett.* **285:** 44-48
- **Turner CE, Glenney JR Jr. & Burridge K. (1990)** Paxillin: a new vinculin-binding protein present in focal adhesions. *J. Cell Biol.* **111:** 1059-1068
- Turner CE & Miller JT. (1994) Primary sequence of paxillin. J. Cell Sci. 107: 1583-1591
- **Turner CE, Schaller MD & Parsons JT. (1993)** Tyrosine phosphorylation of the focal adhesion kinase pp125<sup>FAK</sup> during development: relation to paxillin. *J. Cell Sci.* **105**: 637-645
- Ueda N, Iniguez-Lluhi JA, Lee E, Smrcka AV, Robishaw JD & Gilman AG. (1994) G protein  $\beta\gamma$  subunits. Simplified purification and properties of novel isoforms. *J. Biol. Chem.* **269**: 4388-4395
- Vaillancourt RR, Gardner AM & Johnson GL. (1994) B-Raf-dependent regulation of the MEK-1/mitogen-activated protein kinase pathway in PC12 cells and regulation by cAMP. *Mol. Cell. Biol.* 14: 6522-6530
- **Valius M, Bazenet C & Kazlauskas A. (1993)** Tyrosines 1021 and 1009 are phosphorylation sites in the carboxy terminus of the platelet-derived growth factor receptor  $\beta$  subunit and are required for binding of phospholipase C $\gamma$  and a 64-kilodalton protein, respectively. *Mol. Cell. Biol.* **13**: 133-143
- Valius M & Kazlauskas A. (1993) Phospholipase C-γ1 and phosphatidylinositol 3 kinase are the downstream mediators of the PDGF receptor's mitogenic signal. *Cell* 73: 321-334
- Valverde AM, Sinnett-Smith J, van Lint J & Rozengurt E. (1994) Molecular cloning and characterisation of protein kinase D: A target for diacylglycerol and phorbol esters with a distinctive catalytic domain. *Proc. Natl. Acad. Sci. USA* 91: 8572-8576
- **van Biesen T, Hawes BE, Luttrell DK, Krueger KM, Touhara K, Porfiri E, Sakaue M, Luttrell LM & Lefkowitz RJ. (1995)** Receptor-tyrosine-kinase- and Gβγ-mediated MAP kinase activation by a common signalling pathway. *Nature* **376:** 781-784

- **van Biesen T, Hawes BE, Raymond JR, Luttrell LM, Koch WJ & Lefkowitz RJ. (1996)** G<sub>0</sub>-protein α-subunits activate mitogen-activated protein kinase via a novel protein kinase C-dependent mechanism. *J. Biol. Chem.* **271:** 1266-1269
- van Corven E, Groenink A, Jalink K, Eichholtz T & Moolenaar WH. (1989) Lysophosphatide-induced cell proliferation: Identification and dissection of signaling pathways mediated by G proteins. *Cell* **59**: 45-54
- Van Corven EJ, Hordijk PL, Medema RH, Bos JL & Moolenaar WH. (1993) Pertussis toxin-sensitive activation of p21<sup>ras</sup> by G protein-coupled receptor agonists in fibroblasts. *Proc. Natl. Acad. Sci USA* **90:** 1257-1261
- Van der Bend R, Bruner J, Jalink K, Van Corven EJ, Moolenaar WE & Van Blitterswijk
  WJ. (1992) Identification of a putative membrane receptor for the bioactive phospholipid, lysophosphatidic acid. *EMBO J.* 11: 2495-2501
- van Dop C, Yamanaka G, Steinberg F, Sekura RD, Manclark CR, Stryer L & Bourne HR. (1984a) ADP-ribosylation of transducin by pertussis toxin blocks the light-stimulated hydrolysis of GTP and cGMP in retinal photoreceptors. *J. Biol. Chem.* 259: 23-26
- van Dop CD, Tsubokawa M, Bourne HR & Ramachandran J. (1984b) Amino acid sequence of retinal transducin at the site ADP-ribosylated by cholera toxin. *J. Biol. Chem.* 259: 696-698
- van Lint J, Sinnett-Smith J & Rozengurt E. (1995) Expression and characterisation of PKD, a phorbol ester and diacylglycerol-stimulated serine protein kinase. *J. Biol. Chem.* **270**: 1455-1461
- van Sande J, Raspe E, Perret J, Lejeune C, Maenhaut C, Vassart G & Dumont JE. (1990) Thyrotropin activates both the cyclic AMP and the PIP<sub>2</sub> cascades in CHO cells expressing the human cDNA of TSH receptor. *Mol. Cell. Endocrinol.* **74:** R1-R6
- Vara F & Rozengurt E. (1985) Stimulation of Na<sup>+</sup>/H<sup>+</sup> antiport activity by epidermal growth factor and insulin occurs without activation of protein kinase C. *Biochem. Biophys. Res. Commun.* 130: 646-653
- Vara F, Schneider JA & Rozengurt E. (1985) Ionic responses rapidly elicited by activation of protein kinase C in quiescent Swiss 3T3 cells. *Proc. Natl. Acad. Sci. USA* 82: 2384-2388
- Vasiliev JM. (1985) Spreading of non-transformed and transformed cells. *Biochem. Biophys. Act.* 780: 21-65
- von Schrenck T, Heinz-Erian P, Moran T, Mantey S, Gardner J & Jensen R. (1989) Neuromedin B receptor in esophagus: evidence for subtypes of bombesin receptors. *Am. J. Physiol.* **256**: G747-G758
- von Tscharner V, Prod'hom B, Baggiolini M & Reuter H. (1986) Ion channels in human neutrophils activated by a rise in free cytosolic calcium concentration. *Nature* 324: 369-372

- Vostal JG & Fratantoni JC. (1993) Econozole inhibits thapsigargin-induced platelet calcium influx by mechanisms other than cytochrome P-450 inhibition. *Biochem. J.* 295: 525-529
- **Vouret-Craviari V, van Obberghen-Schilling E, Scimeca JC, van Obberghen E & Pouyssegur J. (1993)** Differential activation of  $p44^{mapk}$  (ERK1) by  $\alpha$ -thrombin and thrombin-receptor peptide agonist. *Biochem. J.* **289:** 209-214
- Voyno-Yasenetskaya T, Conklin BR, Gilbert RL, Hooley R, Bourne HR & Barber DL. (1994a) Gα13 stimulates Na-H exchange. *J. Biol. Chem.* **269**: 4721-4724
- **Voyno-Yasenetskaya TA, Pace AM & Bourne HR. (1994b)** Mutant α subunits of G12 and G13 proteins induce neoplastic transformation of Rat-1 fibroblasts. *Oncogene* 9: 2559-2565
- Wada E, Way J, Shapira H, Kusano K, Lebacq-Verheyden AM, Coy D, Jenson R & Battey J. (1991) cDNA cloning, characterization, and brain region-specific expression of a neuromedin-B-preferring bombesin receptor. *Neuron* 6: 421-430
- **Wahl M & Gruenstein E. (1993)** Intracellular free  $Ca^{2+}$  in the cell cycle in human fibrobasts: Transitions between  $G_1$  and  $G_0$  and progression into S phase. *Mol. Biol. Cell* **4**: 293-302
- Wall MA, Coleman DE, Lee E, Iniguez-Lluhi JA, Posner BA, Gilman AG & Sprang SR. (1995) The structure of the G protein heterotrimer  $G_{i\alpha 1}\beta_{1\gamma_{2}}$ . *Cell* 83: 1047-1058
- Walsh DA & van Patten SM. (1994) Multiple pathway signal transduction by the cAMPdependent protein kinase. *FASEB J.* 8: 1227-1236
- **Wan Y, Kurosaki T & Huang X-Y. (1996)** Tyrosine kinases in activation of the MAPK cascade by G-protein-coupled receptors. *Nature* **380:** 541-544
- Wang Y & Durkin JP. (1995)  $\alpha$ -amino-3-hydroxy-5-methyl-4-isoxazolepropionic acid, but not *N*-methyl-D-aspartate, activates mitogen activated protein kinase through Gprotein  $\beta\gamma$  subunits in rat cortical neurons. *J. Biol. Chem.* **270**: 22783-22787
- Wang Y, Simonson MS, Pouyssegur J & Dunn MJ. (1992) Endothelin rapidly stimulates mitogen-activated protein kinase activity in rat mesangial cells. *Biochem. J.* 287: 589-594
- Wang Z-W & Rozengurt E. (1983) Interplay of cAMP and microtubules in modulating the initiation of DNA synthesis in 3T3 cells. *J. Cell Biol.* 96: 1743-1750
- Weng Z, Taylor JA, Turner CE, Brugge JS & Seidel-Dugan C. (1993) Detection of Src homology 3-binding proteins, including paxillin, in normal and v-src-transformed Balb/c 3T3 cells. J. Biol. Chem. 268: 14956-14963
- West RE Jr, Moss J, Vaughan M, Liu T & Liu T-Y. (1985) Pertussis toxin-catalyzed ADPribosylation of transducin: Cysteine 347 is the ADP-ribose acceptor site. *J. Biol. Chem.* 260: 14428-14430
- **Westermark B & Heldin C-H. (1991)** Platelet-derived growth factor in autocrine transformation. *Cancer Res.* **51:** 5087-5092

- Wharton J, Polak JM, Bloom SR, Ghatei MA, Solcia E, Brown MR & Pearse AGE. (1978) Bombesin-like immunoreactivity in the lung. *Nature* 273: 769-770
- White MF & Kahn CR. (1994) The insulin signalling system. J. Biol. Chem. 269: 1-4
- Willey JC, Lechner JF & Harris CC. (1984) Bombesin and the C-terminal tetradecapeptide of gastrin-releasing peptide are growth factors for normal human bronchial epithelial cells. *Exp. Cell Res.* 153: 245-248
- Wilson L, Carrier MJ & Kellie S. (1995) pp125<sup>FAK</sup> tyrosine kinase activity is not required for the assembly of F-actin stress fibres and focal adhesions in cultured mouse aortic smooth muscle cells. *J. Cell Sci.* 108: 2381-2319
- Winitz S, Russell M, Qian N-X, Gardner A, Dwyer L & Johnson GL. (1993) Involvement of Ras and Raf in the G<sub>i</sub>-coupled acetylcholine muscarinic m2 receptor activation of mitogen-activated protein (MAP) kinase kinase and MAP kinase. *J. Biol. Chem.* 268: 19196-19199
- Withers DJ, Bloom SR & Rozengurt E. (1995) Dissociation of cAMP-stimulated mitogenesis from activation of the mitogen-activated protein kinase cascade in Swiss 3T3 cells. *J. Biol. Chem.* 270: 21411-21419
- Withers DJ, Coppock HA, Seufferlein T, Smith DM, Bloom SR & Rozengurt E. (1996) Adrenomedullin stimulates DNA synthesis and cell prolferation via elevation of cAMP in Swiss 3T3 cells. *FEBS Lett.* **378:** 83-87
- Woll PJ, Coy DH & Rozengurt E. (1988) [Leu<sup>13</sup>-psi(CH<sub>2</sub>NH)Leu<sup>14</sup>]bombesin is a specific bombesin receptor antagonist in Swiss 3T3 cells. *Biochem. Biophys. Res. Commun.* 155: 359-365
- Woll PJ & Rozengurt E. (1988a) Bombesin and bombesin antagonists: Studies in Swiss 3T3 cells and human small cell lung cancer. *Br. J. Cancer* 57: 579-586
- Woll PJ & Rozengurt E. (1988b) Two classes of antagonist interact with receptors for the mitogenic neuropeptides bombesin, bradykinin and vasopressin. *Growth Factors* 1: 75-83
- Woll PJ & Rozengurt E. (1989) Multiple neuropeptides mobilise calcium in small cell lung cancer: effects of vasopressin, bradykinin, cholecystokinin, galanin and neurotensin. *Biochem. Biophys. Res. Commun.* **164:** 66-73
- Wolstenholme GEW & Knight J. (1971) Editors of: Growth Control in Cell Cultures. *A Ciba Foundation Symposium*. Churchill Livingstone, Edinburgh and London.
- Wong WL, Brostrom MA, Kuznetsov G, Gmitter-Yellen D & Brostrom CO. (1993) Inhibition of protein synthesis and early protein processing by thapsigargin in cultured cells. *Biochem. J.* 289: 71-79
- Wood KW, Sarnecki C, Roberts TM & Blenis J. (1992) *ras* mediates nerve growth factor receptor modulation of three signal-transducing protein kinases: MAP kinase, RAF-1, and RSK. *Cell* 68: 1041-1050

- Wood SM, Wood JR, Ghatei MA, Lee YC, O'Shaughnessy D & Bloom SR. (1981) Bombesin, somatostatin and neurotensin-like immunoreactivity in bronchial carcinoma. J. Clin. Endocrinol. Metab. 53: 1310-1312
- Wu C, Whiteway M, Thomas DY & Leberer E. (1995a) Molecular characterization of Ste20p, a potential mitogen-activated protein or extracellular signal-related kinase kinase (MEK) kinase kinase from *Saccharomyces cerevisiae*. J. Biol. Chem. 270: 15984-15992
- Wu D, Jiang H & Simon MI. (1995b) Different α1-adrenergic receptor sequences required for activating different Gα subunits of Gq class of G proteins. *J. Biol. Chem.* 270: 9828-9832
- Xu N, Bradley L, Ambdukar I & Gutkind JS. (1993) A mutant  $\alpha$  subunit of G<sub>12</sub> potentiates the eicosanoid pathway and is highly oncogenic in NIH 3T3 cells. *Proc. Natl. Acad. Sci. USA* 90: 6741-6745
- Xu Y & Ware JA. (1995) Selective inhibition of thrombin receptor-mediated Ca<sup>2+</sup> entry by protein kinase C. *J. Biol. Chem.* **270:** 23887-23890
- Yan K, Kalyanaraman V & Gautam N. (1996) Differential ability to form the G protein  $\beta\gamma$  complex among members of the  $\beta$  and  $\gamma$  subunit families. *J. Biol. Chem.* 271: 7141-7146
- Yeo E-J & Exton J. (1995) Stimulation of phospholipase D by epidermal growth factor requires protein kinase C activation in Swiss 3T3 cells. *J. Biol. Chem.* 270: 3980-3988
- Zachary I, Gil J, Lehmann W, Sinett-Smith J & Rozengurt E. (1991a) Bombesin, vasopressin and endothelin rapidly stimulate tyrosine phosphorylation in intact Swiss 3T3 cells. *Proc. Natl. Acad. Sci. U.S.A.* 88: 4577-4581
- Zachary I, Millar J, Nanberg E, Higgins T & Rozengurt E. (1987) Inhibition of bombesin-induced mitogenesis by pertussis toxin: Dissociation from phospholipase C pathway. *Biochem. Biophys. Res. Commun.* 146: 456-463
- Zachary I & Rozengurt E. (1985) High-affinity receptors for peptides of the bombesin family in Swiss 3T3 cells. *Proc. Natl. Acad. Sci. USA* 82: 7616-7620
- Zachary I & Rozengurt E. (1987) Identification of a receptor for peptides of the bombesin family in Swiss 3T3 cells by affinity cross-linking. *J. Biol. Chem.* 262: 3947-3950
- Zachary I & Rozengurt E. (1987) Internalization and degradation of peptides of the bombesin family in Swiss 3T3 cells occurs without ligand-induced receptor down-regulation. *EMBO J.* 6: 2233-9
- **Zachary I & Rozengurt E. (1992)** Focal Adhesion Kinase (p125<sup>FAK</sup>): A point of convergence in the action of neuropeptides, integrins, and oncogenes. *Cell* **71**: 891-894
- Zachary I, Sinnett-Smith J & Rozengurt E. (1986) Early events elicited by bombesin and structurally related peptides in quiescent Swiss 3T3 cells. I. Activation of protein

kinase C and inhibition of epidermal growth factor binding. *J. Cell. Biol.* **102**: 2211-2222

- Zachary I, Sinnett-Smith J & Rozengurt E. (1991b) Stimulation of tyrosine kinase activity in anti-phosphotyrosine immune complexes of Swiss 3T3 cell lysates occurs rapidly after addition of bombesin, vasopressin, and endothelin to intact cells. *J. Biol. Chem.* 266: 24126-24133
- Zachary I, Sinnett-Smith J & Rozengurt E. (1992) Bombesin, vasopressin and endothelin stimulation of tyrosine phosphorylation in Swiss 3T3 cells: identification of a novel tyrosine kinase as a major substrate. *J. Biol. Chem.* 267: 19031-19034
- Zachary I, Sinnett-Smith J, Turner CE & Rozengurt E. (1993) Bombesin, vasopressin and endothelin rapidly stimulate tyrosine phosphorylation of the focal adhesion associated protein paxillin in Swiss 3T3 cells. *J. Biol. Chem.* **268**: 22060-22065
- Zhang K, Papageorge AG & Lowy DR. (1992) Mechanistic aspects of signaling through ras in NIH 3T3 cells. *Science* 257: 671-674
- **Zheng C-F, Ohmichi M, Saltiel AR & Guan K-L. (1994)** Growth factor induced MEK activation is primarily mediated by an activator different from c-Raf. *Biochem.* **33**: 5595-5599
- **Zurier RB, Kozma M, Sinnett-Smith J & Rozengurt E. (1988)** Vasoactive intestinal peptide synergistically stimulates DNA synthesis in mouse 3T3 cells: Role of cAMP, Ca<sup>2+</sup> and protein kinase C. *Exp. Cell Res.* **176:** 155-61

### **PUBLICATIONS**

Publications arising from this work are:

- **Charlesworth A & Rozengurt E (1994)** Thapsigargin and di-*tert*-butylhydroqunione induce synergistic stimulation of DNA synthesis with phorbol ester and bombesin in Swiss 3T3 cells. *J. Biol. Chem.* **269**: 32528-32535
- **Charlesworth A, Broad S & Rozengurt E (1996)** The bombesin/GRP receptor transfected into Rat-1 fibroblasts couples to phospholipase C activation, tyrosine phosphorylation of p125<sup>FAK</sup> and paxillin and cell proliferation. *Oncogene* **12**: 1337-1345

### Thapsigargin and Di-*tert*-butylhydroquinone Induce Synergistic Stimulation of DNA Synthesis with Phorbol Ester and Bombesin im Swiss 3T3 Cells\*

(Received for publication, August 15, 1994, and in revised form, October 6, 1994)

### Amanda Charlesworth and Enrique Rozengurt‡

From the Imperial Cancer Research Fund, P. O. Box 123, 44 Lincoln's Inn Fields, London, WC2A 3PX, United Kingdom

The specific inhibitors of the endoplasmic reticulum Caa<sup>2+</sup> pump, thapsigargin and 2,5-di-tert-butylhydroquinome (DBHQ), stimulated reinitiation of DNA synthesis in synergy with either phorbol 12.13-dibutyrate or boymbesin in Swiss 3T3 cells. Maximum stimulation was aclhieved at 0.5 nm thapsigargin and 7.5 µm DBHQ. Kineticss of [<sup>3</sup>H]thymidine incorporation were consistent with exiit from G<sub>0</sub> and entry into S phase. Autoradiography of labeled nuclei showed that the increase in [3H]thymidime incorporation was due to an increase in the proportion of cells entering into DNA synthesis. Down-regulattion or selective inhibition of protein kinase C abiolished this synergistic stimulation of DNA synthesis. Thapsigargin and DBHQ did not potentiate protein kinaise C-mediated signals such as direct phosphorylation of 'myristoylated alanine-rich C-kinase substrate, activaticon of mitogen-activated protein kinase, and tyrosine phosphorylation of bands 110,000-130,000 and 70,000-80),000. Thapsigargin and DBHQ caused a marked reducticon in the ability of bombesin to induce a rapid and transient increase in intracellular Ca<sup>2+</sup> via depletion of tottal cellular Ca<sup>2+</sup>, measured by <sup>45</sup>Ca<sup>2+</sup> content. The synerrgistic stimulation of DNA synthesis by DBHQ and phorbol 12,13-dibutyrate was dependent on a high conceentration of extracellular  $Ca^{2+}$  (ED<sub>50</sub> = 410 µm) and was prreferentially inhibited by the inhibitor of Ca<sup>2+</sup> influx ec:onozole. This suggests a role for Ca<sup>2+</sup> entry in growth countrol. This is the first time that either thapsigargin or DIBHQ has been shown to stimulate the reinitiation of DINA synthesis in any target cell.

Fibroblasts reversibly arrested in the  $G_0$  phase of the cell cyccle can be stimulated to reinitiate cell proliferation through multiple signal transduction pathways that act in a synergistic and combinatorial fashion (1, 2). The possibility that  $Ca^{2+}$ fluxes could play a role in the transduction of mitogenic signals has attracted intense interest. Neuropeptides such as bombesiin, vasopressin, endothelin, and bradykinin, which are potent mitogens for Swiss 3T3 cells (1), induce a rapid and transient increase in the cytoplasmic concentration of  $Ca^{2+} ([Ca^{2+}]_i)^1$  fol-

lowed by a persistent depletion of  $Ca^{2+}$  from stores in the ER (3-6). This effect is mediated by the second messenger  $Ins(1,4,5)P_3$  (7), which is generated in response to receptormediated activation of  $G_a$ , the GTP-binding protein that couples neuropeptide receptors to phospholipase C  $\beta$  (8, 9). Accordingly, transfection of receptors that are constitutively coupled to  $G_{a}(10)$  or of activated  $G_{a}$  mutants (11) causes transformation of certain cells. Also, microinjection of antibodies raised against G<sub>a</sub> blocks the mitogenic activity of bradykinin (12). Furthermore, a bacterial toxin from Pasteurella multocida, which causes a striking activation of inositol phosphate production and Ca<sup>2+</sup> mobilization, also promotes cell proliferation including anchorage-independent growth (13-15). All of these findings suggest, but do not prove, a role for Ca<sup>2+</sup> fluxes in the mitogenic activation of quiescent fibroblasts because hydrolysis of phosphatidylinositol 4,5-bisphosphate also generates diacylglycerol, which activates PKC, a well known pathway leading to gene expression and cell proliferation (16-19). Furthermore, insulin and epidermal growth factor can stimulate DNA synthesis in Swiss 3T3 cells without inducing Ca<sup>2+</sup> mobilization (20, 21). Similarly, mutant receptors for plateletderived growth factor and fibroblast growth factor defective in phospholipase C  $\gamma$  activation have been shown to retain mitogenic activity upon ligand activation (22-25). Thus, the contribution of Ca<sup>2+</sup> to mitogenic signaling remains unclear.

An approach for assessing the role of  $Ca^{2+}$  in the transition from  $G_0$  into DNA synthesis is the use of specific inhibitors of the  $Ca^{2+}$  ATPase pump of the ER, which is responsible for the accumulation of  $Ca^{2+}$  into these stores. The tumor promoter thapsigargin and the structurally distinct compound DBHQ have been identified as inhibitors of  $Ca^{2+}$  accumulation in the ER (26–31). Addition of these compounds to intact cells induces mobilization of  $Ca^{2+}$  from internal stores bypassing phospholipase C-mediated formation of  $Ins(1,4,5)P_3$ . Previous studies have shown that the addition of either thapsigargin or DBHQ, at micromolar concentrations, profoundly inhibited cell proliferation or induced cytotoxic effects (32, 33). These findings are difficult to reconcile with the fact that mitogenic neuropeptides and other growth-promoting factors also cause persistent depletion of  $Ca^{2+}$  from the ER.

The experiments presented here were designed to assess the effect of gradual depletion of  $Ca^{2+}$  from the ER on the ability of quiescent cells to exit from  $G_0$  and enter DNA synthesis.

### EXPERIMENTAL PROCEDURES

Cell Culture—Stock cultures of Swiss 3T3 cells were propagated as previously described (34). For experimental purposes, cells were subcultured in 33- or 100-mm Nunc Petri dishes with DMEM supplemented with 10% FBS and incubated in a humidified atmosphere of  $10\% \text{ CO}_2$ , 90% air at 37 °C until they became confluent and quiescent (6-8 days).

ide gel electrophoresis;  $\mathrm{Me}_2\mathrm{So},$  dimethyl sulfoxide; mAb, monoclonal antibodies.

<sup>\*\*</sup> The costs of publication of this article were defrayed in part by the payment of page charges. This article must therefore be hereby marked *"acdvertisement"* in accordance with 18 U.S.C. Section 1734 solely to indicate this fact.

 $<sup>\</sup>ddagger$  To whom all correspondence should be addressed. Tel.: 71-269-3455; Fax: 71-269-3417.

<sup>&</sup>lt;sup>11</sup> The abbreviations used are: [Ca<sup>2+</sup>], intracellular calcium concentration; DBHQ, di-(*tert*)-butylhydroquinone; DMEM, Dulbecco's modified Eagle's medium; ER, endoplasmic reticulum; [Ca<sup>2+</sup>]<sub>out</sub>, extracellular calcium concentration; FBS, fetal bovine serum; fura-2/AME, fura 2-tetratacetoxymethyl ester; Ins(1,4,5)P<sub>3</sub>, inositol 1,4,5-trisphosphate; MAP kimase, mitogen-activated protein kinase; 80K/MARCKS, myristoylated alanine-rich C-kinase substrate; PDB, phorbol 12,13-dibutyrate; PBS, phosphate-buffered saline; PKC, protein kinase C; PAGE, polyacrylam-

*Pretreatments*—Pretreatments with thapsigargin and DBHQ were carried out in conditioned medium, which is the medium on 6–8-day-old cells that have attained quiescence and is thus essentially devoid of growth-promoting activity.

 $[{}^{3}H]$ Thymidine Incorporation Assay—Determinations of DNA synthesis were performed as previously described (35). Cultures were washed twice with DMEM and incubated with DMEM/Waymouth's medium (1:1 (v/v)) containing  $[{}^{3}H]$ thymidine (0.25 µCi/m], 1 µM) and various additions as described in the figure legends. After 40 h, the cultures were washed twice with PBS and incubated in 5% trichloroacetic acid at 4 °C for 30 min to remove acid-soluble radioactivity. Cultures were washed with ethanol, solubilized in 1 ml of 2% Na<sub>2</sub>CO<sub>3</sub>, 0.1 M NaOH, 1% SDS, and the radioactivity in the acid-insoluble pools was determined by scintillation counting in 6 ml of Ultima Gold (Packard).

Autoradiography of Labeled Nuclei—Labeled nuclei were determined by autoradiography as previously described (35). Cultures were washed twice with DMEM and incubated with DMEM/Waymouth's medium (1:1 (v/v)) containing [<sup>3</sup>H]thymidine (2.5  $\mu$ Ci/ml) and various additions as described in the figure legends. After 40 h, the medium was removed, and the cultures were fixed in formol saline (1:9 (v/v)). The cultures were washed twice with Tris-saline, pH 7.4, at 4 °C, fixed twice (for 5 and 2 min) with 5% trichloroacetic acid at 4 °C, and washed three times with 70% ethanol. Autoradiographic film was put on the cultures and developed after a 2-week exposure. After counterstaining with Giemsa, the labeled nuclei were counted.

Measurement of Total Inositol Phosphates-Total inositol phosphates were determined as previously described (14). Briefly, cultures of Swiss 3T3 cells labeled for 16-18 h with myo-[2-3H]inositol (15 µCi/ml) were washed twice with DMEM/Waymouth's medium (1:1 (v/v)) and incubated in this medium with additions as described in the figure legends. LiCl was added to a final concentration of 20 mm for the last 30 min of the incubation. Inositol phosphates were extracted by replacing the medium with 1 ml of ice-cold 3% HClO4. After 20 min at 4 °C, the extract was neutralized with 0.5 m KOH containing 25 mm HEPES, 5 mm EGTA, and 0.01% phenol red. Precipitated KClO4 was removed by centrifugation. Samples were diluted to 10 ml with water then loaded onto 1 ml of Dowex AG 1-X8 (200-400 mesh, HCOO<sup>-</sup> form) in Bio-Rad Econo-columns. Columns were washed with  $3 \times 10$  ml of H<sub>2</sub>O and  $2 \times 10$  ml of 60 mM NH<sub>4</sub>COOH, 5 mM Na<sub>2</sub>B<sub>4</sub>O<sub>7</sub> and inositol phosphates eluted with 6 ml of 1 M NH<sub>4</sub>COOH, 0.1 M HCOOH. An aliquot (1 ml) of eluate was counted in 10 ml of Ultima Gold.

 ${}^{32}P_i$  Labeling of Swiss 3T3 Cells—Quiescent cultures of cells in 33-mm dishes were washed twice with DMEM without phosphate and labeled overnight in this medium with 50 µCi/ml  ${}^{32}P_i$ . Cells were treated with various factors as indicated then lysed in 250 µl of ice-cold lysis buffer (10 mm Tris/HCl, 5 mm EDTA, 50 mm NaCl, 30 mm sodium pyrophosphate, 50 mm NaF, 100 mm Na<sub>3</sub>VO<sub>4</sub>, 1% Triton X-100, and 1 mm phenylmethylsulfonyl fluoride, pH 7.6).

Immunoprecipitations—For immunoprecipitation of phosphotyrosyl proteins, quiescent cultures of cells in 100-mm dishes were washed twice with DMEM, treated with factors in 5 ml of DMEM as indicated, and lysed at 4 °C in 1 ml of lysis buffer. Lysates of these cells and those of the  $^{32}P_1$ -labeled cells were clarified by centrifugation at 15,000 × g for 10 min. Tyrosine-phosphorylated proteins were immunoprecipitated for 2 h at 4 °C with agarose-linked anti-Tyr(P) as previously described (36). 80K/MARCKS was immunoprecipitated for 3 h at 4 °C with 80K/MARCKS antiserum (37) and then 1 h with agarose-linked protein A. Immunoprecipitates were washed three times with lysis buffer, extracted in 2 × sample buffer (200 mM Tris/HCl, 6% SDS, 2 mM EDTA, 4% 2-mercaptoethanol, 10% glycerol, pH 6.8) and then resolved by one-dimensional SDS-PAGE and analyzed as described in the figure legends.

Western Blotting—After SDS-PAGE, proteins were transferred to Immobilon transfer membranes. Membranes were blocked using 5% nonfat dried milk in PBS, pH 7.2, and incubated for 2 h with either the anti-Tyr(P) mAbs (Py20 and 4G10, 1 µg/ml antibody) or the MAP kinase antiserum, raised against a synthetic peptide to the C-terminal domain (1:1000) in PBS containing 3% non-fat dried milk. Immunoreactive bands were visualized by autoradiography using <sup>125</sup>I-labeled sheep antimouse IgG (1:1000) for anti-Tyr(P) mAbs and <sup>125</sup>I-labeled protein A (1:1000) for the MAP kinase antiserum.

Measurement of Intracellular Calcium— $[Ca^{2+}]_i$  was measured with the fluorescent  $Ca^{2+}$  indicator fura-2/AME using procedures previously described (38). Cells on 100-mm dishes were washed twice with DMEM and incubated for 10 min in 5 ml of DMEM containing 1 µm fura-2/AME freshly prepared from a 1 mm stock maintained in Me<sub>2</sub>SO. Cells were washed twice with PBS, gently scraped and resuspended in 2 ml of electrolyte solution (150 mm NaCl, 0.5 mm KCl, 0.9 mm MgCl<sub>2</sub>, 1.8 mm CaCl<sub>2</sub>, 25 mM glucose, 16 mM HEPES (the same concentration of amino acids as those in DMEM), pH 7.1), then transferred to a 1-cm<sup>2</sup> quartz cuvette, which was kept at 37 °C and stirred continuously throughout the experiment. Fluorescence was monitored in a Perkin-Elmer LS-5 luminescence spectrometer with an excitation wavelength of 336 nm and an emission wavelength of 510 nm.  $[Ca^{2+}]_i$  was calculated after sequential addition of Triton X-100 to 0.05% and EGTA to 100 mM to obtain the maximum  $(F_{max})$  and minimum  $(F_{min})$  fluorescence respectively, using the formula  $[Ca^{2+}]_i$  in nm =  $K (F - F_{min})/(F_{max} - F)$ , where F is the fluorescence at the unknown  $[Ca^{2+}]_i$ , and the value of K is 220 nM for fura-2/AME.

Measurement of Total <sup>45</sup>Ca<sup>2+</sup> Content—Confluent quiescent Swiss 3T3 cells were loaded to equilibrium with 4 µCi of <sup>45</sup>Ca<sup>2+</sup>/ml over 16 h. Cells were then washed twice with DMEM and incubated with factors for 10 min at 37 °C. Cells were rapidly washed with  $2 \times 5$  ml of DMEM with 2 mm EGTA and then  $6 \times 5$  ml of DMEM. Cells were dissolved in 1 ml of 2% Na<sub>2</sub>CO<sub>3</sub>, 0.1 N NaOH, 1% SDS, and the total amount of <sup>45</sup>Ca<sup>2+</sup> in the cells was measured by liquid scintillation counting in 10 ml of Ultima Gold.

Materials—Thapsigargin, GF 109203X, and fura-2/AME were obtained from Calbiochem. DBHQ was obtained from Aldrich. PDB, insulin, bombesin, and econozole were obtained from Sigma. [methyl-<sup>3</sup>H]Thymidine (25 Ci/mmol), myo-[2-<sup>3</sup>H]inositol (16 Ci/mmol), <sup>32</sup>P<sub>1</sub> (10 mCi/ml), <sup>125</sup>I-labeled sheep anti-mouse IgG (20  $\mu$ Ci/µg), <sup>125</sup>I-labeled protein A (30 mCi/mg), and <sup>45</sup>CaCl<sub>2</sub> (10–40 mCi/mg calcium) were obtained from Amersham International (United Kingdom). MAP kinase antiserum was a generous gift of Dr. Johan Van Lint of The Katholieke Universiteit Leuven. Anti-mouse IgG-agarose-linked anti-Tyr(P) mAb clone Py72 was obtained from the hybridoma development unit, Imperial Cancer Research Fund (London). Py20 anti-Tyr(P) mAb was from Tissue Culture Supplies (Buckingham, UK). FBS was obtained from Life Technologies, Inc. All other materials were of the purest grade commercially available.

Stock solutions of thapsigargin were made up at 1 mM in Me<sub>2</sub>SO and stored at -20 °C. DBHQ was dissolved fresh each time to 100 mM in Me<sub>2</sub>SO. Dilutions of compounds in Me<sub>2</sub>SO and Me<sub>2</sub>SO controls were added to the medium at 1 µl/ml. This concentration of Me<sub>2</sub>SO had no measurable effects on our cells.

### RESULTS

Effect of Thapsigargin and DBHQ on PDB-stimulated DNA Synthesis—Initial experiments confirmed that thapsigargin (2-10 nM) inhibited DNA synthesis induced by a variety of growth-promoting factors in Swiss 3T3 cells. However, when the dose dependence of this effect was carefully examined, a stimulatory action was discovered at low concentrations of thapsigargin. This prompted a detailed analysis of the effects of Ca<sup>2+</sup>ATPase pump inhibitors on cellular DNA synthesis.

Quiescent cultures of Swiss 3T3 cells were preincubated with a range of concentrations of DBHQ or thapsigargin for 6 h. The cultures were then exposed to fresh medium containing PDB, the same concentrations of thapsigargin or DBHQ and [3H]thymidine. Cumulative [3H]thymidine incorporation was measured after 40 h of incubation. Fig. 1 shows that both DBHQ and thapsigargin caused a striking stimulation of DNA synthesis in the presence of PDB. The optimum effects were observed at 7.5 им DBHQ and 0.5 nм thapsigargin where they reached 63 and 62%, respectively, of the maximum response induced by FBS. The stimulation of [3H]thymidine incorporation by these agents was induced within a narrow concentration range. At higher concentrations of DBHQ or thapsigargin, cell toxicity (detachment) was clearly evident after 40 h of incubation. DBHQ and thapsigargin added alone did not induce any [3H]thymidine incorporation in Swiss 3T3 cells (Fig. 1).

To examine the extent to which the pretreatment with 0.5 nM thapsigargin and 7.5 µM DBHQ affected the resulting stimulation of [<sup>3</sup>H]thymidine incorporation, confluent quiescent cultures of Swiss 3T3 cells were preincubated with these agents for various times. Cultures were then washed, thapsigargin or DBHQ was replaced, and PDB was added. Cumulative [<sup>3</sup>H]thymidine incorporation was measured after 40 h. The results in



FIG. 1. A and B, dose-response curve for the stimulation of DNA synthesis by DBHQ (A) or thapsigargin (TG) (B). Confluent quiescent cultures of Swiss 3T3 cells were pretreated with various concentrations of DBHQ (A) or thapsigargin (B) for 6 h. The cells were washed twice with DMEM and incubated at 37 °C in 2 ml of 1:1 (v/v) DMEM Waymouth's medium containing 0.25 µCi [3H]thymidine/ml and various concentrations of DBHQ (A) or thapsigargin (B) in presence (closed circles) or absence (open circles) of 80 nm PDB. After 40 h, DNA synthesis was assessed by measuring the level of [3H]thymidine incorporated into half of the acid-precipitable material by liquid scintillation counting. Each point is the mean of two determinations from one representative experiment. 10% FBS gave an incorporation of  $197 \times 10^3$ cpm (A) or  $189 \times 10^3$  cpm (B). Similar results were obtained in 17–22 independent experiments. C and D, effect of the length of preincubation with DBHQ and thapsigargin on the subsequent stimulation of DNA synthesis by these compounds in synergy with PDB. Confluent quiescent cultures of Swiss 3T3 cells were pretreated without (-, open bars) or with (closed bars) 7.5 µM DBHQ (C) or 0.5 nM thapsigargin (D) for 0, 2, 6, or 24 h. Then (as above), the cells were washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with [3H]thymidine, 7.5  $\mu$ M DBHQ (C) or 0.5 nM thapsigargin (D), and 20 nM (C) or 80 nM PDB (D). DNA synthesis is expressed as percent of maximum. Each bar is the mean of 3-8 determinations. Error bars are standard error of the mean (S.E.). E and F, dose-response curve for the stimulation of DNA synthesis by PDB. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6-8 h without (open circles) or with (closed circles)  $7.5 \ \mu M \ DBHQ \ (E) \ or \ 0.5 \ nm \ thapsigargin \ (F).$  Then (as above), the cells were washed twice with DMEM and incubated for 40 h in DMEM/ Waymouth's medium with [3H]thymidine and various concentrations of PDB in the absence (open circles) or presence (closed circles) of 7.5 µm DBHQ (E) or 0.5 nm thapsigargin (F). 10% FBS gave  $164 \times 10^3$  cpm (E) and  $168 \times 10^3$  cpm (F). Similar results were obtained in three independent experiments.

Fig. 1, *C* and *D*, reveal that DBHQ or thapsigargin synergized with PDB in stimulating DNA synthesis without preincubation, but the effect was enhanced after a pretreatment of 6 h. Similar enhancement of DNA synthesis was seen when the cells were preincubated with thapsigargin or DBHQ for 8 h instead of 6 h (data not shown).

To further substantiate the observed synergy between PDB and DBHQ or thapsigargin, we examined the effect of a range of concentrations of PDB in the presence of a fixed and optimum concentration of either DBHQ (7.5  $\mu$ M) or thapsigargin (0.5 nM). The results show striking synergistic stimulation of DNA synthesis by PDB in the presence of either DBHQ or thapsigargin (Fig. 1, *E* and *F*, respectively). Maximum effects were produced at a concentration of 80 nM PDB. PDB alone did not induce any significant [<sup>3</sup>H]thymidine incorporation in Swiss 3T3 cells in agreement with previous results (35, 39). The synergistic stimulation of DNA synthesis by the combination of PDB with either DBHQ or thapsigargin was also assessed as a function of time (Fig. 2A). These compounds stimulated [<sup>3</sup>H]thymidine incorporation after a lag period of 20–22 h. This is consistent with the duration of the transition from  $G_0/G_1$  to DNA synthesis stimulated by neuropeptide growth factors (*e.g.* bombesin, vasopressin) in Swiss 3T3 cells (34, 35).

To assess whether PDB and thapsigargin or DBHQ synergistically interact in eliciting initiation of DNA synthesis rather than in changing the specific activity of the [3H]thymidine precursor pool, quiescent cultures of Swiss 3T3 cells were treated with PDB and thapsigargin or DBHQ, and incorporation of [<sup>3</sup>H]thymidine into DNA was quantified by autoradiography of labeled nuclei. As shown in Fig. 2B, thapsigargin and DBHQ caused a marked enhancement of the labeling index (from 7% in cultures treated with PDB alone to 44 and 54% in cultures treated also with 0.5 nm thapsigargin and 7.5 µM DBHQ, respectively). Thus, PDB in combination with either thapsigargin or DBHQ synergistically enhances the proportion of cells that enter into DNA synthesis. This was in very good agreement with the cumulative [<sup>3</sup>H]thymidine incorporation into parallel cultures expressed as a percentage of the response induced by 10% FBS (Fig. 2C).

The results shown in Figs. 1 and 2 demonstrate that DBHQ and thapsigargin, at low concentrations, promote synergistic stimulation of DNA synthesis in combination with PDB.

Effect of DBHQ with Other Mitogens-Next, we examined whether DBHQ could act synergistically with signaling pathways other than those mediated by PKC. As shown in Table I, the effect of DBHQ on DNA synthesis in the presence of various growth-promoting factors was determined. The polypeptides, epidermal growth factor and insulin, bind to receptors endowed with intrinsic, ligand-dependent, tyrosine kinase activity that are not linked to phospholipase C  $\boldsymbol{\gamma}$  and hence to PKC activation in Swiss 3T3 cells (20, 21). These polypeptides failed to act synergistically with DBHQ. Similarly, the cAMP-elevating agents forskolin and 3-isobutyl-1-methylxanthine, which are potent mitogens for Swiss 3T3 cultures in the presence of insulin (40), did not increase DNA synthesis in the presence of DBHQ. Interestingly, bradykinin, which causes a rapid but transient activation of PKC (41), also failed to act synergistically with DBHQ.

Bombesin, a potent mitogen for Swiss 3T3 cells, induces persistent stimulation of PKC even at concentrations sub-maximal for inducing DNA synthesis. As shown in Table I and Fig. 3, 2.5 nM bombesin (a concentration that was sub-maximal) acted synergistically with DBHQ in promoting DNA synthesis in Swiss 3T3 cells. This synergistic effect was also demonstrated when DNA synthesis was determined by autoradiography of [<sup>3</sup>H]thymidine-labeled nuclei (Fig. 3A). Bombesin (2.5 nM) was almost as effective as PDB in promoting DNA synthesis in the presence of DBHQ. This synergy was not evident when saturating doses of bombesin (10 nM) were used (data not shown). These results strongly suggest that DBHQ-mediated Ca<sup>2+</sup> mobilization acts synergistically with agents that induce persistent activation of PKC. Similar results were obtained when thapsigargin was added instead of DBHQ (data not shown).

The preceding results prompted us to verify that the synergistic effects between DBHQ and either PDB or bombesin on stimulation of DNA synthesis are dependent on PKC. It is known that treatment with saturating doses of PDB (800 nm) causes down-regulation of all phorbol ester-sensitive isoforms of PKC present in Swiss 3T3 cells (42), *i.e.* PKCs  $\alpha$ ,  $\delta$ , and  $\epsilon$  (43). As shown in Fig. 3C, down-regulation of phorbol ester-sensitive PKCs strikingly decreased the [<sup>3</sup>H]thymidine incorporation in-



FIG. 2. A, time course for incorporation of [<sup>3</sup>H]thymidine in the presence of thapsigargin and DBHQ. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6–8 h without (*open circles*) or with 0.5 nm thapsigargin (*squares*) or 7.5  $\mu$ M DBHQ (*closed circles*). Then, the cells were washed twice with DMEM and incubated for various times (from 16 to 44 h) in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine, 80 nm PDB, and in the absence (*open circles*) or presence of 0.5 nm thapsigargin (*squares*) or 7.5  $\mu$ M DBHQ (*closed circles*). A similar lag period prior to DNA synthesis was obtained in three independent experiments. *B* and *C*, effect of thapsigargin and DBHQ on DNA synthesis assessed by autoradiography (*B*) and liquid scintillation counting (*C*) of [<sup>3</sup>H]thymidine incorporation. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6–8 h without (*open bars*) or with (*closed bars*) 0.5 nm thapsigargin or 7.5  $\mu$ M DBHQ. Then, as in Fig. 1, the cells were washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with 2.5  $\mu$ Ci [<sup>3</sup>H]thymidine/ml (*B*) or 0.25  $\mu$ Ci/ml (*C*) and with 20 nm PDB either in the absence (*-*, *open bars*) or presence (*closed bars*) of 0.5 nm thapsigargin (*TG*) or 7.5  $\mu$ M DBHQ. For autoradiography using Kodak autoradiographic stripping plate AR10. Films were then developed *in situ*. *Values* in *B* are the percent of nuclei that were labeled, counted from eight microscopic fields (total of 65–80 cells per field) per dish. 10% FBS labeled 93% of nuclei. *Values* in *C* are expressed as percentage of the values obtained with 10% FBS and the area field. *Values* are the mean of two dishes with the *error bars* showing the range.

#### TABLE I

### Effect of DBHQ on DNA synthesis induced by various mitogens

Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 8 h without (–) or with (+) 7.5  $\mu m$  DBHQ. Then, as in Fig. 1, the cells were washed twice with DMEM and incubated for 40 h in DMEM/ Waymouth's medium with [^3H]thymidine and various mitogens (1  $\mu g/ml$  insulin (INS), 5 ng/ml EGF, 50  $\mu m$  forskolin (FOR), 25  $\mu m$  3-isobutyl-l-methylxanthine (IBMX), 50 nm bradykinin (BK), 2.5 nm bombesin (BOM), and 20 nm PDB) in the presence (+) or absence (–) of 7.5  $\mu m$  DBHQ. Values are expressed as percent of [^3H]thymidine incorporation obtained with 10% FBS and are the mean, with the error shown as S.E. of 4 independent experiments performed in duplicate or 27 experiments for PDB.

Mitogen(s)	[ <sup>3</sup> H]Thymidine incorporation (of FBS)	
	-	+ DBHQ
	%	
-	$1.3 \pm 0.9$	$0.8 \pm 0.5$
INS	$10.4 \pm 3.3$	$17.3 \pm 3.8$
EGF	$6.5 \pm 1.3$	$8.0 \pm 2.5$
FOR + IBMX	$4.2 \pm 2.3$	$4.5 \pm 1.0$
BK	$4.1 \pm 1.3$	8.8 ± 4.3
BOM	$9.1 \pm 3.5$	$36.7 \pm 6.5$
PDB	$7.1 \pm 0.7$	$43.7 \pm 3.5$

duced by DBHQ and PDB or DBHQ and bombesin.

To substantiate the requirement of PKC activity for synergistic stimulation of DNA synthesis by DBHQ and PDB or bombesin, we used the inhibitor GF 109203X. This staurosporine-related compound has been reported to be a selective inhibitor of PKC in Swiss 3T3 cells (44). Pretreatment of Swiss 3T3 cells with 3.5  $\mu$ M GF 109203X for 1 h before addition of PDB or bombesin, conditions that have been previously shown to effectively inhibit PKC activity in intact Swiss 3T3 cells (45), also strikingly decreased the [<sup>3</sup>H]thymidine incorporation induced by DBHQ and PDB or DBHQ and bombesin. The results shown in Fig. 3, *C* and *D*, demonstrate that PKC is necessary for the synergistic stimulation of DNA synthesis induced by DBHQ and either PDB or bombesin.

Thapsigargin and DBHQ Do Not Potentiate PDB-mediated Signal Transduction—Treatment with thapsigargin or DBHQ could enhance PDB-mediated signals or converge with them at a downstream locus further into  $G_1$ . To distinguish between these possibilities, we determined whether prior exposure of Swiss 3T3 cells to 0.5 nM thapsigargin or 7.5 µM DBHQ for 6 h had any effect on PDB-mediated events. Fig. 4 (*upper panel*) shows that preincubation with thapsigargin or DBHQ did not affect the ability of PDB to stimulate the phosphorylation of 80K/MARCKS, a prominent and specific substrate of PKC in intact cells (20). Similarly, treatment with thapsigargin or DBHQ did not affect the activation of MAP kinase by exposure to PDB, as judged by a mobility shift assay (Fig. 4, *middle panel*).

Recently, it has been shown that PDB stimulates the tyrosine phosphorylation of multiple substrates including bands of  $M_{\star}$ 110,000-130,000 and M. 70,000-80,000 (45). The novel cytosolic tyrosine kinase p125<sup>FAK</sup> has been identified as a prominent tyrosine-phosphorylated protein of the  $M_r$  110,000-130,000 band, whereas paxillin corresponds to the  $M_r$  70,000-80,000 heterogenous band (36, 45). As shown in Fig. 4 (lower panel), pretreatment with thapsigargin or DBHQ did not change the basal level of tyrosine phosphorylation or affect the PDB-induced increase in tyrosine phosphorylation of the  $M_r$  110,000– 130,000 or M. 70,000-80,000 bands. In accordance with all of these results, treatment of Swiss 3T3 cells with either DBHQ or thapsigargin did not alter the extent to which [<sup>3</sup>H]PDB bound to intact cells (data not shown). These results indicate that thapsigargin and DBHQ do not directly potentiate PKCmediated signals.

Effect of Thapsigargin and DBHQ on Intracellular Calcium Stores—Thapsigargin inhibits the activity of the ER Ca<sup>2+</sup>ATPase, leading to a depletion of the intracellular Ca<sup>2+</sup> stores (26, 27, 29–31). However, most previous studies on the effects of thapsigargin on cell proliferation and Ca<sup>2+</sup> sequestering in the ER have used high concentrations of this compound (0.1–10 µM). Here, we found synergistic stimulation of DNA synthesis at sub-nanomolar concentrations. Consequently, it was important to examine the effects of low concentrations of thapsigargin and DBHQ on intracellular Ca<sup>2+</sup>. Specifically, if treatment with mitogenic concentrations of thapsigargin or DBHQ inhibits the accumulation of Ca<sup>2+</sup> into the Ins(1,4,5)P<sub>3</sub>sensitive pool, it should prevent the increase in [Ca<sup>2+</sup>]<sub>i</sub> induced by a subsequent addition of Ca<sup>2+</sup>-mobilizing agents.

Fig. 5A shows that treatment with 0.5 nm thapsigargin or 7.5  $\mu$ m DBHQ markedly reduced the ability of 2.5 nm bombesin to induce a rapid and transient increase in the [Ca<sup>2+</sup>], from Swiss

### Ca<sup>2+</sup>ATPase Inhibitors Induce DNA Synthesis



FIG. 3. A and B, effect of DBHQ on DNA synthesis induced by bombesin assessed both by autoradiography (A) and liquid scintillation counting (B) of [<sup>3</sup>H]thymidine incorporation as comparison. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6–8 h without (*open bars*) or with 7.5  $\mu$ M DBHQ (*closed bars*). Cells were then washed twice with DMEM and incubated for 40 h in DMEM/Waymouth's medium with 2.5  $\mu$ C [<sup>3</sup>H]thymidine/ml (A) or 0.25  $\mu$ Ci/ml (B) and with 2.5 nM bombesin in the absence (*open bars*) or presence of 7.5  $\mu$ M DBHQ (*closed bars*). Values in A are the percent of nuclei that were labeled, counted from eight microscopic fields (total of 65–80 cells per field) per dish. 10% FBS labeled 93% of nuclei. Values in B are expressed as mean percentage of the values obtained with 10% FBS with error as S.E. Effect of down-regulating (C) or inhibiting (D) PKC on [<sup>3</sup>H]thymidine incorporation induced by DBHQ and either PDB or bombesin. C, confluent quiescent cultures of Swiss 3T3 cells were treated in the absence (*open bars*) or presence (*striped bars*) of 800 nm PDB for 48 h to down-regulate PKC. Then, the cells were pretreated without (–) or with 7.5  $\mu$ M DBHQ (+). After 6 h, the cells were washed twice with DMEM and incubated in DMEM/Waymouth's medium with [<sup>3</sup>H]thymidine in the absence (*open bars*) or presence of 80 nm PDB or 1 nm bombesin (*Bom*) and without (–) or with (+) 7.5  $\mu$ M DBHQ. *D*, confluent quiescent cultures of Swiss 3T3 cells were pretreated without (–) or with (+) 7.5  $\mu$ M DBHQ. *D*, confluent quiescent cultures of Swiss 3T3 cells were pretreated without (–) or with (+) 7.5  $\mu$ M DBHQ. *D*, confluent quiescent cultures of Swiss 3T3 cells were pretreated without (–) or with (+) 7.5  $\mu$ M DBHQ. *D*, confluent quiescent cultures of Swiss 3T3 cells were pretreated without (–) or with (+) 7.5  $\mu$ M DBHQ. *D*, confluent quiescent cultures of Swiss 3T3 cells were pretreated without (–) or with (+) 7.5  $\mu$ M DBHQ. *D*, confluent quiescent cultures of Swiss 3T3 cells were pretre

3T3 cells. Preincubation with 0.5 nM thapsigargin or 7.5  $\mu$ M DBHQ for 6–8 h inhibited bombesin-induced increase in  $[Ca^{2+}]_i$  by 79 ± 2% (n = 7) and 90 ± 1% (n = 9), respectively (Fig. 5B).

Next, we examined the reduction in bombesin-induced  $Ca^{2+}$ mobilization by thapsigargin or DBHQ as a function of length of pretreatment time. Fig. 5*C* demonstrates that maximum inhibition of the increase in  $[Ca^{2+}]_i$  was reached within 1 h of exposure to thapsigargin or DBHQ. We verified that treatment with these agents did not prevent the production of inositol phosphates in response to bombesin (Table II).

To further demonstrate that the decrease in bombesin-induced Ca<sup>2+</sup> mobilization was due to depletion of Ca<sup>2+</sup> stores, cells were loaded with <sup>45</sup>Ca<sup>2+</sup>, and the total Ca<sup>2+</sup> remaining in the cells was determined after various treatments. This method measures Ca<sup>2+</sup> in intracellular stores, which is accumulated to millimolar concentrations rather than cytoplasmic Ca<sup>2+</sup>, which is maintained at nanomolar concentrations. The results in Fig. 5D demonstrate that 8 h of pretreatment by either 0.5 nmthapsigargin or 7.5 µM DBHQ, conditions that are optimum for DNA synthesis, reduced total cellular <sup>45</sup>Ca<sup>2+</sup> content to a level comparable with that reached after 10 min of exposure to 10 nm bombesin, a maximum concentration. When thapsigargin- or DBHQ-treated cells were further challenged with 10 nm bombesin for 10 min, a small amount of additional <sup>45</sup>Ca<sup>2+</sup> was released, which is in agreement with the measurements of  $[Ca^{2+}]_i$ shown in Fig. 5B. Complete depletion of the Ins(1,4,5)P<sub>3</sub>-responsive pool was obtained by exposure of cells to 30 nm thapsigargin for 30 min since subsequent exposure to bombesin caused no further reduction in cellular <sup>45</sup>Ca<sup>2+</sup>. For comparison, an 8-h treatment with 10 nm bombesin induced a depletion in  $^{45}Ca^{2+}$  content similar to that achieved by an 8-h pretreatment with 0.5 nm thapsigargin. The results in Fig. 5 demonstrate that treatment of Swiss 3T3 cells with thapsigargin or DBHQ, at concentrations that induce synergistic stimulation of DNA synthesis with PDB, causes a partial and persistent depletion in  $Ca^{2+}$  from the Ins(1,4,5)P<sub>3</sub>-responsive Ca<sup>2+</sup> pool.

Effect of Extracellular Calcium on DNA Synthesis—The Ca<sup>2+</sup> response to agents that mobilize Ca<sup>2+</sup> through formation of Ins(1,4,5)P<sub>3</sub> is typically biphasic. The first event is a large transient increase in  $[Ca^{2+}]_i$  originating from intracellular stores, and the second is a sustained influx of Ca<sup>2+</sup> through the plasma membrane, which causes a prolonged elevation of cytoplasmic Ca<sup>2+</sup> (46). Because this sustained Ca<sup>2+</sup> influx through the



FIG. 4. Effect of thapsigargin and DBHQ on PDB-mediated signal transduction. Upper panel, 80K/MARCKS phosphorylation. Confluent quiescent cultures of Swiss 3T3 cells were loaded with <sup>32</sup>P, overnight and then pretreated for 6 h without or with 0.5 nm thapsigargin or 7.5 µM DBHQ as indicated. Cells were incubated 10 min at 37 °C with or without 80 nm PDB as indicated. Cells were lysed and immunoprecipitated with 80K/MARCKS antiserum and resolved by SDS-PAGE. Bands were visualized by autoradiography. Middle panel, MAP kinase activation. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6 h with or without 0.5 nm thapsigargin or 7.5 µm DBHQ as indicated. Cells were then washed twice with DMEM and incubated in DMEM for 10 min at 37 °C without (-) or with 0.5 nm thapsigargin (TG) or 7.5 µM DBHQ in the presence or absence of 80 nM PDB as indicated for 5, 20, or 60 min. Cells were lysed with 2 x sample buffer, and proteins were separated by SDS-PAGE. Proteins were transferred to Immobilon membranes and Western blotted with MAP kinase antiserum. Bands were visualized by autoradiography. Results shown are after 20 min of stimulation. Similar results were obtained when PDB was added for 5 or 60 min instead of 20 min (results not shown). Lower panel, tyrosine phosphorylation of proteins M, 110,000-130,000 and M, 70,000-80,000. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6 h with or without 0.5 nм thapsigargin or 7.5 µм DBHQ as indicated. Then, cells were washed twice with DMEM and incubated in DMEM for 10 min at 37  $^\circ\mathrm{C}$  with or without (-) 0.5 nm thapsigargin or 7.5 µM DBHQ and in the presence or absence of 80 nm PDB as indicated. Cells were lysed and immunoprecipitated with agarose-linked anti-Tyr(P) mAb and resolved by SDS-PAGE. Phosphotyrosyl proteins were transferred to Immobilon membranes and Western blotted with 4G10 and Py20 anti-Tyr(P) mAbs. Bands were visualized by autoradiography.





FIG. 5. Effect of Thapsigargin and DBHQ on hormonal responsive calcium pool in Swiss 3T3 cells. A and B, reduction of bombesin (Bom)-induced  $Ca^{2+}$  mobilization by thapsigargin (TG) and DBHQ. Confluent, quiescent Swiss 3T3 cells in 100-mm Nunc dishes were incubated at 37 °C for 8 h without (-) or with 0.5 nm thapsigargin or 7.5 им DBHQ. The cells were then washed twice with DMEM and incubated with DMEM containing 1 µM fura-2/AME in the absence (-) or presence of 0.5 nм thapsigargin or 7.5 µм DBHQ for 10 min at 37 °C The cells were washed twice with PBS and gently scraped into 2 ml of electrolyte solution containing 0.5 nm thapsigargin or 7.5 µm DBHQ as necessary and transferred to a quartz cuvette, which was kept at 37 °C and stirred continuously throughout the experiment. Fluorescence was monitored. After 1 min control time, bombesin was added to 2.5 nm (shown by *arrow*). Values on the left side of each tracing (A) are  $[Ca^{2+}]_i$ in nm. B shows the mean increase in  $[Ca^{2+}]_i$  induced by 2.5 nm bombesin in control cells (-, open bars) or in cells pretreated (closed bars) with thapsigargin or DBHQ. Errors are presented as S.E. The number of samples are n = 17 (-), n = 7 (TG), n = 9 (DBHQ). C, time course of depletion of bombesin-induced calcium response by thapsigargin and DBHQ. Confluent quiescent Swiss 3T3 cells were incubated at 37 °C for various times with 0.5 nm thapsigargin (squares) or 7.5 µm DBHQ (circles). The cells were then washed twice with DMEM and loaded with fura-2/AME (as above) in the presence of 0.5 nm thapsigargin (squares) or 7.5 µM DBHQ (circles). The cells were then washed twice in PBS and scraped into 2 ml of electrolyte solution containing 0.5 nm thapsigargin (squares) or 7.5 µм DBHQ (circles). After 1 min, 2.5 nm bombesin was added. Values are expressed as a mean percentage of the difference in  $[Ca^{2+}]_i$  induced by 2.5 nm bombesin in untreated cells, with errors as S.E. from 2-3 experiments. D, effect of thapsigargin and DBHQ on total intracellular calcium measured by <sup>45</sup>Ca<sup>2+</sup>. Confluent quiescent Swiss 3T3 cells loaded with <sup>45</sup>Ca<sup>2+</sup> were then pretreated for 8 h without (-) or with 0.5 nm thapsigargin (TG 8h), 7.5 µм DBHQ (DBHQ 8h), 10 nm bombesin (Bom 8h), or for 30 min with 30 nm thapsigargin (TG 30). Cells were then washed twice with DMEM, the same factors added back as for the pretreatment, and then incubated for 10 min at 37 °C in the presence (striped bars) or absence (open bars) of 10 nm bombesin. In the case of bombesin pretreatment, due to down-regulation of the bombesin response, further stimulation was by 10 min of 20 nM vasopressin (reverse striped bar). Residual  $\rm ^{45}Ca^{2+}$  in the cells was measured as described under "Experimental Procedures." Values are mean of three determinations with errors shown as S.E.

plasma membrane is dependent on extracellular  $Ca^{2+}$ , we examined the effect of extracellular  $Ca^{2+}$  concentration on DNA synthesis induced by a variety of growth-promoting factors. The results in Fig. 6A show that DBHQ and PDB-induced [<sup>3</sup>H]thymidine incorporation is strikingly sensitive to extracellular  $Ca^{2+}$ . The concentration of  $Ca^{2+}$  in the medium ( $[Ca^{2+}]_{out}$ ) that is required to achieve 50% of the maximum response (ED<sub>50</sub>) induced by PDB and DBHQ was 410 µM. A similar result was obtained for thapsigargin (data not shown.) In sharp contrast, the ED<sub>50</sub> for PDB and insulin, a combination that induces DNA synthesis but does not stimulate  $Ca^{2+}$  mobilization, was

#### TABLE II

### Effect of thapsigargin on total inositol phosphates generated by various mitogens

Confluent quiescent cultures of Swiss 3T3 cells were equilibrated overnight with 15  $\mu$ Ci/ml myo-[2-<sup>3</sup>H]inositol and then pretreated for 8 h without (-) or with 0.5 nM thapsigargin (+ TG). Cells were washed twice and incubated in DMEM/Waymouth's for 10 min at 37 °C with either 20 nM PDB or 2.5 or 10 nM bombesin (BOM) in the absence (-) or presence of 0.5 nM thapsigargin (+ TG). Total inositol phosphates were extracted and collected on Dowex AG 1-X8 columns, HCOO<sup>-</sup> form, and eluted with 1 m NH<sub>4</sub>COOH, 0.1 m HCOOH as described under "Experimental Procedures." Values are expressed as cpm. The experiment was performed in triplicate, with the error shown as S.E.

Mitogen	F Brit- , beth	+ TG
their militial in foreits	cpm	cpm
distant of the second second second	$209 \pm 21$	$263 \pm 26$
20 nm PDB	194 ± 6	$199 \pm 10$
2.5 nм ВОМ	$482 \pm 21$	$560 \pm 15$
10 nм ВОМ	822 ± 21	$800 \pm 48$



FIG. 6. A, effect of extracellular calcium on [<sup>3</sup>H]thymidine incorporation. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6–8 h with 7.5 µm DBHQ (closed circles). Cells were then washed twice with DMEM and incubated in DMEM/Waymouth's medium containing [<sup>3</sup>H]thymidine with various amounts of Ca<sup>2+</sup> buffered by 2 mm EGTA in the presence of 80 nm PDB (open circles), 80 nm PDB, and 7.5 µm DBHQ (closed circles), or 80 nm PDB and 1 µg/ml insulin (closed triangles). Similar results were obtained in four independent experiments. B, effect of econozole on [<sup>3</sup>H]thymidine incorporation induced by different mitogens. Confluent quiescent cultures of Swiss 3T3 cells were pretreated for 6 h with or without 7.5 µm DBHQ. Cells were then washed twice with DMEM and incubated in DMEM/Waymouth's medium containing [<sup>3</sup>H]thymidine. Cultures were stimulated with either 80 nm PDB (-), 80 nm PDB, and 7.5 µm DBHQ or with 80 nm PDB and 1 µg/ml insulin (*INS*) in the absence (open bars) or presence (striped bars) of 1 µm econozole.

6 µM. To substantiate the involvement of  $Ca^{2+}$  influx in the synergistic stimulation of DNA synthesis, we used the inhibitor econozole. This compound inhibits  $Ca^{2+}$  influx in several cell types (47–50) through a mechanism independent of cytochrome P-450 inhibition (49). The results in Fig. 6B show that DNA synthesis stimulated by PDB and DBHQ was reduced by 55% in the presence of 1 µM econozole (a concentration that has been shown to inhibit capacitative  $Ca^{2+}$  influx (47)). In contrast, DNA synthesis induced by PDB and insulin was only reduced by 13%. The results in Fig. 6 demonstrate that the synergistic stimulation of DNA synthesis by DBHQ with PDB requires capacitative  $Ca^{2+}$  influx.

### DISCUSSION

The findings presented here demonstrate that thapsigargin and DBHQ synergistically stimulate DNA synthesis with PDB or bombesin in cultures of Swiss 3T3 cells. At the low concentrations that induce DNA synthesis, thapsigargin and DBHQ cause a slow but persistent depletion of the intracellular Ca<sup>2+</sup> stores. To our knowledge, this is the first time that either thapsigargin or DBHQ has been shown to stimulate the reinitiation of DNA synthesis in any target cell.

It is recognized that depletion of ER  $Ca^{2+}$  stores, e.g. by thapsigargin, DBHQ, or neuropeptides, activates Ca<sup>2+</sup> influx from the extracellular medium of many cell types (46). This has been termed capacitative  $Ca^{2+}$  entry (51) and is thought to be activated by a diffusable messenger released from the ER when the  $Ca^{2+}$  is depleted (48, 52, 53). How does depletion of intracellular Ca<sup>2+</sup> stores act in signaling DNA synthesis? Two alternative mechanisms could be envisaged. The diffusable messenger released from the depleted Ca<sup>2+</sup> stores could, as well as activating  $Ca^{2+}$  entry, interact with other signaling components to promote synergistic stimulation of cell cycle entry from G<sub>0</sub>. Alternatively, capacitative Ca<sup>2+</sup> influx may lead to a persistent increase in  $[Ca^{2+}]$ . We attempted to distinguish between these possibilities. If the stimulation of DNA synthesis elicited by thapsigargin and DBHQ depends on Ca2+ influx, we reasoned that DNA synthesis should be critically dependent on [Ca<sup>2+</sup>]<sub>out</sub>. Indeed, this was the case for the mitogenic combination of DBHQ and PDB. The  $ED_{50}$  of  $[Ca^{2+}]_{out}$  required to support DNA synthesis stimulated by DBHQ and PDB was 100-fold higher than that required for DNA synthesis induced by insulin and PDB. The high [Ca<sup>2+</sup>]<sub>out</sub> in DBHQ-treated cells is consistent with the apparent  $K_m$  (in the millimolar range) of the channel that mediates capacitative Ca<sup>2+</sup> influx (54). In support of this, econozole, which inhibits Ca<sup>2+</sup> influx (47-50), inhibited DNA synthesis induced by PDB and DBHQ to a much greater extent than DNA synthesis induced by PDB and insulin. Taking these data together, we conclude that the induction of DNA synthesis by DBHQ and thapsigargin in the presence of PDB is dependent on the activity of the capacitative Ca<sup>2+</sup> influx pathway.

In theory, treatment with thapsigargin or DBHQ leading to persistent depletion of  $Ca^{2*}$  stores could enhance the ability of PDB to induce cellular responses through PKC. Several lines of evidence indicate that this is not the case. Specifically, treatment with either thapsigargin or DBHQ did not enhance the ability of PDB to induce a variety of responses such as direct PKC-mediated phosphorylation of 80K/MARCKS, stimulation of the kinase cascade leading to MAP kinase activation and tyrosine phosphorylation of multiple bands previously identified as p125<sup>FAK</sup> and paxillin (36, 45). We conclude that  $Ca^{2+}$  and PKC signals act in parallel and converge downstream to induce synergistic stimulation of DNA synthesis.

Our results contrast with previous studies demonstrating that thapsigargin and DBHQ profoundly inhibit cell proliferation or induce cytotoxic effects (32, 33). Although different cell types were used, an obvious difference between the present and previous studies is the concentration of thapsigargin employed. In fact, we can readily obtain inhibition of DNA synthesis and cytotoxic effects in Swiss 3T3 cells at higher concentrations of thapsigargin and DBHQ. The existence of stimulatory and inhibitory effects could explain the bell-shaped dose response of these agents on DNA synthesis found in this study. The inhibitory effects are likely to be caused by extensive depletion of  $Ca^{2+}$  from the ER, thereby interfering with its function. The high Ca<sup>2+</sup> content in the lumen of the ER is thought to play a critical role in the post-translational processing and trafficking of newly synthesized secretory, lysosomal, and membrane proteins (55-57). Furthermore, Ca<sup>2+</sup> sequestered in the ER appears to be required for maintaining a normal rate of protein synthesis (58, 59). The disruption of these important processes by extensive depletion of ER Ca<sup>2+</sup> could explain the inhibition of cell proliferation induced by thapsigargin and DBHQ at higher concentrations than those used in the present study. We hypothesize that thapsigargin and DBHQ, at optimum mitogenic concentrations, reduce the Ca<sup>2+</sup> content from internal stores in Swiss 3T3 cells to a critical level that leads to entry into DNA synthesis but does not impair essential ER functions.

In conclusion, the results presented here demonstrate for the first time that thapsigargin and DBHQ cause stimulation of DNA synthesis in synergy with phorbol ester and bombesin. Synergistic induction of DNA synthesis by these  $Ca^{2+}$ -mobilizing agents requires a functional capacitative  $Ca^{2+}$  influx pathway. Thus, it is plausible that a persistent depletion of  $Ca^{2+}$  from internal stores, as induced by mitogenic neuropeptides and growth factors, may play a role as one of the synergistic signals that contribute to stimulating the transition from  $G_0$  to DNA synthesis.

#### REFERENCES

- 1. Rozengurt, E. (1986) Science 234, 161-166
- 2. Rozengurt, E. (1992) Curr. Opin. Cell Biol. 4, 161-165
- Mendoza, S. A., Schneider, J. A., Lopez-Rivas, A., Sinnett-Smith, J., and Rozengurt, E. (1986) J. Cell Biol. 102, 2223–2233
- Lopez-Rivas, A., and Rozengurt, E. (1984) Am. J. Physiol. 247, C156-C162
   Lopez-Rivas, A., Mendoza, S. A., Nanberg, E., Sinnett-Smith, J., and Rozengurt, E. (1987) Proc. Natl. Acad. Sci. U. S. A. 84, 5768-5772
- 6. Nanberg, E., and Rozengurt, E. (1988) EMBO J. 7, 2741-2747
- 7. Berridge, M. J. (1993) Nature 361, 315-325
- Lee, C. H., Park, D., Wu, D., Rhee, S.-G., and Simon, M. I. (1992) J. Biol. Chem. 267, 16044–16047
- Berstein, G., Blank, J. L., Jhon, D.-Y., Exton, J. H., Rhee, S. G., and Ross, E. M. (1992) Cell 70, 411–418
- Allen, L. F., Lefkowitz, R. J., Caron, M. G., and Cotecchia, S. (1991) Proc. Natl. Acad. Sci. U. S. A. 88, 11354–11358
   K. K. M. S. M. Sci. L. S. (1992) Proc. Natl. Acad. Sci. U. S. A. 88, 11354–11358
- Kalinec, G., Nazarali, A. J., Hermouet, S., Xu, N., and Gutkind, J. S. (1992) Mol. Cell. Biol. 12, 4687–4693
   LaMorte, V. J., Harootunian, A. T., Spiegel, A. M., Tsien, R. Y., and Feramisco,
- J. R. (1993) J. Cell Biol. 121, 91–99
- Rozengurt, E., Higgins, T., Chanter, N., Lax, A. J., and Staddon, J. M. (1990) Proc. Natl. Acad. Sci. U. S. A. 87, 123–127
- Murphy, A. C., and Rozengurt, E. (1992) J. Biol. Chem. 267, 25296–25303
   Higgins, T. E., Murphy, A. C., Staddon, J. M., Lax, A. J., and Rozengurt, E.
- (1992) Proc. Natl. Acad. Sci. U. S. A. 89, 4240–4244
   Rozengurt, E., and Sinnett-Smith, J. (1988) Prog. Nucleic Acid Res. Mol. Biol.
- 35, 261–295
   Rozengurt, E., Rodriguez-Pena, A., Coombs, M., and Sinnett-Smith, J. (1984) Proc. Natl. Acad. Sci. U. S. A. 81, 5748–5752
- 18. Nishizuka, Y. (1986) Science 233, 305–312
- 19. Nishizuka, Y. (1988) Nature 334, 661–665
- Rozengurt, E., Rodriguez-Pena, A., and Smith, K. A. (1983) Proc. Natl. Acad. Sci. U. S. A. 80, 7224–7248
- 21. Vara, F., and Rozengurt, E. (1985) Biochem. Biophys. Res. Commun. 130, 646-653
- 22. Valius, M., and Kazlauskas, A. (1993) Cell 73, 321-334
- Valius, M., Bazenet, C., and Kazlauskas, A. (1993) Mol. Cell. Biol. 13, 133-143
   Mohammadi, M., Dionne, C. A., Li, W., Li, N., Spivak, T., Honegger, A. M., Jone, M. and Schlargier, J. (1909) Nature 254 621 624
- Jaye, M., and Schlessinger, J. (1992) Nature 358, 681–684
   25. Peters, K. G., Marie, J., Wilson, E., Ives, H. E., Escobedo, J., Del Rosario, M., Mirda, D., and Williams, L. T. (1992) Nature 358, 678–681
- Thastrup, O., Dawson, A. P., Scharff, O., Foder, B., Cullen, P. J., Drobak, B. K., Bjerrum, P. J., Christensen, S. B., and Hanley, M. R. (1989) Agents Actions 27, 17–23
- Thastrup, O., Cullen, P. J., Drobak, B. K., Hanley, M. R., and Dawson, A. P. (1990) Proc. Natl. Acad. Sci. U. S. A. 87, 2466-2470
- Moore, G. A., McConkey, D. J., Kass, G. E. N., O'Brien, P. J., and Orrenius, S. (1987) FEBS Lett. 224, 331–336
- Takemura, H., Hughes, A. R., Thastrup, O., and Putney, J. W., Jr. (1989) J. Biol. Chem. 264, 12266-12271
- 30. Sagara, Y., and Inesi, G. (1991) J. Biol. Chem. 266, 13503-13506
- Bian, J., Ghosh, T. K., Wang, J.-C., and Gill, D. L. (1991) J. Biol. Chem. 266, 8801–8806
- Ghosh, T. K., Bian, J., Short, A. D., Rybak, S. L., and Gill, D. L. (1991) J. Biol. Chem. 266, 24690–24697
- Short, A. D., Bian, J., Ghosh, T. K., Waldron, R. T., Rybak, S. L., and Gill, D. L. (1993) Proc. Natl. Acad. Sci. U. S. A. 90, 4986–4990
- Rozengurt, E., and Sinnett-Smith, J. (1983) Proc. Natl. Acad. Sci. U. S. A. 80, 2936–2940
- 35. Dicker, P., and Rozengurt, E. (1980) Nature 287, 607-612
- Zachary, I., Sinnett-Smith, J., Turner, C. E., and Rozengurt, E. (1993) J. Biol. Chem. 268, 22060–22065
   Brooks, S. F., Herget, T., Erusalimsky, J. D., and Rozengurt, E. (1991) EMBO
- J. 10, 2497–2505
- Millar, J. B. A., and Rozengurt, E. (1990) J. Biol. Chem. 265, 12052–12058
   Dicker, P., and Rozengurt, E. (1978) Nature 276, 723–726
- Rozengurt, E., Legg, A., Strang, G., and Courtenay-Luck, N. (1981) Proc. Natl. Acad. Sci. U. S. A. 78, 4392–4396
- 41. Issandou, M., and Rozengurt, E. (1990) J. Biol. Chem. 265, 11890-11896
- Rodriguez-Pena, A., and Rozengurt, E. (1984) Biochem. Biophys. Res. Commun. 120, 1053-1059
- 43. Olivier, A. R., and Parker, P. J. (1992) J. Cell. Physiol. 152, 240-244
- Toullec, D., Pianetti, P., Coste, H., Bellevergue, P., Grand-Perret, T., Ajakane, M., Baudet, V., Boissin, P., Boursier, E., Loriolle, F., Duhamel, L., Charon, D., and Kirilovsky, J. (1991) J. Biol. Chem. 266, 15771-15781
- Sinnett-Smith, J., Zachary, I., Valverde, A. M., and Rozengurt, E. (1993) J. Biol. Chem. 268, 14261-14268

- Putney, J. W., Jr., and Bird, G. St. J. (1993) Endocr. Rev. 14, 610–631
   Alvarez, J., Montero, M., and Garcia-Sancho, J. (1991) Biochem. J. 274, 193– 197
- Randriamampita, C., and Tsien, R. Y. (1993) Nature. 364, 809–814
   Vostal, J. G., and Fratantoni, J. C. (1993) Biochem. J. 295, 525–529
- 50. Breittmayer, J.-P., Berthe, P., Cousin, J.-L., Bernard, A., and Aussel, C. (1993)
- Cell. Immunol. 152, 143–151 51. Putney, J. W., Jr. (1986) Cell Calcium 7, 1–12 52. Montero, M., Alvarez, J., and Garcia-Sancho, J. (1992) Biochem. J. 288, 519– 525
- Parekh, A. B., Terlau, H., and Stuhmer, W. (1993) Nature 364, 814–818
   Mauger, J.-P., Poggioli, J., Guesdon, F., and Claret, M. (1984) Biochem. J. 221, 121–127 55. Kuznetsov, G., Brostrom, M. A., and Brostrom, C. O. (1992) J. Biol. Chem. 267,
- 3932-3939
- Sob2-350
   Sambrook, J. F. (1990) Cell 61, 197–199
   Schutzbach, J. S., and Forsee, W. T. (1990) J. Biol. Chem. 265, 2546–2549
   Wong, W. L., Brostrom, M. A., Kuznetsov, G., Gmitter-Yellen, D., and Brostrom, C. O. (1993) Biochem. J. 289, 71–79
   Brostrom, C. O., and Brostrom, M. A. (1990) Annu. Rev. Physiol. 52, 577–590

# The bombesin/GRP receptor transfected into Rat-1 fibroblasts couples to phospholipase C activation, tyrosine phosphorylation of $p125^{FAK}$ and paxillin and cell proliferation

Amanda Charlesworth, Simon Broad and Enrique Rozengurt

Imperial Cancer Research Fund, PO Box 123, 44 Lincoln's Inn Fields, London, WC2A 3PX, UK

Bombesin elicits multiple signalling pathways in various cell types. It is not clear, however, whether these responses are mediated by a single receptor subtype or by different subtypes that couple preferentially to specific pathways. To resolve this we transfected the mouse bombesin/GRP receptor into Rat-1 fibroblasts and investigated the pathways activated by bombesin. Expression of the transfected receptors was verified by binding of [125I]GRP and two clones were selected, BOR5 and BOR15. Bombesin stimulation of BOR5 and BOR15 cells caused intracellular Ca<sup>2+</sup> mobilisation and increased the phosphorylation of 80K/MARCKS, a prominent protein kinase C substrate. The transfected receptor conferred a proliferative response to bombesin demonstrated by incorporation of [3H]thymidine after 18 h and an increase in total cell numbers after 1-2days. In BOR5 and BOR15 cells, bombesin rapidly stimulated the tyrosine phosphorylation of multiple proteins Mr 110 000-130 000 and 70 000-80 000 including p125FAK and paxillin, at low concentrations (half maximum 0.3 nM). The specific bombesin/GRP receptor antagonist, D-F5-Phe6, D-Ala11-Bombesin(6-13)OMe, inhibited all the above responses. These results show that phospholipase C activation, cell growth and tyrosine phosphorylation emanate from a single class of bombesin receptor.

**Keywords:** neuropeptides; growth factors;  $Ca^{2+}$  mobilisation; protein kinase C

### Introduction

Neuropeptides stimulate DNA synthesis and cell proliferation in cultured cells and are implicated as cellular growth factors in a variety of fundamental biological processes including development, tissue regeneration, inflammation and tumourigenesis (Rozengurt, 1986; Zachary *et al.*, 1987; Rozengurt, 1995a). In particular, the peptides of the bombesin family including Gastrin Releasing Peptide (GRP) are potent mitogens for Swiss 3T3 cells (Rozengurt and Sinnett-Smith, 1983; Zachary and Rozengurt, 1985) and various tumour cell types including small cell lung cancer (Cuttitta *et al.*, 1985; Carney *et al.*, 1987; Sethi *et al.*, 1991). Bombesin binds to specific high affinity receptors on the cell surface (Zachary and Rozengurt, 1985, 1987a; Battey *et al.*, 1991) and elicits a variety of early

Correspondence: E Rozengurt

biochemical responses including rapid phospholipase C (PLC) mediated hydrolysis of polyphosphoinositides leading to mobilisation of intracellular calcium and activation of protein kinase C (PKC) (Zachary *et al.*, 1986; Nanberg and Rozengurt, 1988; Rozengurt, 1992).

Tyrosine phosphorylation has been implicated in the intracellular signalling of bombesin and other neuropeptides (Rozengurt, 1995b). Bombesin stimulates tyrosine phosphorylation of multiple proteins in Swiss 3T3 cells, including broad bands migrating with the apparent molecular masses of 110 000-130 000 and 70 000-80 000 (Zachary et al., 1991a,b). More recently the focal adhesion associated proteins p125 focal adhesion kinase (p125<sup>FAK</sup>) and paxillin have been identified as prominent tyrosine phosphorylated proteins in Swiss 3T3 cells after stimulation by bombesin (Zachary et al., 1992, 1993; Sinnett-Smith et al., 1993). p125FAK is a novel cytosolic tyrosine kinase that lacks Src Homology 2 (SH2) and SH3 domains (Hanks et al., 1992; Schaller et al., 1992) but associates with signalling proteins including pp60<sup>src</sup> (Cobb et al., 1994) and paxillin (Hildebrand et al., 1995). Recent molecular cloning of paxillin revealed a multidomain protein that may function as an adaptor capable of associating with p125FAK, Crk, Src and vinculin (Turner and Miller, 1994; Salgia et al., 1995). Bombesin induces tyrosine phosphorylation of p125FAK and paxillin via a PKC and Ca<sup>2+</sup>-independent pathway that requires the integrity of the actin cytoskeleton (Sinnett-Smith et al., 1993; Zachary et al., 1993) and functional p21<sup>tho</sup> (Rankin et al., 1994; Seckl et al., 1995). Tyrphostin, an inhibitor of tyrosine kinases, markedly reduces the level of p125<sup>FAK</sup> tyrosine phosphorylation and DNA synthesis induced by bombesin in Swiss 3T3 cells suggesting that the tyrosine kinase pathway plays a role in signalling the mitogenic response elicited by bombesin in these cells (Seckl and Rozengurt, 1993).

At present three mammalian members of the bombesin-like receptor family have been cloned and shown to belong to the G protein-linked receptor superfamily. They are the bombesin/GRP-preferring (Spindel *et al.*, 1990; Battey *et al.*, 1991), the neuromedin B (NmB)-preferring (Corjay *et al.*, 1991; Wada *et al.*, 1991) and BRS-3 (Gorbulev *et al.*, 1992; Fathi *et al.*, 1993), a bombesin-like receptor whose physiological ligand is not yet known. It has been generally assumed that all the early signals and the subsequent mitogenic response elicited by bombesin are mediated by occupancy of the bombesin/GRP receptor. This assumption has recently been questioned. Bold *et al.* (1994) reported that antagonists that inhibit bombesin-induced Ca<sup>2+</sup> mobilisation do not prevent

Received 24 August 1995; revised 13 November 1995; accepted 13 December 1995

bombesin-induced cell growth in human gastric adenocarcinoma cells. However tyrphostin blocked bombesin-induced proliferation of cells. Bold *et al.* (1994). suggested that a bombesin receptor subtype couples to  $Ca^{2+}$  mobilisation and a different bombesin receptor subtype, not yet identified, couples to the protein tyrosine phosphorylation pathway and cell growth. However, no direct evidence supporting this hypothesis has been provided.

The experiments presented here were designed to elucidate whether the bombesin/GRP-preferring receptor could couple to both PLC activation as well as to p125<sup>FAK</sup> and paxillin tyrosine phosphorylation. Initially, we show that expression of the bombesin/GRP preferring receptor in Rat-1 cells, a cell line that can be reversibly arrested in the  $G_0/G_1$  phase of the cell cycle, confers a proliferative response to bombesin in the receptor transfected cells. Using this model system, we found that agonist activation of the bombesin/GRP receptor elicits both PLC activation as well as tyrosine phosphorylation of multiple proteins including p125<sup>FAK</sup> and paxillin. Our results demonstrate, for the first time, that the bombesin/GRP receptor couples to both signalling pathways and to cell proliferation.

### Results

### Expression of the bombesin GRP receptor in Rat-1 cells

The mouse bombesin/GRP-preferring receptor cDNA cloned into the mammalian expression vector pCD2, and under the transcriptional control of the SV40 early region promoter, was used for transfection experiments. Rat-1 fibroblasts were chosen as the recipient cell line because these cells can be reversibly arrested in the  $G_0/G_1$  phase of the cell cycle (Higgins *et al.*, 1992) and do not express endogenous functional bombesin/GRP receptors (see below). After transfection, 21 G418 resistant colonies were isolated and propagated; six of these cell lines expressed the bombesin/GRP-preferring receptor compared with the control, untransfected Rat-1 cells.

The functional expression of bombesin/GRP receptors in the transfected Rat-1 cells was verified using <sup>125</sup>I-labelled GRP. Figure 1A shows that the transfectants have specific binding sites for [125I]GRP. Parental Rat-1 cells did not specifically bind [125I]GRP. The apparent affinity of [125I]GRP for the transfected receptor in BOR 5 and BOR 15 cells ( $K_d = 1 -$ 2 nM) was similar to the endogenous receptor expressed in Swiss 3T3 cells ( $K_d = 0.8$  nM). The number of receptors on the transfected cells was 30-60% lower than that of Swiss 3T3 cells, about 10<sup>5</sup> per cell (Zachary and Rozengurt, 1985), under identical experimental conditions (data not shown). When BOR 5 cells were incubated with [125I]GRP, cell associated radioactivity increased rapidly reaching a maximum value after 30 min and decreasing afterwards (Figure 1B). This is consistent with internalisation and degradation of the peptide as reported previously in Swiss 3T3 cells (Zachary and Rozengurt, 1987b) and in transformed rat kidney cells (Grady et al., 1995).

Addition of unlabelled GRP, bombesin, litorin, the structurally related neuromedin B (NmB), or the bombesin C-terminal fragment (amino acids 8-14)

inhibited specific [<sup>125</sup>I]GRP binding to BOR 5 cells in a concentration dependent manner (Figure 1C). The IC<sub>50</sub> values for bombesin, GRP and litorin were approximately 1-3 nM. In contrast NmB inhibited binding with much lower affinity (IC<sub>50</sub> = 500 nM). Figure 1 shows that BOR 5 has specific binding sites with high affinity for bombesin/GRP and low affinity for NmB demonstrating that the transfected receptor in Rat-1 cells showed the characteristic ligand recognition of the bombesin/GRP-preferring receptor.

### The transfected bombesin/GRP receptor functionally couples to PLC-activated signalling pathways

In order to verify that the transfected bombesin/GRP receptor expressed in Rat-1 cells could couple to PLC, we measured the effect of bombesin on the intracellular calcium concentration ( $[Ca^{2+}]$ ) and on the phosphor-



Figure 1 (A) Specific [125I]GRP binding to different clones. Confluent, quiescent cultures of transfected and parental (WT) Rat-1 cells were washed twice with DMEM and incubated for 30 min at 37°C in DMEM containing [1251]GRP (0.05 nm, 0.01 µCi and 0.95 nm cold GRP) and 1 mg/ml BSA. Cultures were then placed on ice and rapidly washed 4 × with PBS containing 1 mg ml BSA. Cultures were dissolved in 0.1 N NaOH/2% NaCO3/1% SDS and counted in a gamma counter. Results are expressed as specific binding per  $10^6$  cells. Non-specific binding was determined in the presence of 1 µM bombesin. Specific binding was 60-80% of the total binding. (B) Time-course of [1251]GRP binding to BOR 5 cells. Confluent, quiescent cultures of BOR 5 cells were washed twice with DMEM and incubated at 37°C in DMEM with [125I]GRP (as above) in the presence of 1 mg/ml BSA (closed circles). Non-specific binding was determined by adding  $1\,\mu M$  bombesin (open circles). The binding of [<sup>125</sup>I]GRP to BOR 5 cells was terminated after various times as indicated. (C) Competition binding of [<sup>125</sup>I]GRP to BOR 5 cells. Confluent, quiescent cultures of BOR 5 cells were washed twice with DMEM and incubated for 30 min at 37°C in DMEM with [125I]GRP (as above) and various concentrations of competing peptides in the presence of 1 mg/ml BSA. Peptides were bombesin (open squares), GRP (open circles), litorin (open triangles), NmB (closed squares) and bombesin fragment 8-14 (closed circles). Cultures were then placed on ice and washed and counted as above. Results are expressed as a percentage of the total specific binding of  $[^{125}I]GRP$ 

ylation of the prominent PKC substrate 80K/ MARCKS (Myristylated Alanine Rich C Kinase Substrate) (Rodriguez-Pena and Rozengurt, 1986; Brooks *et al.*, 1991).

Bombesin increased  $[Ca^{2+}]_i$  in a concentration dependent manner in BOR 5 cells (Figure 2A). The maximum increase, elicited by 100 nM bombesin, was 4.0-fold over the basal level (186 nM  $\pm$  5.5, n = 62). In contrast bombesin caused no increase in  $[Ca^{2+}]_i$  in parental Rat-1 cells even up to 1  $\mu$ M (data not shown). Addition of the specific bombesin/GRP receptor antagonist, D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe (Coy *et al.*, 1992), inhibited bombesin-induced increases in  $[Ca^{2+}]_i$  (Figure 2B) but had no effect on the increase in  $[Ca^{2+}]_i$  induced by endothelin which binds to receptors endogenously expressed in Rat-1 cells. Studies of  $Ca^{2+}$  mobilisation with cultures of BOR 15 cells produced exactly the same results (data not shown).

Addition of bombesin to cultures of BOR 5 cells prelabelled with  ${}^{32}P_i$  markedly increased the phosphorylation of the prominent PKC substrate, 80K/MARCKS, in a concentration dependent manner (Figure 2C). The level of phosphorylation was

comparable to that induced by 200 nM PDB, a direct activator of PKC. In contrast 80K/MARCKS was not phosphorylated in response to bombesin (up to 100 nM) in parental Rat-1 cells. The specific PKC inhibitor GF 109203X (Toullec *et al.*, 1991) prevented bombesin stimulation of 80K/MARCKS phosphorylation substantiating the role of PKC in the action of bombesin, (data not shown). The results in Figure 2 therefore show that the transfected bombesin/GRP receptor is functionally coupled to intracellular signalling pathways initiated by PLC activation.

## Expression of the bombesin/GRP receptor confers a growth response to bombesin in Rat-1 cells

Next, we examined whether agonist activation of the transfected bombesin/GRP receptor leads to reinitiation of DNA synthesis in Rat-1 cells. Quiescent cultures of BOR 5 cells were incubated with bombesin and at various times after stimulation the cells were labelled with [<sup>3</sup>H]thymidine for 3 h. The rate of [<sup>3</sup>H]thymidine incorporation reached a maximum between 18-24 h after the addition of bombesin (Figure 3A). This is consistent with previous results



Figure 2 (A and B) Bombesin-induced increase in  $[Ca^{2+}]_i$  in BOR 5 cells. Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and detached from the dishes by a 10 min incubation in DMEM with 2.5 µg/ml dispase. Cells were loaded with 1 µM fura-2/AME while in suspension. Loaded cells were pelleted and resuspended in 2 ml electrolyte solution and transferred to a quartz cuvette. Cells were kept at 37°C and stirred continuously while fluorescence was monitored. After a 1 min control period different concentrations of bombesin were added. (A). Results are expressed as the difference in  $[Ca^{2+}]_i$  between unstimulated and stimulated levels. (B). After 3 min of monitoring fluorescence bombesin was added to 1 nM (Bn) and at 5 min endothelin was added to 50 nM (En). At 1 min the bombesin antagonist D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6–13)OMe was added to 10 nM as indicated (Atg). Shown are the traces from one representative experiment with  $[Ca^{2+}]_i$  in nM on right. (C). Phosphorylation of 80K/MARCKS. Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM without phosphate and incubated in this medium with 5µCi carrier-free <sup>32</sup>P<sub>i</sub> at 37°C for 18 h. Various concentrations of bombesin (0.1 nM, 1 nM, 10 nM) or 200 nM PDB were added for 10 min. Cultures were put on ice and rapidly washed with cold DMEM without phosphate and lysed with 250 µl lysis buffer. 80K/MARCKS

of bombesin-induced DNA synthesis in Swiss 3T3 cells (Rozengurt and Sinnett-Smith, 1983).

DNA synthesis induced by bombesin in cultures of BOR 5 cells was inhibited by the specific bombesin/ GRP receptor antagonist, D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6–13)OMe, (Figure 3B) with half maximum and maximum inhibition at 1 nM and 30 nM respectively. When BOR 5 cultures were incubated with various concentrations of bombesin, maximum [<sup>3</sup>H]thymidine incorporation was elicited at 1 nM (data not shown). These results substantiate that bombesin-stimulated reinitation of DNA synthesis in BOR 5 cells was elicited through a bombesin/GRP preferring receptor.

The mitogenic activity of bombesin in cultures of BOR 5 cells could also be demonstrated by monitoring cell number (rather than [<sup>3</sup>H]thymidine incorporation into acid precipitable material) over a period of several days (Figure 3C). The addition of bombesin to BOR 5 in serum-free medium increased cell number compared with control cultures as early as day 1 of incubation and reached a maximum after 3 days of bombesin addition. In other experiments we also verified that bombesin induced DNA synthesis and cell proliferation in cultures of BOR 15 cells. Thus, the transfected bombesin/GRP receptor is coupled to signal tranduction pathways which stimulate DNA synthesis and subsequently cell proliferation.

# Bombesin induces tyrosine phosphorylation of multiple proteins in BOR 5 and BOR 15 cells

In view of the preceding results we examined whether bombesin can stimulate tyrosine phosphorylation in Rat-1 cells transfected with the bombesin/GRP receptor. The experiments were performed with cultures of both BOR 5 and BOR 15 cells. Lysates of quiescent cultures of BOR 5 and BOR 15 cells treated with bombesin were immunoprecipitated with the antiphosphotyrosine (anti-Tyr(P)) monoclonal antibody (mAb) Py72 and the immunoprecipitates were analysed by Western blotting with anti-Tyr(P) mAb 4G10. As shown in Figure 4, bombesin induced a dosedependent increase in the phosphotyrosine content of multiple components including bands migrating with an apparent  $M_r$  of 110 000-130 000 and 70 000-80 000. Scanning densitometry of these bands revealed that half maximum and maximum tyrosine phosphorylation was achieved at bombesin concentrations of 0.1-0.3 nM and 1-3 nM, respectively. In contrast, parental Rat-1 cells did not show any increase in tyrosine phosphorylation in response to bombesin (100 nM) in these and the following experiments (data not shown).

The time course of tyrosine phosphorylation induced by bombesin in Rat-1 cells transfected with the bombesin/GRP receptor is shown in Figure 5. The increase in tyrosine phosphorylation of the  $M_r$ 110 000–130 000 and the  $M_r$  70 000–80 000 bands was detectable within seconds and reached a maximum after 1 min of incubation with 10 nM bombesin in cultures of both BOR 5 and BOR 15 cells. Increased protein tyrosine phosphorylation persisted for at least 3 h.

To further substantiate the involvement of the bombesin/GRP receptor in tyrosine phosphorylation we used the bombesin/GRP receptor-specific antagonist D-F<sub>3</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6–13)OMe. We have already shown that this antagonist inhibits bombesin-induced Ca<sup>2+</sup> mobilisation, 80K/MARCKS phosphorylation (Figure 2B) and DNA synthesis (Figure 3B). We therefore examined whether it can also inhibit bombesin-induced tyrosine phosphoryla-



**Figure 3** (A) Kinetics of  $[{}^{3}H]$ thymidine incorporation induced by bombesin. Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at 37°C in DMEM/Waymouth's [1:1] with (solid bars) or without (open bars) 2 nm bombesin. After the times indicated below each bar  $[{}^{3}H]$ thymidine was added for 3 h and the incorporation was stopped as described in Materials and methods. Results are expressed as the mean c.p.m. from two dishes from a representative experiment. (B) Effect of antagonist on bombesin-induced  $[{}^{3}H]$ thymidine incorporation. Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at 37°C in DMEM/Waymouth's [1:1] with various concentrations of antagonist D-F<sub>5</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6–13)OMe in the presence (closed squares) or absence (open squares) of 1 nM bombesin. [ ${}^{3}H$ thymidine was added after 18 h and its incorporation was stopped after 6 h of additional incubation as described in Materials and methods. Results are expressed as the mean c.p.m. from two dishes from a representative experiment. (C). Cell proliferation induced by bombesin. Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at 37°C DMEM/Waymouth's [1:1] with (closed squares) or without (open squares) 1 nM bombesin. (C). Cell proliferation induced by bombesin. Confluent, quiescent cultures of BOR 5 cells were washed twice in DMEM and incubated at 37°C DMEM/Waymouth's [1:1] with (closed squares) or without (open squares) 1 nM bombesin. Cells were trypsinised off the dishes and counted with a Coulter counter than the symbol



Figure 4 Bombesin-induced tyrosine phosphorylation of multiple proteins. Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at  $37^{\circ}$ C in DMEM with various concentrations of bombesin for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with anti Tyr(P) mAb Py72. Immunoprecipitates were resolved by SDS-PAGE and Western blotted with 4G10 anti-Tyr(P) mAb. Bands were visualised by autoradiography. Autoradiograms were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area. The 70000-80000 band is depicted by closed circles and the 110000-130000 band by open circles. Results shown are the average of three independent experiments. Values are expressed as a percentage of the maximum tyrosine phosphorylation after subtraction of the control

tion. Cultures of BOR 5 and BOR 15 cells were pretreated with the bombesin/GRP receptor-specific antagonist for 10 min then stimulated by bombesin or endothelin and cell lysates were analysed for tyrosine phosphorylated proteins. Figure 6A shows that the antagonist completely blocked bombesin-induced tyrosine phosphorylation of the  $M_r$  110 000–130 000 and the  $M_r$  70 000–80 000 bands. In contrast, the antagonist did not interfere with the stimulation of tyrosine phosphorylation of similar bands induced by endothelin.

It has been shown that bombesin induces tyrosine phosphorylation in Swiss 3T3 cells through a PKCindependent pathway that requires the integrity of the actin cytoskeleton (Sinnett-Smith *et al.*, 1993). We examined whether the stimulation of tyrosine phosphorylation by agonist activation of the transfected bombesin/GRP receptor exhibit similar signalling characteristics. We verified that protein tyrosine phosphorylation was not downstream of PKC. Although the selective PKC inhibitor bisindolylmaleimide (GF 109203X) (Toullec *et al.*, 1991) did prevent



Figure 5 Kinetics of bombesin induction of tyrosine phosphorylation of multiple proteins. Upper Section. Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with 10 nM bombesin for various times. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 mAb. Immunoprecipitates were resolved by SDS-PAGE and Western blotted with 4G10 anti-Tyr(P) mAb. Lower section. Autoradiograms were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area. Results shown are of the 70000–80000 band from a representative experiment of BOR 15 which was repeated at least three times with similar results. Values expressed as a percentage of the maximal tyrosine phosphorylation after subtraction of the control

PDB-induced tyrosine phosphorylation of M<sub>r</sub> 110 000-130 000 and 70 000-80 000 bands, it did not prevent bombesin-induced tyrosine phosphorylation in the transfected Rat-1 cells (data not shown). We also examined the effect of cytochalasin D, which selectively disrupts the network of actin filaments, on bombesin-induced tyrosine phosphorylation. Figure 6B shows that pretreating BOR 5 and BOR 15 cultures with cytochalasin D for 2 h, dramatically inhibited the tyrosine phosphorylation of all bands stimulated by bombesin. In contrast, cytochalasin D did not inhibit EGF-induced tyrosine phosphorylation of its receptor (data not shown) indicating that this drug specifically inhibits the tyrosine phosphorylation of bands M<sub>r</sub> 110 000-130 000 and 70 000-80 000. Thus, bombesin-induced tyrosine phosphorylation in BOR 5 and BOR 15 cells is dependent on the integrity of the actin cytoskeleton.

### Bombesin stimulates tyrosine phosphorylation of $p125^{FAK}$ and paxillin in BOR 5 and BOR 15 cells

The cytosolic tyrosine kinase p125FAK and the adaptor protein paxillin which localise to focal adhesion plaques have been identified as prominent tyrosine phosphorylated proteins in bombesin-stimulated Swiss 3T3 cells. It was important to therefore assess whether agonist activation of the transfected bombesin/GRP receptor stimulates tyrosine phosphorylation of p125FAK and paxillin in Rat-1 cells. Lysates of bombesin-treated cultures of BOR 5 and BOR 15 were immunoprecipitated with anti-Tyr(P) mAb and the immunoprecipitates were analysed by Western blotting with anti-p125FAK mAb. Figure 7 shows that bombesin induced tyrosine phosphorylation of p125FAK in a concentration dependent manner; half maximum and maximum effects were achieved at 0.3 nM and 3 nM respectively.

Parallel cultures of BOR 5 and BOR15 cells treated with various concentrations of bombesin were lysed and the lysates were immunoprecipitated with antipaxillin mAb. The immunoprecipitates were analysed by Western blotting with anti-Tyr(P) mAb. Figure 7 shows that bombesin caused tyrosine phosphorylation of paxillin in a concentration dependent manner. The concentrations of bombesin that induced half-maximum and maximum stimulation were identical to those required to induce tyrosine phosphorylation of p125<sup>FAK</sup> (i.e. 0.3 and 3 nM respectively).



Figure 6 Left side. Effect of antagonist on bombesin-induced tyrosine phosphorylation of multiple proteins. Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with (+) or without (-) 100 nM D-F5-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe (Ant) for 10 min. Then 10 nm bombesin (Bn) or 50 nm endothelin (En) was added as indicated for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 mAb and Western blotted with 4G10 anti-Tyr(P) mAb. Bands were visualised by autoradiography. Right side. Effect of Cytochalasin D on bombesin-induced tyrosine phosphorylation of multiple proteins. Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with (+) or without (-)  $2 \mu M$  cytochalasin D (Cyt D) for 2h. Then 10 nm bombesin (Bn) was added as indicated for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 mAb and Western blotted with 4G10 anti-Tyr(P) mAb. Bands were visualised by autoradiography



Figure 7 Bombesin-induced tyrosine phosphorylation of  $p125^{FAK}$  and paxillin. Upper section,  $p125^{FAK}$ . Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with various concentrations of bombesin for 10 min. Cells were lysed and tyrosine phosphorylated proteins immunoprecipitated with Py72 mAb. Immunoprecipitates were resolved by SDS-PAGE and Western blotted with anti-p125<sup>FAK</sup> mAb. Paxillin. Confluent, quiescent cultures of BOR 5 and BOR 15 cells were washed twice in DMEM and incubated at 37°C in DMEM with various concentrations of bombesin for 10 min. Cells were lysed and paxillin immunoprecipitated with anti-paxillin mAb 165 and Western blotted with 4G10 anti-Tyr(P) mAb. Lower section. Autoradiograms were scanned with an LKB Ultrascan XL densitometer to quantify tyrosine phosphorylation in terms of peak area. BOR 5 is represented by squares and BOR 15 by circles. Results shown are from a representative experiment which was repeated at least three times with similar results. Values expressed as a percentage of the maximum tyrosine phosphorylation after subtraction of the control

### Discussion

The experiments presented here were designed to determine whether the bombesin/GRP preferring receptor couples to both PLC activation and tyrosine kinase pathway. To examine this possibility, the cDNA encoding the bombesin/GRP receptor has been expressed in Rat-1 fibroblasts, a cell line that can be reversibly arrested in the  $G_0/G_1$  phase of the cell cycle. The relative abilities of bombesin related peptides to displace <sup>125</sup>I-GRP in these transfected Rat-1 cells are consistent with the binding properties of endogenous bombesin/GRP receptors found in Swiss 3T3 fibroblasts and other cell types. Furthermore the apparent affinity of GRP for the transfected receptor is similar to that for the endogenous receptor. It should be pointed out that the Rat-1 cell lines transfected with the bombesin/GRP receptor used in this study (BOR 5 and BOR 15) expressed fewer receptors per cell than Swiss 3T3 cells. This is of importance in our analysis of cellular responses because overexpression of G protein linked receptors could overcome the specificity of coupling to effector systems (Ashkenazi et al., 1987).

Thus, the clones selected in this study, BOR 5 and BOR 15, provide a useful system to examine the signalling pathways activated by the bombesin/GRP receptor.

Our results demonstrate that agonist binding to the transfected bombesin/GRP receptor elicits multiple responses in Rat-1 cells: (1) mobilisation of Ca<sup>2+</sup> from internal stores leading to a rapid increase in  $[Ca^{2+}]_i$  and increased phosphorylation of the 80K/MARCKS protein mediated by PKC, (2) reinitiation of DNA synthesis and cell proliferation in serum free medium and (3) tyrosine phosphorylation of multiple proteins including broad bands of M<sub>r</sub> 110 000-130 000 and 70 000-90 000. We identified two major substrates in the transfected cells as  $p125^{FAK}$  and paxillin. None of these responses were elicited by bombesin in the parental Rat-1 cells.

The characteristics of protein tyrosine phosphorylation induced by agonist binding to the bombesin/GRP receptor transfected into Rat-1 cells were examined in detail in two clones, BOR 5 and BOR 15. An increase in the tyrosine phosphorylation of multiple proteins including p125<sup>FAK</sup> and paxillin occurred at low concentrations of bombesin, the half maximum being 0.3 nM. This is similar to that observed in Swiss 3T3 cells (half maximum 0.08-0.3 nM) (Zachary et al., 1992; Sinnett-Smith et al., 1993). Both Swiss 3T3 cells (Zachary et al., 1991a) and transfected Rat-1 cells show similar time dependence of tyrosine phosphorylation, increases are detected within seconds of bombesin addition, reach a maximum at 1 min and remain elevated for hours. Bombesin stimulation of tyrosine phosphorylation in BOR 5 and BOR 15 cells, as in Swiss 3T3 cells, is not prevented by the specific PKC inhibitor GF 109203X showing that this pathway is not downstream of PKC. p125FAK and paxillin are localised in the focal adhesions which form at the termini of actin stress fibres and disruption of the actin cytoskeleton by cytochalasin D prevents tyrosine phosphorylation of these proteins in Swiss 3T3 cells (Sinnett-Smith et al., 1993; Zachary et al., 1993). In BOR 5 and BOR 15 cells bombesin-induced tyrosine phosphorylation of multiple proteins was also prevented by cytochalasin D. Therefore the protein tyrosine phosphorylation pathway activated by the transfected bombesin/GPR receptor in Rat-1 cells shows the same characteristics as that induced by bombesin stimulation of Swiss 3T3 cells.

In the transfected Rat-1 cells the increases in  $Ca^{2+}$  mobilisation, protein tyrosine phosphorylation and DNA synthesis were inhibited by the specific bombesin/GRP receptor antagonist, D-F<sub>3</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6–13)OMe (Coy *et al.*, 1992), further substantiating that all these responses emanate from the same receptor. This is the first report demonstrating that ligand activation of a transfected bombesin/GRP receptor elicits multiple signalling pathways and leads to cell proliferation.

The results presented here raise important questions regarding the mechanism(s) by which a single seven transmembrane receptor subtype can couple to multiple signalling pathways. While the mechanism by which bombesin/GRP receptor and other seven-transmembrane domain receptors couple to PLC through heterotrimeric G proteins of the  $G_q$  subfamily is increasingly understood (Lee and Rhee, 1995), little is

known about the pathways coupling the bombesin/ GRP receptor to tyrosine phosphorylation of  $p125^{FAK}$ and paxillin. Recent work has implicated activation of the monomeric G protein of the  $p21^{rho}$  subfamily as one step in the signalling pathway leading to tyrosine phosphorylation of these proteins (Rankin *et al.*, 1994; Seckl *et al.*, 1995). Thus, our results suggest that the bombesin/GRP receptor couples to both heterotrimeric (G<sub>q</sub>) and monomeric (p21<sup>rho</sup>) G proteins.

The demonstration that certain seven transmembrane domain receptors couple to  $p21^{ras}$  via  $\beta y$  subunits of the heterotrimeric G protein of the G<sub>i</sub> subfamily (Koch et al., 1994) suggests a possible coupling mechanism between the bombesin/GRP receptor and p21<sup>tho</sup>. By analogy, the  $\beta\gamma$  subunits generated by activation of  $G_q/G_{11}$  could be preferentially coupled to activation of p21<sup>rho</sup> and thereby to the tyrosine phosphorylation pathway whereas the  $\alpha$  subunit of  $G_{\alpha}$ stimulates PLC (Lee and Rhee, 1995). Alternatively, one domain of the bombesin/GRP receptor may couple to  $G_a$  and thereby to PLC whereas a separate domain could lead to activation of p21<sup>tho</sup> and to the tyrosine phosphorylation of p125FAK and paxillin through interaction with a different heterotrimeric G-protein. In this context the recent demonstration that  $G\alpha_{12}$  and  $G\alpha_{13}$ , but not  $G\alpha_{q}$ , regulate Rho-dependent actin polymerisation is highly relevant (Buhl et al., 1995). This suggests that the bombesin/GRP receptor could couple to Rho and tyrosine phosphorylation of p125FAK and paxillin via different heterotrimeric Gproteins than those which couple it to PLC. Further experimental work will be required to distinguish between these models of signal transduction.

In conclusion, our results provide compelling evidence that ligand activation of the transfected bombesin/GRP receptor activates tyrosine phosphorylation of  $p125^{FAK}$ , paxillin and other proteins as well as couple to PLC-mediated events and induce cell proliferation.

### Materials and methods

### Transfection of Rat-1 cells

A 1.4 kb insert containing the complete cDNA coding region of the murine bombesin/GRP preferring receptor cloned into the mammalian expression vector pCD2-neo (BNR-pCD2-neo) and under the transcriptional control of the SV40 early region promoter, was kindly provided by Dr Jim Battey (Laboratory of Biological Chemistry, National Cancer Institute, Bethesda, Maryland, USA). For transfection, Rat-1 cells were plated at a density of 1  $\times$  10<sup>6</sup> cells/100 mm dish. The following day, cultures were washed twice in Dulbecco's Modified Eagle's Medium (DMEM) and 20 µg DNA from BNR-pCD2-neo expression vector, mixed with 40  $\mu$ g lipofectin reagent, was added to the cultures in 5 ml DMEM at 37°C. After 14 h incubation, 5 ml of 20% foetal bovine serum (FBS) in DMEM were added and incubation extended for a further day. Then, cultures were trypsinised, divided into four and selected in DMEM supplemented with 1 mg/ml neomycin derivate G418. The medium was changed every 3-4 days. After 2-3 weeks, resistant colonies were cloned by ring isolation and propagated in DMEM supplemented with 5% FBS and 0.5 mg/ml G418.



### Cell culture

Stock cultures of Rat-1 cells were maintained as described previously (Higgins *et al.*, 1992) except that transfected cell lines were propagated in the presence of 0.5 mg/ml G418. For experimental purposes,  $5 \times 10^4$  or  $3 \times 10^5$  cells were subcultured in 33 or 100 mm Nunc Petri dishes respectively with DMEM supplemented with 5% FBS and 0.5 mg/ml G418 as necessary and incubated in a humidified atmosphere of 10% CO<sub>2</sub>, 90% air at 37°C until they became 50-70% confluent (2-3 days), then switched down to 0.5% FBS for 3-4 days until they were quiescent.

### [<sup>125</sup>I]GRP binding assay

Confluent, quiescent, cultures of Rat-1 cells in 33 mm dishes were washed twice with DMEM and incubated for the indicated times at 37°C in DMEM with [125I]GRP (0.05 nM, 0.01  $\mu$ Ci per ml and 0.95 nM cold GRP) in the presence of 1 mg/ml bovine serum albumin (BSA). For the competition studies various concentrations of competing peptides (bombesin, GRP, litorin, NmB and bombesin fragment 8-14) were also added to the binding medium. Non-specific binding was determined by adding  $1 \, \mu M$ bombesin. Cultures were then placed on ice and rapidly washed four times with ice cold phosphate buffered saline (PBS) containing 1 mg/ml BSA. Cultures were dissolved in 0.1 N NaOH/2% NaCO<sub>3</sub>/1% SDS and counted in a gamma counter. Results are expressed as specific binding per 106 cells. Specific binding in the transfected cells was 60-80%of the total binding. No statistically significant binding was oberved in untransfected cells in this assay which is able to detect as little as 200 receptors per cell.

### Measurement of $[Ca^{2+}]_i$

 $[Ca^{2+}]_i$  was measured with the fluorescent  $Ca^{2+}$  indicator fura-2/acetoxymethylester (fura-2/AME). Confluent, quiescent, cultures of transfected Rat-1 cells in 100 mm dishes were washed twice with DMEM and incubated for 10 min at 37°C in 2.5 mg/ml Dispase/DMEM to digest the intercellular matrix. Cells  $(4-5 \times 10^6)$  in single cell suspension were then incubated at 37°C, 10 min, in 5 ml of DMEM containing 1  $\mu$ M fura-2/AME freshly prepared from a 1 mM stock maintained in dimethylsulphoxide. Cells were collected by centrifugation and resuspended in 2 ml electrolyte solution (150 mM NaCl, 0.5 mM KCl, 0.9 mM MgCl<sub>2</sub>, 1.8 mM CaCl<sub>2</sub>, 25 mM glucose, 16 mM HEPES, the same amino acid composition as DMEM, pH 7.1) and transferred to a 1 cm<sup>2</sup> quartz cuvette which was maintained at 37°C and stirred continuously throughout the experiment. Fluorescence was monitored in a Perkin-Elmer LS-5 luminescence spectrometer with an excitation wavelength of 336 nm and emission wavelength of 510 nm. Intracellular calcium concentration was calculated by sequential addition of Triton X-100 to 0.05% and EGTA to 100 mM to obtain the maximum  $(F_{max})$  and minimum  $(F_{min})$  fluorescence respectively, using the formula  $[Ca^{2+}]_i$  in  $nM = K(F - F_{min})/(F_{max} - F)_i$ , where F is the fluorescence at the unknown  $[Ca^{2+}]_i$  and the value of K is 220 nM for fura2/AME.

### <sup>32</sup>P<sub>i</sub> labelling of Swiss 3T3 cells

Confluent, quiescent, cultures of cells in 33 mm dishes were washed twice with DMEM without  $P_i$  and labelled overnight in this medium with 50  $\mu$ Ci/ml <sup>32</sup> $P_i$ . Cells were treated with various factors as indicated then lysed in 250  $\mu$ l of ice cold lysis buffer (10 mM Tris/HCl, 5 mM EDTA, 50 mM NaCl, 30 mM sodium pyrophosphate, 50 mM NaF, 100  $\mu$ M Na<sub>3</sub>VO<sub>4</sub>, 1% Triton X-100 and 1 mM phenylmethylsulphonyl fluoride, pH 7.6).

### *Immunoprecipitations*

For immunoprecipitation of phosphotyrosyl proteins and paxillin, confluent, quiescent, cultures of Rat-1 cells in 33 mm dishes  $(2.5 \times 10^6$  cells total) were washed twice with DMEM, treated with factors in 5 ml of DMEM as indicated and lysed at 4°C in 1 ml lysis buffer. Lysates of these cells and those of the <sup>32</sup>P<sub>i</sub>-labelled cells were clarified by centrifugation at 15 000 g for 10 min. For immunoprecipitation of tyrosine phosphorylated proteins or paxillin, clarified lysates were incubated with anti-Tyr(P) mAb Py72 for 4 h at 4°C or with anti-paxillin mAb 165 for 2 h at 4°C as indicated. These immune complexes were precipitated with anti-mouse IgG (whole molecule)-agarose for 1 h at 4°C. For immunoprecipitation of 80K/MARCKS clarified lysates were incubated for 3 h at 4°C with 80K/MARCKS antiserum (Brooks et al., 1991) then 1 h with Protein A agarose. Immunoprecipitates were washed three times with lysis buffer, extracted in 2  $\times$  sample buffer (200 mM Tris-HCl, 6% SDS, 2 mM EDTA, 4% 2-mercaptoethanol, 10% glycerol, pH 6.8) and then resolved by one-dimensional SDS-polyacrylamide gel electrophoresis (SDS-PAGE) and analysed as described in the figure legends.

### [<sup>3</sup>H]thymidine incorporation assay

Confluent, quiescent, cultures of Rat-1 cells in 33 mm dishes were washed twice with DMEM and incubated at 37°C with DMEM/Waymouth's medium [1:1(vol/vol)] and various additions as described in the figure legends. After times specified [<sup>3</sup>H]thymidine (0.25  $\mu$ Ci/ml, 1  $\mu$ M) was added for 3 h. Cultures were then washed twice with PBS and incubated in 5% trichloroacetic acid at 4°C for 30 min to remove acid-soluble radioactivity. Cultures were washed with ethanol, solubilised in 1 ml of 2% Na<sub>2</sub>CO<sub>3</sub>, 0.1 M NaOH, 1% SDS and the radioactivity in the acid-insoluble pools determined by scintillation counting in 6 ml Ultima Gold (Packard).

### Measurement of cell proliferation

Confluent, quiescent, cultures of Rat-1 cells in 33 mm dishes were washed twice with DMEM and incubated at 37°C in DMEM/Waymouth's with additions as specified in the figure legends. After various times cell number was determined by removing the cells from the dish with a trypsin solution (0.05% trypsin in  $Ca^{2+}$  and  $Mg^{2+}$  free PBS with EDTA) and counting a portion of the resulting cell suspension in a Coulter counter.

### Western blotting

After SDS-PAGE, proteins were transferred to Immobilon transfer membranes. Membranes were blocked using 5% non-fat dried milk in PBS, pH 7.2, and incubated for 2 h with anti-Tyr(P) mAb (4G10 1  $\mu$ g/ml) or anti-p125<sup>FAK</sup> mAb (1:1000) as indicated in PBS containing 3% non-fat dried milk. Immunoreactive bands were visualised by autoradiography using <sup>123</sup>I-labelled sheep anti-mouse IgG (1:1000).

### Materials

FBS was obtained from Gibco BRL, Life Technologies, GRP, BSA, litorin, NmB, bombesin fragment amino acids 8-14, endothelin, insulin, bombesin, anti-mouse IgG (whole molecule)-agarose and Protein A-agarose were obtained from Sigma. Dispase was obtained from Boehringer Mannheim. Fura-2/AME was obtained from Calbiochem. The bombesin/GRP receptor-specific antagonist D-F<sub>3</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe was a kind gift of Dr David H Coy of Peptide Research Laboratories, Tulane University Medical Center, New Orleans, Louisiana, USA. [methyl-<sup>3</sup>H]thymidine (25 Ci/mmol), <sup>32</sup>P<sub>i</sub> (10 mCi/ml), <sup>125</sup>I-labelled sheep anti-mouse IgG (20  $\mu$ Ci/ $\mu$ g) and (3-[<sup>125</sup>I]iodotyrosyl<sup>15</sup>) Gastrin releasing peptide (2000 Ci/mmole) were obtained from Amersham International, UK. Anti-Tyr(P) mAb clone Py72 was obtained from the hybridoma development unit, Imperial Cancer Research Fund, London, United Kingdom. The 4G10 anti-Tyr(P) mAb was from Tissue Culture Supplies, Buckingham, UK. Anti FAK mAb was obtained from AFFINITI Research Products Ltd, Nottingham, UK. Anti-paxillin mAb 165 was from ICN, High Wycombe, UK. All other materials were of the purest grade commercially available.

### References

- Ashkenazi M, Winslow JW, Peralta EG, Peterson GL, Schimerlik MI, Capon DJ and Ramachandran J. (1987). Science, 238, 672-675.
- Battey JF, Way JM, Corjay MH, Shapira H, Kusano K, Harkins R, Wu JM, Slattery T, Mann E and Feldman RI. (1991). Proc. Natl. Acad. Sci. USA., 88, 395-399.
- Bold RJ, Lowry PS, Ishizuka J, Battey JF, Townsend CM Jr
- and Thompson JC. (1994). J. Cell. Physiol, 161, 519-525. Brooks SF, Herget T, Erusalimsky JD and Rozengurt E.
- (1991). EMBO J., 10, 2497-2505. Buhl AM, Johnson NL, Dhanasekaran N and Johnson GL.
- (1995). J. Biol. Chem., 270, 24631-24634. Carney DN, Cuttitta F, Moody TW and Minna JD. (1987).
- Cancer Res., 47, 821–825.
- Cobb BS, Schaller MD, Leu T-H and Parsons JT. (1994). *Mol. Cell. Biol.*, 14, 147-155.
- Corjay MH, Dobrzamski DJ, Way JM, Viallet J, Shapira H, Worland P, Sausville EA and Battey JF. (1991). J. Biol. Chem., 266, 18771-18779.
- Coy D, Mungan Z, Rossowski W, Cheng B, Lin JT, Mrozinski J and Jensen R. (1992). Peptides, 13, 775-781.
- Cuttitta F, Carney DN, Mulshine J, Moody TW, Fedorko J, Fischler A and Minna JD. (1985). *Nature*, **316**, 823-826.
- Fathi Z, Corjay MH, Shapira H, Wada E, Benya R, Jensen R, Viallet J, Sausville EA and Battey JF. (1993). J. Biol. Chem., 268, 5979-5984.
- Gorbulev V, Akhundova A, Buchner H and Farrenholz F. (1992). Eur. J. Biochem., 208, 405-410.
- Grady EF, Slice LW, Brant WO, Walsh JH, Payan DG and Bunnett NW. (1995). J. Biol. Chem., 270, 4603-4611.
- Hanks SK, Calalb MB, Harper MC and Patel SK. (1992). Proc. Natl. Acad. Sci. USA., 89, 8487-8491.
- Higgins TE, Murphy AC, Staddon JM, Lax AJ and Rozengurt E. (1992). Proc. Natl. Acad. Sci. USA., 89, 4240-4244.
- Hildebrand JD, Schaller MD and Parsons JT. (1995). Mol. Biol. Cell., 6, 637-647.
- Koch WJ, Hawes BE, Allen LF and Lefkowitz RJ. (1994). Proc. Natl. Acad. Sci. USA., 91, 12706-12710.
- Lee SB and Rhee SG. (1995). Curr. Opin. Cell. Biol., 7, 183-189.
- Nanberg E and Rozengurt E. (1988). EMBO J., 7, 2741-2747.
- Rankin S, Morii N, Narumiya S and Rozengurt E. (1994). FEBS Lett., 354, 315-319.
- Rodriguez-Pena A and Rozengurt E. (1986). EMBO J., 5, 77-83.
- Rozengurt E. (1986). Science, 234, 161-166.
- Rozengurt E. (1992). Curr. Opin. Cell Biol., 4, 161-165.
- Rozengurt E. (1995a). Oxford Textbook of Oncology.
  Peckham M, Pinedo H and Veronesi U. (ed.). Oxford University Press. pp. 12-20.

### Acknowledgements

We would like to thank Dr Jim Battey for his generous gift of the bombesin/GRP receptor cDNA, BNR-pCD2-neo and Dr David H Coy for his generous gift of the specific bombesin/GRP receptor antagonist, D-F<sub>3</sub>-Phe<sup>6</sup>, D-Ala<sup>11</sup>-Bombesin(6-13)OMe.

- Rozengurt E. (1995b). Cancer Surveys, 24, 81-96.
- Rozengurt E and Sinnett-Smith J. (1983). Proc. Natl. Acad. Sci. USA., 80, 2936-2940.
- Salgia R, Li J-L, Lo SH, Brunkhorst B, Kansas GS, Sobhany ES, Sun Y, Pisick E, Halleck M, Ernst T, Tantravahi R, Chen LB and Griffin JD. (1995). J. Biol. Chem., 270, 5039-5047.
- Schaller MD, Borgman CA, Cobb BS, Vines RR, Reynolds AB and Parsons JT. (1992). Proc. Natl. Acad. Sci. USA., 89, 5192-5196.
- Seckl M and Rozengurt E. (1993). J. Biol. Chem., 268, 9548 9554.
- Seckl MJ, Morii N, Narumiya S and Rozengurt E. (1995). J. Biol. Chem., 270, 6984-6990.
- Sethi T, Langdon S, Smyth J and Rozengurt E. (1991). Cancer Res., 52, 2737s-2742s.
- Sinnett-Smith J, Zachary I, Valverde AM and Rozengurt E. (1993). J. Biol. Chem., 268, 14261-14268.
- Spindel ER, Giladi E, Brehm P, Goodman RJ and Segerson TP. (1990). Mol. Endocrinol., 4, 1956-1963.
- Toullec D, Pianetti P, Coste H, Bellevergue P, Grand-Perret T, Ajakane M, Baudet V, Biossin P, Boursier E, Loriolle F, Duhamel L, Charon D and Kirilovsky J. (1991). J. Biol. Chem., 266, 15771-15781.
- Turner CE and Miller JT. (1994). J. Cell. Sci., 107, 1583-1591.
- Wada E, Way J, Shapira H, Kusano K, Lebacq-Verheyden AM, Coy D, Jenson R and Battey J. (1991). Neuron, 6, 421-430.
- Zachary I and Rozengurt E. (1985). Proc. Natl. Acad. Sci. USA., 82, 7616-7620.
- Zachary I and Rozengurt E. (1987a). J. Biol. Chem., 262, 3947-3950.
- Zachary I and Rozengurt E. (1987b). EMBO J., 6, 2233-2239.
- Zachary I, Sinnett-Smith J and Rozengurt E. (1986). J. Cell. Biol., 102, 2211-2222.
- Zachary I, Woll PJ and Rozengurt E. (1987). Dev. Biol., 124, 295-308.
- Zachary I, Gil J, Lehmann W, Sinett-Smith J and Rozengurt E. (1991a). Proc. Natl. Acad. Sci. USA., 88, 4577-4581.
- Zachary I, Sinnett-Smith J and Rozengurt E. (1991b). J. Biol. Chem., 266, 24126-24133.
- Zachary I, Sinnett-Smith J and Rozengurt E. (1992). J. Biol. Chem., 267, 19031-19034.
- Zachary I, Sinnett-Smith J, Turner CE and Rozengurt E. (1993). J. Biol. Chem., 268, 22060-22065.

